

Quality of life and other outcomes of breast reduction surgery

by

Tamara Ann Crittenden

Thesis Submitted to Flinders University for the degree of

Doctor of Philosophy

College of Medicine and Public Health

November 2020

TABLE OF CONTENTS

TABLE	E OF CONTENTS	I	
LIST C	OF FIGURES	VII	
LIST C	OF TABLES	IX	
	OF APPENDICES	XI	
	OF ABBREVIATIONS	XII	
THESI	IS SUMMARY	XIV	
DECLA	ARATION	XVI	
	OWLEDGEMENTS	XVII	
PUBLI	ICATIONS	xıx	
PRESE	ENTATIONS	XX	
1		1	
1.			
1.1		1	
1.2	BREAST HYPERTROPHY	2	
	1.2.1 Symptoms of breast hypertrophy	2	
1.3	BREAST REDUCTION SURGERY	4	
1.4	INDICATIONS FOR SURGERY AND VARIATIONS BETWEEN JURISDICTIONS		
1.5	VARYING ELECTIVE SURGERY RESTRICTIONS BETWEEN JURISDICTIONS – METHODS A	ND _	
1.6		8	
1.7	SURGICAL COMPLICATIONS	9	
	1.7.1 Risk factors for complications		
1.8	HEALTH-RELATED QUALITY OF LIFE AND PATIENT-REPORTED OUTCOME MEASURES		
	1.8.1 Health-related quality of life (HRQoL)		
	1.8.2 Measurement of HRQoL with patient-reported outcome measures	19	
	1.8.3 Challenges of interpreting HRQoL data		
1.9	REVIEW OF HRQOL RESEARCH IN BREAST REDUCTION SURGERY		
	1.9.1 Generic PROMs	23	
	1.9.1.1 Short Form-36		
	1.9.1.2 MBSRQ	24	
	1.9.1.3 Rosenberg Self-Esteem Scale	25	
	1.9.2 Condition or surgery-specific PROMs	25	
	1.9.2.1 BREAST-Q	25	

		1.9.2.2	BRSQ	28
		1.9.2.3	BRASS	29
		1.9.2.4	mBEQ	29
	1.9.3	Preferer	nce-based measures	30
	1.9.4	Factors	influencing the HRQoL of women following breast reduction surgery	31
	1.9.5	Compari	ison of women with breast hypertrophy with the general population	32
1.10	Asses	SMENT OF	BREAST VOLUME AND BODY SHAPE	34
	1.10.1	Breast v	olume measurement	34
	1.10.2	Body sh	ape assessment	35
1.11	ECON	OMIC EVAL	UATION OF BREAST REDUCTION SURGERY	35
2.	A PRO	OSPECTI	VE STUDY OF WOMEN UNDERGOING BREAST REDUCTION	
SURGE	ERY: CI		OUTCOMES AND HEALTH-RELATED QUALITY OF LIFE	38
2.1	Intro	DUCTION .		38
2.2	Parti	CIPANTS A	ND METHODS	39
	2.2.1	Design a	and participants	39
		2.2.1.1	Surgical cohort	39
		2.2.1.2	Breast hypertrophy control cohort	40
	2.2.2	Outcome	e measures	41
		2.2.2.1	Short-Form 36 (SF-36)	41
		2.2.2.2	Multidimensional Body-Self Relations Questionnaire (MBSRQ)	42
		2.2.2.3	Adjunct questionnaire	43
		2.2.2.4	Assessment of surgical complications	43
	2.2.3	Distribut	ion-based estimates of MID	44
	2.2.4	Statistica	al analysis	45
2.3	RESUL	.TS		46
	2.3.1	Study pa	articipants	46
		2.3.1.1	Surgical cohort	46
		2.3.1.2	Breast hypertrophy control cohort	47
	2.3.2	Clinical of	outcomes in the surgical cohort	49
	2.3.3	Health-re	elated quality of life outcomes using the SF-36	51
		2.3.3.1	Distribution-based estimates of MID for the SF-36	57
		2.3.3.2	Comparing the improvement in health-related quality of life with other	•
			surgical interventions	58
		2.3.3.3	The impact of participant characteristics on health-related quality of lin	fe
			and benefit of surgical intervention	61
	2.3.4	Quality of	of life outcomes using the MBSRQ	64
	2.3.5	Adjunct	questions	67

2.4	DISCUSSION			
2.5	CONC	CONCLUSION		
3.	GENERATION OF AUSTRALIAN GENERAL POPULATION REFERENCE DATA FOR			
INTER	PRETIN	G THE BREAST-Q	72	
3.1	Intro	DUCTION	72	
3.2	Метно	DDS	73	
	3.2.1	Participant recruitment	73	
	3.2.2	Data collection	73	
	3.2.3	The BREAST-Q	73	
	3.2.4	Normative data from the United States	74	
	3.2.5	Statistical analysis	74	
3.3	RESUL	TS	75	
	3.3.1	Participant characteristics	75	
	3.3.2	Australian normative BREAST-Q Reduction module scores	78	
	3.3.3	Predictors of BREAST-Q scale scores	86	
	3.3.4	Comparison of socio-demographic characteristics between normative populat	ion	
		samples	88	
	3.3.5	Comparison of normative BREAST-Q scores between normative population		
		samples	90	
3.4	Discu	SSION	94	
3.5	CONC	LUSION	96	
4	ουτα	COMES OF BREAST REDUCTION SURGERY USING THE BREAST-O' A		
PROS	PFCTIV	E STUDY AND COMPARISON WITH NORMATIVE DATA	97	
4.1	INTRO	DUCTION	97	
4.2	Parti	CIPANTS AND METHODS	98	
	4.2.1	Surgical cohort	98	
	4.2.2	Breast hypertrophy control cohort	98	
	4.2.3	The BREAST-Q	98	
	4.2.4	Minimal important difference estimates	99	
	4.2.5	Statistical analysis	99	
4.3	RESUL	TS	100	
	4.3.1	Study participants	100	
		4.3.1.1 Surgical cohort	100	
		4.3.1.2 Breast hypertrophy control cohort	100	
	4.3.2	BREAST-Q scores before and after breast reduction surgery	102	
	4.3.3	BREAST-Q scores in women with breast hypertrophy awaiting surgery	105	
	4.3.4	Distribution-based estimates of MID	105	

	4.3.5	Comparison of BREAST-Q scores between women with breast hypertr	ophy
		and women in the general Australian population	
4.4	DISCUSSION		
4.5	CONCLUSION		
5.	LONG	G-TERM OUTCOMES OF BREAST REDUCTION SURGERY	113
5.1	Intro	DUCTION	113
5.2	Parti	CIPANTS AND METHODS	114
	5.2.1	Participant recruitment	114
	5.2.2	Patient-reported outcome measures	114
	5.2.3	Adjunct questionnaires	115
	5.2.4	Statistical analysis	115
5.3	RESUL	_TS	116
	5.3.1	Study participants	116
	5.3.2	Participant characteristics	117
	5.3.3	Non-responders	119
	5.3.4	Long-term patient-reported outcome scores	119
		5.3.4.1 Short Form-36	119
		5.3.4.2 BREAST-Q	122
		5.3.4.3 MBSRQ	
		5.3.4.4 Adjunct questionnaire	127
		5.3.4.5 Other clinical outcomes	127
	5.3.5	Factors influencing long-term quality of life	128
5.4	Discu	ISSION	132
	5.4.1	Strengths and limitations	135
5.5	CONC	LUSION	135
6.	RELA	TIONSHIP BETWEEN BREAST AND BODY SIZE AND THE IMPROVE	EMENT IN
HEAL	TH-REL	ATED QUALITY OF LIFE FOLLOWING BREAST REDUCTION SURGI	ERY 137
6.1	Intro	DUCTION	137
6.2	Parti	CIPANTS AND METHODS	138
	6.2.1	Surgical participants	138
	6.2.2	Anthropometric body shape assessment	139
		6.2.2.1 Surgical participants	
		6.2.2.2 Normative anthropometry data	
	6.2.3	Measuring breast and body volume using a 3D laser scanner	139
	6.2.4	Water displacement for measuring breast volume	140
	6.2.5	Statistical analysis	141
6.3	RESUL	_TS	142

	6.3.1	Participants	142
	6.3.2	Missing data	142
	6.3.3	Anthropometric body shape assessment before and after breast reduction	
		surgery	142
	6.3.4	3D laser scanner	144
		6.3.4.1 Relationship between breast volume and bra cup size	146
		6.3.4.2 Relationship between breast volume and anthropometric variables .	147
		6.3.4.3 Breast asymmetry	148
		6.3.4.4 Breast size as a proportion of body size	149
		6.3.4.5 Relationship between body volume and BMI	151
	6.3.5	Water displacement	151
	6.3.6	Comparison of breast volume measurement using 3D scan and water	
		displacement	152
	6.3.7	Comparison of breast volume measurement using 3D scan and anthropome	tric
		formula	154
6.4	Discu	SSION	156
6.5	CONC	LUSION	159
7.	ECON	IOMIC EVALUATION OF BREAST REDUCTION SURGERY IN AUSTRALIA	160
7.1	Intro	DUCTION	160
7.2	Parti	CIPANTS AND METHODS	161
	7.2.1	Participants	161
	7.2.2	Healthcare costs	161
	7.2.3	Health-related quality of life	162
	7.2.4	Cost-utility analysis	162
	7.2.5	Sensitivity analysis	164
	7.2.6	Statistical analysis	165
7.3	RESUL	_TS	165
	7.3.1	Healthcare costs	166
	7.3.2	Outcomes	168
	7.3.3	Cost-utility analysis	171
		7.3.3.1 Base case analysis	171
		7.3.3.2 Sensitivity analysis	174
7.4	Discu	SSION	175
	7.4.1	Strengths and limitations	178
7.5	CONC	LUSION	179
8.	THES	IS CONCLUSIONS	180
8.1	SUMM	ARY OF THESIS	180

8.2	LIMITATIONS		
8.3	ORIGINAL CONTRIBUTION TO KNOWLEDGE	183	
8.4	FUTURE DIRECTIONS		
8.5	CONCLUSION		
REFERENCES			
APPENDICES			

LIST OF FIGURES

Figure 1.1.	Shoulder grooving due to breast hypertrophy	3	
Figure 1.2.	Common instruments to measure HRQoL outcomes of breast surgery	20	
Figure 1.3.	BREAST-Q conceptual framework		
Figure 2.1	Comparison of mean pre-operative and post-operative SF-36 scores in surgical		
	participants with age-standardised female population norms	53	
Figure 2.2	Comparison of mean baseline and 12-months post-baseline SF-36 scores in the		
	breast hypertrophy control group with age-standardised female population norms .	54	
Figure 2.3	Mean change in SF-36 scores from baseline to 12-months for surgical and breast		
	hypertrophy control cohorts in comparison to MID estimates (0.5 SD)	58	
Figure 2.4	Mean improvement in SF-36 Physical Component Summary (PCS) scores at		
	12m post-operative	60	
Figure 2.5	Mean improvement in SF-36 Mental Component Summary (MCS) scores at 12m		
	post-operative	60	
Figure 2.6	Relationship between body mass index and tissue resection weight at surgery	61	
Figure 2.7	Baseline SF-36 PCS scores for BMI categories	63	
Figure 2.8	Mean MBSRQ scores for surgical participants	66	
Figure 3.1	Item response frequencies 'Satisfaction with Breasts'	79	
Figure 3.2	Item response frequencies 'Psychosocial Wellbeing'	80	
Figure 3.3	Item response frequencies 'Sexual Wellbeing'	81	
Figure 3.4	Item response frequencies 'Physical Wellbeing'	82	
Figure 3.5	Normative BREAST-Q Reduction module pre-operative scores	84	
Figure 4.1	Pre- and post-operative photos of a patient who underwent an inferior pedicle		
	Wise pattern breast reduction with corresponding BREAST-Q scores	103	
Figure 4.2	Pre- and post-operative photos of a patient who underwent a Hall-Findlay pattern		
	breast reduction with corresponding BREAST-Q scores	103	
Figure 4.3	Mean change in BREAST-Q scores following surgery	104	
Figure 4.4	Mean change in BREAST-Q scores from baseline to 12-months for surgical and		
	breast hypertrophy control groups in comparison to MID estimate	106	
Figure 4.5	Comparison of surgical patient group mean BREAST-Q scores before and after		
	reduction surgery to normative scores for Australia and the United States	107	
Figure 4.6	Post-operative only BREAST-Q Reduction Module satisfaction scale scores	108	
Figure 5.1	Study flow-chart	117	
Figure 5.2	Mean SF-36 subscale scores over time compared to female general population		
	reference values	121	
Figure 5.3	Mean BREAST-Q scores for surgical participants over time in comparison to		
	population norms	123	

Figure 5.4	Mean MBSRQ scores pre-operatively versus scores at 12 months post-operative	
	and long-term follow-up	126
Figure 5.5	Bra cup size frequency distribution at pre-operative compared to long-term	
	follow-up	128
Figure 6.1	Volumetric analysis of the breast with Cyberware WBX 3D scanner	140
Figure 6.2	Water displacement technique	141
Figure 6.3	Relationship between the change in breast volume and weight of tissue resection	
	at surgery	145
Figure 6.4	Distribution of measured breast volume by bra cup size	146
Figure 6.5	Frequency distribution of the degree of breast asymmetry at baseline	
	assessment	148
Figure 6.6	Frequency distribution of proportion of breast to body size	149
Figure 6.7	Relationship between disproportionality and change in SF-36 PCS score	150
Figure 6.8	Relationship between disproportionality and change in SF-36 MCS score	150
Figure 6.9	Relationship between total body volume and BMI	151
Figure 6.10	Pearson correlation of breast volume measurement – water displacement versus	
	3D laser scan	152
Figure 6.11	Bland-Altman plot of breast volume measured by water displacement and 3D	
	scanning techniques	153
Figure 6.12	The level of agreement between the two methods	154
Figure 6.13	Pearson correlation of breast volume measurement – anthropometric formula	
	versus 3D laser scan	155
Figure 6.14	Bland-Altman plot of breast volume measured by anthropometric formula and 3D	
	scanning techniques	156
Figure 7.1	The cost-effectiveness plane	164
Figure 7.2	Mean SF-6D utility scores for surgical intervention and control cohorts in	
	comparison to age-adjusted normative data	169
Figure 7.3	Cost-Effectiveness Planes	172
Figure 7.4	Cost-Effectiveness Acceptability Curves	173
Figure 7.5	Comparison of health state using different algorithms	175

LIST OF TABLES

Table 1.1.	Summary of literature of major risk factors for complications	12
Table 2.1	Dimensions of the SF-36v2	42
Table 2.2	Dimensions of the MBSRQ	43
Table 2.3	Clavien-Dindo classification of surgical complications	44
Table 2.4	Baseline characteristics of participants.	48
Table 2.5	Association between participant and clinical characteristics and any complication.	50
Table 2.6	Logistic regression of risk factors for complications	51
Table 2.7	Mean (SD) SF-36 scores for participants in the surgical cohort, hypertrophy contr	ol
	cohort and normative female population	52
Table 2.8	Mean (95% CI) change in SF-36 scores in the breast hypertrophy control cohort	
	and surgical cohort	56
Table 2.9	Summary of distribution-based estimates of clinical significance (MID)	57
Table 2.10	Mean improvement in SF-36 PCS and MCS scores between surgical	
	interventions	59
Table 2.11	Comparison of mean change in SF-36 scores following surgery in non-obese and	1
	obese participants	62
Table 2.12	Comparison of mean change in SF-36 scores by complication status	64
Table 2.13	Mean (SD) MBSRQ scores for participants in the surgical and breast hypertrophy	,
	control cohorts in comparison with populations norms	65
Table 3.1	Participant socio-demographic characteristics	77
Table 3.2	Australian normative BREAST-Q Reduction module scores	78
Table 3.3	Australian normative BREAST-Q scores by socio-demographic characteristics	85
Table 3.4	Multiple linear regression models for BREAST-Q scales	87
Table 3.5	Comparison of participant socio-demographics between normative population	
	samples	89
Table 3.6	Comparison of unadjusted normative BREAST-Q Reduction module scores for th	е
	United States of America and Australia	91
Table 3.7	Comparison of age-adjusted normative BREAST-Q Reduction module scores	93
Table 4.1	Baseline characteristics of participants.	101
Table 4.2	BREAST-Q scores before and after breast reduction surgery	102
Table 4.3	Significance of surgical complications on mean change (SD) in BREAST-Q	
	scores	105
Table 4.4	BREAST-Q scores in the breast hypertrophy control group	105
Table 4.5	BREAST-Q Reduction module minimal important difference estimates	. 106
Table 4.6	Correlation between Satisfaction with Outcome and post-operative BREAST-Q	
	scales	108

Table 4.7	Validity of traditional insurance selection criteria for coverage of breast reduction	
	surgery and 'Satisfaction with Outcome' BREAST-Q Scores	109
Table 5.1	Participant characteristics	118
Table 5.2	Comparison of participant characteristics between responders and	
	non-responders	119
Table 5.3	Mean (SD) SF-36 scores for surgical participants over time in comparison to	
	normative data	120
Table 5.4	Summary of mean (SD) BREAST-Q scores in surgical participants over time	122
Table 5.5	Summary of post-operative BREAST-Q scale scores in surgical participants	123
Table 5.6	Comparison of mean (SD) MBSRQ scores over time for surgical participants	125
Table 5.7	Summary of adjusted long-term BREAST-Q scores by BMI category	128
Table 5.8	Summary of unadjusted long-term BREAST-Q scores by BMI category	129
Table 5.9	Summary of adjusted long-term SF-36 scores by BMI category	130
Table 5.10	Summary of adjusted long-term BREAST-Q scores by age category	130
Table 5.11	Summary of unadjusted long-term BREAST-Q scores by age category	131
Table 5.12	Summary of adjusted long-term SF-36 scores by age category	131
Table 6.1	Anthropometric measurements for surgical participants in comparison to	
	normative general population sample	143
Table 6.2	Breast and body volumes measured by 3D scanner pre- and 12 months post-	
	operative	144
Table 6.3	Relationship between pre-operative breast volume and anthropometric and 3D	
	scan volume variables	147
Table 6.4	Breast volumes for surgical participants pre- and post-operatively	152
Table 7.1	Baseline characteristics of participants	166
Table 7.2	Costs per participant for intervention and control groups over 12 months (\$AUD)	167
Table 7.3	Utility scores and QALYs gained for the intervention and control groups	170

LIST OF APPENDICES

Appendix A.	BMJ Open journal publication	. 208
Appendix B.	Ethics extension approval for study 118.056	. 217
Appendix C.	Ethics approval for study 73.17	. 218
Appendix D.	Short Form-36 (SF-36)	. 219
Appendix E.	Multidimensional Body-Self Relations Questionnaire (MBSRQ)	. 225
Appendix F.	Surgical Complications Checklist	. 231
Appendix G.	Ethics approval for study 7848	. 232
Appendix H.	Pre-operative BREAST-Q Reduction module	. 233
Appendix I.	Post-operative BREAST-Q Reduction module	. 236
Appendix J.	Data Usage Agreement	. 243
Appendix K.	Descriptive statistics and internal consistency of normative BREAST-Q scores	. 244
Appendix L.	BREAST-Q scores for each of the four scales stratified by socio-demographic	
	characteristics for Australian and United States populations	. 246
Appendix M.	Plastic and Reconstructive Surgery journal published article	. 250
Appendix N.	Ethics approval for study 299.18	. 251
Appendix O.	Letter of invitation for study 299.18	. 252
Appendix P.	Participant information sheet for study 299.18	. 253
Appendix Q.	Patient Health Questionnaire	. 254
Appendix R.	Australasian Journal of Plastic Surgery original article	. 258
Appendix S.	Anthropometric measurements	. 266

LIST OF ABBREVIATIONS

ASA	American Society of Anesthesiologists
AUD	Australian dollars
BMI	Body mass index
BP	Bodily Pain (SF-36 scale)
BSA	Body surface area
CEAC	Cost-effectiveness acceptability curve
CEP	Cost-effectiveness plane
CI	Confidence interval
CUA	Cost-utility analysis
ES	Effect size
GH	General Health (SF-36 scale)
HRQoL	Health-related quality of life
ICER	Incremental cost-effectiveness ratio
MBSRQ	Multidimensional Body-Self Relations Questionnaire
MCS	SF-36: Mental Component Summary
MH	Mental Health (SF-36 scale)
MID	Minimal important difference
MSAC	Medical Services Advisory Committee
Ν	Number
NICE	National Institute of Clinical Excellence
OR	Odds ratio
PBAC	Pharmaceutical Benefits Advisory Committee

PCS	SF-36: Physical Component Summary
PF	Physical Function (SF-36 scale)
PRO	Patient-reported outcome
PROM	Patient-reported outcome measure
QALY	Quality-adjusted life year
RE	Role Emotional (SF-36 scale)
RP	Role Physical (SF-36 scale)
SD	Standard deviation
SEM	Standard error of measurement
SF	Social Functioning (SF-36 scale)
SF-6D	Short Form six-dimensional
SF-36	Short Form-36 questionnaire
SF-36v2	Short Form-36 version 2 questionnaire
SRM	Standardised response mean
UK	United Kingdom
US	United States of America
VT	Vitality (SF-36 scale)
WHO	World Health Organisation

THESIS SUMMARY

The effectiveness of breast reduction surgery for women with symptomatic mammary hypertrophy has been established by previous studies but questions have remained about whether restrictions to accessing breast reduction surgery, based on particular criteria, are appropriate. This thesis is a comprehensive body of work on the health benefits of breast reduction surgery, evidence for restrictions and cost-effectiveness.

The introductory chapter of this thesis presents a review of the literature relating to the health burden associated with breast hypertrophy; and the reported health gains following breast reduction surgery. A review of the assessment of outcomes following surgery through postoperative complications and patient-reported outcomes is included. This chapter highlights the inconsistencies and limitations described in existing research studies. The final section of Chapter 1 details the increasing importance of economic evaluations in weighing up the costs and benefits of medical interventions.

This thesis reports on research which aimed to provide insight into the health burden of breast hypertrophy and the clinical- and cost-effectiveness of breast reduction surgery in improving health-related quality of life (HRQoL) within the Australian healthcare system. Chapter 2 describes a prospective outcome study in Australian women before and up to 12-months after breast reduction surgery. This study details the recruitment of a control group of women with breast hypertrophy who did not undergo surgery for comparison. HRQoL was assessed using validated patient-reported outcome measures (PROMs) including the Short Form-36 (SF-36) and Multi-Dimensional Body-Self Relations Questionnaire (MBSRQ). This chapter describes the comparison of outcomes to normative data and demonstrates the effectiveness of surgery in improving HRQoL to levels of the general population. The analysis extended to assess whether women benefit from surgery regardless of participant and clinical factors, including commonly scrutinised factors such as body mass index and resection weight.

Normative data provides a valuable clinical reference point for clinicians to compare scores from women undergoing breast reduction surgery, enabling a better understanding of the health burden of breast hypertrophy and the health benefits of breast reduction surgery. Chapter 3 was designed to derive normative BREAST-Q data from women within the Australian general population; prior to this study, normative BREAST-Q data was limited to a United States population. A comparison of normative population data demonstrated that differences exist when comparing HRQoL between populations, emphasizing the importance of using country-specific normative data wherever possible.

Chapter 4 examines the satisfaction and wellbeing in women before and 12-months after breast

reduction surgery using the BREAST-Q Reduction module. The use of a condition-specific questionnaire in combination with the generic instruments described in Chapter 2 provides a comprehensive assessment of patient-reported outcomes following surgery. Minimal important difference estimates remain undetermined for this PROM and were established in this study to further enhance interpretability and provide clinical relevance of BREAST-Q scores.

Limited evidence exists as to the long-term health benefits of breast reduction surgery. A prospective study establishing the long-term outcomes following breast reduction surgery is detailed in Chapter 5. This study provides a follow-up of women who underwent surgery and were included in the studies described in Chapter 2 and Chapter 4.

Chapter 6 describes the estimation of breast and body volume in women with breast hypertrophy before and after surgery using a series of techniques including anthropometric measurements, water displacement and three-dimensional laser scanning.

In an era of limited healthcare resources, evidence assessing the cost-effectiveness of breast reduction surgery is important to support the maintenance of funding by healthcare decision-makers. Chapter 7 details an economic evaluation of breast reduction surgery within the Australian public healthcare setting. HRQoL gains were assessed using the SF-6D and effectiveness of surgery was measured in terms of cost per quality-adjusted life years (QALYs) gained.

The final chapter summarises the evidence presented in this thesis that breast hypertrophy is a chronic and painful health condition and establishes the effectiveness of surgery in improving HRQoL to levels of the general population. This thesis established that women benefit from surgery regardless of patient and clinical characteristics, including those commonly used as restrictive criteria for access to surgery and insurance coverage in many countries and jurisdictions worldwide. This thesis provides strong evidence to demonstrate the clinical- and cost-effectiveness of surgery, supporting the inclusion of this procedure in publicly funded health systems.

DECLARATION

I certify that this thesis:

- 1. does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university; and
- 2. to the best of my knowledge and belief, does not contain any material previously published or written by another person except where due reference is made in the text.

Signed:

Tamara Crittenden

ACKNOWLEDGEMENTS

I would like to sincerely thank and acknowledge my principal supervisor, Associate Professor Nicola Dean, for providing me with the opportunity to undertake this degree and fulfill a career goal. I am also incredibly grateful for your valuable input and unwavering support during my candidature. Your ongoing commitment to providing timely feedback and guidance was instrumental in the completion of my Doctorate and greatly appreciated.

I would also like to thank my supervisors, Professor David Watson and Professor Julie Ratcliffe, for their ongoing support during the course of this candidature. Your practical guidance, research expertise and insightful comments have been greatly appreciated. I would also like to thank Professor Ratcliffe for sharing her knowledge of health economics and whose advice was invaluable in undertaking the economic evaluation study as part of this body of work.

I am incredibly grateful to all members of the AFESA Research Group for their contribution to the surgical study. This research would not have been possible without your time and efforts and I am genuinely appreciative to all of those involved. I would especially like to thank Dr Philip Griffin, who was the principal investigator and one of drivers of the original study. Thank you to Professor Maciej Henneberg and Daisy Veitch for their valuable contribution to the breast and body volume measurement aspect of this work. I would also like to thank Daisy for her expertise and involvement in the anthropometric measurement and 3D laser scanning of surgical participants (Chapter 6).

To my colleagues within the Department of Plastic and Reconstructive Surgery at Flinders Medical Centre, I would like to thank you all for your ongoing support of this research. I would especially like to thank Suzanne Patten, Andrea Smallman and Phillipa van Essen for their ongoing support during this candidature.

I would like to sincerely thank Dr Lily Mundy and colleagues for their collaboration in providing the United States normative BREAST-Q dataset for comparison to the derived Australian normative population data (Chapter 3). I would also like to thank Professor Anne Taylor, Dr Tiffany Gill and Dr Jodie Avery at the Population Research and Outcome Studies Unit at the University of Adelaide for providing the Australian SF-36 normative dataset and scoring algorithms (Chapter 2).

I would like to thank Christine Mpundu-Kaambwa for her assistance with the final analysis for the economic evaluation study in Chapter 7. I would also like to thank David Banham and Professor Robert Goldney for providing the dataset used to calculate Australian SF-6D normative values; and Dr Gang Chen for providing the syntax for the SF-6D scoring algorithms.

I would like to say a very special thank you to all of the participants, whose commitment and

involvement made these research studies possible. I hope that this research provides the evidence and awareness of the health benefits of breast reduction surgery so that their contribution is rewarded in a meaningful way by supporting the ongoing access to surgery for women going forward.

I would like to acknowledge the financial support of the Australian Government Research Training Program Scholarship that has enabled me to undertake this research.

Lastly, I would like to sincerely thank my family and friends, who were an ongoing source of support and encouragement in the pursuit of this degree. To my parents, David and Barbara, who have always believed in me and whose constant love and support are with me in anything I have ever strived to achieve. Most importantly, I wish to thank my husband Adam and my children, Isaac and Harry, for their understanding and support, and for providing me with the inspiration to accomplish my goals.

PUBLICATIONS

The following is a list of publications which were generated from the work contained within this thesis:

- 2020Does breast reduction surgery improve health-related quality of life? A
prospective cohort study in Australian women.Crittenden T, Watson DI, Ratcliffe, J, Griffin, P, Dean, NR on behalf of the AFESA
Research Group
BMJ Open 2020; 10(2): e031804.
- 2019 Outcomes of Breast Reduction Surgery using the BREAST-Q: A Prospective Study and Comparison with Normative Data. Crittenden T, Griffin P, Watson DI, Ratcliffe JR, Dean NR *Plastic and Reconstructive Surgery* 2019; Nov; 144: 1034-1044.
- 2019 Breast reduction surgery improves health-related quality of life in women with breast hypertrophy: a comparison to other surgical interventions. Crittenden T, Watson DI, Ratcliffe, J, Dean, NR *Quality of Life Research* 2019; 28: S115-S115.
- 2018 Measuring breast volume in hypertrophy: laser scanning or water displacement?
 Crittenden T, Veitch D, Henneberg M, Burford K, van Essen P, Deut K, Lomax E, Griffin P, Dean NR
 Australasian Journal of Plastic Surgery 2018; 1(2): 33-40.

2018 Normative data for the BREAST-Q Reduction module in an Australian population. Crittenden T, Watson D, Ratcliffe J, Dean NR *Quality of Life Research* 2018; 27(S1): S56-S57.

PRESENTATIONS

The following is a list of presentations which were generated from the work contained within this thesis:

- Breast reduction surgery provides symptomatic relief and improves healthrelated quality of life in women with breast hypertrophy.
 Crittenden, T., Watson, D., Ratcliffe, J., Griffin, P., & Dean, N.R.
 Presenter: Tamara Crittenden
 Oral presentation at the Plastic Surgery Congress (PSC) 2019, Melbourne, Australia.
- Quality of life in women who undergo breast reduction surgery: a prospective outcome study and comparison with normative data.
 Crittenden, T., Watson, D., Ratcliffe, J., Griffin, P., & Dean, N.R.
 Presenter: Tamara Crittenden
 Oral presentation at the Plastic Surgery Congress (PSC) 2019, Melbourne, Australia.
- Breast reduction surgery improves health-related quality of life in women with breast hypertrophy: a comparison to other surgical interventions.
 Crittenden, T., Watson, D., Ratcliffe, J. & Dean, N.R.
 Presenter: Tamara Crittenden
 Poster presentation at the International Society for Quality of Life Research (ISOQOL)
 26th Annual Conference 2019, San Diego, USA (International meeting).
- 2018 Normative data for the BREAST-Q Reduction Module in an Australian Population. Crittenden, T., Watson, D., Ratcliffe, J. & Dean, N.R.
 Presenter: Tamara Crittenden
 Poster presentation at the International Society for Quality of Life Research (ISOQOL)
 25th Annual Conference 2018, Dublin, Ireland (International meeting).
 Awarded Student Poster Presentation Finalist.
- 2017 Does Bilateral Breast Reduction Surgery Improve the Health Burden of Macromastia?
 Crittenden, T., Watson, D., Ratcliffe, J., Griffin, P., & Dean, N.R.
 Presenter: Tamara Crittenden
 Oral presentation at Flinders Health Research Week 2017, Flinders Centre for Innovation in Cancer, Adelaide, Australia.

1. LITERATURE REVIEW

1.1 Introduction

A body of literature has established that breast hypertrophy can have a significant impact on daily activities and the quality of life of affected women. It is a cause of considerable physical and psychosocial impairment. Effects of breast hypertrophy include persistent back, neck, shoulder and breast pain; painful bra strap grooving into shoulders; chronic skin rashes; poor body posture; hand tingling; difficulty performing physical activities including exercise; and psychological symptoms such as poor body image and low self-esteem (Blomqvist et al., 2000, Gonzalez et al., 1993, Guthrie et al., 1998, Kerrigan et al., 2001, Sigurdson et al., 2007b). Breast reduction surgery provides symptomatic relief in most cases and considerably improves the health-related quality of life (HRQoL) and wellbeing in women suffering from the functional symptoms of breast hypertrophy (Chao et al., 2002, Klassen et al., 1996, Mello et al., 2010, Singh and Losken, 2012, Thoma et al., 2007). Despite the evidence, breast reduction surgery is often regarded more as a cosmetic rather than a functional procedure by the general public and many medical professionals (Blomqvist et al., 2000, Frey et al., 2014, Taylor et al., 2004).

Traditionally surgical outcomes are measured by morbidity, mortality and hospital complications data (Devlin, 2010). However, when the primary goal of a surgical intervention is to improve patient quality of life and function, it would seem logical to measure function and quality of life to assess how well the surgery is achieving those goals. Patient-reported outcome measures (PROMs) enable the measurement of outcome according to the patient's own experience and opinion. Healthcare providers and clinicians worldwide are increasingly recognising the valuable information that PROMs provide for clinical practice improvement, patient advocacy, in support of research studies, and for the comparison of outcomes using alternative techniques or medical interventions (Cano et al., 2009, Pusic et al., 2011). There is considerable literature investigating surgical outcomes, risk factors for complications and HRQoL after breast reduction. However, there are contradictory findings and comparison difficulties that remain, particularly in the measurement of post-operative complications. In addition, there have been no outcome studies in Australian women and no comprehensive comparison between objective physical body measurements and patient-reported outcomes in the current literature.

The increasing demand for breast reduction surgery and increasing pressure to constrain healthcare spending has led to lengthy waiting times and restrictions placed on surgery in numerous countries and jurisdictions worldwide (Breuning et al., 2010, Klassen et al., 1996, Koltz et al., 2013, Nguyen et al., 2008, O'Blenes et al., 2006, Seitchik, 1995). Often these restrictions or denials are centred on body mass index or a minimum amount of resected breast tissue in order to qualify for surgery and are poorly supported by evidence. Clinical comparative effectiveness

1

research (CER) is an inclusive model of research that utilises best evidence to assess "what works best", "for whom" and "under what circumstances?" (Chalkidou et al., 2009, Selby et al., 2012, Sox, 2010). Understanding the clinical effectiveness of breast reduction surgery would be useful at both a patient and clinician level but also for healthcare funders and the wider community. Furthermore, establishing the cost-effectiveness of breast reduction surgery would be valuable; that is, to assess whether this surgery significantly improves HRQoL and at what cost to the Australian healthcare system. There remains the need for reliable, high quality studies to assist clinicians and health care decision-makers in determining appropriate levels of access to breast reduction surgery.

This chapter provides a comprehensive literature review on the health burden of breast hypertrophy, the effectiveness of breast reduction surgery, and the measure of outcomes including complications and health-related quality of life. Also included in this chapter is an overview of breast volume assessment and the economic evaluation of breast reduction surgery.

1.2 Breast hypertrophy

Breast hypertrophy, otherwise known as macromastia, is poorly defined in the literature and has no universally accepted definition. Intrinsic to the concept of this condition is the overdevelopment of breasts such that the weight of the breasts causes significant symptoms. Usually it is implied that both breasts are affected, but the term can also be used in the more unusual situation of one breast overdeveloping (developmental breast asymmetry). Juvenile macromastia is a subcategory of breast hypertrophy where an alarmingly rapid and continued breast growth occurs during puberty, usually as an abnormal response to hormonal stimuli. Gigantomastia is used to refer to cases of extreme breast enlargement (Dancey et al., 2008, Wolfswinkel et al., 2013).

Symptomatic breast hypertrophy causes a physical functional deficit and is associated with psychological distress and a reduced quality of life in women. Eighty percent of women with breast hypertrophy have tolerated symptoms that originated during puberty (Corriveau and Jacobs, 1990). The aetiology of breast hypertrophy is not well understood but its development is thought to be a multifactorial process involving factors such as hormonal change including adolescence and pregnancy, genetics, medication and obesity. Although studies exploring the relationship between body mass and breast size have presented inconsistent findings (Benditte-Klepetko et al., 2007, Brown et al., 2012, Findikcioglu et al., 2013, Vandeput and Nelissen, 2002), obesity is very common in adult women with breast hypertrophy with prevalence ranging from 24% to 43.5% (Cruz-Korchin et al., 2002, Roehl et al., 2008, Wagner and Alfonso, 2005). Although there is therefore evidence of a relationship between obesity and breast hypertrophy, pre-operative weight loss does not provide effective relief from symptoms of breast hypertrophy (Collins et al., 2002).

1.2.1 Symptoms of breast hypertrophy

Breast hypertrophy is a cause of considerable physical and psychosocial impairment in women.

Women often endure physical symptoms such as chronic back, neck, shoulder and breast pain; difficultly in carrying out daily activities and exercise; painful bra strap grooving into shoulders (Figure 1.1); chronic skin rashes; headaches; poor body posture; and numbness or tingling in their hands. Ducic and colleagues noted that 69% of patients with macromastia reported chronic headaches or migraines at their surgical consultation (Ducic et al., 2010). Starley et al. found women with breast hypertrophy experience reduced lung function and suggested that heavy breasts decreases chest wall compliance (Starley et al., 1998). It is also been shown that breast hypertrophy alters a woman's centre of gravity (Letterman and Schurter, 1980), increasing spinal curvature which can lead to degenerative changes in the spine (Benditte-Klepetko et al., 2007) and musculoskeletal pain (Coltman et al., 2019b). Coltman and colleagues found that women with large hypertrophic breasts participated in significantly less total physical activity than their counterparts, and suggested that breast size should be acknowledged as a potential barrier to women participating in physical activity (Coltman et al., 2019a).



Figure 1.1. Shoulder grooving due to breast hypertrophy

Psychological symptoms in women with breast hypertrophy may include poor body image, embarrassment, low self-esteem, poor psychosexual function, depression and anxiety (Beraldo et al., 2016, Singh and Losken, 2012). Up to one third of women who present for breast reduction surgery have clinical evidence of anxiety or depression, or both (Faria et al., 1999, Guthrie et al., 1998, Iwuagwu et al., 2006a, Saariniemi et al., 2011a, Saariniemi et al., 2009). Breast mass has been correlated with depression, with the finding that the higher the breast mass was in the study population, the greater the incidence of depressive symptoms according to studies using the Beck Depression Inventory (BDI) (Benditte-Klepetko et al., 2007). Social problems related to the large size of the breast include poor fitting clothing, unwanted attention, low self-esteem and feeling uncomfortable in body image and sexual relationships.

Breast hypertrophy is a chronic health problem that has been proven to significantly affect health and quality of life (Blomqvist et al., 2000). The combination of physical pain and psychological symptoms in breast hypertrophy produce a substantial negative impact on quality of life.

1.3 Breast reduction surgery

Breast reduction surgery, also known as reduction mammoplasty, has been shown to be the most effective treatment for breast hypertrophy in women; weight loss, hormonal therapy and physical therapy have had very little success (Benditte-Klepetko et al., 2007, Collins et al., 2002). Relief of disabling symptoms is the primary motivator for most women who are pursuing breast reduction surgery. As an example, Davis et al. reported that 91% of patients in their study stated that relief of their symptoms was the primary reason for deciding to have the operation (Davis et al., 1995). Surgery provides almost immediate symptomatic relief in most cases including decreased shoulder, back, and neck pain, reduced headaches and considerably improves the health-related quality of life and wellbeing in women suffering from the functional symptoms of breast hypertrophy (Berberoglu et al., 2015, Blomqvist et al., 2000, Foreman et al., 2009, Mello et al., 2010). Freire et al. reported that surgery improved functional capacity and reduced pain in the lower back, shoulders, and neck of patients with breast hypertrophy (Freire et al., 2007). Studies have also reported that surgery results in a significant improvement in the body posture of patients (Chao et al., 2002, Sahin et al., 2013) and a reduced or discontinued use of analgesia (Miller et al., 1995, Schnur et al., 1997). Breast reduction surgery has also been shown to improve oncologic surveillance by facilitating breast self-examinations (Brown et al., 2008), as an option in women with breast cancer and macromastia (Losken et al., 2017), and as a risk-reduction surgery option (Tarone et al., 2004).

Having large breasts often makes exercise difficult and leads to a more sedentary lifestyle; reduction surgery facilitates an increase in physical activity and exercise (Boschert et al., 1996). In fact, Brown and colleagues reported that 100 percent of women found exercising easier and were able to participate in more rigorous forms of exercise post-operatively (Brown et al., 2008). Breast reduction surgery has also been shown to increase work capacity and productivity in patients with breast hypertrophy (Atterhem et al., 1998, Cabral et al., 2017b).

Psychologically, breast reduction surgery has a positive impact on the symptoms of anxiety and depression (Beraldo et al., 2016, Berberoglu et al., 2015, Cerovac et al., 2005, Chadbourne et al., 2001, Chahraoui et al., 2006, Faria et al., 1999, Iwuagwu et al., 2006a, Perez-Panzano et al., 2017, Romeo et al., 2010, Saariniemi et al., 2009, Saariniemi et al., 2011b). Saariniemi et al.

conducted a prospective randomised trial to assess levels of anxiety and depression in an operative group and compared this to women who underwent conservative treatment. They found a significant reduction in anxiety and depression in the surgical group (Saariniemi et al., 2009). In a subsequent study, women were followed for up to 5 years after surgery and it was found that the improvement in anxiety and depression was long lasting (Saariniemi et al., 2011b). Improvement in self-esteem (Mello et al., 2010, Saariniemi et al., 2009, Sabino Neto et al., 2008, Shakespeare and Postle, 1999) and psychosexual function (Cerovac et al., 2005, Romeo et al., 2010) have also been reported following breast reduction surgery.

Despite the reported health benefits of breast reduction surgery, some members of the general public, medical professionals and health insurance providers often mistakenly think of the surgery as a "cosmetic" procedure rather than a functional operation (Frey et al., 2014). This is in spite of the finding that breast reduction surgery not only reduces pain and functional problems, but also increases and normalises the quality of life in these patients (Blomqvist et al., 2000, Miller et al., 2005, Mundy et al., 2017c, O'Blenes et al., 2006). This highlights the effectiveness of surgery; quality of life and wellbeing are significantly improved to a level that is equal to that of the general population. Further research into the "normalisation" effect after surgery is warranted, as there are a limited number of studies on this comparison in the literature.

1.4 Indications for surgery and variations between jurisdictions

Whilst the precise prevalence of breast hypertrophy is unknown, it is believed to be increasing. Reduction mammoplasty is a common procedure in plastic surgery and is one of the most frequently performed breast surgeries worldwide. In the United States of America, the American Society of Plastic Surgeons (ASPS) reported that 100,969 breast reductions were performed in 2016 (American Society of Plastic Surgeons, 2016). This number represents a 19% increase since 2000. In Australia, breast reduction surgery statistics are difficult to collate, as surgery may take place in the public hospital, private hospital or cosmetic setting and there are currently no reporting requirements from surgeons. Data from the Australian Institute of Health and Welfare (AIHW) indicates that 7102 bilateral breast reduction surgeries were performed in 2017/18, an 104% increase since 2000/01 in which 3481 surgeries were recorded (Australian Institute of Health and Welfare, 2019b). However, the release of surgical data by Australian state and territory or by clinical setting is currently restricted outside of the Department of Health and related agencies, and therefore it is difficult to ascertain where these increases have occurred.

The demand for elective surgery procedures including breast reduction surgery is increasing in many countries worldwide and health systems do not have the resources to meet this need (Curtis et al., 2010). Consequently, access to surgery is often limited and waiting lists are long. A study of insurance company medical policies revealed they are "uniformly inconsistent with peer-reviewed,

5

published indications for breast reduction surgery", resulting in women with a medical need for surgery denied coverage (Nguyen et al., 2008). Health care funders and third-party providers have placed restrictions on access to surgery that commonly focus on the obese population or a minimum weight of reduction. For example, in Sweden only women at a normal weight (body mass index (BMI) 19 to 24.9 kg/m² in the age group <50 years and BMI \leq 26.9 kg/m² in the age group ≥50 years) and large breasts (volume ≥800 mL per breast) have access to breast reduction surgery through the public health system (Eggert et al., 2009, Hansson et al., 2014). This is also the situation in Spain and many other countries. In the United Kingdom, reduction mammoplasty is a "postcode lottery" and is available in some areas of the National Health Service (NHS), whilst in many areas it is a restricted procedure (Klassen et al., 1996, Royal College of Surgeons of England, 2016, Wraight et al., 2007). In the province of Nova Scotia, a patient with symptoms of mammary hyperplasia must have a BMI of less than 27 kg/m² to qualify for insurance coverage. In other regions, there are different insurance criteria (O'Blenes et al., 2006). In the United States and Canada, health care insurance providers often stipulate a minimum amount of breast tissue be resected in order to qualify for surgery with many using a body surface area (BSA)-adjusted value scale described by Schnur and colleagues (Schnur et al., 1991). However, evidence from the literature relating to a BMI cut-off point or a minimum resection weight does not support these restrictions, as it has been shown that symptom relief and improvement in health-related quality of life are independent of these factors (Blomqvist et al., 2000, Cole and Shakespeare, 1998, Collins et al., 2002, Dabbah et al., 1995, Gonzalez et al., 2012, Nguyen et al., 2008, Spector and Karp, 2007, Spector et al., 2008, Strong and Hall-Findlay, 2015, Wagner and Alfonso, 2005). As an example, a minimum 500 grams per breast of resected tissue is commonly required for reimbursement despite the fact that clinical studies have shown that amounts as little as 205 grams per breast result in significant clinical improvement (Nguyen et al., 2008).

In Australia, an 'Elective surgery access policy' was released by the Victorian Government for indications for surgery in Victorian public health services. In the case of breast reduction surgery, surgery is indicated for patients with a BMI less than 30 kg/m² where clinical symptoms are present (Victorian Government Department of Human Services, 2015). Whilst in Tasmania this procedure is restricted to patients with a BMI less than 35 kg/m² and requires exceptional clinical indications for surgery; chronic head, neck and back ache (where pain is due to breast size) and/or chronic intertrigo (Tasmanian Government Department of Health and Human Services, 2015). In New South Wales, bilateral breast reduction surgery is listed as a cosmetic surgical procedure that should not routinely be performed in public hospitals, except in the instance of severe disability due to breast size (NSW Government Department of Health, 2012). Similarly, in Western Australia breast reduction (not performed as part of cancer treatment) is on the excluded procedures list; the surgical procedure should not be routinely performed unless approval is given by the appropriate executive (Government of Western Australia, 2017). Whereas in the Australian Capital Territory

6

reduction mammoplasty is an excluded procedure and should not be performed in public hospitals with the exception of gross breast asymmetry in patients under 21 and virginal (juvenile) hypertrophy (ACT Government, 2016). In South Australia, reduction mammoplasty is also on the restricted procedures list. However, it may be performed if justified on true clinical grounds; to rectify poor posture, to relieve back or shoulder pain or to relieve discomfort caused by grooves from the brassiere strap (Government of South Australia SA Health, 2018). In short, there is inequitable access to breast reduction surgery within Australian public hospitals and access for patients is ultimately reliant upon state and local policies. Furthermore, the Australian Medicare Benefits Schedule (MBS) does allow breast reduction surgery for a rebate in private hospitals, therefore recognising that it is a functional procedure that justifies tax-payer funding.

As rising economic pressure continues to enforce changes and restrictions on the health care system, it has become increasingly important to demonstrate the clinical effectiveness and long-term health benefits of breast reduction surgery. No such studies currently exist in the Australian health care setting.

1.5 Varying elective surgery restrictions between jurisdictions – methods and evidence

Worldwide, a wide variety of systems exists to allocate resources in health care, none of which provide the perfect model. It is impossible for public health care systems to fund every procedure for every patient. This has led to 'rationing' of health care in public health systems. Whilst this is an essential process in health care and is driven by an excess in demand for scarce or limited resources, it has the potential to create inequity and raises a number of complex moral, legal and ethical concerns. The term when used in health care often has negative connotations as many people believe that it is not reasonable or ethical to place arbitrary limits on access to care.

Procedures performed for non-life threatening conditions are the first to be scrutinised when services have to be rationed (Breuning et al., 2010). Rationing means that some patients who, despite having a surgically treatable condition, will not receive surgical treatment and may be forced to endure symptoms and a reduced quality of life. In the United Kingdom, there have been reports on the rationing of surgery by the NHS on the basis of geographical location, resulting in a "postcode lottery" (Royal College of Surgeons of England, 2014). The report found that 73% of Clinical Commissioning Groups (CCGs) reviewed were not following NICE and clinical guidelines for referral for hip replacements and 44% required patients to meet various thresholds before they were referred for hip replacements. A subsequent report in 2016 by the Royal College of Surgeons of England (RCS) was prompted by concerns of further evidence of rationing of surgery by the NHS and that some patients, smokers and the overweight, were soft targets for NHS savings (Royal College of Surgeons of England, 2016). Whilst surgeons and medical professionals recognise that ceasing smoking and weight loss is potentially beneficial to surgical

outcomes, they have never proposed a ban to such patients. As an example, clinical guidance by NICE and the RCS states that 'patient-specific factors such as age, gender, smoking, obesity and co-morbidity should not be barriers to referral' for joint surgery (National Institute for Health and Clinical Excellence, 2014b). Despite this, 39% of clinical commissioning groups have a policy on BMI for hip and knee replacement surgery, with the cut-off threshold varying by CCG location. Given that 65% of men and 58% of women were reported as either overweight or obese in England in 2014, a significant proportion of the population could be affected by these BMI restrictions. Withholding treatment based on the basis of age has been reported by Evans, and is "best documented in substandard treatment of acute myocardial infarction and other forms of heart disease, where it leads to premature deaths and unnecessary disability" (Evans, 1997). A report by RCS and Age UK found that despite a population that is older and fitter than ever before, elective surgical treatment rates across a range of common conditions declined steadily for the over-65's (Royal College of Surgeons of England, 2012). In the United States (US), most health care is privately funded and therefore is rationed based on the ability to pay; you get what you or your employer can afford. Whereas, in the public system in the US, rationing is by long waiting lists, high patient co-payments and limited payments to doctors and hospitals. Rationing by price means that there is no triaging system according to need; privately insured patients have access to expensive surgeries and treatment, whilst the uninsured are denied access to even the most basic care (Singer, 2009).

In the Australian public health system, access to elective surgery is prioritised through the use of waiting lists using broad urgency categories; surgeons are principally responsible for making decisions for placing patients on waiting lists, urgency of care and selecting patients from the waiting list for surgery. The lack of uniform and specific guidelines about patients' need for surgery is a shortcoming of the Australian health system and does not assure equity of access (Curtis et al., 2010). In Victoria, substantial variability has been observed in the use of urgency categories across surgeons and hospitals (Russell et al., 2003). Furthermore, variation in waiting times for surgery according to socioeconomic status and remoteness have been observed in Australia and reveal an inconsistency in access to elective surgery (Australian Institute of Health and Welfare, 2008).

Clinicians are increasingly faced with moral and professional challenges in making appropriate decisions for a patient's individual treatment needs. Limited resources in the public health setting has forced rationing of health care resources and restrictions that are often not evidence-based; both have the potential to threaten equity of access to surgical treatment.

1.6 Surgical techniques

Surgery to reduce overly large breasts was first reported upon in the sixteenth century by German

8

Hans Schaller who is believed to have carried out the first successful breast amputation. However, it was Johann Friedrich Dieffenbach in 1848 who was the first to describe a breast reduction resection using an incision in the inframammary fold. Over the years, surgical techniques have been modified and refined in order to achieve both aesthetic and functional outcomes.

There is no single breast reduction technique that can be applied to all breasts and each technique has its advantages and limitations. Each individual patient is unique, and a surgeon must adapt their technique appropriately in order to achieve the functional improvement and most aesthetically pleasing result (Antony et al., 2013, Dex et al., 2008, Greco and Noone, 2017, Hall-Findlay, 2004, McCulley and Hudson, 2001).

Various techniques have been described in the literature with different skin incision and pedicle patterns. The Wise "keyhole" pattern incision was developed in 1956 (Wise, 1956) and further described in 1977 (Robbins, 1977). The inferior pedicle Wise breast reduction (inverted T-scar) is still one of the most frequently used procedures (Okoro et al., 2008, Rohrich et al., 2004). However, the disadvantages of the technique are the potential for "bottoming-out" of the lower pole over time and long scars that may detract from the overall aesthetic result. Claude Lassus reported the first limited incision technique in 1964 using a vertical reduction technique (Lassus, 1970). In the last decade, some surgeons have adopted modifications on the vertical or limited incision techniques that have become popular (Hall-Findlay, 2004, Lejour et al., 1990). However, limited incision techniques are preferred less by surgeons as the degree of hypertrophy and ptosis increases (Okoro et al., 2008). Additional limited incision techniques have been described using periareolar approaches (Goes, 1996, Hammond, 1999) and L-shaped scars (Bozola, 1990).

1.7 Surgical complications

Reduction mammoplasty is a very effective surgical procedure to improve function, quality of life and relieve painful symptoms associated with large breasts. However, the surgery does not come without its risk of complications. Risk factors for complications are a major determinant in surgical planning and this has become an important area of research. Post-operative complications after breast reduction can include wound healing problems, wound infection, skin necrosis, fat necrosis, seroma, haematoma, altered nipple sensation, nipple loss, asymmetry and abnormal scarring.

The reported rate of complications in the literature is highly variable, from 4% to as high as around 50% (Atterhem et al., 1998, Bauermeister et al., 2019, Bikhchandani et al., 2007, Blomqvist, 1996, Cogliandro et al., 2017a, Cunningham et al., 2005, Dabbah et al., 1995, Fischer et al., 2014, Guemes et al., 2016, Gust et al., 2013, Hanwright et al., 2013, Hernanz et al., 2016, Lewin et al., 2014, Manahan et al., 2015, Mello et al., 2010, Miller et al., 2005, Nelson et al., 2014a, O'Blenes et al., 2006, Platt et al., 2003, Roehl et al., 2008, Scott et al., 2005, Setala et al., 2009, Shah et al., 2011, Shakespeare and Cole, 1997, Simpson et al., 2019, Srinivasaiah et al., 2014, Young et al.,

2019, Zubowski et al., 2000). However, this sizeable variation may be explained by the lack of comparability between studies: many researchers evaluate limited numbers of patients; studies may not include a range of surgical techniques; results may be based on a single surgeon's experience; there is a lack of clarity on what constitutes a 'complication'; and the majority of the larger studies are retrospective surgical database searches and do not necessarily capture all complication data specific to reduction mammoplasty surgery. Parikh and colleagues conducted a literature review to analyse the quality of complication reporting in plastic surgery, including reduction mammoplasty (Parikh et al., 2018). The authors demonstrated inconsistencies in the reporting of complication outcomes in the plastic surgery literature and highlighted the need for the development and implementation of standardised reporting guidelines to accurately, efficiently, and reproducibly report complication data for core plastic surgery procedures.

In a large single-centre retrospective review of 2142 consecutive breast reduction procedures, Manahan et al. reported that common complications were related to wounds (14.9%) and scars (14.5%), along with fat necrosis (8.2%), infections (7.3%) and seroma (1.2%) (Manahan et al., 2015). In another retrospective study, it was found that complications occurred in 32% of the patients within 30 days of surgery, the most common being infection at the surgical site (16%) followed by delayed wound healing (10%) (Lewin et al., 2014). Whilst Bikhchandani and colleagues reported a lower complication rate of 19%, with T-junction necrosis (11.2%), infection (5.5%) and wound dehiscence (2.2%) being the most common complications (Bikhchandani et al., 2007). However, the authors highlighted that delayed healing, one of the commonly reported complications in other studies, was not recorded in their study due to the difficulties in accurately defining delayed healing in a retrospective study (Bikhchandani et al., 2007).

In a prospective study by a Spanish team of researchers, a complication rate of 27.3% was reported with partial cutaneous dehiscence (11.6%) the most common complication followed by haematoma (3.3%) (Perez-Panzano et al., 2016). In the prospective Breast Reduction Assessment: Value and Outcomes (BRAVO) study of 179 patients, the complication rate was 43%, with healing complications representing the most common complication (21.6%) (Cunningham et al., 2005). The BRAVO study originated to address deficiencies in previous research describing outcomes of breast reduction surgery by using a prospective, controlled, multi-centre study design with a comprehensive set of standardised outcome measures. Outcome studies, both retrospective and prospective, predominantly agree that wound healing complications are the primary problem following breast reduction surgery.

In large population-based analyses of reduction mammoplasty surgeries using the American College of Surgeons National Surgical Quality Improvement Program (NSQIP) surgical database, studies have reported the overall incidence of complications between 4% and 8.7% (Fischer et al., 2014, Gust et al., 2013, Nelson et al., 2014a, Simpson et al., 2019, Young et al., 2019). However, whilst use of the NSQIP database may facilitate large study numbers, it has limitations in that it only captures early post-operative complications within 30 days and does not include many complications specific to breast surgery such as delayed wound healing, haematoma, seroma, hypertrophic scarring, and nipple necrosis. Given that delayed wound healing is widely reported to be the most common complication following reduction mammoplasty, the lower complication rates reported by studies using the NSQIP database are perhaps not surprising.

The Clavien-Dindo Classification (CDC) is a validated grading tool that allows the classification of surgical complications. The CDC grades complications based on the medical intervention required to treat the complications following operative procedures (Dindo et al., 2004). This tool provides a simple, standardised and reproducible approach of objectively ranking surgical outcomes and is valid and applicable across a wide range of surgical specialties (Clavien et al., 2009). The CDC is widely accepted ranking system for classifying surgical complications and overcomes the inconsistency in reporting of complications using terms such as "minor", "major", or "severe" complications following breast reduction surgery. Winter et al. reported an overall complication rate of 63%; 48% had a Grade I, 9% a Grade II, 1% a Grade IIIA, and 5% a Grade IIIB operative complication (Winter et al., 2017). The use of an objective classification of complications following breast reduction surgery would allow a more accurate and objective comparison of the incidence and severity of complications following this procedure.

1.7.1 Risk factors for complications

The study of factors affecting surgical outcome is an area of research that is widely reported in the literature. A number of factors such as body mass index, age, resection weight, smoking, diabetes, operative technique and chronic steroid use have been implicated as risk factors for complications following breast reduction surgery (Table 1.1). When present, these factors can delay the healing process, increase morbidity, and affect the overall aesthetic result. However, the number of prospective, adequately powered studies examining these factors is limited and there are comparison difficulties between many of the existing studies. Furthermore, almost all of the risk factors associated with complications have contradictory outcomes in the literature (Table 1.1).

Diak faatar	Supporting evidence			Refuting evidence				
RISK factor	Study	n	Туре	Study	n	Туре		
Smoking	(Young et al., 2019)	9110	R	(Bauermeister et al., 2019)	938	R		
	(Simpson et al., 2019)	16,812	R	(Guemes et al., 2016)	121	P		
	(Winter et al., 2019)	804	R	(Roehl et al., 2008)	179	Р		
	(Theocharidis et al., 2018)	4912	R	(Karamanos et al., 2015)	2//9	R		
	(\square and \square et al., 2017) (Baltodano et al. 2017)	7068	R	(Henry et al., 2009) (Henry right et al., 2013)	400 2507	R		
	(Fischer et al. 2014)	3538	R	(Hanwight et al., 2010)	2007			
	(Zhang et al., 2016)	10.593	M					
	(Manahan et al., 2015)	2142	R					
	(Cunningham et al., 2005)	179	Р					
	(Bikhchandani et al., 2007)	402	R					
	(Bartsch et al., 2007)	50	Р					
	(Schumacher, 2005)	/1 67	R					
	(1 ewin et al 2014)	512	R					
	(Gravante et al., 2008)	87	P					
	(Shah et al., 2011)	306	R					
	(Chan et al., 2006)	173	R					
Body Mass	(Bauermeister et al., 2019)	938	R	(Eggert et al., 2009)	65	Р		
Index	(Simpson et al., 2019)	16,812	R	(Roehl et al., 2008)	179	Р		
	(Winter et al., 2019) (Poltodopo et al., 2017)	804 7069	R	(Cunningnam et al., 2005)	179	P		
	(Manahan et al., 2017)	2142	R	(Wagner and Alfonso	186	R		
	(Nelson et al., 2014a)	4545	R	2005)	100			
	(Henry et al., 2009)	485	R	(Setala et al., 2009)	273	R		
	(Zubowski et al., 2000)	395	R	(Guemes et al., 2016)	121	Р		
	(Platt et al., 2003)	30	P	(Webb et al., 2012)	67	R		
	(Zhang et al., 2016)	10,593	R	(Tadiparthi and Liew, 2008)	206	R		
	(Sinivasaian et al., 2014) (Eischer et al. 2014)	07 3538	P R	(Gamboa-Bobadilla and Killingsworth 2007)	80	ĸ		
	(Karamanos et al., 2014)	2779	R	(Hanwright et al., 2013)	2507	R		
	(Shah et al., 2011)	306	R	(,,,				
	(Myung and Heo, 2017)	6904	Μ					
	(Gust et al., 2013)(BMI ≥39)	2492	R					
	(Chen et al., 2011)	8000	R					
	(Chun et al., 2012) (Labiri et al., 2006)	075 13	R D					
	(Platt et al., 2000)	30	P					
	(Atterhem et al., 1998)	242	R					
	(Gulcelik et al., 2011)	286	R					
	(Stevens et al., 2008)	444	R					
	(Baldwin et al., 2010)	40	Р					
	(O'Grady et al., 2005) (Plomaviat, 1006)	133	R					
۸de	(Young et al. 2019)	9110	R	(Baltodano et al. 2017)	7068	R		
Age	(Simpson et al., 2019)	16.812	R	(Nelson et al., 2014b)	3537	R		
	(Srinivasaiah et al., 2014)	67	Р	(Zhang et al., 2016)	10,593	R		
	(Shermak et al., 2011)	1192	R	(Henry et al., 2009)	485	R		
				(Roehl et al., 2008)	179	P		
				(Schumacher, 2005) (Braig et al., 2016)	71 50	R D		
				(Draig et al., 2010) (Setala et al. 2009)	273	R		
				(Guemes et al., 2009)	121	P		
				(Shah et al., 2011)	306	R		
				(Cunningham et al., 2005)	179	Р		
				(Hanwright et al., 2013)	2507	R		
				(Hunter-Smith et al., 2012)	283	ĸ		

Table 1.1. Summary of literature of major risk factors for complications

Dick factor	Supporting evidence			Refuting evidence		
RISK lactor	Study	n	Туре	Study	n	Туре
Resected specimen weight	(Bauermeister et al., 2019) (Winter et al., 2019) (Zubowski et al., 2000) (Srinivasaiah et al., 2014) (Lewin et al., 2014)	938 804 395 67 512	R R R P R	(Wagner and Alfonso, 2005) (Zhang et al., 2016)(>1000g) (Roehl et al., 2008)	186 10,593 179	R R P
	(Gravante et al., 2008) (Shah et al., 2011) (Setala et al., 2009) (Henry et al., 2009) (Cunningham et al., 2005) (Schnur et al., 1997)	87 306 273 485 179 328	P R R R R R	(Hunter-Smith et al., 2012)	283	R
Diabetes	(Young et al., 2019) (Simpson et al., 2019) (Lewin et al., 2014) (Eggert et al., 2009)	9110 16,812 512 65	R R P	(Bauermeister et al., 2019) (Manahan et al., 2015) (Fischer et al., 2014) (Hanwright et al., 2013) (Henry et al., 2009) (Roehl et al., 2008)	938 2142 3538 2507 485 179	R R R R R P
Cardiac	(Manahan et al., 2015)	2142	R	(Fischer et al., 2014)	3538	R
Nipple-to- sternal notch distance	(Bauermeister et al., 2019) (Manahan et al., 2015) (Lewin et al., 2014)	938 2142 512	R R R			
Operating time	(Simpson et al., 2019) (Hanwright et al., 2013)	16,812 2507	R R			
Operative technique	(Derby et al., 2016) (pedicle technique)	317	R	(Bauermeister et al., 2019) (Kemaloglu and Ozocak, 2018)	938 50	R P
				(Ogunleye et al., 2017) (Roehl et al., 2008) (Antony et al., 2013) (Hunter-Smith et al., 2012)	124 179 150 283	P P R R
Chronic steroid use	(Simpson et al., 2019) (Barcha and Ranzer, 2018) (Hillam et al., 2017) (Baltodano et al., 2017)	16,812 17,058 13,503 7068	R R R R	(Hanwright et al., 2013)	2507	R

P, prospective study; R, retrospective study; M, meta-analysis; n, number of participants

Zhang and colleagues performed a meta-analysis of 16 unique studies, including 10,593 patients, examining risk factors for complications in order to address the inconsistencies surrounding the predictors of complications after reduction mammoplasty (Zhang et al., 2016). They found that smoking and a BMI greater than or equal to 30 kg/m² increased the risk of complications. Patients who were obese or irradiated were found to be more likely to develop infections, and smokers experienced a higher incidence of wound dehiscence than non-smokers. The authors found no association between age \geq 50 years or a combined tissue resection weight \geq 1000 grams and post-operative complications (Zhang et al., 2016).

Smoking

In previous outcome studies in plastic surgery, smoking has been shown to significantly increase post-operative morbidity. Chang et al. found a 4.4% incidence of flap necrosis after free TRAM flap reconstruction in smokers compared with 0.8% in non-smokers (Chang et al., 2000); and Manassa et al. reported that smokers had 3.2-fold increased chance of developing wound healing complications than non-smokers following abdominoplasty surgery (Manassa et al., 2003). Perhaps surprisingly, disagreement exists in the literature regarding the effect of smoking on patient outcomes following reduction mammoplasty.

Simpson and colleagues conducted the largest retrospective analysis of 16,812 reduction mammoplasties from 2006 to 2015 using the NSQIP database and found that smoking was associated with an increased risk of total complications (OR, 1.39; p = 0.001) but not with major complications (Simpson et al., 2019). In another study of 9110 patients, Young and colleagues aimed to identify pre-operative risk factors and to identify any increased complication risk in patients older than 60 years and found that smoking was found to be a statistically significant risk factor for superficial surgical site infection and deep space infection regardless of age (Young et al., 2019). Hillam and colleagues conducted a multicentre retrospective analysis of 13,503 reduction mammoplasties from 2009-2014 using the NSQIP database to specifically evaluate smoking as a risk factor. They found that smokers had a higher likelihood of any wound complication following surgery compared to non-smokers (OR 1.72; p < 0.001) (Hillam et al., 2017). This finding was supported by the study by Winter and colleagues using the Clavien-Dindo Classification system (Winter et al., 2019). In another study using the NSQIP database, Fischer et al. retrospectively reviewed 3538 reduction mammoplasty patients from 2005-2010. They concurred with Hillam et al. that smoking increased the risk of overall surgical complications (OR, 1.7; p < 0.001), and that active smoking was a strong predictor of major surgical complications (Fischer et al., 2014). Interestingly, findings from another study of 2779 patients using the NSQIP database are contradictory to this. The authors found no significant association between smoking and wound complication (Karamanos et al., 2015). However, as mentioned previously, a limitation of these studies and others using the NSQIP database is that follow-up data is only for 30 days post-operatively; surgical technique information is not collected including laterality and tissue

resection weight; and the database captures general complications and therefore does not capture surgery-specific complications such as delayed wound healing, haematomas, nipple necrosis or scarring. In a retrospective review of 938 patients at a single institution, Bauermeister and colleagues found smoking history was not associated with an increased risk of complications (Bauermeister et al., 2019).

Further studies have indicated that smokers are at increased risk of developing complications compared to non-smokers (Bartsch et al., 2007, Cunningham et al., 2005, Gravante et al., 2008, Srinivasaiah et al., 2014). Bikhchandani et al. found that smokers were 2.3 times more likely to develop a complication (Bikhchandani et al., 2007). Similarly, Schumacher and colleagues reported the incidence of wound complications to be 3.4 times higher in smokers (Schumacher, 2005). In a recent meta-analysis and systematic review on the role of smoking in plastic surgery elective procedures, Theocharidis and colleagues found that tobacco use is found to significantly increase the overall number of post-operative complications following breast reduction surgery (OR, 2.35; p < 0.001) (Theocharidis et al., 2018). Furthermore, the authors found that smoking was linked to an increased incidence of skin necrosis, infection, wound separation, delayed wound healing and re-operation. In a retrospective review of 173 consecutive patients who underwent bilateral reduction mammoplasty, Chan et al. found that smokers were 1.6 times more likely to have a wound healing problem, but interestingly patients who ceased smoking at least 4 weeks before surgery had the same complication rate as non-smokers. As a result, the authors suggested the introduction of urine nicotine testing at the preadmission clinic and prior to the operation to provide objective verification of patients' smoking history, minimize morbidity, and enable healthcare cost savings (Chan et al., 2006).

Body mass index

Numerous studies have evaluated the incidence of post-operative complications following reduction mammoplasty in the obese patient. However, reported findings have been somewhat inconsistent and contradictory, creating confusion as to whether surgery be postponed in the obese patient or whether pre-operative counselling regarding risk factors is more appropriate. Whilst many studies indicate that there is a higher rate of overall complications with increasing BMI, the majority of these complications are often minor and do not affect satisfaction and overall outcome of the patient (Bauermeister et al., 2019, Fischer et al., 2014, Nelson et al., 2014a, Shah et al., 2011).

Myung and Heo recently conducted a systematic review and meta-analysis in order to clarify the relationship between obesity and surgical complications after reduction mammoplasty (Myung and Heo, 2017). The review found 26 studies, 22 of which were retrospective, that reported surgical complication risk and patient body weight, 11 concluded that obesity is not a risk factor and 15

15
reporting that high BMI increases risk of surgical complications. A meta-analysis was conducted on 12 studies which met the quality criteria, including 6904 patients (3752 obese and 3152 nonobese), undergoing reduction mammoplasty. The authors found that obese patients had a higher relative risk of surgical complications (RR 1.38) and tissue necrosis (RR 2.01) and that the risk gradually increases with an increase in the severity of obesity. However, the authors concluded that the risk is not extremely high when compared to the risk for other types and surgeries and that there is no reason to exclude or to postpone surgery in obese patients (Myung and Heo, 2017).

Population based studies of reduction mammoplasty using the NSQIP database have all concurred on the finding that BMI is an independent risk factor for complications (Fischer et al., 2014, Hillam et al., 2017, Karamanos et al., 2015, Nelson et al., 2014a, Simpson et al., 2019). Furthermore, when stratified by WHO obesity classification, morbidly obese patients (BMI >40 kg/m²) were twice as likely as normal weight patents to experience surgical complications. Interestingly, obesity did not appear to be an independent risk factor for major surgical complications (deep infection and/or unplanned return to the operating room) (Fischer et al., 2014). In a study investigating obesity and early complications in 4545 reduction mammoplasty patients, Nelson et al. found a significant increase with increasing obesity class, with morbidly obese patients at highest risk of complications, particularly wound infection and dehiscence. The authors encouraged appropriate pre-operative counselling as opposed to declining surgery in the obese patient (Nelson et al., 2014a).

Age

As women age, the ability to heal and respond to injury lessens. Several studies have reported on the detrimental effect of age on outcomes following breast reduction surgery. In a retrospective study of 1192 consecutive patients undergoing reduction mammoplasty over a 10-year period, Shermak et al. proposed that women older than 50 years were more likely to experience infection (OR, 2.7; p = 0.003) and there was a trend towards wound healing problems and re-operative wound debridement. The authors hypothesised that hormonal deficiency may partially account for this finding (Shermak et al., 2011). Young and colleagues conducted a retrospective study of 9110 patients using the NSQIP database to identify any increased complication risk in patients aged 60 years and above and found that age does have an impact on outcomes and complication risk, with a higher rate of total complications and an increased risk of cerebral vascular accidents, myocardial infarction, and readmission (Young et al., 2019). In a prospective study 67 patients, the authors found that the probability of a patient developing a complication increases for older age patients (Srinivasaiah et al., 2014). In contrast, the BRAVO study found that delayed healing correlated inversely with age, with every year of increasing age decreasing the likelihood of having wound-healing complications by 7% (Cunningham et al., 2005). Interestingly, Roehl et al. found that complication rate was most frequent in those aged 30 to 39 years, although the authors could

not explain the reason for this finding (Roehl et al., 2008).

A meta-analysis of 16 studies demonstrated that an age of 50 years and older bears no relationship to complications in reduction mammoplasty compared to younger patients less than 50 years (Zhang et al., 2016). Additionally, Braig et al. conducted a study specifically evaluating the impact of increasing age on breast reduction surgery and found that age is not a contraindication for surgery, although noted that patient counselling is essential around age-related expectations and specific complications (Braig et al., 2016). The finding that age is not an independent risk factor for complications is supported by several other studies (Table 1.1) (Baltodano et al., 2017, Hanwright et al., 2013, Henry et al., 2009, Hillam et al., 2017, Hunter-Smith et al., 2012, Nelson et al., 2014b, Schumacher, 2005, Setala et al., 2009, Shah et al., 2011).

Co-morbidities and other risk factors

Whilst the majority of the research into risk factors for complications following reduction mammoplasty explores the effect of smoking and obesity, there are limited studies into the effect of co-morbidities such as diabetes mellitus. In a large retrospective review of records from 9110 women undergoing reduction mammoplasty between 2013 and 2015, Young and colleagues found that diabetes was found to be a significant risk factor for readmission, regardless of age (Young et al., 2019). In a prospective study of 65 women undergoing reduction mammoplasty, Eggert et al. found that diabetes was a risk factor for wound infection. However, small numbers limit this study to some degree (Eggert et al., 2009). In a larger retrospective study of 512 consecutive women who underwent bilateral breast reduction, diabetes was found to be an independent risk factor for necrosis of the areola (OR, 8.22; p = 0.003). The authors also found that a longer suprasternal notch-to-nipple distance (SND) and tissue resection weight were significant and independent predictors of postsurgical complications (Lewin et al., 2014). This is supported by the findings of Setala and colleagues (Setala et al., 2009) and Bauermeister and colleagues (Bauermeister et al., 2019) in a retrospective review of 273 and 938 consecutive cases, respectively. In contrast, in a retrospective analysis of 2142 reduction mammoplasties, Manahan et al. did not find associations with comorbidities including diabetes and hypertension. However, they found that a greater degree of pre-operative ptosis was associated with increased rates of fat necrosis (OR, 2.14; p = 0.03) and wound healing problems (OR, 2.6; p = 0.002). Nipple-to-sternal notch distances greater than 41cm were associated with infections (OR, 4.3; p = 0.02) and major wounds (OR 13.3; p = 0.04), while SND more than 43cm led to increased seroma rates (OR, 24.4; p = 0.006) (Manahan et al., 2015). Using the NSQIP database, Fischer et al. reported that diabetes, alcohol use, COPD and cardiovascular risk factors were not significantly associated with complications (Fischer et al., 2014). Unfortunately, Zhang and colleagues stated that it was not possible to evaluate diabetes in their meta-analysis of risk factors in reduction mammoplasty as there were too few related studies (Zhang et al., 2016). This reinforces the importance of further study into the impact of diabetes on

surgical outcomes following breast reduction. In a retrospective analysis of 94,140 plastic surgery cases comprising 17,058 reduction mammoplasties, Barcher and Ranza found that chronic steroid use was a significant risk factor for post-operative wound complications (OR, 2.2; p = 0.001) (Barcha and Ranzer, 2018). This finding is further supported by several other published studies (Baltodano et al., 2017, Hillam et al., 2017, Simpson et al., 2019).

Despite the risks of postsurgical complication, breast reduction surgery remains a beneficial intervention for symptomatic relief and improved quality of life in women suffering from symptomatic macromastia. Although there is contradictory evidence regarding the factors which are associated with an increased risk of complications, the overwhelming majority of studies ultimately support that reduction mammoplasty should be accessible, with the appropriate pre-operative counselling, to all women with symptomatic breast hypertrophy.

The clinical effectiveness of breast reduction surgery has been established by previous research studies. Likewise, the risks of post-operative morbidity following surgery have been highlighted. What is a far more complex concept and one that is not well reported in the literature is how these two fields relate to one another. That is, how much importance does the patient place on shortterm complications when considering long-term outcomes? A 'trade-off' in decision-making exists which is largely dependent upon an individual's perspective and what their priorities are in what are they willing to accept or give up in order to have something else (Gawande, 2015, McNutt, 2004). Trade-offs happen every day in health care. Clinicians have to make choices every day on the trade-off between positive outcomes and potential risks for their patients. They also may need to make trade-offs in following regulations of their organisation versus doing what they think is right for their patients. Patients themselves make trade-offs and have to choose between their own health care options that they often know very little about. Breast reduction surgery entails risks and sacrifices, but the expected benefits include improved long-term 'quality of life' and wellbeing. However, ultimately it is the patient alone who truly knows how they will feel about adverse outcomes and how much they are willing to compromise for long-term gain. It is anticipated that this thesis will provide quantitative evidence as to both the risks and benefits of surgery to assist women in making informed decisions prior to undergoing surgery.

High levels of satisfaction are reported by the vast majority of women following breast reduction surgery. Between 78% and 100% of patients who have undergone surgery report they would have the surgery again or would recommend it to others (Atterhem et al., 1998, Collins et al., 2002, Dabbah et al., 1995, Davis et al., 1995, Faria et al., 1999, Godwin et al., 2014, Godwin et al., 1998, Makki and Ghanem, 1998, Schnur et al., 1997). This is despite a reported complication rate of up to 53%; the majority of patients seem to accept the 'trade-off' between sometimes less than ideal outcomes and aesthetic results because the overall positive health benefits outweigh the short-term negative issues that may arise. For example, Godwin and colleagues conducted a study to

evaluate the difference between patient and surgeon opinion on the aesthetic outcome of reduction mammoplasty. Thirty-four women who were at least one-year post-surgery were asked to evaluate the appearance of their breasts. Patients' satisfaction with their outcome of surgery significantly exceeded that of the surgical panel. Scars were found to be the biggest issue for patients; 35% of women said the appearance of post-operative scars was 'unacceptable' and 53% of these patients underwent additional corrective procedures. Yet remarkably, 94% of women stated that they would have the surgery again or recommend it to a friend (Godwin et al., 1998). The functional benefits and long-term satisfaction gained by women following breast reduction surgery appear to compensate for less than perfect outcomes in the majority of women.

1.8 Health-related quality of life and patient-reported outcome measures

1.8.1 Health-related quality of life (HRQoL)

Surgical outcomes are traditionally measured by morbidity, mortality and rate of complications. Whilst these measures continue to be important, the aims of the majority of procedures performed in reconstructive surgery are to improve a patient's health-related quality of life (HRQoL) and improve or restore function. In 1948, the Constitution of the World Health Organization (WHO) defined health in its broader sense as "a state of complete physical, mental, and social well-being and not merely the absence of disease or infirmity" (WHO, 1947). Quality of life and, more specifically, 'health-related quality of life' (HRQoL) is a broad and complex concept for which there is no universally accepted definition. HRQoL relates to factors directly affecting an individual's health and wellbeing; it is multi-dimensional and describes the physical, role functional and emotional aspects of functioning and wellbeing. HRQoL is subjective and is therefore best reported from the patient's own perspective.

An important outcome measure of the impact of a health condition and the success of surgical intervention is the patient's own perception of the outcome. In reconstructive and aesthetic surgery, there is often a profound difference in the patient's assessment of their outcome when compared to that of the surgeon (Cano et al., 2009). Breast hypertrophy is not a life-threatening condition and quality of life outcome studies have become an important way to assess the beneficial effects of such medical interventions.

1.8.2 Measurement of HRQoL with patient-reported outcome measures

Patient-reported outcome measures (PROMs) are questionnaire-based and enable the patient to have a "voice" in the clinical setting. They measure HRQoL and examine the physical, psychological, emotional, social and functional outcomes of reconstructive surgery. This is in contrast to the traditional aesthetic assessment of outcome according to the surgeon. PROMs are increasingly becoming an important part of clinical practice in plastic surgery internationally for

several reasons. Firstly, to provide strong evidence to support that surgery improves function and quality of life. This information is valuable in an era of increasing restrictions on access to certain medical procedures. Secondly, the utilisation of outcome instruments enables the comparison to new or alternative surgical techniques, some of which are more costly, as to whether they are actually superior from a patient perspective. Thirdly, PROMs are being adopted by regulatory bodies which recognise the validity in capturing patient-centred data which can only be provided by PROMs (Cano et al., 2009, Voineskos et al., 2018).

There are several different types of PROMs: generic questionnaires that examine the quality of life and general wellbeing across patient and population groups, e.g. the Short Form-36 (SF-36); surgery-specific instruments are more sensitive to capture symptoms and outcomes specific to a condition, e.g. BREAST-Q; and preference-based measures such as the Short Form Six-Dimension (SF-6D) (derived from the SF-36), the EuroQol (EQ-5D) and the Health Utilities Index (HUI) which generate utility scores for the calculation of quality-adjusted life years (QALYs) for application in economic evaluation (Figure 1.2).

Generic instruments			
	Short Form-36 (SF-36)		
	Multidimensional Body-Self Relations Questionnaire (MBSRQ)		
	Rosenberg Self-Esteem Scale		
Surgery-specific instruments			
	BREAST-Q		
	Breast-Related Symptoms Questionnaire (BRSQ)		
	Breast Reduction Assessed Severity Scale (BRASS)		
	Modified Breast Evaluation Questionnaire (mBEQ)		
Preference-based measures			
	Short Form Six-Dimension (SF-6D)		
	Health Utilities Index (HUI)		
	15-dimensional (15D)		
	EuroQol EQ-5D (EQ-5D)		

Figure 1.2. Common instruments to measure HRQoL outcomes of breast surgery

1.8.3 Challenges of interpreting HRQoL data

The multi-dimensional and subjective nature of HRQoL data can make interpretation of change in scores difficult. Many factors can predict an individual's HRQoL and it is unlikely that all of these will be captured in any one or several HRQoL instruments, no matter how comprehensive they may be. Other contributing factors, both known and unknown, and life events will undoubtedly affect responses. This becomes increasingly relevant in longitudinal HRQoL research studies

conducted over multiple time points.

An important factor in interpretation of HRQoL data is what is the "minimal important difference" (MID)? The MID, from the patient-perspective, can be defined as 'the smallest difference in score in the domain of interest which patients perceive as beneficial and which would mandate, in the absence of troublesome side effects and excessive cost, a change in the patient's management' (Jaeschke et al., 1989, p. 408). Whilst is useful to obtain quantitative scores to measure and compare quality of life before and after a health intervention, the value is somewhat limited if we do not know what is considered a meaningful change in the context of the clinical setting. For example, there is no known universal magnitude to establish a clinically important difference on the SF-36 subscales. However, a rule of thumb of a 10-point change on 100-point quality of life scales has been recommended (King, 1996). Estimates of minimal important clinical difference are particularly important when HRQoL instruments are used as the primary measure of outcome in randomised clinical trials and for power calculations in clinical research. However, this threshold may vary depending on the context of disease, disease severity, by population characteristics and by type of outcome instruments; highlighting the importance of estimating MID based on the specific clinical setting (Jayadevappa et al., 2017). Two broad strategies for estimation of MID are commonly used to interpret changes in HRQoL scores following a medical intervention or treatment: (1) anchor-based; and (2) distribution-based approaches (Crosby et al., 2003, King, 2011, Lydick and Epstein, 1993). Anchor-based methods express changes in HRQoL scores by linking specific scales to known variables of clinical relevance or to patient- or clinician-derived ratings of change (Guyatt et al., 2002, Jaeschke et al., 1989, Musoro et al., 2018). Distributionbased methods are based on the statistical distribution of HRQoL scores (Wyrwich et al., 2005), and are often suggested to support anchor-based estimates (Revicki et al., 2008). To date, there are no established estimates of what represents a clinically meaningful change for any of the validated generic or condition-specific patient-reported outcome measures utilised in studies of women with breast hypertrophy undergoing breast reduction surgery.

Another challenge relating to the interpretation of quality of life scores in longitudinal outcome studies is the "response shift" phenomenon. Schwartz and Sprangers define this as when individuals undergo a change in health state they may change their "internal standards, values, or the conceptualization of quality of life" (Schwartz and Sprangers, 1999). That is, due to the self-report and subjective nature of HRQoL measurements, the mechanism by which people assess or quantify their HRQoL could change over time. These changes may be reflected in perceived quality of life scores to be greater or smaller than what they actually are. Response shift is a natural process that has the potential to distort the interpretation of change in HRQoL scores over time and affect what represents a 'true' change. For example, patients undergoing cancer treatment may adapt to their increased symptom level or impaired HRQoL that the disease or treatment cause, and patients' internal standard of measurement may be changed. Response

shifts in this situation may therefore occur during the course of the disease, or during the course of a treatment (Brazier et al., 2017, Sprangers, 1996).

An issue in longitudinal HRQoL studies is the likelihood of missing data and loss to follow-up of participants. In observational studies over time, participants may miss appointments or fail to complete questionnaires at different study time points. The interpretation of the change in HRQoL scores can be significantly affected in this scenario. For example, missing HRQoL data in cancer trials can be problematic as it is likely that the data are not missing randomly; if patients whose condition has deteriorated are not completing their HRQoL questionnaires, HRQoL may be overestimated and may not capture the full side-effects of treatment. Statistical approaches are employed to handle missing data, from simple approaches such as complete case analysis, to more sophisticated imputation-based approaches (Brazier et al., 2017, pp. 271-275).

In summary, the interpretation is HRQoL scores in research studies should be carefully made with respect to these challenges and biases.

1.9 Review of HRQoL research in breast reduction surgery

Research studies have highlighted that breast reduction surgery is of significant benefit to women with macromastia, in terms of improving health-related quality of life (HRQoL). In 1996, Klassen and colleagues introduced the use of validated and self-reported instruments for the assessment of outcomes following breast reduction surgery (Klassen et al., 1996). Validated HRQoL tools such as the SF-36 and the BREAST-Q questionnaire have been valuable for measuring the improvement in quality of life following breast reduction surgery. However, there have been no outcome studies performed in Australian women, and no comprehensive comparison between physical assessment and patient-reported outcomes.

A systematic review of the literature of studies conducted between 1966 and 2006 by Pusic et al., exploring the use of patient-reported health outcomes instruments in breast surgery studies, identified 227 health outcomes questionnaires. Of these, only one patient-reported outcomes instrument, the Breast-Related Symptoms Questionnaire (BRSQ), was developed and validated for use in the cosmetic and reconstructive breast surgery population. The authors concluded that there was a lack of availability of validated, condition or surgery specific instruments for breast surgery and that whilst generic instruments play an important complementary role, they are not sensitive enough to detect the full spectrum of changes after breast surgery (Pusic et al., 2007a). To fill this gap, Pusic and colleagues developed the four modules of the BREAST-Q (augmentation, reconstruction, mastectomy and reduction) in order to study the unique outcomes of breast surgery from the patients' perspective (Pusic et al., 2009). Subsequently, a systematic review of patient-reported outcome measures used in 95 studies of reduction mammoplasty patients was conducted; the authors emphasizing that using a combination of validated generic

measures and breast-specific surveys is preferred and provide a more accurate assessment of patient outcomes (Lonie et al., 2019).

1.9.1 Generic PROMs

1.9.1.1 Short Form-36

The Short Form-36 Health Survey (SF-36) is a generic instrument which has broad applicability and is the most commonly used instrument globally to measure the general health across population and patient groups. The SF-36 was developed as part of the Medical Outcomes Study (MOS), with the main objective of developing a measure to monitor the outcomes of medical care (Ware and Sherbourne, 1992). An advantage of the SF-36 is that it can be used to compare the burden of various disease states and interventions, for comparison to normative populations, and individual responses can be transformed with a SF-6D scoring algorithm to facilitate economic evaluations (Brazier et al., 2017, Brazier et al., 2002, Norman et al., 2014). In a systematic review of PROMs in breast surgery, the SF-36 was found to be the most widely used instrument in cosmetic and reconstructive breast surgery studies. However, the authors stated that whilst generic instruments play an important complementary role in patient-reported outcome studies, they may not be sensitive or responsive enough to detect changes as a result of surgery or capture all aspects of outcome after breast surgery (Pusic et al., 2007a).

Perez-Panzano et al. conducted a prospective study of quality of life and patient satisfaction after breast reduction surgery in a short- and long-term evaluation of 121 patients with symptomatic macromastia (Perez-Panzano et al., 2016). Symptom-specific guestionnaires were used to assess patient satisfaction and the SF-36 was used to assess quality of life. They found an improvement in quality of life in all SF-36 domains in the long-term at one year after surgery. At the one-month assessment, all of the domain scores were significantly higher except physical functioning and social functioning, which did improve but not significantly. Role Physical was the only domain which deteriorated one month after surgery, although this was regarded as foreseeable due to the convalescence period after surgery (Perez-Panzano et al., 2016). Blomqvist and colleagues conducted a prospective study to assess the HRQoL in 49 women with breast hypertrophy who underwent reduction mammoplasty (Blomgvist et al., 2000). They found that reduction mammoplasty resulted in significantly improved quality of life in all eight domains of the SF-36 post-operatively. They also found that results were similar at 6- and 12-months post-surgery. indicating long-term improvement. Furthermore, they found that these patients were normalised in health-related quality of life, with no statistically significant differences between the patients and age-matched control population (Blomqvist et al., 2000). Other studies had similar findings of improvement in HRQoL across all domains of the SF-36 when assessing outcomes beyond the initial convalescence period following surgery (Behmand et al., 2000, Collins et al., 2002, Klassen et al., 1996, Mello et al., 2010). Thoma et al. conducted a prospective study of 52 patients using the SF-36 pre-operatively and again at 1-, 6- and 12-months after breast reduction surgery. The

authors found that the improvement from 1 month after surgery was maintained to 1 year after surgery (Thoma et al., 2007). However, the authors only reported physical and mental summary scores as opposed to all SF-36 domains. Iwuagwu et al. conducted a randomised-controlled trial of a surgical group compared to a non-surgical group who received physiotherapy. Using the SF-36, they found that surgery resulted in considerable improvement in physical and mental summary scores whilst conservative treatment was not effective in providing relief from symptoms of breast hypertrophy and improving HRQoL (Iwuagwu et al., 2006b).

Despite the apparent breadth of literature on HRQoL in women undergoing breast reduction surgery using the SF-36 at various follow-up time points, there remain comparison problems and gaps in the literature. This study will address some of these limitations by being a prospective, longitudinal study with adequate participant numbers with control groups for comparison and the collection of HRQoL data as the primary measure of outcome using several validated patient-reported outcome measures.

1.9.1.2 MBSRQ

The Multi-Dimensional Body-Self Relations Questionnaire (MBSRQ) is a validated patient-reported outcome measure for the assessment of body image. The full, 69-item version consists of seven factor subscales: (1) appearance evaluation; (2) appearance orientation; (3) fitness evaluation; (4) fitness orientation; (5) health evaluation; (6) health orientation; and (7) illness orientation. There are also three additional MBSRQ subscales: body areas satisfaction scale; (2) overweight preoccupation; and (3) self-classified weight (Cash and Pruzinsky, 1990). Two forms of the MBSRQ are available, the full version and the MBSRQ-Appearance Scales (MBSRQ-AS).

The MBSRQ has been used in combination with additional PROMs in a number of studies investigating breast hypertrophy and the outcome of breast reduction surgery (Collins et al., 2002, Cunningham et al., 2005, Kerrigan et al., 2001, Thoma et al., 2013, Thoma et al., 2005). Comparison studies of women with breast hypertrophy who are seeking breast reduction surgery, hypertrophy control subjects, and normal control subjects have revealed that, pre-operatively, surgical candidates scored worse on the appearance evaluation of the MBSRQ questionnaire than the two control groups. In addition, the hypertrophy control subjects scored worse than the normal control subjects did. The authors concluded that breast hypertrophy has a significant impact on women's quality of life as measured by the MBSRQ and other validated self-report instruments (Collins et al., 2002, Kerrigan et al., 2001). In a prospective study, Thoma et al. found that the mean MBSRQ score before surgery was 2.3 and at 12 months after surgery had improved to 3.2, nearing the adult female normative score of 3.36 (Thoma et al., 2007).

A systematic review of patient-reported outcome instruments in cosmetic, functional and reconstructive breast surgery, conducted by Pusic and colleagues, criticised the use of the MBSRQ as a standalone measure of outcome in this patient population as the questions do not

address relevant body image concerns such as scarring and breast size. However, the authors argued that generic instruments such as the MBSRQ play an important complementary role in outcome studies (Pusic et al., 2007a).

1.9.1.3 Rosenberg Self-Esteem Scale

The Rosenberg Self-Esteem scale (RSE) is a widely-used self-report measure of self-esteem (Rosenberg, 1965). It is comprised of 10-items answered on a four-point Likert scale. The scale ranges from a score of 0 to 30, in which higher scores indicate higher self-esteem. Klassen and colleagues (Klassen et al., 1996) first described using the RSE to demonstrate a significant improvement in self-esteem after breast reduction surgery. Similar findings have since been reported in other outcome studies (Hermans et al., 2005, Kececi et al., 2015, Mello et al., 2010, Miller et al., 2005, O'Blenes et al., 2006, Shakespeare and Cole, 1997).

1.9.2 Condition or surgery-specific PROMs

1.9.2.1 BREAST-Q

The BREAST-Q is a condition specific patient-reported outcome questionnaire that was developed and validated to measure patient satisfaction and quality of life following cosmetic, functional and reconstructive breast surgery (Pusic et al., 2009). Individual modules were generated to assess unique outcomes related to specific types of breast surgery; breast reconstruction, breast augmentation, mastectomy, and breast reduction. The BREAST-Q underwent full development and validation using the three-stage approach described previously by Cano et al., including item generation (individual questions are developed from patient interviews, experts and literature), item reduction (field test to finalise questions), and psychometric evaluation (final questionnaire administered to a large population to assess data quality, validity, reliability and responsiveness) (Cano et al., 2009). The conceptual framework of the BREAST-Q was developed following literature review and qualitative interviews with breast surgery patients in order to identify six key themes and form the framework of patient satisfaction and quality of life among breast surgery patients (Figure 1.3) (Klassen et al., 2009, Pusic et al., 2009).

Image removed due to copyright restriction. Original can be accessed at:

https://journals.lww.com/plasreconsurg/Abstract/2009/08000/Development_of_a_New_Patient_R eported_Outcome.1.aspx

Figure 1.3. BREAST-Q conceptual framework Source: (Pusic et al., 2009)

Satisfaction with breasts — Relates to women's satisfaction with their breasts with reference to size, shape, symmetry, cleavage, positioning, how natural their breasts look and feel both clothed and unclothed, and how their breasts sit in proportion to the rest of their body. There are also post-operative only items such as the location and appearance of scars.

Satisfaction with outcome — Relates to an overall level of satisfaction with the outcome of breast surgery.

Psychosocial wellbeing — Relates to the effect of breast surgery on their psychosocial wellbeing with items that ask about body image and confidence in social settings, emotional health and self-esteem.

Sexual wellbeing — Relates to the way a woman's breast condition and surgery impact on her sexual life and body-image issues when clothed and unclothed.

Physical wellbeing — Relates mainly to chest and upper body symptoms and how these impact on physical function and participation in activities before and after breast surgery.

Satisfaction with care — Relates to patient satisfaction with the overall process of care. This domain represents a key area in the patient's overall assessment of the surgery. There are three main subthemes: satisfaction with information pre-operatively; satisfaction with the care provided by the plastic surgeon; and satisfaction with office staff and other members of the medical team.

Satisfaction with information — Items cover possible risks and complications, healing and recovery time, and breast appearance and scarring. The patient's relationship with the plastic surgeon was a key area and items cover whether or not they felt comfortable, had their questions answered,

understood what they wanted, and whether they felt the surgeon had provided adequate follow-up after surgery.

A review of the literature on the use of the BREAST-Q in surgical research from 2009 to 2015, conducted by Cohen and colleagues, yielded 49 peer-reviewed articles that met the inclusion criteria (Cohen et al., 2016b). Of these, four references used the BREAST-Q Reduction module questionnaire (Carty et al., 2012, Coriddi et al., 2013, Gonzalez et al., 2012, Gurunluoglu et al., 2013). In the largest of these studies, Carty and colleagues conducted a retrospective study of 279 women who had undergone breast reduction surgery between 1995 and 2007. They found that older patients were more satisfied with their breasts than younger patients after reduction mammoplasty, and they also found no difference in satisfaction with outcome and/or breasts based on surgeon's age or experience (Carty et al., 2012). Gonzalez and colleagues conducted a 10year retrospective analysis of patient satisfaction and surgeon experience in 178 women who had previously undergone breast reduction surgery. The authors found that breast reduction surgery had a positive impact on quality of life and that no difference was found in satisfaction with outcome by quantity of breast tissue removed. However, the limitations of this study were a low response rate, the lack of pre-operative BREAST-Q data, and that only the item responses (i.e. the raw data) rather than scored BREAST-Q data was used in the analysis, which was not in line with the design or the intended use of the instrument (Gonzalez et al., 2012). Retrospective studies are, by their very nature, unable to provide pre-operative HRQoL levels and therefore assume that all respondents had an approximately equal baseline level of satisfaction and wellbeing before surgery. In addition, these studies lack the longitudinal tracking of patients following surgery and often include a broad range of follow-up time following surgery that may alter patient satisfaction responses.

Coriddi and colleagues completed the only prospective study that was included in the systematic review by Cohen et al (Cohen et al., 2016b, Coriddi et al., 2013). Using the BREAST-Q before and 6 weeks after surgery they found that breast reduction significantly improves satisfaction with breast, psychosocial, sexual, and physical wellbeing, and that overall satisfaction with outcome is highly correlated with satisfaction with breasts after surgery. However, this study was limited by a small sample size and, furthermore, only a small subset of participants completed both pre- and post-operative questionnaires.

Cogliandro and colleagues in Italy conducted the largest outcome study using the BREAST-Q in 414 breast reduction patients (Cogliandro et al., 2017a). However, this study was also retrospective in design and therefore even though the authors reported a high level of patient satisfaction following surgery, there was no pre-operative data to serve as a baseline for comparison. In addition to this, there was a very broad range of time since surgery and this could have affected patient responses. Following on from this study, the authors presented data in a

short letter from a prospective study of 156 women who completed the BREAST-Q pre- and 6 months post-surgery. Whilst the authors reported a statistically significant improvement across three of the four common BREAST-Q domains, an unusual finding in this study was the significant decrease in physical wellbeing scores following surgery (Cogliandro et al., 2017b). In contrast, prospective studies by Cohen and colleagues (Cohen et al., 2016a) and Cabral and colleagues (Cabral et al., 2017a) found a significant improvement in all four of the BREAST-Q scales. Whilst the study by Cohen et al. has the advantage of being prospective in nature, the results are limited by a small sample size and furthermore, not all of these patients completed the BREAST-Q at each study time point. During the course of this thesis, a prospective study of outcomes following reduction mammoplasty using the BREAST-Q was published. The authors reported a significant improvement in all four BREAST-Q domains when comparing pre- to one year post-surgery in 156 women in Sweden (Lewin et al., 2019). However, a limitation of this study was the restrictive inclusion criteria as per the Swedish national guidelines for breast reduction (BMI <25 kg/m² if younger than 50 years and BMI <27 kg/m² if older than 50 years; plus breast volume \geq 800 cm³ with physical problems) and a low response rate among women who answered both the pre-operative and post-operative questionnaires (45%).

Despite there being several studies published using the BREAST-Q to assess patient outcomes following breast reduction surgery, there still remains the need for high quality, longitudinal, and most importantly, prospective studies. Further to this, there have been no outcome studies using the BREAST-Q Reduction questionnaire in Australian women.

1.9.2.2 BRSQ

Breast-Related Symptoms Questionnaire (BRSQ) is a validated 13-item scale uniquely developed to measure symptoms associated with breast hypertrophy. The focus of the instrument is on breast symptoms and physical symptoms and/or limitations due to breast hypertrophy and does not include general QoL issues or satisfaction with breast appearance. The BRSQ was developed for use in the BRAVO study and subsequently underwent further validation studies (Kerrigan et al., 2001). The instrument is scored from 0 (corresponds to having all 13 symptoms all the time) to 100 (corresponds to no symptoms). In a prospective multi-centre study using the BRSQ, Valtonen et al. found that breast reduction surgery significantly improved breast-related symptoms and HRQoL (Valtonen et al., 2014). Thoma and colleagues utilised the BRAVO instruments (BRSQ, MBSRQ, SF-36) to assess HRQoL before and at 1, 6, and 12 months after surgery. The authors also found that breast-related symptom scores significantly improved after surgery (Thoma et al., 2007). This is supported by the findings of other studies (Collins et al., 2002, Kerrigan et al., 2002, Lewin et al., 2019).

In a randomised-controlled trial, the Finnish version of the BRSQ was used to compare outcomes of a surgical group versus a control group. The authors found that the surgical patients had

significantly less reported breast-associated symptoms or pain when compared with the nonsurgical group at the six month follow-up (Saariniemi et al., 2008a).

Whilst well-validated, the BRSQ does not capture the full spectrum of HRQoL related to breast reduction surgery as it fails to address issues around psychosocial and sexual functioning, body image, aesthetics, satisfaction with care and overall quality of life.

1.9.2.3 BRASS

The Breast Reduction Assessed Severity Scale (BRASS) is a 39-item questionnaire using the five point Likert scale for measuring the burden of breast hypertrophy (Sigurdson et al., 2007a). The instrument comprises five subscales: (1) physical implications; (2) poor self-concept; (3) body pain; (4) negative social interactions; and (5) physical appearance. In a test of criterion validity, the BRASS subscale analysis showed moderate though significant correlations with conceptually similar domains from the Short Form-36 and the Rosenberg Self-Esteem scale and correlated strongly with the Breast-Related Symptoms (BRSQ) score. Despite its reliability, the instrument has been criticised for focusing solely on capturing physical symptoms and psychological issues and the absence of items or scales that measure post-operative outcomes including scarring and pain (Reavey et al., 2011).

Kececi and colleagues conducted a prospective cohort study of 94 patients with breast hypertrophy who underwent breast reduction and found a significant post-operative improvement in all domains of the BRASS (Kececi et al., 2015). However, this study utilised a Turkish translation of the BRASS, which had been validated in a group of women presenting with breast hypertrophy (Kececi et al., 2013). However, it has been argued that this version of the BRASS has limited validity due to by the absence of a post-operative cohort to ensure that the questionnaire is responsive to change and lacks the additional dimensions to address surgery-specific post-operative issues which are included in instruments such as the BREAST-Q (Kerrigan, 2013).

1.9.2.4 mBEQ

The Breast Evaluation Questionnaire (BEQ) was initially developed and validated on 1244 women seeking breast augmentation surgery and is comprised of 55-item scale (Anderson et al., 2006). The instrument addresses three subscales of breast comfort and satisfaction: (1) Comfort not fully dressed; (2) comfort fully dressed; and (3) satisfaction. The scores range from 0 (very dissatisfied/uncomfortable) to 100 (very satisfied/comfortable). However, the applicability in using the BEQ to assess outcomes following other types of breast surgery as recommended by the developers, such as breast reduction and reconstruction, has been questioned (Pusic et al., 2007b). Subsequently, a modified Breast Evaluation Questionnaire (mBEQ) version was developed and validated to specifically measure the outcomes of breast reduction surgery (Lewin et al., 2018). The authors state that the mBEQ assesses important aspects of breast hypertrophy and may provide additional information about psychosocial morbidity in comparison with the

BREAST-Q Reduction module. Lewin and colleagues presented findings of a prospective study using the modified mBEQ to demonstrate that reduction mammoplasty improves psychosocial issues associated with breast hypertrophy (Lewin et al., 2019).

1.9.3 Preference-based measures

Preference-based measures provide a single index score that quantifies a patient or population's state of health-related quality of life. These are typically represented on the quality adjusted life years (QALY) scale whereby a utility score of 0 corresponds to being dead and a utility score of 1 corresponds to full health (Brazier et al., 2017). Utilities facilitate the classification of the HRQoL associated with different diseases or conditions on a common scale and are especially useful in economic evaluations of health interventions. The Short Form Six-Dimensional (SF-6D), 15-dimensional (15D), EuroQol-5D (EQ-5D), Quality of Well-Being Scale Self-Administered (QWB-SA), and Health Utilities Index Mark 2 (HUI2) and Mark 3 (HUI3) are commonly used generic preference-based instruments.

Preference-based measures of quality of life, were first described in the area of plastic surgery almost two decades ago. Kerrigan and colleagues reported a utility score for living with breast hypertrophy of 0.86, which was lower (i.e. worse) than expected and was comparable with the reported burden of living with other health conditions, such as moderate angina (0.90) and a kidney transplant (0.84) (Kerrigan et al., 2000). Similarly, Thoma et al, using the HUI3, reported a HRQoL utility score of 0.76 before surgery and 0.89 at one year after surgery, with a difference of 0.13 (Thoma et al., 2007). The burden of breast hypertrophy is comparable to reports for other chronic conditions: 0.77 for heart disease; 0.78 for arthritis; 0.78 for epilepsy; 0.79 for diabetes; and 0.81 for back problems (Mittmann et al., 1999).

In a prospective study of 80 patients using the 15D instrument before and 6 months after surgery, Tykka et al. found that mean HRQoL score increased from 0.916 to 0.939 (Tykka et al., 2010). In another study, Saariniemi et al. also conducted a cost-utility analysis using the 15D pre-operatively and at a mean follow-up time point of 4 years on 73 patients who underwent reduction mammoplasty. They describe a utility score of 0.847 at baseline and 0.930 at follow-up, with a mean improvement of 0.083 (Saariniemi et al., 2012). The minimal clinically important difference in the 15D and the HUI is \geq 0.03, highlighting that these studies all agree that reduction mammoplasty significantly improves HRQoL (Drummond, 2001, Sintonen, 2001). Araujo described a considerably lower pre-operative utility index of 0.63, increasing to 0.73 six months postoperatively, using the SF-6D (Araujo et al., 2014).

There have been no utility studies performed on women with breast hypertrophy in Australia in order to compare with the burden with other health conditions and to perform a health economic evaluation of breast reduction surgery.

1.9.4 Factors influencing the HRQoL of women following breast reduction surgery

Studies have shown that patients express a high level of satisfaction and improved quality of life following breast reduction (Carty et al., 2012, Cohen et al., 2016a, Coriddi et al., 2013, Davis et al., 1995, Gonzalez et al., 2012, Makki and Ghanem, 1998, Perez-Panzano et al., 2016, Scott et al., 2005). However, there are several factors which have the potential to influence the level of change in HRQoL in women with breast hypertrophy following surgery: body mass index; tissue resection weight; age; surgical technique; surgical complications; time since surgery; pre-operative breast symmetry; and degree of hypertrophy.

In the prospective study of patents undergoing breast reduction surgery using the BRAVO instruments, a positive relationship was found between breast resection weight and body mass index (Thoma et al., 2007). However, the authors established that the patients' body mass index did not affect the change in HUI scores. This demonstrates that women with breast hypertrophy of all body weights, obese or non-obese, can benefit from reduction surgery. The authors also found that tissue resection weight was not significantly associated with the change in HUI scores. That is, all women benefit from surgery regardless of the amount of tissue removed at surgery (Thoma et al., 2007). This is supported by Collins et al. in the finding that all patients benefited regardless of body weight or tissue resection weight (Collins et al., 2002). In a retrospective chart review, Wagner et al. also found that there was no correlation between BMI or the volume of tissue resected and the relief of symptoms (Wagner and Alfonso, 2005). Guemes and colleagues evaluated the effect of reduction mammoplasty on quality of life using the SF-36 in obese patients compared with non-obese. They found that quality of life significantly improved following surgery in both obese and non-obese groups (Guemes et al., 2016). This is supported by the findings of Eggert and colleagues (Eggert et al., 2009).

In 2017, a retrospective study was conducted to determine if the quantity of tissue removed at surgery had any influence on patient satisfaction as measured by the BREAST-Q; the researchers found that quality of life and satisfaction were similar irrespective of resection weight, BMI, and surgical technique (Menéndez-Cardo et al., 2017). Cabral and colleague's study also supported these findings and found no association between BMI, age or resection weight on BREAST-Q satisfaction and wellbeing scores (Cabral et al., 2017a). In addition, overall satisfaction with outcome has been shown to be associated with satisfaction with breast appearance, further supporting the importance of thorough pre-operative counselling in relation to patients ideals on post-operative breast size and shape (Cabral et al., 2017a, Coriddi et al., 2013).

Cogliandro and colleagues conducted the largest study of 414 participants using the BREAST-Q Reduction module. They found that participants who presented with severe asymmetry and breast hypertrophy were more satisfied than others were. Limitations of this study include a retrospective

design and lack of pre-operative baseline BREAST-Q data (Cogliandro et al., 2017a).

In a retrospective study, Carty and colleagues found that BREAST-Q 'satisfaction with breasts' and 'satisfaction with outcome' scores were unaffected by gains in operative efficiency with increasing surgeon experience (Carty et al., 2012). They also found that patients older than 40 years demonstrated significantly higher satisfaction with breasts scores than younger patients. This is believed to be due to a variety of factors including more realistic expectations following surgery and that older patients have lived with symptoms of macromastia for a longer duration and therefore may find a greater degree of relief. Furthermore, this study demonstrated that patients who had post-operative skin necrosis demonstrated significantly reduced satisfaction scores. In contrast, studies have found no relationship between post-operative complications and satisfaction scores using the SF-36 and MBSRQ instruments (Cunningham et al., 2005). A recent article by Cohen et al. conducted the only study using the BREAST-Q to determine the temporal relationship on satisfaction and wellbeing scores (Cohen et al., 2016a). In a study of 20 patients, they found that HRQoL scores significantly improved following reduction mammoplasty and that this was maintained during the post-operative period.

Whilst a large number of studies have investigated factors that influence the satisfaction and HRQoL outcomes following reduction surgery, there exists a research gap. There have been no outcome studies performed in Australian women, and no studies to date have provided a comprehensive comparison between physical assessment of body shape and patient-reported outcomes. In addition, the quantity of prospective studies in the current literature with sufficient participant numbers for subgroup analyses is lacking.

1.9.5 Comparison of women with breast hypertrophy with the general population

Research that describes the health-related quality of life of women with symptomatic breast hypertrophy who undergo breast reduction surgery is useful in its own right. However, the comparison with scores from a reference population, such as normative data from the general population, provides a clinical reference point and enables a better understanding of the health burden of breast hypertrophy and the success of breast reduction surgery. The availability of normative values provides a valuable benchmark and allows findings from research studies to be placed into context for what is considered 'normal' for a given population. Population-based reference data that can be further categorised according to variables such as gender and age demographics is particularly useful for describing populations. However, normative data is not readily available for many PROMs.

Normative values for the general population are available for the original Short-Form 36 (SF-36) in Australia (Australian Bureau of Statistics, 1997, Watson et al., 1996) and in numerous countries worldwide as part of the International Quality Of Life Assessment (IQOLA) project including

Denmark, France, Germany, Italy, Japan, the Netherlands, Norway, Sweden, Spain, the United Kingdom, the United States (Gandek et al., 1998), and in Canada (Hopman et al., 2000). Population-based normative data for the revised "international version" of the SF-36, the SF-36 Version 2 (SF-36V2), are available for Australia (Hawthorne et al., 2007, Marin et al., 2009), the United States (Maglinte et al., 2012, Ware et al., 2000), New Zealand (Frieling et al., 2013), Sweden (Taft et al., 2004), and the United Kingdom (Jenkinson et al., 1999). Normative values for condition-specific HRQoL instruments may be advantageous over those from a more generic instrument as they focus on relevant disease and treatment-specific issues. The recent publication of normative data for the most commonly used breast surgery-specific instrument, the BREAST-Q, for the United States population provides a valuable comparison for assessing the outcome of reconstructive breast surgery (Mundy et al., 2017a, Mundy et al., 2017b, Mundy et al., 2017c). Despite some recognised limitations of the study in regard to population diversity, this HRQoL normative data provides a benchmark or reference point and provides a greater context to the burden of disease and the effectiveness of surgical intervention when interpreting HRQoL scores. Additional normative data from a more diverse and different population, such as that of the Australian population, would further enhance the knowledge for ongoing and future outcome studies in reconstructive breast surgery. Furthermore, evidence has shown that there are potentially important differences when comparing the HRQoL within the Australian population to other populations (Frieling et al., 2013, Hawthorne et al., 2007). This highlights the importance of utilising country-specific HRQoL population norm data for accurate comparison in outcome studies wherever possible. The availability of normative breast-related quality of life data in Australian women is lacking and is an important area for further research and investigation.

Evidence from several international research studies demonstrates that women with symptomatic breast hypertrophy have a considerable health deficit and impaired quality of life compared to women in the general population (Blomqvist and Brandberg, 2004, Kececi et al., 2015, Klassen et al., 1996, Miller et al., 2005, Mundy et al., 2017c, O'Blenes et al., 2006, Saariniemi et al., 2008b, Shakespeare and Cole, 1997). These studies also report that surgical intervention provides symptomatic relief and improves health-related quality of life to such an extent that they 'normalise' post-operatively to that of the general population. For example, O'Blenes and colleagues conducted a prospective study to assess HRQoL in 57 women pre-operatively and at 6- and 21.5- months following reduction mammoplasty. When comparing to age-matched, female Canadian normative SF-36 scores, the authors found that health deficits were eliminated at 6 months following surgery and showed a normalisation effect within the 21.5 months after surgery (O'Blenes et al., 2006). This finding is supported by other published studies comparing HRQoL in women undergoing breast reduction surgery to population norms using generic HRQoL instruments (Behmand et al., 2000, Blomqvist and Brandberg, 2004, Blomqvist et al., 2000, Collins et al., 2002, Eggert et al., 2009, Faria et al., 1999, Hernanz et al., 2016, Kececi et al., 2015, Kerrigan et al.,

2001, Klassen et al., 1996, Lewin et al., 2019, Shakespeare and Cole, 1997). In the publication by Mundy and colleagues in 2017, a comparison was made between the generated United States normative BREAST-Q Reduction module data and previously published pre- and post-operative data by Coriddi and colleagues (Coriddi et al., 2013, Mundy et al., 2017c). The authors highlighted the health burden and reduced quality of life associated with breast hypertrophy, and in turn demonstrated the success of breast reduction surgery to increase breast-related HRQoL to normative levels. Notably, all of these studies comparing HRQoL in women with breast hypertrophy and following breast reduction surgery have been conducted overseas, predominantly in the United States (Behmand et al., 2000, Collins et al., 2002, Coriddi et al., 2013, Kerrigan et al., 2001, Mundy et al., 2017c), Canada (Miller et al., 2005, O'Blenes et al., 2006) or Sweden (Blomqvist and Brandberg, 2004, Blomqvist et al., 2000, Eggert et al., 2009, Lewin et al., 2019).

1.10 Assessment of breast volume and body shape

1.10.1 Breast volume measurement

A variety of techniques have been described in the literature to fulfil the need for an accurate and objective measurement of breast volume in the clinical setting: magnetic resonance imaging, mammography, plaster casting, Grossman-Roudner plastic cups, water-displacement, anthropometric measurement and 3D surface imaging (Bulstrode et al., 2001, Howes et al., 2017, Kovacs et al., 2007, Kwong et al., 2020, Lee et al., 2016, Longo et al., 2013, Losken et al., 2005, Muslu et al., 2019, Qiao et al., 1997, 1986, Sigurdson and Kirkland, 2006, Smith et al., 1986, Wang et al., 2019, Wesselius et al., 2018). The accurate and objective assessment of breast volume plays an important role in breast reduction, reconstruction, developmental asymmetry and augmentation. In women with breast hypertrophy this is important for pre-operative planning and may aid intraoperative decision-making regarding the amount of tissue to be taken from each breast to achieve breast symmetry, or in cases where removal of a minimum amount of breast tissue is required to justify proceeding to surgery (Boukovalas et al., 2019, Descamps et al., 2008, Dvoracek et al., 2019, Eder et al., 2013, Kocak et al., 2011, Sommer et al., 2002, Wamalwa et al., 2018, Wampler et al., 2019). Accurate estimation of breast volume and predicted tissue resection weight also promotes improved counsel to the patient and provides a valuable guide to training surgeons.

Traditionally, the pre-operative measurement of breast volume and the amount of tissue to be resected in breast reduction surgery is estimated by jugular notch to nipple distance or other measurements, which are not necessarily well validated. In a situation where a woman is undergoing post-mastectomy breast reconstruction the most accurate method for measuring breast tissue volume for reconstruction is the Archimedes method of water displacement of the mastectomy specimen (Yip et al., 2012). In terms of measuring breast tissue volume in the intact state such as candidates for breast reduction surgery, MRI is reported as being the most accurate.

However, MRI is expensive and not accessible to all. Three-dimensional laser scanning has been demonstrated to be a valid method of breast volume measurement, but some centres may lack access to either a 3D laser scanner or other forms of 3D imaging. Water displacement of the intact breast as a method has been used in some centres since 1970, but has not previously been validated against other more modern methods (Bouman, 1970).

1.10.2 Body shape assessment

Body mass index (BMI) is commonly used as a criterion for surgery in patients presenting for breast reduction surgery. However, there is not always a clear relationship between BMI and breast size and the appropriateness of BMI as a selection criterion in this patient group is debatable (Benditte-Klepetko et al., 2007, Findikcioglu et al., 2013, Katch et al., 1980, Vandeput and Nelissen, 2002). Distinct clinical anthropometric measurements of fixed skeletal and soft tissue landmarks provide a useful tool in the evaluation of body and breast size pre-operatively and following surgical procedures. As an example, the linear measurement of the suprasternal notch to nipple distance is widely used by surgeons to evaluate breasts pre-operatively and to assess the outcome of breast surgery. In women with breast hypertrophy, increasing breast mass is associated with the inferior migration of the nipple, resulting in a larger measurement (Khan and Bayat, 2008). Whilst previous studies have investigated the relationship of suprasternal notch to nipple distance and surgical outcomes, there are limited studies evaluating other anthropometric variables related to breast and body size. In addition, studies exploring the relationship between BMI and the physical symptoms associated with large breasts have presented conflicting findings (Dabbah et al., 1995, Gonzalez et al., 1993, Miller et al., 1995, Netscher et al., 2000). In women with breast hypertrophy, the influence of body shape on physical symptoms and health-related quality of life is scarcely reported in the literature. Furthermore, the effect of disproportionality (breast size as a proportion of overall body size) is an important area for further research.

1.11 Economic evaluation of breast reduction surgery

In addition to clinical effectiveness, evidence to assess the cost effectiveness of interventions is becoming increasingly important for health care decision-makers due to limited resources and the comparatively larger number of treatment options available. Making rational decisions requires decision-makers to weigh up the costs and benefits of a treatment option and compare to other interventions in order to decide which are the most beneficial for informing optimal treatment pathways for patients and for driving health system efficiencies. Measurement and valuation of health-related quality of life using validated outcome measures is increasingly being employed to measure treatment success and provide comparative data between different medical interventions. As an example, the generic SF-36 questionnaire is commonly utilised worldwide across a wide variety of specialties as a measure of health outcome. However, it may be unclear how improvement or change in one area of health compares to an improvement or change in another.

For this reason, a single measure of quality of life may be employed to enable simpler comparisons between treatments. In the case of the SF-36, this led to the development of the SF-6D, providing a single preference-based utility score (on the 0 to 1 being dead to full health quality-adjusted life years (QALY) scale) from the 36-items specifically for application in economic evaluations (Brazier et al., 2002). Whilst the National Institute of Clinical Excellence (NICE) in the United Kingdom, which provides guidance on which treatments and care types are available from the National Health Service (NHS) in England and Wales, has mandated the EQ-5D as the preferred measure of HRQoL (National Institute for Health and Clinical Excellence, 2014a), guidelines from the Pharmaceutical Benefits Advisory Committee (PBAC) and Medical Services Advisory Committee (MSAC) in Australia are not so prescriptive and advocate the use of any generic preference-based health outcome measure as long as it is justified (Medical Services Advisory Committee, 2016, Pharmaceutical Benefits Advisory Committee, 2016). In this context, outcomes using generic measures, as opposed to condition-specific measures, are in principle valued consistently across all healthcare treatments and programmes.

Health utility scores are commonly used to value the benefits of health care which are measured in terms of quality-adjusted life years (QALYs). QALYs attempt to provide a measure which combines both length of life and health-related quality of life into a single measure and are used in cost-utility analyses (Brazier et al., 2017). In the United Kingdom, NICE has adopted the cost per QALY as its main measure of outcome. This ratio incorporates the incremental cost associated with an intervention with its incremental benefits (defined in terms of QALYS) (Dakin et al., 2015, Rawlins and Dillon, 2005, Rawlins and Culyer, 2004). It is then possible to directly compare a new intervention with existing treatments to determine which is the most cost-effective option. However, this approach may be limited by the lack of information for many medical interventions. Despite the published evidence on the health gains of breast reduction surgery, it is possible that without comprehensive evidence relating to its cost-effectiveness it may be mistaken as a cosmetic or lifestyle intervention rather than a functional operation.

Research assessing the cost-effectiveness of breast reduction surgery is scarce and has only been described in a few countries worldwide including the United Kingdom (Taylor et al., 2004), Finland (Saariniemi et al., 2012, Tykka et al., 2010), Canada (Thoma et al., 2014) and Brazil (Araujo et al., 2014). In the study by Taylor and colleagues in the United Kingdom, the cost per QALY for reduction mammoplasty was found to be between £4733 and £5729. However, the HRQoL data was obtained from a Swedish study, whilst the cost data was from the United Kingdom (Taylor et al., 2004). In addition, this study only reported the cost per QALY and did not separately report the number of QALYs gained. In Finland, the average direct hospital costs for reduction mammoplasty was approximately €3601 and the mean cost per QALY gained was €1180 (Saariniemi et al., 2012). In another Finnish study, the mean direct costs were found to be comparable at €3383, yet the cost per QALY of €3638 was considerably higher using the same 15D utility instrument but

over a shorter 6 month follow-up period (Tykka et al., 2010). Thoma and colleagues performed a prospective study on 52 women using the Health Utilities Index Mark 3 scores and found that patients who undergo breast reduction surgery have an average lifetime gain of 5.32 quality-adjusted life years, which is equivalent to each patient living an extra 5.32 years in perfect health (Thoma et al., 2007). Whilst this study advanced the understanding of the health burden of breast hypertrophy on health-related quality of life health burden in comparison to other chronic health conditions, and the expected lifetime benefits of breast reduction surgery, the authors did not present the utility value or cost per QALY data to enable further comparison. Subsequently, the authors conducted a randomised-controlled trial to compare the cost-effectiveness of vertical scar reduction in comparison to inverted T-shaped reduction mammoplasty (Thoma et al., 2014). The authors found that inverted T-shaped reduction dominated vertical scar reduction from the Ministry of Health perspective by being slightly less costly (\$3090.06 versus \$3106.58 in 2012 Canadian dollars) and slightly more effective (0.87 quality-adjusted life-years versus 0.86 quality-adjusted life-years).

Despite some limitations in current published studies in this area, preliminary evidence argues that the cost-effectiveness and cost per QALY for breast reduction surgery compares favourably to other medical interventions. Whilst NICE in the United Kingdom rejects the use of an absolute cost per QALY threshold (Rawlins and Culyer, 2004), it's stated 'range of acceptable cost effectiveness' is between £20,000-£30,000 (Devlin and Parkin, 2004). The cost per QALY ratio for reduction mammoplasty of between £4733 and £5729 was found to be very favourable when compared with interventions approved by NICE; for example Etanercept and infliximab for rheumatoid arthritis at a cost per QALY of £27,000 - £35,000, Infliximab for Crohn's disease at £27,500 (Taylor et al., 2004) and £5623 for a total knee replacement (Dakin et al., 2012).

Given the reported gains in health-related quality of life in many international studies, a thorough economic evaluation of breast reduction surgery in Australia is warranted to provide the necessary information that decision-makers need to objectively maintain Australian government funding of this surgery within the healthcare system.

2. A PROSPECTIVE STUDY OF WOMEN UNDERGOING BREAST REDUCTION SURGERY: CLINICAL OUTCOMES AND HEALTH-RELATED QUALITY OF LIFE

A version of this chapter has been accepted as an original article for publication in *BMJ Open* journal (Crittenden et al., 2020); attached as Appendix A. A comparison of the magnitude of change in SF-36 scores between surgical interventions was presented at the 2019 International Society for Quality of Life Research (ISOQOL) 26th annual conference in San Diego, USA (Crittenden et al., 2019a).

2.1 Introduction

Breast reduction surgery is a common plastic surgery procedure and it has previously been shown to be effective for relieving pain and functional problems associated with breast hypertrophy (Blomqvist et al., 2000, Freire et al., 2007, Klassen et al., 1996, Mello et al., 2010, Thoma et al., 2007), whereas conservative approaches to treatment such as physiotherapy, hormonal therapy and weight loss have much less impact (Collins et al., 2002, Iwuagwu et al., 2006b). However, despite existing evidence to support the efficacy of breast reduction surgery, this has not translated to policy in many countries and jurisdictions worldwide. As described in Chapter one, within the Australian public hospital system, access to breast reduction surgery for patients is variable and is ultimately reliant upon state and local policies. Similarly, in the United Kingdom, reports on the rationing of surgery by the National Health Service (NHS) on the basis of geographical location has resulted in a "postcode lottery" (Royal College of Surgeons of England, 2014, Wraight et al., 2007). In 2018, reports from the NHS England 'Evidence-Based Interventions Programme' proposed to restrict funding for procedures it considers 'unnecessary', to save money and eliminate unwarranted clinical variation (lacobucci, 2018). The inclusion of breast reduction surgery as a 'procedure of limited effectiveness' implies that it is a marginal and low priority procedure in comparison to other medical interventions (NHS England, 2018). However, labelling breast reduction surgery an 'ineffective' and 'unnecessary' procedure might be misleading and inaccurate, with little evidence to support this claim. Furthermore, restrictive access policies are in place in both public and private sectors in many countries and jurisdictions worldwide; often these restrictions are based on body mass index or a minimum weight of breast resection at surgery (Breuning et al., 2010, Frey et al., 2014, Klassen et al., 1996, Koltz et al., 2013, Nguyen et al., 2008, NHS England, 2018, O'Blenes et al., 2006, Royal College of Surgeons of England, 2014, Seitchik, 1995, Taylor et al., 2004, Wraight et al., 2007). The validity of such criteria might not be evidence-based, resulting in women with a medical need for surgery being denied access to it.

The Short Form-36 (SF-36) is a well-established and widely-utilised indicator of patient-reported outcome for evaluating the burden of disease states and the outcomes of medical interventions

and was therefore chosen as the primary outcome measure for this study (Ware and Sherbourne, 1992); with the Multidimensional Body-Self Relations Questionnaire (MBSRQ) used to further assess body image and health status (Brown et al., 1990, Cash and Pruzinsky, 1990). An ongoing challenge of interpreting changes in health-related quality of life (HRQoL) measures following medical treatments is defining a threshold as to what constitutes a clinically meaningful change. The interpretation of HRQoL scores based solely on statistical significance may be misleading because small differences in scores can be statistically significant even in the absence of the change being clinically meaningful (Lydick and Epstein, 1993, Osoba et al., 1998). The minimal important difference (MID) has become a standard approach for interpreting the clinical relevance of changes in scores of patient-reported outcomes such as HRQoL and can be defined as the smallest difference in HRQoL score that is perceived as beneficial or important by a patient or clinician (Jaeschke et al., 1989, p. 408). However, there is no universal measure of MID as this threshold may vary due to many factors including; population groups, type of disease, disease severity, between different health measures or clinical context (Jaeschke et al., 1989, King, 2011, Musoro et al., 2018). MID estimates for interpreting SF-36 scores in women undergoing breast reduction surgery are currently undetermined and are limited to rule-of-thumb approaches such as a 10-point change on 100-point subscales (King, 1996), and universal estimates for the SF-36 summary measures recommended by the developers (Frendl and Ware, 2014, Ware et al., 1995). Given that the SF-36 is the most frequently used patient-reported outcome measure in studies of women undergoing breast reduction surgery worldwide (Lonie et al., 2019), condition-specific estimates of MIDs provide valuable information for use in both the clinical setting, and for interpretation of findings from research studies.

The primary aim of this study was to longitudinally assess health-related quality of life (HRQoL) in women with breast hypertrophy before and after breast reduction surgery, and to compare these outcomes to control groups of women with breast hypertrophy not undergoing surgery, and also to a normative female reference population. Secondly, this study aimed to assess the impact of patient demographics and surgical characteristics including, but not limited to, those commonly used as selection criteria for access to surgery and insurance coverage on pre-operative HRQoL scores and the long-term improvement in HRQoL following surgery. Finally, this study aimed to estimate thresholds of minimal important difference for the SF-36 in patients for the interpretation of the change in scores in women with breast hypertrophy who underwent breast reduction surgery

2.2 Participants and methods

2.2.1 Design and participants

2.2.1.1 Surgical cohort

A prospective cohort study was performed at Flinders Medical Centre in Adelaide, Australia. Ethics approval was obtained for this project from the local Southern Adelaide Clinical Human

Research Ethics Committee (SAC HREC approval number 118.056) (Appendix B). All women aged 18 years and over with symptomatic breast hypertrophy who were assessed for bilateral breast reduction surgery between April 2007 and February 2016 were informed of the study. Women who were unable to complete written questionnaires or refused to participate, were excluded from the study. Participants were provided with the information sheet and written informed consent was obtained.

Participants who consented to the study were assessed prospectively pre-operatively and at 3, 6 and 12 months following bilateral breast reduction surgery. Participants were asked to complete the Short-Form-36 version 2 (SF-36v2), Multi-Dimensional Body-Self Relations Questionnaire (MBSRQ) and adjunct study-specific questionnaire at each of the four study time points. Participant characteristics including age, height, weight and smoking status were recorded at the time of enrolment. Intraoperatively, the surgical technique and weight of tissue resected from each breast was documented. Hospital records were used to determine the American Society of Anesthesiologists (ASA) Physical Status Classification System status (American Society of Anesthesiologists, 2014), length of hospital stay, number of outpatient clinic appointments relating to the surgery, and complications leading to re-hospitalisation or a further operative procedure within the 12 months follow-up period. Standardised clinical photography was performed at all four study timepoints.

Breast and body measurements were performed pre-operatively and 12 months post-operatively using a Cyberware WBX three-dimensional laser body scanner (Cyberware, Monterey, USA), anthropometric measurements and water displacement. These measurements are described in detail in Chapter 6.

2.2.1.2 Breast hypertrophy control cohort

A control group of women with breast hypertrophy who were actively seeking breast reduction surgery through Flinders Medical Centre but had not yet had surgery were recruited for comparison. Women aged 18 years and above who were referred to Flinders Medical Centre by their general practitioner for consultation for bilateral breast reduction surgery were invited to participate in the study. The breast hypertrophy control group of women were identified from two sources: either new referrals on the waiting list for an outpatient consultation with a consultant plastic surgeon, or those who were on the surgical admissions waiting list. This non-surgical cohort represents an appropriate reference to compare HRQoL to those in the surgical cohort as they too were presenting with symptoms of breast hypertrophy and were followed up for the same 12-month timeframe without undergoing surgical intervention. Women who were unable to complete written questionnaires, had breast reduction surgery during the study, or did not return study questionnaires, were excluded from the study. Ethics approval was obtained for this study from the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC approval

Eligible participants were mailed study information and questionnaires. Consenting participants were required to self-complete and return the SF-36v2 and MBSRQ questionnaire and provide information for relevant variables such as age, height, weight, bra cup size and health status at baseline and again at 12 months after enrolment.

2.2.2 Outcome measures

2.2.2.1 Short-Form 36 (SF-36)

The Short Form-36 Health Survey (SF-36) is a generic instrument which has broad applicability and is widely used in the medical literature to measure the general health-related quality of life across population and patient groups (Ware and Sherbourne, 1992). The SF-36 version 2 (SF-36v2) used in this study is an updated version of the original SF-36 questionnaire and contains 36 items which form and assess health across eight subscales: (1) physical functioning; (2) role physical; (3) bodily pain; (4) general health; (5) vitality; (6) social functioning; (7) role emotional; and (8) mental health (Appendix D) (Table 2.1). Questionnaire responses were transformed using QualityMetric Health Outcomes[™] Scoring Software for Microsoft Windows and as per the SF-36v2 scoring manual to provide the eight subscales a score between 0 and 100, with higher scores indicating better health (Ware et al., 2000). The subscales were converted into two summary scores: Physical Component Score (PCS) and Mental Component Score (MCS) using norm-based methods and scoring coefficients from the Australian general population (Hawthorne et al., 2007).

For comparison purposes, general female population normative scores were obtained from the 2008 South Australian Health Omnibus Survey dataset (Marin et al., 2009, Population Research and Outcome Studies Unit, 2008), with scores weighted to correspond to the age distribution of the study participants.

Table 2.1 Dimensions of the SF-36v2

Dimonsion	Description	No.	
Dimension	Description	items	
Physical Functioning (PF)	Limitations in carrying out a range of physical activities because of health problems.	10	
Role Physical (RP)	Limitations in performing regular daily activities because of physical health problems.	4	
Bodily Pain (BP)	Intensity of bodily pain or discomfort and the extent to which pain interferes with their normal activities.	2	
General Health (GH)	General health perceptions.	5	
Vitality (VT)	Energy and fatigue.	4	
Social Functioning (SF)	Limitations in social activities due to physical or emotional problems.	2	
Role Emotional (RE)	Limitations in usual role activities because of emotional problems.	3	
Mental Health (MH)	Psychological distress and well-being.	5	
Physical Component (PCS)	Physical health summary.	35	
Mental Component (MCS)	Mental health summary.	35	

2.2.2.2 Multidimensional Body-Self Relations Questionnaire (MBSRQ)

The Multidimensional Body-Self Relations Questionnaire (MBSRQ) is a validated patient-reported outcome measure for the assessment of self-attitude aspects of body image, physical activity and health (Brown et al., 1990, Cash and Pruzinsky, 1990). Two forms of the MBSRQ are available, the full version and the MBSRQ-Appearance Scales (MBSRQ-AS). The full, 69-item version consists of seven factor subscales: (1) appearance evaluation; (2) appearance orientation; (3) fitness evaluation; (4) fitness orientation; (5) health evaluation; (6) health orientation; and (7) illness orientation. There are also three additional MBSRQ subscales: body areas satisfaction scale (BASS); (2) overweight preoccupation; and (3) self-classified weight (Cash and Pruzinsky, 1990) (Table 2.2). In this study, the full version of the MBSRQ questionnaire was utilised and scores were generated from one to five according to the revised algorithms provided by Cash (Cash, 2000) (Appendix E). Higher scores on evaluation scales indicate the person feels more satisfied in relation to either the construct (i.e. appearance, fitness or health), whereas higher scores for orientation scales indicates a greater importance that a respondent places on the construct. Previously published normative data from the United States population was used for comparison (Cash, 2000).

Table 2.2 Dimensions of the MBSRQ

Dimension	Description	No. items	
Appearance evaluation	Feelings of physical attractiveness or unattractiveness; satisfaction or dissatisfaction with one's looks.	7	
Appearance orientation	Extent of investment in one's appearance.	12	
Fitness evaluation	Feelings of being physically fit or unfit.	3	
Fitness orientation	Extent of investment in being physically fit or athletically competent.	13	
Health evaluation	Feelings of physical health and/or the freedom from physical illness.	6	
Health orientation	Extent of investment in a physically healthy lifestyle.	8	
Illness orientation	Extent of reactivity to being or becoming ill.	5	
Body areas satisfaction scale	Similar to the Appearance Evaluation subscale, except that the BASS taps satisfaction with discrete aspects of one's appearance.	9	
Overweight preoccupation	This scale assesses a construct reflecting fat anxiety, weight vigilance, dieting, and eating restraint.	4	
Self-classified weight	This scale reflects how one perceives and labels one's weight, from very underweight to very overweight.	2	

2.2.2.3 Adjunct questionnaire

Supplementary study-specific questionnaires were completed at each study timepoint. Preoperatively, participants were asked to state: (1) number of days off work in the last six months from symptoms relating to their breast condition; and (2) spending on medications and treatments (creams, pain relief etc.) due to symptoms from their breast condition. Post-operatively, in addition to the aforementioned questions participants were also asked: (3) whether they would choose to have the surgery again if they had their time over; (4) whether they experienced any new health issues or a worsening of an existing health issue that was not related to the surgery; and (5) whether or not they experienced any complications from the surgery.

2.2.2.4 Assessment of surgical complications

A comprehensive complications checklist was completed prospectively by the treating doctor at the following approximate timepoints; 4 weeks, 3 months, 6 months, and 12 months following surgery (Appendix F). Complications were subsequently graded according to the Clavien-Dindo Classification (CDC) (Clavien et al., 2009, Dindo et al., 2004), a standardised grading system for surgical complications based on the medical intervention required to treat the complication (Table 2.3). The CDC provides a validated approach to classifying surgical complications and enables an objective, standardised and reproducible comparison of surgical complications; avoiding subjective

interpretation and overcoming limitations with the comparison of surgical complications between surgeons, across institutions and treatment modalities.

Grade	Definition
Grade I	Any deviation from the normal postoperative course without the need for pharmacological treatment or surgical, endoscopic, and radiological interventions. Allowed therapeutic regimens are: drugs as antiemetics, antipyretics, analgetics, diuretics, electrolytes, and physiotherapy. This grade also includes wound infections opened at the bedside
Grade II	Requiring pharmacological treatment with drugs other than such allowed for grade I complications. Blood transfusions and total parenteral nutrition are also included.
Grade III Grade IIIa Grade IIIb	Requiring surgical, endoscopic or radiological intervention. Intervention not under general anaesthesia Intervention under general anaesthesia
Grade IV Grade IVa Grade IVb	Life-threatening complication (including CNS complications)* requiring IC/ICU management Single organ dysfunction (including dialysis) Multiorgan dysfunction
Grade V	Death due to the intervention

Table 2.3 Clavien-Dindo classification of surgical complications

*Brain haemorrhage, ischemic stroke, subarachnoidal bleeding, but excluding transient ischemic attacks. CNS, central nervous system; IC, intermediate care; ICU, intensive care unit.

For analysis purposes, surgical complications graded using the CDC system in this study were further categorised into two groups: (1) CDC grades I and II were classified as minor complications; and (2) CDC grades III and above were classified as major complications.

2.2.3 Distribution-based estimates of MID

The minimal important difference (MID) was estimated for the SF-36 scores by applying distribution-based techniques. The most common distribution-based approaches use sample variability, either at baseline, after follow-up or using the change between timepoints. Half a standard deviation of the baseline scores has been suggested to serve as a default value for important patient-perceived change on HRQoL measures (Norman et al., 2003). This approach is the most frequently reported distribution-based criterion for determination of the MID (Ousmen et al., 2018). The 0.2 SD and 0.3 SD have also been suggested as a useful starting point to estimate MID, with a corresponding effect size (ES) calculated as the mean difference (or change) in score divided by the standard deviation of the baseline score, with a treatment effect size of 0.2 to 0.5

suggested as minimally important (Cohen, 1988, Crosby et al., 2003, Kazis et al., 1989, King, 2011, Samsa et al., 1999). This MID can be used to evaluate change both within-persons and between-persons.

The standardised response mean (SRM) is similar to the effect size but is defined as the mean score change divided by the standard deviation of that change score (Liang et al., 1990). This method therefore factors in the variability of the change between measurements whereas the effect size does not, and therefore the SRM is not as dependent on the heterogeneity of the sample at baseline. As per interpretation of the effect size, values of 0.2, 0.5 and 0.8 SRM have been proposed to represent small, moderate and large change, respectively (Crosby et al., 2003).

Another commonly used method that can be used to benchmark the change in HRQoL scores by approximating the MID is the standard error of measurement (SEM). The SEM is calculated using the sample standard deviation and the reliability coefficient (*r*) of the instrument (SEM = $SD\sqrt{1-r}$); in this study the Cronbach's alpha was used as an estimate of reliability for each individual SF-36 scale and summary measure in the calculation of SEM. A value of 1 SEM change threshold has been recommended for defining the MID (Norman et al., 2003, Wyrwich et al., 1999a, Wyrwich et al., 1999b).

2.2.4 Statistical analysis

Statistical analyses were performed using SPSS v25.0 statistical software (IBM Corp., Armonk, N.Y.). Descriptive statistics including mean, standard deviation and 95% confidence interval were computed for continuous variables. Comparisons between groups were made using *t* tests for continuous data and chi-square tests for categorical data, with Fisher's exact test as appropriate.

Linear mixed-models were used to assess the significance of changes in SF-36 scores for the surgical cohort pre-operatively and 3, 6, and 12 months post-operatively, where the dependent variable was each SF-36 scale and time was the fixed effect. For each SF-36 scale, a change score was calculated using the score obtained at the last available assessment, with a higher score representing a greater improvement from baseline.

Pearson correlation coefficients were calculated to assess the linear association between SF-36 scores and baseline participant and clinical characteristics; variables that showed a significant association were entered into the regression model. For any two independent variables with high correlation (r > 0.8), one was chosen to be included in the regressions model rather than both. Candidate variables included age, body mass index (BMI), pre-operative breast volume, bra cup size, tissue resection weight (grams), breast asymmetry, and ratio of breast to body volume. Variables were continuous except for bra cup size which was categorised into six groups as follows: D, DD, E, F, G and \geq H cup. Multiple linear regression was used to assess whether any of

the collected socio-demographic or clinical variables were predictive of firstly, SF-36 PCS score at baseline, and secondly, with the change in SF-36 PCS scores from baseline to 12-months after surgery.

The analysis of surgical complications was conducted using several approaches, including the grade of complications and the presence or absence of any complication. Any complication was defined by whether or not a patient experienced at least one of the complications listed on the comprehensive assessment checklist, with the incidence reported as the number of cases (percentages). Bivariate analysis was performed between any complication and potential baseline participant and clinical variables using an independent samples *t*-test for continuous variables or a chi-square or Fisher's exact test for categorical variables. Logistic regression was then conducted to explore potential relationships between any complication (dependent dichotomous variable) and candidate predictor variables (independent variables). Variables of interest included patient age, BMI, smoking status, presence of diabetes, total tissue resection weight at surgery, operative time, bra cup size, pre-operative breast volume and sternal-notch-to-nipple distance. From this model, adjusted odds ratios with associated 95% confidence interval were estimated for each explanatory variable as measures of association with any complication.

The World Health Organisation (WHO) definition of obesity was used to classify patients with a BMI <30 kg/m² as non-obese and those with BMI \geq 30 kg/m² as obese (World Health Organization, 2018). BMI was further divided into six categories for statistical analysis based on the WHO classification: underweight (<18.5 kg/m²); normal weight (18.5–24.9 kg/m²); overweight (25–29.9 kg/m²); obese class I (30–34.9 kg/m²); obese class II (35–39.9 kg/m²); and obese class III (\geq 40 kg/m²). A one-way analysis of variance (ANOVA) was used to compare mean SF-36 scores between BMI groups, with a Dunnett's two-tailed post-hoc test used to compare participants of a normal weight to higher BMI categories.

For all analyses, statistical significance was established at a *p*-value of less than 0.05.

Sample size was determined *a priori* and a minimum sample size of 98 patients per group was calculated to give 80% power at a two-sided significance level of 5% to detect a mean difference of 10-points with an estimated standard deviation of 25-points in the SF-36 questionnaire score.

2.3 Results

2.3.1 Study participants

2.3.1.1 Surgical cohort

Of 251 participants who completed a baseline assessment and underwent bilateral breast reduction surgery, 209 (83.3%) completed at least one post-operative follow-up assessment and were included in the study group for analysis. Missing data were due to participants repeatedly not

attending appointments or choosing to not complete and return the study questionnaires at some time points. Twenty-three participants formally withdrew from the study following surgical intervention.

Participant demographics for the surgical cohort are summarised in Table 2.4. The mean age at surgery was 42.6 ± 13.4 years. The mean sternal notch-to-nipple distances were 32.7 ± 4.5 cm and 32.4 ± 4.5 cm for the right and left breasts, respectively. The majority of participants (64%) were classified as obese with a body mass index greater than or equal to 30 kg/m^2 . The mean total weight of breast tissue resected at surgery was $1338 \text{ g} \pm 817 \text{ g}$. An inferior pedicle breast reduction technique was the most commonly used approach (161/209, 77%), followed by a superior pedicle technique (35/209, 17%). The average hospital stay was 2.3 days.

Baseline characteristics were compared between participants who were lost to follow-up or declined to attend subsequent study assessments and those who completed at least one post-operative assessment. No difference was observed for age, body mass index, tissue weight resected or pre-operative SF-36 scales and summary scores except for the mental health scale, where non-respondents had a lower mean score of 6.8-points less than responders (p = 0.034).

2.3.1.2 Breast hypertrophy control cohort

Study questionnaires were initially posted to 350 women with breast hypertrophy who were not scheduled for surgery; 160 (46%) completed and returned the questionnaires at baseline, and of these 124 responded again 12 months later. Twenty-four of those contacted to participate in the study underwent breast reduction surgery during the study timeframe and were therefore excluded. Participant demographics for the hypertrophy control cohort are summarised in Table 2.4.

There were no statistically significant differences in characteristics including age, body mass index, obesity status, or smoking history between the surgical group and hypertrophy control group.

	Surgical appart	Hypertrophy control	<i>p</i> -value of
	Surgical conort	cohort	difference ⁺
No. of participants	209	124	
Mean (SD; range) age (years)	42.6 (13.4; 18 to 72)	45.3 (13.1; 20 to 79)	0.079
Age group (years):			
18–24	24 (12)	12 (10)	
25–34	38 (18)	15 (12)	
35–44	64 (31)	26 (21)	
45–54	41 (20)	43 (34)	
55–64	31 (15)	21 (18)	
≥ 65	11 (5)	7 (6)	
Mean (SD) BMI (kg/m²)	32.7 (6.0)	32.2 (6.1)	0.468
Obesity status:			
Non-obese (<30 kg/m²)	71 (34)	48 (39)	0.326
Obese (≥30 kg/m²)	138 (66)	74 (61)	
Missing	0 (0)	2 (0)	
Smoking status:			
Non-smoker	108 (52)	78 (63)	0.243
Current smoker	35 (17)	14 (11)	
Ex-smoker <12 months	15 (7)	5 (5)	
Ex-smoker >12 months	47 (23)	25 (20)	
Missing	4 (0)	2 (0)	
Bra cup size:			
≤D	13 (6)	4 (3)	
DD	43 (21)	13 (11)	
E	50 (24)	19 (15)	
F	46 (22)	27 (22)	
G	35 (17)	37 (30)	
≥H	19 (10)	19 (15)	
Missing	3 (0)	5 (0)	
ASA status:			
1 (normal healthy)	82 (39)	-	
2 (mild systemic disease)	101 (48)	-	
3 (severe systemic	19 (9)	-	
alsease)			
	168 (38)	-	
(min)			

Table 2.4 Baseline characteristics of participants.

Values are numbers (percentages) unless stated otherwise.

⁺ Using independent samples *t*-test or chi-square test as appropriate.

2.3.2 Clinical outcomes in the surgical cohort

Overall, 92 participants (44%) experienced at least one complication within the 12 months following breast reduction surgery; with 117 patients (56%) experiencing no surgical complications. Classification of surgical complications for each of the 209 patients according to the Clavien-Dindo Classification (CDC) system were as follows: Grade I (n = 39, 19%); Grade II (n = 38, 18%); Grade IIIa (n = 5, 2%); and Grade IIIb (n = 10, 5%). There were no Grade IV or Grade V complications. Minor complications, classified as CDC Grade I and Grade II, accounted for the majority (77/92, 83.7%) of all post-operative complications.

Wound healing problems were the most frequent cause of post-operative complications (n = 39, 19%). These were predominantly wound healing problems requiring dressings (n = 34, 16.3%). Importantly, this study was very inclusive of surgical complications and given that incomplete wound healing at 4 weeks may not be reported as a complication in other studies, if these cases were excluded from the reported complications then the incidence reduced to 28% (58 participants). The most common cause of Grade II complications was wound infections requiring oral antibiotics (n = 36, 17%), prescribed by either one of the hospital doctors in the treating team or by a community general practitioner. It should be noted that the reporting of a wound infection did not require a positive bacterial growth on laboratory testing or a raised white cell count; therefore, it may be that some women who were prescribed antibiotics by their general practitioner may have had wound inflammation rather than true infection. Small wound debridement and minor scar excision under local anaesthesia were the treatments for Grade IIIa complications (n = 5, 2%). Revisional surgery under general anaesthesia (Grade IIIb) was required for five patients (2.4%) for evacuation of a haematoma, and nine patients (4.3%) had subsequent procedures for revision of surgical scars or to correct standing cone deformities or 'dog-ears'. As for post-operative sequelae, a total of 20 patients (9.6%) reported noticeable breast asymmetry; and 19 patients (9.1%) noted altered nipple sensation at their final post-operative review, with 5 patients reporting hyper-sensation and 14 stating reduced nipple sensation following surgery.

Using bivariate analysis, smoking status, sternal notch-to-nipple distance and total tissue resection weight were significantly associated with an increased incidence of any complication (Table 2.5). In contrast, there was no significant association between obesity status and the incidence of surgical complications, with the incidence of complications in non-obese participants (25/71, 35.2%) and obese participants (67/138, 48.6%). There were also no differences in the incidence of major complications (CDC grade III and above) based on obesity status, with incidences of 5/71 (7.0%) in non-obese and 10/138 (7.2%) in obese cases ($X^2 = 3.7$, p = 0.158). The increased prevalence of any complication in obese participants was due to minor complications (CDC grade I and II), with 57 (41%) of obese participants experiencing a minor complication in comparison to 20 (28%) of non-obese participants. Age, ASA status, diabetes and operating time were not associated with an increased incidence of complications.

	No Complication	Complication	<i>p</i> -value of
	(<i>n</i> = 117)	(<i>n</i> = 92)	difference ⁺
Age, years	42.5 (13.4)	42.8 (13.4)	0.839
Pre-operative breast volume, [‡] ml	1710.6 (590)	1751 (650)	0.661
SND distance, [¥] cm	32.0 (4.2)	33.4 (4.8)	0.046
Total resection weight, g	1192.3 (637.4)	1506.8 (974.4)	0.006
Operative time, min	167 (39)	169 (38)	0.770
Obesity (BMI ≥30 kg/m²)	71 (61%)	67 (73%)	0.078
Smoking	11 (9%)	24 (26%)	0.002
Diabetes	4 (2%)	6 (3%)	0.349
ASA status			
1	43 (37%)	39 (14%)	0.638
2	58 (50%)	43 (47%)	
3	12 (13%)	7 (8%)	

Table 2.5 Association between participant and clinical characteristics and any complication

⁺ Using an independent two-sample *t*-test for continuous variables and a chi-square test for categorical variables.

^{*} Average of left and right breast volumes.

[¥] Average of left and right breast sternal notch-to-nipple (SND) distances.

A logistic regression analysis was performed to assess the impact of baseline participant and clinical characteristics on the likelihood that participants will experience any complication. The model contained nine independent variables at baseline: age, smoking status, diabetes, BMI, sternal notch-to-nipple distance, pre-operative breast volume, bra cup size, total weight of tissue resection, and operating time. The full logistic regression model containing all predictors was statistically significant ($X^2 = 40.6$, p = 0.001), indicating that the model was able to distinguish between patients who did and did not experience any post-operative complication. The model explained between 26.5% (Cox and Snell R²) and 35.4% (Nagelkerke R²) of the variance in any complication, and correctly classified 72.7% of cases. As shown in Table 2.6, smoking status was the only independent variable that made a unique statistically significant contribution to the model. Current smokers and those who were reformed less than 12-months were both 9.7-times more likely to have any complication.

Variable	В	SE	р	Odds	95% CI for
				ratio	odds ratio
Age, years	0.025	0.02	0.175	1.02	0.99 to 1.06
BMI, kg/m²	0.123	0.13	0.310	1.13	0.89 to 1.43
Total resection weight, g	0.001	0.001	0.400	1.00	0.99 to 1.02
Operative time, min	0.0001	0.0001	0.582	1.00	1.0 to 1.0
Pre-operative breast volume, ml	-0.001	0.001	0.234	1.00	0.99 to 1.0
Sternal notch-to-nipple distance, cm	0.146	0.091	0.103	1.16	0.97 to 1.38
Smoking⁺					
Current smoker	2.27	0.68	0.001	9.72	2.6 to 36.7
Ex-smoker <12 months	2.27	0.75	0.003	9.73	2.2 to 42.5
Ex-smoker >12 months	0.32	0.54	0.556	1.38	0.48 to 3.98
Diabetes [‡]					
Yes	0.98	1.52	0.284	2.67	0.44 to 16.1

Table 2.6 Logistic regression of risk factors for complications

+ Compared with non-smoker.

* Compared with non-diabetic.

2.3.3 Health-related quality of life outcomes using the SF-36

The SF-36 was completed pre-operatively and at least once post-operatively by 209 surgical participants; 191 (91%) completed the post-operative questionnaires at 3 months, 183 (88%) at 6 months and 193 (92%) at 12 months. When compared with previously published age-adjusted normative data for the female Australian population (Marin et al., 2009), mean baseline SF-36 scores for the surgical cohort were significantly lower across all scales (p < 0.001) (Table 2.7). A comparison of mean pre-operative and 3 month post-operative SF-36 scores showed that scores were significantly higher across all eight SF-36 subscales (p < 0.001) (Table 2.7), such that they reached the level of the normative population (Figure 2.1). Mean SF-36 PCS and MCS scores significantly improved following surgery, increasing by 10.2 (95% CI; 8.2 to 12.1) and 9.2 points (95% CI; 6.9 to 11.6), respectively (p < 0.001) (Table 2.7 and Table 2.8). SF-36 scores were stable at 6- and 12-months post-surgery and linear mixed-model analysis showed no significant difference from those at 3 months post-surgery.

Mean baseline SF-36 scores for women in the breast hypertrophy control group were significantly lower than the normative population across all dimensions (Table 2.7). At 12 months post-baseline, SF-36 scores showed no significant improvement and remained significantly lower than population norms (Figure 2.2) and post-operative scores for women in the surgical cohort (Table 2.7). Mean SF-36 PCS and MCS summary scores for women in the breast hypertrophy control group were significantly lower than those who underwent breast reduction surgery, with a mean difference of 10.6 (95% CI; 8.3 to 12.8) and 11.1 points (95% CI; 8.2 to 13.9), respectively (p < 0.001) (Table 2.7).
	Normative ⁺	Hypertrop Coł	hy Control Nort	Surgical Cohort			
SF-36 scale	(<i>n</i> = 1551)	Baseline (<i>n</i> = 160)	12 months (<i>n</i> = 124)	Pre-operative (<i>n</i> = 209)	3 months post-operative (<i>n</i> = 190)	6 months post-operative (<i>n</i> = 181)	12 months post-operative (n = 191)
Physical function	84.2 (19.1)	64.7 (24.2)	61.1 (25.5)	61.0 (25.4)	80.1 (22.3)	80.8 (24.2)	83.4 (22.1)
Role physical	82.0 (24.8)	58.3 (28.7)	58.1 (29.2)	56.0 (28.3)	79.5 (23.7)	81.1 (25.4)	81.3 (25.1)
Bodily pain	73.0 (21.4)	39.8 (22.5)	37.9 (21.2)	38.5 (21.5)	67.4 (24.1)	67.6 (27.6)	71.6 (25.9)
General health	70.2 (22.2)	49.7 (21.8)	49.8 (22.4)	57.9 (21.7)	69.1 (19.8)	69.5 (19.8)	70.4 (19.1)
Vitality	57.3 (20.9)	36.7 (19.7)	35.1 (21.0)	39.7 (20.2)	57.7 (20.4)	58.6 (19.1)	58.9 (19.7)
Social function	82.6 (24.3)	55.2 (29.0)	55.1 (27.4)	57.1 (27.9)	78.8 (25.9)	79.4 (25.9)	81.4 (23.4)
Role emotional	88.3 (20.3)	62.8 (30.3)	60.2 (28.6)	61.7 (28.9)	80.1 (25.1)	82.3 (23.1)	84.6 (22.2)
Mental health	77.0 (18.2)	58.8 (22.8)	56.1 (21.0)	59.8 (20.2)	73.7 (18.7)	73.8 (18.4)	74.3 (18.6)
Physical Component Score	49.7 (9.7)	39.6 (9.3)	39.3 (9.9)	39.7 (9.7)	48.9 (9.2)	49.0 (10.3)	49.9 (9.9)
Mental Component Score	47.6 (11.4)	36.2 (14.6)	35.1 (13.2)	37.0 (13.2)	45.4 (11.9)	45.7 (11.5)	46.2 (11.6)

Table 2.7 Mean (SD) SF-36 scores for participants in the surgical cohort, hypertrophy control cohort and normative female population

[†] Source: Age-standardised normative data from the South Australian female population (Marin et al., 2009).



Figure 2.1 Comparison of mean pre-operative and post-operative SF-36 scores in surgical participants with age-standardised female population norms



Figure 2.2 Comparison of mean baseline and 12-months post-baseline SF-36 scores in the breast hypertrophy control group with age-standardised female population norms

Table 2.8 summarises the mean changes in SF-36 subscales and summary scores from baseline to each follow-up timepoint in the surgical cohort and the breast hypertrophy control cohort. Within the surgical cohort, there were statistically significant improvements in SF-36 scores across all scales and summary scales. Corresponding effect sizes at 12-months post-operative ranged from a moderate effect of 0.58 for General Health, with the remaining seven subscales and both summary measures exhibiting a large effect size according to Cohen's criteria, ranging up to 1.53 for Bodily Pain. In contrast, scores did not improve within the breast hypertrophy control cohort, with scores for the Physical Functioning scale significantly lower by 4.4-points (95% Cl, 0.67 to 7.9) when comparing baseline to 12-months (p = 0.02).

	Hypertrophy control cohort		Surgical cohort		
SF-36 scale	Baseline to 12m post-baseline	Baseline to 3m post-operative	Baseline to 6m post-operative	Baseline to 12m post-operative	Effect size (95% Cl) [‡]
Physical function	-4.3 (-7.9 to -0.7)**	19.1 (14.5 to 23.7)*	19.8 (15.1 to 24.5)*	22.3 (17.7 to 27.0)*	0.94 (0.64–1.24)
Role physical	-0.82 (-5.3 to 3.6)	23.5 (18.4 to 28.6)*	25.0 (19.9 to 30.2)*	25.3 (20.2 to 30.4)*	0.95 (0.65–1.25)
Bodily pain	-1.97 (-5.7 to 1.7)	28.9 (24.0 to 33.7)*	29.1 (24.2 to 34.0)*	33.0 (28.2 to 37.9)*	1.39 (1.08–1.71)
General health	0.55 (-2.5 to 3.6)	11.2 (7.2 to 15.2)*	11.6 (7.6 to 15.6)*	12.5 (8.5 to 16.4)*	0.61 (0.32–0.90)
Vitality	-1.6 (-4.5 to 1.2)	18.0 (14.1 to 21.9)*	18.8 (14.9 to 22.8)*	19.2 (15.3 to 23.1)*	0.96 (0.66–1.26)
Social function	0.32 (-4.3 to 4.9)	21.6 (16.5 to 26.7)*	22.3 (17.1 to 27.5)*	24.3 (19.2 to 29.4)*	0.94 (0.65–1.24)
Role emotional	-1.71 (-6.6 to 3.2)	18.4 (13.5 to 23.4)*	20.6 (15.5 to 25.6)*	22.8 (17.9 to 27.8)*	0.89 (0.59–1.19)
Mental health	-2.9 (-6.2 to 0.5)	13.9 (10.2 to 17.7)*	14.1 (10.3 to 17.9)*	14.5 (10.8 to 18.3)*	0.75 (0.45–1.04)
Physical Component Score	-0.7 (-2.1 to 0.7)	9.2 (7.3 to 11.2)*	9.3 (7.3 to 11.3)*	10.2 (8.2 to 12.1)*	1.04 (0.74–1.34)
Mental Component Score	-1.08 (-3.2 to 1.0)	8.4 (6.0 to 10.8)*	8.7 (6.3 to 11.2)*	9.2 (6.9 to 11.6)*	0.74 (0.45–1.03)

Table 2.8 Mean (95% CI) change in SF-36 scores in the breast hypertrophy control cohort and surgical cohort

* Differences between pre-operative versus post-operative scores in the surgical cohort were significant at the p < 0.001 level.

^{**} Differences between baseline versus 12 months post-baseline scores in the breast hypertrophy control cohort were significant at the *p* <0.05 level.

⁺ Effect size was calculated as the mean difference between groups (12m post-operative – baseline), divided by the pooled standard deviation for the two groups.

2.3.3.1 Distribution-based estimates of MID for the SF-36

Minimal important difference (MID) estimates from the distribution-based analyses for each scale of the SF-36 are presented in Table 2.9. The SEM, 0.2 SD, 0.3 SD and 0.5 SD are presented based on baseline SF-36 scores, and the 0.2 SRM and 0.3 SRM were computed as a measure of effect size based on the change in SF-36 scores from baseline to 12 months following surgery. MID estimates for the SF-36 instrument varied for each individual subscale, ranging from 6 to 9-points using the 0.3 SRM, and 10 to 14-points using the one-half a standard deviation approach. MID estimates were lower using the standard error of measurement analyses based on the reliability of each subscale and ranged from 6 to 10-points. The distribution-based estimates for change scores were similar to those based on baseline scores, mostly within 1-point range (i.e. when comparing 0.2 SD to 0.2 SRM, and 0.3 SD to 0.3 SRM). For the norm-based SF-36 physical and mental summary scales, MID estimates were 5-points and 7-points using half a standard deviation, respectively; SRM estimates were slightly smaller at 3 and 4 points, respectively.

	Pre-surgery			Change from post-s	n pre- to 12m surgery	
SF-36 scale	0.2SD	0.3 SD	0.5 SD	1 SEM	0.2 SRM	0.3 SRM
Physical function	5.1	7.6	12.7	6.2	4.8	7.2
Role physical	5.7	8.5	14.2	6.6	5.7	8.5
Bodily pain	4.3	6.5	10.8	8.3	5.2	7.9
General health	4.3	6.5	10.9	8.9	4.3	6.4
Vitality	4.0	6.1	10.1	7.8	4.4	6.7
Social function	5.6	8.4	14.0	9.7	5.6	8.4
Role emotional	5.8	8.7	14.5	7.6	5.7	8.5
Mental health	4.0	6.1	10.1	7.8	4.2	6.3
Physical component score	1.9	2.9	4.9	2.2	2.0	3.1
Mental component score	2.6	4.0	6.6	3.5	2.8	4.1

Table 2.9 Summar	y of distribution-based	l estimates of clinica	l significance (N	/ID)
------------------	-------------------------	------------------------	-------------------	------

The mean change in SF-36 scores from baseline to 12 months in participants who underwent breast reduction surgery was in excess of MID threshold estimates based on a rule-of-thumb 10-point change on 100-point quality of life scales (King, 1996) or a 0.5 SD default value for patient-perceived important change (Norman et al., 2003), as well as additional distribution-based MID estimates derived in this study, in all eight SF-36 subscales. In addition, the mean change in SF-36 PCS and MCS scores was in excess of both the developer-recommended 3-point MID threshold, as well as condition-specific distribution-based estimates derived in this study. This is in contrast to scores for women in the breast hypertrophy control cohort which remained significantly lower and did not improve over the course of the study across all SF-36 subscales and summary measures. This finding is shown in Figure 2.3 by plotting the mean change from baseline to 12-

months in both surgical and hypertrophy control groups in comparison to the one-half standard deviation estimate of MID.



Figure 2.3 Mean change in SF-36 scores from baseline to 12-months for surgical and breast hypertrophy control cohorts in comparison to MID estimates (0.5 SD).

Error bars represent 95% confidence intervals.

2.3.3.2 Comparing the improvement in health-related quality of life with other surgical interventions

The improvement in SF-36 physical and mental summary scores in women who underwent surgery in this study were compared to existing studies which describe 12-month post-operative outcomes from other surgical interventions (Table 2.10). Breast reduction surgery provided a greater gain in SF-36 PCS scores than a coronary artery bypass graft and hemia repair and the improvement was similar to that experienced by patients undergoing total knee replacement surgery (Figure 2.4). The improvement in SF-36 MCS scores following breast reduction surgery exceeded that of all other surgical procedures (Figure 2.5).

Reference	Surgical Intervention	Pre-op PCS	Post-op PCS	Mean change in SF-36 PCS	Pre-op MCS	Post-op MCS	Mean change in SF-36 MCS	n
This study	Bilateral breast reduction	39.7	49.9	10.2	37.0	46.2	9.2	191
(Pivec et al., 2015)	Total knee replacement	33.0	47.8	14.8	52.9	55.9	3.0	281
(Stickles et al., 2001)	Total hip replacement	28.0	41.2	13.2	51.2	53.9	2.7	551
(Muller-Nordhorn et al., 2004)	Coronary artery bypass graft	36.0	43.0	7.3	45.0	50.0	4.3	412
(Polly et al., 2007)	Lumbar fusion (spine)	26.6	40.0	13.4	n/a	n/a	n/a	1826
(Rogmark et al., 2016)	Incisional hernia repair	41.6	49.5	8.1	50.2	52.3	1.7	124
(Faulconbridge et al., 2013)	Bariatric surgery	37.7	46.4	8.7	43.1	45.5	2.4	36

Table 2.10 Mean improvement in SF-36 PCS and MCS scores between surgical interventions

PCS, SF-36 Physical Component Score; MCS, SF-36 Mental Component Score; mean change in SF-36 score from pre-operative to 12 months post-operative; n/a, not applicable; *n*, number of participants.



Figure 2.4 Mean improvement in SF-36 Physical Component Summary (PCS) scores at 12m post-operative.



Figure 2.5 Mean improvement in SF-36 Mental Component Summary (MCS) scores at 12m post-operative.

2.3.3.3 The impact of participant characteristics on health-related quality of life and benefit of surgical intervention

There was a significant positive correlation between baseline body mass index (BMI) and the total amount of breast tissue resected at surgery. That is, as BMI increased there was an associated increase in the amount of breast tissue removed (Pearson r = 0.645, p < 0.001) (Figure 2.6).



Figure 2.6 Relationship between body mass index and tissue resection weight at surgery

When exploring the change in individual SF-36 subscale scores following surgery, scores for obese women (BMI \geq 30 kg/m²) improved equally, if not greater than, their non-obese counterparts (BMI <30 kg/m²) following surgery, reaching statistical significance for the physical functioning subscale (Table 2.11).

	Mean chan	ge (95% CI)	Difforance in means	n-value of
SF-36 scale	Non-obese (<i>n</i> = 71)	Obese (<i>n</i> = 138)	(95% CI)	difference [‡]
Physical function	16.1 (11.2 to 22.1)	23.4 (19.5 to 27.3)	6.8 (0.03 to 13.5)	0.050
Role physical	19.4 (12.4 to 26.3)	25.9 (21.2 to 30.5)	6.5 (-1.7 to 14.7)	0.121
Bodily pain	28.6 (22.8 to 34.5)	32.3 (27.8 to 36.9)	3.7 (-3.9 to 11.4)	0.337
General health	10.2 (6.0 to 14.3)	12.2 (8.4 to 16.0)	2.0 (-4.1 to 8.2)	0.516
Vitality	18.9 (14.8 to 23.1)	18.3 (14.2 to 22.4)	-0.7 (-7.2 to 5.9)	0.842
Social function	23.6 (17.8 to 29.4)	21.9 (16.6 to 27.2)	-1.7 (-10.2 to 6.8)	0.701
Role emotional	18.9 (12.8 to 25.0)	22.5 (17.2 to 27.8)	3.6 (-5.0 to 12.2)	0.409
Mental health	14.9 (11.4 to 18.5)	13.0 (9.2 to 16.9)	-1.9 (-7.9 to 4.1)	0.532

Table 2.11 Comparison of mean change in SF-36 scores following surgery in non-obese and obese participants

⁺ Using an independent *t*-test.

Obesity status: non-obese (<30 kg/m²), obese (≥30 kg/m²)

When exploring baseline SF-36 PCS scores, a significant negative correlation was found between SF-36 PCS scores and age (r = -0.13), BMI (r = -0.30), tissue resection weight (r = -0.26), degree of breast hypertrophy (r = -0.28) and ratio of breast to body volume (r = -0.19). Multivariate regression of candidate variables against baseline SF-36 PCS scores found BMI to be the only variable significantly related to pre-operative SF-36 PCS scores ($R^2 = 0.16$, p < 0.001). Following on from the regression analysis, this finding was supported with the results of the one-way ANOVA which was conducted to explore the impact of pre-operative BMI class on baseline SF-36 PCS scores. Participants were divided into five groups according to the WHO classification: normal weight, overweight, obese class I, obese class II, and obese class III. No participants were in the underweight category and therefore this was not included in the analysis. There was a statistically significant difference in baseline SF-36 PCS scores between the BMI categories (p < 0.001), with Dunnett's post-hoc test (two-tailed) revealing baseline SF-36 PCS scores were lower for patients within each of the 3 classes of obesity when compared to participants of a 'normal weight': (i) normal weight vs obese class I (mean difference = 6.3-points; 95% CI: 1.6 to 11.1, p = 0.006); (ii) normal weight vs obese class II (mean difference = 10.0-points; 95% CI: 4.8 to 15.1; p < 0.001); and (iii) non-obese vs obese class III (mean difference = 7.8-points; 95% CI: 2.2 to 13.5, p = 0.003). Scores were lower for 'overweight' participants in comparison to those of a normal weight, but the mean difference of 4.3-points but this did not reach statistical significance. The results comparing baseline SF-36 PCS score based on BMI classification are shown graphically in Figure 2.7.



Figure 2.7 Baseline SF-36 PCS scores for BMI categories

Multivariate linear regression analysis was also used to analyse predictors of the change in SF-36 PCS score following surgery and showed that improvement in SF-36 PCS scores was not significantly associated with any of these factors.

When stratifying the surgical participants by post-operative complication status, patients with and without a complication had significant improvements in SF-36 scores between baseline and 12 months in all eight SF-36 subscales as well as physical and mental summary scores (Table 2.12). When comparing the improvements in SF-36 scores following surgery between the groups based on the presence or absence of any complication, there were no significant differences in change in SF-36 scores across all eight subscales and summary measures (p > 0.05).

	Mean chang	ge (95% CI)	Difforence in	n-value of
SF-36 scale	No complication (<i>n</i> = 117)	Any complication (<i>n</i> = 92)	means (95% CI)	difference [‡]
Physical function	20.1 (15.9 to 24.2)	22.4 (17.3 to 27.6)	-2.4 (-8.9 to 4.1)	0.475
Role physical	25.0 (19.8 to 30.4)	21.9 (15.9 to 27.8)	3.2 (-4.7 to 11.1)	0.429
Bodily pain	33.3 (28.4 to 38.2)	28.3 (22.8 to 33.7)	5.0 (-2.3 to 12.3)	0.176
General health	11.2 (7.4 to 15.0)	11.9 (7.4 to 16.5)	-0.75 (-6.6 to 5.2)	0.801
Vitality	18.7 (14.4 to 23.1)	18.3 (13.9 to 22.6)	0.48 (-5.7 to 6.7)	0.879
Social functioning	24.1 (19.0 to 29.3)	20.4 (13.8 to 26.9)	3.8 (-4.4 to 11.9)	0.364
Role emotional	20.2 (14.9 to 25.5)	22.6 (16.2 to 29.1)	-2.5 (-10.7 to 5.8)	0.556
Mental health	13.1 (9.2 to 17.0)	14.4 (10.2 to 18.6)	-1.3 (-7.0 to 4.5)	0.662
Physical summary	10.3 (8.3 to 12.3)	9.1 (6.9 to 11.3)	1.2 (-1.7 to 4.2)	0.418
Mental summary	9.2 (6.5 to 12.0)	9.5 (6.5 to 12.4)	-0.2 (-4.2 to 3.8)	0.915

Table 2.12 Comparison of mean change in SF-36 scores by complication status

* Mean change between 12 month post-operative and pre-operative scores.

⁺ Independent samples *t*-test.

2.3.4 Quality of life outcomes using the MBSRQ

Mean MBSRQ scores for surgical and breast hypertrophy control participants are summarised in Table 2.13. When compared with previously published normative data from women within the United States general population, women with breast hypertrophy have significantly lower baseline scores for all scales. Within the surgical cohort, a comparison of mean pre-operative and 3 month post-operative MBSRQ scores showed that scores significantly improved in Appearance Evaluation, Fitness Evaluation, Health Evaluation, Health Orientation, Body Areas Satisfaction and Self-classified Weight (p < 0.05) (Table 2.13 and Figure 2.8). MBSRQ scores were stable at 6-and 12-months post-surgery and linear mixed-model analysis showed no significant difference from those at 3 months post-surgery.

Women with breast hypertrophy who were awaiting breast reduction surgery displayed lower mean scores across all subscales of the MBSRQ instrument in comparison to population norms (Table 2.13). Whereas women who underwent surgery displayed a significant improvement in scores across six scales of the MBSRQ, a comparison of mean baseline and 12-month scores for women in the breast hypertrophy control group showed that scores did not improve over time. In fact, mean scores for women in the breast hypertrophy control group remained low and unchanged over the course of the study, with the exception of Appearance Evaluation and Appearance Orientation which significantly declined over the 12-month period (p < 0.05) (Table 2.13). When comparing the surgical cohort and the breast hypertrophy control cohort at their final assessment, women who underwent surgery were more satisfied with their appearance, fitness levels, reported being in better health and were more satisfied with their body than those who did not undergo surgery (Table 2.13).

		Surgic	al group	Hypertrophy control		Normative [*]	
	Preop (n = 209)	3m postop (n=190)	6m postop (n = 183)	12m postop (n = 191)	Baseline (n = 161)	12m post (n = 124)	(n = 1070)
Appearance Evaluation	1.96 (0.64)	2.81 (0.70)*	2.83 (0.79)	2.87 (0.77) †	2.02 (0.66)	1.93 (0.58) [‡]	3.36 (0.87)
Appearance Orientation	3.57 (0.75)	3.53 (0.60)	3.53 (0.61)	3.50 (0.60)	3.63 (0.64)	3.53 (0.64) [‡]	3.91 (0.60)
Fitness Evaluation	3.10 (0.78)	3.31 (0.79)*	3.30 (0.86)	3.35 (0.80) †	3.05 (0.85)	2.95 (0.84)	3.48 (0.97)
Fitness Orientation	2.94 (0.64)	3.03 (0.61)	3.03 (0.65)	3.06 (0.68) †	2.86 (0.68)	2.86 (0.58)	3.20 (0.85)
Health Evaluation	2.99 (0.72)	3.29 (0.78)*	3.34 (0.73)	3.32 (0.75) †	2.86 (0.75)	2.84 (0.74)	3.86 (0.80)
Health Orientation	3.62 (0.62)	3.44 (0.59)*	3.54 (0.56)	3.50 (0.60) †	3.38 (0.64)	3.31 (0.64)	3.75 (0.70)
Illness Orientation	3.04 (0.78)	3.14 (0.75)	3.14 (0.80)	3.13 (0.77)	3.19 (0.82)	3.22 (0.74)	3.21 (0.84)
Body Areas Satisfaction	2.47 (0.56)	2.93 (0.61)*	2.99 (0.63)	3.00 (0.62) †	2.48 (0.58)	2.46 (0.58)	3.23 (0.74)
Overweight Preoccupation	3.31 (0.83)	3.17 (0.85)	3.16 (0.91)	3.23 (0.80)	3.34 (0.94)	3.34 (0.90)	3.03 (0.96)
Self-classified Weight	4.14 (0.71)	4.00 (0.64)*	3.98 (0.64)	4.00 (0.62)	4.00 (0.82)	4.09 (0.62)	3.57 (0.73)

Table 2.13 Mean (SD) MBSRQ scores for participants in the surgical and breast hypertrophy control cohorts in comparison with populations norms

* denotes significant difference (p < 0.05) between pre-operative and 3m post-operative scores using linear mixed-model analysis. † denotes significant difference (p < 0.05) between 12m post-operative scores for the surgical group and 12m post-baseline scores for the hypertrophy control group using an independent samples *t*-test.

[‡] denotes significant difference (p < 0.05) between baseline and 12m post-baseline scores in the hypertrophy control group using a paired *t*-test. [¥] Normative scores from females within the United States population (Cash, 2000).



Figure 2.8 Mean MBSRQ scores for surgical participants

2.3.5 Adjunct questions

The majority of participants (204/209, 97.6%) responded in the post-operative questionnaire that they would have the surgery again, whilst others were either unsure (4/209, 1.9%) or would not have surgery again (1/209, 0.5%). Following surgery, participants on average spent less money on medications and treatments (AU\$26.41 vs AU\$5.73 per month, p < 0.001) and took fewer days off work (4.5 days vs 0.1 days in the previous 6-month period, p = 0.009) when compared to before surgery. In contrast, no significant differences were observed in the breast hypertrophy control cohort when comparing spending on medications and number of days off work between baseline and 12 months following enrolment, with both remaining significantly higher than post-operative surgical participants (p < 0.001).

Surgical participants were asked to self-report whether or not they experienced any complications from the surgery; 115 (55%) stated they did not, 61 (29%) reported they did have a complication, and 11 (5%) were unsure. Surgical participants were also asked whether they experienced any new health issues or a worsening of an existing health issue that was not related to the surgery; 138 (66%) did not, 41 (20%) stated yes, and 8 (4%) were unsure. There were 22 cases (10%) of missing data for this supplementary study questionnaire.

2.4 Discussion

Findings from this prospective cohort study demonstrate that women with symptomatic breast hypertrophy have impaired quality of life compared to those in the general population. At baseline, participants in both the surgical and control breast hypertrophy groups scored significantly lower than the female general population in all SF-36 subscales, with pain being the most prominent. Surgical participants' quality of life improved following breast reduction to such an extent that the health deficits were eliminated at 3 months following surgery and quality of life was 'normalised' to levels equivalent to that of the general population across all dimensions. This normalisation effect was stable across 12 months follow-up. The SF-36 health gain ranged from 14.5 to 33.1 points for individual scale scores, and this exceeded the minimally important difference threshold estimates based on a rule-of-thumb of a 10-point change on 100-point sub-scales (King, 1996). Furthermore, estimates of the minimal important difference were derived from this study using a series of distribution-based analyses for each SF-36 subscale, with MIDs using the 0.5 SD and SEM approaches close to 10-points for many scales. Accordingly, the improvement in HRQoL exceeded all MID estimates in women who underwent surgery across all eight SF-36 subscales, supporting the proposition that breast reduction surgery provides a clinically relevant health benefit to women.

The finding that breast reduction surgery improves health-related quality of life to levels of the general population is supported by existing studies using the SF-36 within the literature (Blomqvist and Brandberg, 2004, Blomqvist et al., 2000, Kececi et al., 2015, Klassen et al., 1996, Miller et al., 2005, O'Blenes et al., 2006, Shakespeare and Cole, 1997). Interestingly, findings from a Swedish study in 2019 are contradictory to this and report that scores for General Health, Physical Functioning and Mental Health decreased at 1 year following surgery; with scores remaining significantly lower than population norms for these scales as well as the Vitality scale (Lewin et al., 2019). The authors did not comment on possible reasons for the post-operative decrease in scores across these scales. In addition, these findings are unexpected given that large gains were reported in physical and psychosocial wellbeing using another patient-reported outcome instrument in this study.

Secondary aims of this study were to investigate factors which have the potential to influence the level of improvement in quality of life following surgery: body mass index; degree of hypertrophy; bra cup size; age; pre-operative breast symmetry, disproportionality and weight of tissue resection at surgery. Several of these factors are frequently used to restrict access to breast reduction surgery, none of which are based on high quality evidence. In this study the improvement in health-related quality of life was independent of these factors, suggesting that all women with symptomatic breast hypertrophy can benefit from this surgery regardless of commonly scrutinised factors. This is of clinical relevance as it highlights the fact that women with a higher body mass

index or those with a lower weight of resection benefit as much as others requesting this surgery and should not be discriminated against based on criteria-based restrictions. The finding that symptom relief and improvement in health-related quality of life are not impacted by BMI and weight of resection is consistent with existing studies using the SF-36 (Collins et al., 2002, Eggert et al., 2009, Guemes et al., 2016, Lewin et al., 2019). In spite of these findings access restrictions for breast reduction surgery are in place in many countries including Australia, despite a lack of supporting evidence. The current study aims to contribute to the gap in the literature in the comprehensive assessment of the impact of patient and clinical characteristics on health-related quality of life outcomes following surgery.

When assessing the summary measures of health for the SF-36, the intervention effect of breast reduction surgery in our study was well in excess of the minimal clinically important difference for SF-36 physical and mental component score, which has been recommended by the developers as a 3-point change (Frendl and Ware, 2014, Ware et al., 1995, Ware et al., 1994). Furthermore, the mean change in SF-36 PCS and MCS scores following breast reduction surgery in this study were greater than the study-specific MID estimates of 4.9-points and 6.6-points using the half a standard deviation approach which has been recognised as a more universally accepted important difference (Norman et al., 2003). These MIDs for the SF-36 PCS and MCS were similar to published findings in patients undergoing surgery for degenerative cervical myelopathy, with scores reported as 4.63 and 6.76-points, respectively (Badhiwala et al., 2018). The improvements in the SF-36 physical component score at one year following surgery were comparable to those of other widely-accepted surgical interventions such as total hip and total knee replacement (Pivec et al., 2015), spinal fusion (Polly et al., 2007), bariatric surgery (Faulconbridge et al., 2013), and coronary artery bypass graft surgery (Muller-Nordhorn et al., 2004). The improvements in the mental component score following breast reduction surgery actually exceeded those of all other interventions cited. Breast reduction surgery is a relatively inexpensive procedure, and the improvement in health-related quality of life provides evidence as to the comparative effectiveness of this intervention in relieving the health burden and the functional symptoms of breast hypertrophy.

When assessing patient-reported outcomes using the MBSRQ, surgical patients rated their appearance significantly higher after breast reduction and rated themselves as being more physically fit post-operatively, despite their fitness orientation remaining unchanged. Importantly women who underwent surgery rated their bodies to be in better health following surgery and were more invested in leading a healthier lifestyle post-operatively. They rated themselves to be more content with areas of their body and viewed themselves as less overweight following surgery. These findings are consistent with an existing study using the MBSRQ in patients undergoing breast reduction surgery (Collins et al., 2002), and with prior outcome studies that used only the appearance evaluation subscale (Kerrigan et al., 2001, Thoma et al., 2013, Thoma et al., 2005).

To the author's knowledge, this study is the first to report the prospective assessment of surgical complications following breast reduction surgery using the standardised Clavien-Dindo Classification (CDC) system. An overall surgical complication rate of 44% was reported in this study, this is slightly lower than the rate of 63% reported in the retrospective study by Winter and colleagues using the CDC grading (Winter et al., 2017), and comparable with existing prospective studies such as the multi-institutional BRAVO study (Cunningham et al., 2005). However, this rate is slightly higher than those reported by previous studies (Guemes et al., 2016, Perez-Panzano et al., 2016, Srinivasaiah et al., 2014). This finding reiterates the need for standardised reporting of surgical complications, with the discrepancies likely to be attributable to inconsistencies in the collection and reporting of complications data. Whilst this study reported a relatively high infection rate of 17%, this figure included the prescription of oral antibiotics by General Practitioners outside of the hospital system, which may have been provided for presumed infection and without confirmation of bacterial infection by microbiological tests. It is acknowledged this may have overestimated the occurrence of surgical site infection in this study. This study also supports previous findings of no significant difference in the complication rate based on obesity status (Cunningham et al., 2005, Guemes et al., 2016, Wagner and Alfonso, 2005); further advocating that surgery should not be restricted to only those patients with a low BMI as the health benefits were found to outweigh the risks of post-operative complication. Lastly, whilst 44% of patients in this study experienced a surgical complication, no differences were found in the improvement in health-related quality of life across all dimensions between those who experienced a complication and those who did not.

A potential limitation of this study was the participant response rate for the breast hypertrophy control cohort was relatively low at 46 percent, which may be due to the recruitment process via postal questionnaire. Furthermore, whilst the total follow-up period for this cohort was 12 months, the intermediate timepoints of 3 and 6 months that were collected in the surgical cohort were not included in this cohort, although the consistency of outcomes at baseline and 12 months suggest that 3 and 6 month outcomes are likely to have been similar. Finally, estimation of the minimal important difference for the SF-36 patient-reported outcome in this study was limited to distribution-based approaches. Whilst there is a lack of consensus as to the best approach to approximate the minimal important difference, a more optimal approach may have been a combination of both anchor- and distribution-based methods if suitable clinical anchors were available (Revicki et al., 2008).

The strengths of this study were the prospective design, the relatively large sample size, and the inclusion of a non-surgical control sample of women with breast hypertrophy who were recruited from the same waiting list as those in the surgical cohort. Given the non-surgical hypertrophy control group were found to have a similar health deficit to the surgical cohort at baseline and showed no improvement in HRQoL over time, this cohort provided a meaningful reference to

accurately assess the impact of surgery on HRQoL. In addition, the post-operative outcomes described in this study included multiple time points over a 12-month period. Additionally, our surgical cohort were not biased by restrictions that have been reported in previous studies based on a minimum weight of resection or body mass index and therefore includes a broad spectrum across these variables. This is particularly important as it enables the accurate assessment of these factors as potential predictors of the change in health-related quality of life and outcomes of surgery and overcomes these limitations.

2.5 Conclusion

Breast hypertrophy is a painful condition which is effectively treated by breast reduction surgery. The marked improvement in quality of life following breast reduction surgery is comparable to other widely accepted and publicly funded surgical interventions. This study highlights that the improvement in quality of life following surgery is independent of traditionally used criteria based on body mass index or a minimum weight of resection and demonstrates the health benefits of surgery occur regardless of these factors. This confirms the clinical effectiveness of breast reduction surgery and supports the proposition that there is no justification for excluding women based on criteria such as body mass index or the extent of breast hypertrophy.

3. GENERATION OF AUSTRALIAN GENERAL POPULATION REFERENCE DATA FOR INTERPRETING THE BREAST-Q

A summary of the normative BREAST-Q data generated in this study was presented at the International Society for Quality of Life Research (ISOQOL) 25th annual conference in Dublin, Ireland (Crittenden et al., 2018b) and a version of this chapter has been published as an original article in the *Plastic and Reconstructive Surgery* journal (Crittenden et al., 2019b); attached as Appendix M.

3.1 Introduction

Health-related quality of life (HRQoL) data from a reference population such as the general population provides a valuable clinical reference point or as a benchmark for the interpretation of HRQoL data in outcome studies. General population reference values are available for generic instruments such as the Short Form-36 and have provided a valuable comparator for the assessment and interpretation of outcomes following breast reduction surgery. However, research studies have shown that potentially important differences exist when comparing HRQoL values between countries (Frieling et al., 2013, Hawthorne et al., 2007, Hopman et al., 2000, Mercieca-Bebber et al., 2019, Nolte et al., 2019, Norman et al., 2013). This highlights the importance of utilising country-specific population norms wherever possible for the accurate interpretation of HRQoL data in the clinical setting and for research studies. Furthermore, age and gender have been shown to influence patient-reported outcome scores and therefore is important to incorporate these key characteristics and distributions for general population reference values (Hjermstad et al., 1998).

Normative data for the BREAST-Q Reduction questionnaire provides valuable information as to the levels of breast-related satisfaction and quality of life within the general female population, enabling a better understanding of the health burden associated with breast hypertrophy and the health benefits of breast reduction surgery. However, normative BREAST-Q data are currently limited to a single population, the United States of America (Mundy et al., 2017c). Given that the BREAST-Q is one of the most widely used patent-reported outcome measures in breast surgery worldwide, the generation and reporting of Australian population-based reference data for comparison to women undergoing surgery provides an important contribution to the literature and a valuable reference for clinical care and future research studies.

The primary aim of this study was therefore to derive Australian general population reference values for the BREAST-Q Reduction module. This data will provide a population-specific comparison for the interpretation of HRQoL outcome data in women with breast hypertrophy who underwent breast reduction surgery (described in Chapter 4). Secondary aims were to explore the

association between socio-demographic characteristics and BREAST-Q scale scores, and to compare Australian values with United States BREAST-Q general population reference values.

3.2 Methods

3.2.1 Participant recruitment

Australian women aged 18 years and above who were registered members of a national online survey panel (Pureprofile Pty Ltd, Sydney, New South Wales, Australia; <u>www.pureprofile.com</u>) were invited to participate in the study in April 2018. Consenting respondents who self-reported as being without a history of breast cancer or breast surgery and not actively seeking breast surgery were eligible to participate in this study. Participants who were accepted into the study after answering the eligibility screening question and completed the questionnaire received a small monetary compensation (AUD\$3). Ethics approval was obtained from the Social and Behavioural Research Ethics Committee at Flinders University (SB REC approval number 7848) (Appendix G).

3.2.2 Data collection

Participants were required to self-complete the pre-operative version of the BREAST-Q Reduction module and provide information for the following relevant variables: age, height, weight, bra cup size, post code, employment status and health status. Participants within the normative study cohort were distributed across age and geographical variables in a manner consistent with the national representation population breakdown from the 2016 Census conducted by the Australian Bureau of Statistics (Australian Bureau of Statistics, 2016a). The corresponding target percentages for the age groups, taken from the Census, were as follows: 18–24 years (11.8%), 25–34 years (18.5%), 35–44 years (17.3%), 45–54 years (17.1%), 55–64 years (15.1%) and 65 years and above (20.2%). Target distribution across geographical variables according to the census breakdown were as follows: New South Wales (32.0%); Victoria (25.6%); Queensland (20.0%); South Australia (7.0%); Western Australia (10.5%); Australian Capital Territory (1.7%); Tasmania (2.1%); and Northern Territory (1.0%).

A Socio-Economic Indexes for Areas (SEIFA) measure of relative socio-economic status, the Index of Relative Socio-economic Disadvantage (IRSD), was obtained from postcodes; where a low index score indicates relatively greater disadvantage and a high index score indicates relative lack of disadvantage within an area (Australian Bureau of Statistics, 2016c). The SEIFA measure refers to the level of socio-economic status of the area in which the respondent lives, so is a broad measure of socio-economic status. IRSD scores were stratified into quintiles for further statistical analysis.

3.2.3 The BREAST-Q

The BREAST-Q Reduction module is a validated condition-specific patient-reported outcome

measure that was developed to specifically evaluate the outcomes of breast reduction surgery (Pusic et al., 2009). The BREAST-Q Reduction pre-operative module was used in this study to evaluate breast-related satisfaction and HRQoL in women within the general population (Appendix H). The instrument comprises Likert-type item responses across independently functioning patient satisfaction and wellbeing scales. The pre-operative version of the BREAST-Q comprises the following scales: Satisfaction with Breasts (n = 11 items), Psychosocial Wellbeing (n = 9 items), Sexual Wellbeing (n = 5 items) and Physical Wellbeing (n = 14 items). Questionnaire responses were transformed using Q-Score software (New York, N.Y.) to generate a score between 0 and 100, where a score of 0 is the minimum score and 100 is the maximum score and indicates the highest level of satisfaction or wellbeing. Administration of the electronic version of the BREAST-Q has previously been found to be psychometrically valid (Fuzesi et al., 2017).

3.2.4 Normative data from the United States

Normative data for the BREAST-Q Reduction module was first described in women within the United States population (Mundy et al., 2017c). Participants were recruited via the Army of Women (AOW), an online community of women, with and without breast cancer, started in 2008 by the Dr. Susan Love Research Foundation, with a mission to promote breast cancer research. An electronic survey link was sent to registered members and participants who self-reported they met the inclusion criteria were able to complete the pre-operative version of the BREAST-Q questionnaire and study-specific questions including demographic information, bra cup size, height and weight. Inclusion criteria included women aged 18 years and above, with no prior history of breast cancer or breast surgery, and the ability to complete an online questionnaire in English. A limited raw dataset from the Army of Women United States study was kindly made available by the authors under formal agreement with Dartmouth-Hitchcock Clinic, New Hampshire, United States of America (Appendix J). The dataset was therefore able to be used in a comprehensive comparison to generated Australian norms and analysis of the association between socio-demographic variables and BREAST-Q scores described in this chapter.

A comparison of age composition was made between the United States normative population sample and published age and sex composition data from the 2016 United States Census (United States Census Bureau, 2016).

3.2.5 Statistical analysis

Descriptive statistics for the four BREAST-Q scales were calculated, including the mean, standard deviation, 95% confidence intervals and the percent scoring at the lowest value (floor) and the highest value (ceiling). Cronbach's alpha coefficient was used to assess internal consistency (reliability) of the four multi-item BREAST-Q scales. A coefficient of 0.70 and above was considered acceptable. Comparisons of BREAST-Q scores were made between groups using an independent samples *t*-test, or a one-way analysis of variance (ANOVA) for three or more groups.

Socio-demographic variables were presented using descriptive summary statistics, or categorised and expressed as frequencies and percentages. Body mass index (BMI) was calculated from reported height and weight values (kg/m²) and categorised into non-obese (less than 30 kg/m²) and obese (greater than or equal to 30 kg/m²) status according to the World Health Organisation (WHO) classification (World Health Organization, 2018). Participant age was stratified into the following six subgroups: 18–24 years, 25–34 years, 35–44 years, 45–54 years, 55–64 years and 65 years and above. Bra cup size data were categorised into the following groups: less than A cup, A, B, C, D, DD, and greater than DD cup size. Bra cup size and age were further stratified into 2 groups for analysis: less than D cup and greater than or equal to D cup size, and less than 40 years versus greater than or equal to 40 years. Categorical variables were compared using Chi-square (X^2) statistics or Fisher's exact test as appropriate. Socio-demographic data from the population sample were compared with corresponding Australian general population reference values from the Australian Bureau of Statistics 2016 (Australian Bureau of Statistics, 2016a) using X^2 test to examine the representativeness of the study sample.

Multiple linear regression analysis was conducted to determine which socio-demographic variables were predictors for individual BREAST-Q scale scores derived from women in the Australian general population. Candidate variables included age, BMI, bra cup size, presence of a chronic health condition/s, employment status and Index of Relative Socio-economic Disadvantage (IRSD). Age and BMI were entered into the regression model as continuous variables; all other socio-demographic variables were categorical.

Socio-demographic characteristics and unadjusted mean BREAST-Q scores were compared between Australia and the United States population samples using a X^2 test or independent samples *t*-test, as appropriate. Cohen's *d* was used to provide a measure of effect size and was computed as the mean difference between groups divided by the pooled standard deviation for the two groups. Effect sizes were interpreted according to the thresholds established by Cohen as follows: <0.20, trivial; 0.20 to 0.50, small; >0.50 to 0.80, moderate; and >0.80, large (Cohen, 1988). To account for the differences in participant age distributions between the Australian and United States population samples when investigating differences in BREAST-Q scale scores, a one-way analysis of covariance (ANCOVA) model was used with age included as a potential confounder.

Statistical significance was established at a *p*-value of less than 0.05. Statistical analyses were performed using SPSS v25.0 statistical software (IBM Corp., Armonk, N.Y.).

3.3 Results

3.3.1 Participant characteristics

Normative BREAST-Q Reduction Module data were obtained from 513 female panellists within the Pureprofile Australia organisation. A total of 18 (3.4%) women were either ineligible or chose not

to proceed with participation in the study after viewing the study information page and eligibility screening question. Participant demographics from the Australian normative cohort are summarised in Table 3.1. The median patient age was 45 years (range, 18 to 88 years) with a mean body mass index of $27.8 \pm 7.0 \text{ kg/m}^2$. The majority of women were non-obese with a body mass index of less than 30 kg/m² (*n* = 356, 69%) and 43% of women had a bra size of at least a D cup. A chronic health condition was reported by 29% of respondents (*n* = 148). Most women were employed in either a part-time (*n* = 143, 28%) or full-time basis (*n* = 119, 23%).

The age distribution for the normative study is summarised in Table 3.1. Distribution of participants across the geographical variable were as follows: New South Wales (n = 164, 32%); Victoria (n = 133, 25.9%); Queensland (n = 101, 19.7%); South Australia (n = 36, 7.0%); Western Australia (n = 55, 10.7%); Australian Capital Territory (n = 8, 1.6%); Tasmania (n = 13, 2.5%); and Northern Territory (n = 3, 0.6%). Concerning age and geographical distribution, the sample was statistically representative of the Australian general female population, [$X^2(5) = 0.123, p = 1.0$] and [$X^2(5) = 0.195, p = 1.0$], respectively. Furthermore, the proportion of participants who were obese in the population sample was 31%; this was not significantly different from the Australian population data in which 30.2% of females were classed as obese (BMI ≥30 kg/m²) (Australian Bureau of Statistics, 2018).

Characteristic	No. (%)
Sample size	513
Age, years	
Mean age ± SD	46.1 ± 17.0
Median (range)	45 (18–88)
Age group (years):	
18–24	62 (12)
25–34	99 (19)
35–44	88 (17)
45–54	87 (17)
55–64	75 (15)
≥65	102 (20)
Mean BMI \pm SD, kg/m ²	27.8 ± 7.0
Obesity status:	
Non-obese (BMI < 30 kg/m²)	356 (69)
Obese (BMI \ge 30 kg/m ²)	157 (31)
Bra cup size:	
<a .<="" td=""><td>1 (0.2)</td>	1 (0.2)
Α	33 (6)
В	96 (19)
С	153 (30)
D	111 (22)
DD	69 (13)
>DD	50 (10)
Chronic health condition?	
Yes	148 (29)
No	365 (71)
Employment status	
Full-time	119 (23)
Part-time	143 (28)
Voluntary work	3 (1)
Homemaker	84 (16)
Student	26 (5)
Retired	85 (17)
Unable to work/disabled	21 (4)
Unemployed	24 (5)
Other	8 (2)
Index of Relative Socio-economic Disadvantage	
Lowest quintile	78 (15)
Low quintile	94 (18)
Middle quintile	122 (24)
High quintile	106 (21)
Highest quintile	113 (22)

Table 3.1 Participant socio-demographic characteristics

Values are numbers (percentage) unless stated otherwise. *BMI*, body mass index

3.3.2 Australian normative BREAST-Q Reduction module scores

Mean normative BREAST-Q scores from the general Australian population derived in this study are summarised in Table 3.2 and ranged from 49 to 71 on a transformed scale from 0 to 100.

Scale	n	Mean	SD
Satisfaction with Breasts	513	52	17
Psychosocial Wellbeing	513	55	22
Sexual Wellbeing	473	49	24
Physical Wellbeing	513	71	13

Table 3.2 Australian normative BREAST-Q Reduction module scores

Satisfaction with breasts, Psychosocial Wellbeing and Physical Wellbeing were scored for all 513 (100%) study participants. The Sexual Wellbeing scale was completed by 473 participants (92.2%). It is important to note that the Sexual Wellbeing scale was the only scale in which participants could respond 'not applicable' to some or all individual items.

Summary statistics for individual items and Cronbach's alpha coefficients for the multi-item scales are presented for reference in Appendix K. Internal consistency reliability coefficients, estimated by Cronbach's alpha, ranged from 0.89 to 0.97 on the four scales. These high coefficients supported scale reliability and validity. There did not appear to be a strong floor effect (proportion of subjects receiving the minimum possible score) or ceiling effect (proportion of subjects receiving the maximum possible score) in any of the four BREAST-Q scales. Item response frequencies from the Australian normative study are shown for each BREAST-Q scale: Satisfaction with Breasts (Figure 3.1), Psychosocial Wellbeing (Figure 3.2), Sexual Wellbeing (Figure 3.3) and Physical Wellbeing (Figure 3.4).



Figure 3.1 Item response frequencies 'Satisfaction with Breasts'



Figure 3.2 Item response frequencies 'Psychosocial Wellbeing'



Figure 3.3 Item response frequencies 'Sexual Wellbeing'



Figure 3.4 Item response frequencies 'Physical Wellbeing'

Normative BREAST-Q scores with relevant co-variables across the four scales which constitute the pre-operative version are shown in Figure 3.5 and are summarised in Table 3.3. An independent samples *t*-test was conducted to compare BREAST-Q scale scores between socio-demographic categorical variables. Participants with a higher BMI of 30 kg/m² or greater were found to have lower BREAST-Q scores across all scales when compared to those with a lower BMI of less than 30 kg/m² (p < 0.001). Participants who self-reported having a chronic health condition were also found to have significantly lower BREAST-Q scores across all scales (p < 0.001). Participants aged 40 years and above were found to have higher BREAST-Q scores in Satisfaction with Breasts (p = 0.014), Psychosocial Wellbeing (p < 0.001) and Physical Wellbeing scales (p =0.038). In contrast, no significant differences were observed for age in the Sexual Wellbeing scale (p = 0.404). When comparing bra cup size, those with a bra cup size at least D cup had significantly lower scores for Physical Wellbeing when compared to those with a smaller cup size (p = 0.002), whilst scores for the other three scales were not significantly different. An Index of Relative Socio-economic Disadvantage (IRSD) quintile was recorded for each participant and no significant differences were found when comparing BREAST-Q scores across all four scales using a one-way ANOVA.



Figure 3.5 Normative BREAST-Q Reduction module pre-operative scores

Error bars represent 95% confidence intervals. *Orange line* indicates mean score for individual BREAST-Q scales; *BMI*, body mass index; *Bra cup*, bra cup size; *Chronic*?, chronic health condition?

	Satis	Satisfaction with breasts		Psychosocial Wellbeing		Sexual /ellbeing	V	Physical Vellbeing
Variable	n	mean (SD)	N	mean (SD)	n	mean (SD)	n	mean (SD)
Age								
18–24	62	49.8 (13.7)	62	50.0 (19.8)	56	50.0 (24.1)	62	70.0 (12.6)
25–34	99	50.2 (16.1)	99	51.9 (19.9)	98	50.9 (23.1)	99	68.9 (13.7)
35–44	88	48.8 (18.4)	88	49.1 (21.6)	84	43.7 (23.6)	88	70.3 (12.8)
45–54	87	49.7 (16.1)	87	52.0 (22.3)	83	47.5 (23.4)	87	70.5 (12.8)
55–64	75	56.6 (18.1)	75	59.9 (21.3)	68	50.5 (24.8)	75	74.4 (12.7)
65+	102	56.1 (18.6)	102	64.8 (21.6)	84	49.0 (25.0)	102	73.2 (13.0)
BMI								
<30 kg/m ²	356	54.0 (16.7)	356	57.7 (21.3)	336	51.4 (23.4)	356	73.0 (12.9)
≥30 kg/m²	157	47.4 (17.8)	157	48.6 (21.7)	137	41.4 (23.9)	157	67.2 (12.5)
Bra cup size								
< D cup	283	52.8 (17.0)	283	55.8 (21.8)	258	50.1 (24.2)	283	72.9 (13.0)
≥ D cup	230	50.9 (17.6)	230	53.9 (21.9)	215	46.7 (23.6)	230	69.2 (12.9)
Chronic health cond	lition?							
Yes	148	48.6 (17.0)	148	49.4 (23.0)	131	39.6 (25.0)	148	66.6 (11.7)
No	365	53.3 (17.3)	365	57.2 (21.0)	342	51.9 (22.7)	365	73.1 (13.1)
Employment status								
Full-time	119	52.9 (17.1)	119	55.0 (23.2)	118	48.9 (23.1)	119	71.2 (14.1)
Part-time	143	50.8 (17.8)	143	54.8 (20.0)	135	50.8 (23.2)	143	71.2 (13.2)
Voluntary work	3	58.7 (11.0)	3	61.3 (27.7)	3	49.7 (18.3)	3	67.7 (4.2)
Homemaker	84	51.0 (19.2)	84	50.8 (22.7)	83	48.3 (24.5)	84	71.9 (13.4)
Student	26	49.7 (10.6)	26	51.8 (17.4)	19	44.2 (24.1)	26	71.2 (11.0)
Retired	85	56.1 (17.3)	85	64.6 (19.6)	69	50.5 (25.1)	85	73.6 (11.1)
Unable to work	21	48.0 (18.0)	21	45.4 (24.1)	20	33.0 (25.0)	21	63.9 (12.1)
Unemployed	24	45.8 (13.0)	24	45.5 (23.5)	20	44.7 (25.4)	24	67.6 (14.1)
Other	8	55.5 (12.6)	8	60.8 (13.5)	6	50 .0 (22.6)	8	70.4 (14.4)
IRSD								
Lowest quintile	78	51.9 (17.6)	78	54.1 (22.3)	70	46.6 (22.4)	78	71.2 (13.2)
Low quintile	94	49.2 (16.4)	94	52.8 (23.1)	83	47.8 (25.0)	94	72.0 (14.1)
Middle quintile	122	52.7 (18.2)	122	55.1 (21.9)	114	45.0 (25.2)	122	70.6 (12.8)
High quintile	106	53.1 (18.0)	106	56.4 (21.9)	99	51.2 (22.3)	106	71.2 (13.0)
Highest quintile	113	52.3 (16.1)	113	55.8 (20.5)	107	51.7 (24.0)	113	71.4 (12.7)
All participants	513	51.9 (17.3)	513	55.0 (21.8)	473	48.5 (24.0)	513	71.2 (13.1)

Table 3.3 Australian normative BREAST-Q scores by socio-demographic characteristics

3.3.3 Predictors of BREAST-Q scale scores

Multiple linear regression analysis was employed to determine which independent sociodemographic variables, if any, predicted each of the four individual BREAST-Q scale scores from the Australian general population sample. Regression models for each of the four BREAST-Q scales are shown in Table 3.4. Multiple regression models were run to predict BREAST-Q scale scores from candidate variables including age, BMI, bra cup size and presence of a chronic health condition. Results of the multiple linear regression indicated that age, BMI and the presence of a chronic health condition were statistically significantly independent predictors of 'Satisfaction with Breasts', F(4, 508) = 14.6, p < 0.001, R^2 of 0.103, explaining 10.3% of the variance in the data. Age, BMI and chronic health condition were significant independent predictors of 'Psychosocial Wellbeing', F(4, 508) = 29.3, p < 0.001, R^2 of 0.187, explaining 18.7% of variance in the BREAST-Q scale. Of the four candidate variables, BMI and chronic health condition were significant independent predictors of 'Sexual Wellbeing', F(4, 468) = 13.2, p < 0.001, R^2 of 0.101, explaining 10.1% of the variation in the wellbeing scale. Age, BMI, bra cup size and chronic health condition were significant independent predictors of 'Physical Wellbeing', F(4, 508) = 22.7, p < 0.011, R^2 of 0.152, explaining 15.2% of variance in the data.

Variable	Coefficient (β)	Standard error	95% CI	P-value
Satisfaction with breasts				
Intercept	49.6	5.16	39.5 to 59.7	< 0.001
Age	-0.24	0.05	0.15 to 0.33	< 0.001
BMI	-0.65	0.12	-0.88 to 0.41	< 0.001
Bra cup size	0.49	0.58	-0.65 to 1.63	0.400
Chronic health condition	4.19	1.67	0.90 to 7.47	0.013
Psychosocial Wellbeing				
Intercept	45.5	6.21	33.3 to 57.7	< 0.001
Age	0.45	0.05	0.34 to 0.55	< 0.001
BMI	-1.01	0.14	-1.29 to -0.73	< 0.001
Bra cup size	0.83	0.70	-0.54 to 2.20	0.233
Chronic health condition	7.74	2.01	3.79 to 11.7	< 0.001
Sexual Wellbeing				
Intercept	47.1	7.46	32.5 to 61.8	< 0.001
Age	0.11	0.07	-0.02 to 0.24	0.109
BMI	-0.81	0.17	-1.15 to -0.47	< 0.001
Bra cup size	0.34	0.83	-1.29 to 1.96	0.685
Chronic health condition	10.1	2.44	5.33 to 14.92	< 0.001
Physical Wellbeing				
Intercept	67.9	3.79	60.5 to 75.4	< 0.001
Age	0.17	0.03	0.11 to 0.24	< 0.001
BMI	-0.36	0.09	-0.53 to -0.18	< 0.001
Bra cup size	-1.26	0.43	-2.10 to -0.43	0.003
Chronic health condition	6.41	1.23	4.00 to 8.83	< 0.001

Table 3.4 Multiple linear regression models for BREAST-Q scales
3.3.4 Comparison of socio-demographic characteristics between normative population samples

Socio-demographic characteristics were compared between the previously published United States study (Mundy et al., 2017c) and the Australian population sample described in this study. Table 3.5 summarises the comparison of socio-demographic characteristics between population samples. The demographic characteristics of respondents were significantly different between the two population samples across all described variables including age, body mass index, employment status, and presence of a chronic health condition. Notably, the mean age for the US and Australian population sample was 54.7 years (SD 12.5) and 46.1 years (SD 17.0), respectively, with a mean difference of 8.6 years (95% CI, 7.2 to 10.1 years) (p < 0.001). Furthermore, the age distribution across categories was significantly different across the two groups as summarised in Table 3.5 (p < 0.001).

Whilst the normative BREAST-Q data derived for the Australian population in this study is considered representative of the general population by age and geographical breakdown (refer Section 3.3.1), a comparison of the published age and sex composition data from the 2016 United States Census (United States Census Bureau, 2016) indicates that this may not be the case for the United States data. According to the United States Census, the age distribution of females per age group were as follows: 9.2% (18–24 years); 13.5% (25–34 years); 12.5% (35–44 years); 13.4% (45–54 years); 13.1% (55–64 years); and 16.2% (65 years and above). In contrast, the age breakdown for the US normative study was: 0.5% (18–24 years); 7.7% (25–34 years); 14.2% (35–44 years); 20.0% (45–54 years); 33.3% (55–64 years); and 24.0% (65 years and above). A X^2 test demonstrated the age distribution of the population sample in the US study was therefore significantly different from the United States Census population breakdown and is potentially biased and represented by a higher proportion of participants of older age (X^2 (5) = 17.4; p = 0.004).

Variable	Australia	United States	<i>p</i> -value
No.	513	1206	
Mean age (SD), years	46.1 (17.0)	54.7 (12.5)	< 0.001
Median age (range), years	45 (18 – 88)	57 (20 – 86)	
Age group (years):			< 0.001
18–24	62 (12)	6 (0.5)	
25–34	99 (19)	93 (8)	
35–44	88 (17)	171 (14)	
45–54	87 (17)	241 (20)	
55–64	75 (15)	402 (33)	
65+	102 (20)	289 (24)	
Mean BMI (SD), kg/m²	27.8 (6.96)	26.8 (6.03)	0.004
Obesity status	х <i>у</i>	· · · ·	
Non-obese (BMI <30kg/m²)	356 (69)	912 (76)	0.004
Obese (BMI ≥30 kg/m²)	157 (31)	286 (24)	
Employment status			< 0.001
Full-time	119 (23)	511 (43)	
Part-time	143 (28)	168 (14)	
Voluntary	3 (1)	33 (3)	
Homemaker	84 (16)	85 (7)	
Student	26 (5)	16 (1)	
Retired	85 (17)	326 (27)	
Unable to work or disabled	21 (4)	12 (1)	
Unemployed or seeking employment	24 (5)	19 (2)	
Other	8 (2)	32 (3)	
Bra cup size			0.022
<a< td=""><td>1 (0.2)</td><td>22 (2)</td><td></td></a<>	1 (0.2)	22 (2)	
A	33 (6)	90 (8)	
В	96 (19)	287 (24)	
С	153 (30)	320 (27)	
D	111 (22)	232 (19)	
DD	69 (13)	149 (12)	
>DD	50 (10)	100 (8)	
Chronic health condition?			< 0.001
Yes	148 (29)	596 (50)	
No	365 (71)	606 (50)	

Table 3.5 Comparison of participant socio-demographics between normative population samples

Values are number (%) unless stated otherwise.

3.3.5 Comparison of normative BREAST-Q scores between normative population samples

Normative BREAST-Q Reduction module data were compared between the United States (Mundy et al., 2017c) and Australian general population reference values derived in this study. Table 3.6 compares the unadjusted mean scores for the four BREAST-Q scales. An independent *t*-test confirmed that the Australian mean scores were significantly lower than United States norms across all four pre-operative BREAST-Q scales, in particular for the Psychosocial Wellbeing scale with a mean difference of 13-points (95% CI, 10.5 to 14.6 points). Effect size calculations show a small effect in Satisfaction with Breasts, Sexual Wellbeing and Physical Wellbeing when comparing normative reference values between the two population samples. A moderate effect size was found when comparing population norms in the Psychosocial Wellbeing scale.

United States of America ⁺			Australia				Difference in		Effect size [‡]		
	N	Mean	SD	95% CI	N	Mean	SD	95% CI	means (95% CI)	<i>p</i> -value	(95% CI)
Satisfaction with Breasts	1205	56.7	16.3	56.1 – 57.9	513	51.9	17.3	50.4 - 53.4	4.7 (3.0 – 6.4)	<0.001	0.29 (0.19 – 0.39)
Psychosocial Wellbeing	1205	67.5	19.0	66.9 – 69.1	513	55.0	21.8	53.1 – 56.8	12.5 (10.5 – 14.6)	<0.001	0.63 (0.52 – 0.73)
Sexual Wellbeing	1024	55.2	19.4	53.8 – 56.2	473	48.5	24.0	46.4 - 50.7	6.7 (4.4 – 9.0)	<0.001	0.32 (0.21 – 0.43)
Physical Wellbeing	1205	75.7	10.9	75.4 – 76.6	513	71.2	13.1	70.1 – 72.4	4.5 (3.3 – 5.7)	<0.001	0.39 (0.28 – 0.49)

Table 3.6 Comparison of unadjusted normative BREAST-Q Reduction module scores for the United States of America and Australia

+ Source: (Mundy et al., 2017c).

‡ Effect size was calculated as the mean difference between groups, divided by the pooled standard deviation for the two groups.

The difficulty when comparing mean scores across multiple normative studies is that there are differences in the age distribution of the samples. Presentation of overall mean scores for health-related quality of life in a population sample may mask important differences in the distribution of scores across different socio-demographic variables. Given the significant differences in the age distribution between the two normative population samples (Table 3.5), a one-way ANCOVA was conducted to compare each of the four BREAST-Q scale scores between Australian and United States population samples whilst controlling for age. Results of the ANCOVA found a statistically significant difference remained when comparing the two normative studies across all four BREAST-Q scales after adjustment for age (Table 3.7).

A comprehensive comparison of BREAST-Q scores for the four scales stratified by sociodemographic characteristics for the Australian and United States population samples is shown in supplementary tables (Appendix L).

	Unit	ed States	s of America		Aust	ralia	Difference in	<i>p</i> -value of
	N	Mean	95% CI	N	Mean	95% CI	means (95% Cl)	difference
Satisfaction with Breasts	1205	56.5	55.5 – 57.4	513	52.3	50.9 - 53.8	4.1 (2.4 – 5.9)	<0.001
Psychosocial Wellbeing	1205	66.7	65.6 - 67.8	513	56.6	54.8 - 58.3	10.1 (8.0 – 12.2)	<0.001
Sexual Wellbeing	1024	55.3	54.0 - 56.6	473	48.3	46.4 - 50.3	7.0 (4.6 – 9.4)	<0.001
Physical Wellbeing	1205	75.6	74.9 – 76.2	513	71.7	70.6 – 72.7	3.9 (2.7 – 5.1)	<0.001

Table 3.7 Comparison of age-adjusted normative BREAST-Q Reduction module scores

3.4 Discussion

This study is the first to report Australian general population reference values for the BREAST-Q Reduction module. These values establish the levels of breast-related satisfaction and quality of life within the female general population and provide a valuable population-specific comparison for the interpretation of BREAST-Q scores for women with breast hypertrophy undergoing surgery. Such comparisons facilitate the ability to quantify the impact of a condition or disease on health-related quality of life, and also the success of a surgical intervention in comparison to levels within the general population.

Within the normative study, participants of an older age were found to report higher scores in three out of the four scales: Satisfaction with Breasts, Psychosocial Wellbeing and Physical Wellbeing. Furthermore, a larger body mass index and presence of a chronic health condition was found to be associated with lower scores in all BREAST-Q scales of participant satisfaction and quality of life than their respective comparison groups. Increased bra cup size was associated with lower Physical Wellbeing scores. These findings are in agreement with the study by Mundy and colleagues (Mundy et al., 2017c) and highlight the importance of capturing socio-demographic information and comorbidities when conducting health-related quality of life studies.

Prior to this study, published general population reference values for the BREAST-Q Reduction module were limited to a single study and were based on data from the United States population. An interesting finding from this study was that sizeable differences were found in mean BREAST-Q scores when comparing Australian values to reported United States norms, in particular for Psychosocial Wellbeing. A number of reasons for this difference between general population reference values might be postulated, whether it is something related to the Australian population or whether it is due to variations in the populations sampled within each study. It is possible that participants in the United States study recruited through the Army of Women, with a primary focus of connecting breast cancer researchers with women with and without breast cancer, had a different awareness and motivation for participating in the study which may have impacted their responses. When comparing socio-demographic characteristics between the two normative studies pronounced differences were observed, particularly in the age distribution of the population samples. Whilst it was plausible that this may have accounted for the observed differences in BREAST-Q scores as age is often an important predictor of self-reported health status (Hjermstad et al., 1998), on comparison of age-adjusted BREAST-Q scores the significant differences remained between the two general population samples across all four BREAST-Q scales. This finding highlights the importance of utilising country-specific norms for guality of life studies wherever possible, as potentially important differences do exist between populations. Furthermore, this finding is consistent with previous studies which have identified differences between reference values in Australia and those from other countries for various health-related

94

quality of life measures including the Short Form-36, Short Form-6D, EQ-5D and EORTC QLQ-C30 (Frieling et al., 2013, Hawthorne et al., 2007, McCaffrey et al., 2016, Mercieca-Bebber et al., 2019, Norman et al., 2013, Roser et al., 2019, Watson et al., 1996). As an example of the importance of using country-specific reference values, the threshold for a small but clinically important difference has been estimated by a rule-of-thumb approach as 10-points on a 100-point scale (King, 1996). When comparing 'Psychosocial Wellbeing', the United States reference value for this scale is 10.1-points higher than the Australian reference value generated in this study; which therefore represents a clinically meaningful difference; the interpretation of a mean score for a patient presenting for and undergoing breast reduction surgery in Australia would therefore be inaccurate if compared with the United States reference values.

Strengths of this study include the generation of population-specific reference values for the BREAST-Q in order to provide an accurate comparison for the interpretation of health-related quality of life between surgical patients and population norms (see Chapter 4). Furthermore, the population sample described in this study was representative of the general Australian population breakdown by age and geographical location. This enforces the likelihood that this study captured an accurate representation of women in the general population, and overcomes limitations described in the normative study from the US population sample which was potentially biased by not being representative of the age distribution of the United States general population. This is particularly important given that this study demonstrated that age was a significant predictor of BREAST-Q scores in three out of the four pre-operative scales. Finally, given that the BREAST-Q is the most widely used patient-reported outcome measure in breast surgery, this study is valuable because it provides general population reference data for the BREAST-Q Reduction module from a second, diverse population set, and highlights that within this context potentially important differences exist in health-related quality of life between populations.

This study has several potential limitations that should be noted. Firstly, the Pureprofile panel used for participant recruitment in this study consists exclusively of members of the public who voluntarily enrol to complete surveys, and this might introduce an element of selection bias. Bias could be attributable to various factors including: the self-selection of a biased population sample by Pureprofile; or participants who elect to complete online surveys may not represent the population at large. Secondly, data was not collected on socio-demographic characteristics including income, country of birth, marital status and highest level of education, so we could not assess the representativeness of our population sample across these variables. Despite these limitations, online survey organisations represent an efficient method of obtaining a large, representative sample of the general population and consistent with country-specific population breakdown distribution across factors including age and geographic variables. Finally, whilst we have reported normative data from a second diverse population sample, the normative values were derived from another relatively wealthy country. The absence of data from low- and middle-income

95

counties means it is unlikely that this can be representative of worldwide normative data.

3.5 Conclusion

This study reports the first BREAST-Q Reduction module general population reference values specific to Australia, and only the second set of population reference values reported globally. These values will assist in the interpretation of BREAST-Q scale scores for Australian women with breast hypertrophy who undergo reduction mammaplasty, both in existing data and in future studies. This study confirmed that general population reference values for HRQoL measures may differ between populations and highlights the importance of utilising country-specific normative data wherever possible for the accurate interpretation of HRQoL scores in the clinical setting. Lastly, findings from this study highlight the importance of presenting socio-demographic data together with patient-reported outcome data and note which characteristics have a significant influence on HRQoL outcome data.

4. OUTCOMES OF BREAST REDUCTION SURGERY USING THE BREAST-Q: A PROSPECTIVE STUDY AND COMPARISON WITH NORMATIVE DATA

A version of this chapter has been published as an original article in the Plastic and Reconstructive Surgery journal (Crittenden et al., 2019b); attached as Appendix M.

4.1 Introduction

The BREAST-Q is a condition-specific patient-reported outcome measure that was developed and validated by Pusic and colleagues to assess patient satisfaction and HRQoL following cosmetic, functional and reconstructive breast surgery (Pusic et al., 2009). Most reported studies using the BREAST-Q Reduction module to assess outcomes following breast reduction surgery have been limited by their retrospective design, and lack of comparison of scores with a pre-operative baseline (Cabral et al., 2017a, Cogliandro et al., 2017a, Cohen et al., 2016a, Coriddi et al., 2013, Correa et al., 2018, Gonzalez et al., 2012). The generation of Australian population reference values for the BREAST-Q described in Chapter 3 provides a population-specific benchmark for the interpretation of scores derived from the surgical and breast hypertrophy control cohorts detailed in this chapter. Finally, in order to further provide a meaningful interpretation of HRQoL scores, an estimate for the minimal important difference (MID) for the BREAST-Q Reduction module was established. To date, no published MID estimates exist for the reduction version of the BREAST-Q patient-reported outcome measure.

This chapter describes a prospective study using the BREAST-Q to assess outcomes following breast reduction surgery. The primary objective was to assess breast-related satisfaction and health-related quality of life in Australian women with breast hypertrophy before and after breast reduction surgery. Secondary objectives were to compare levels of breast-related satisfaction and wellbeing in a control group of women with breast hypertrophy who did not undergo surgery; and compare to normative data derived from an Australian population (Chapter 3) and to previously published United States norms. Finally, many providers commonly determine the medical necessity for reduction mammaplasty using the Schnur sliding scale based on body surface area-adjusted minimum resection weight calculations; a scale which was never intended for this purpose and has since been misused (Dvoracek et al., 2019, Koltz et al., 2013, Schnur, 1999, Schnur et al., 1991). This study therefore aimed to evaluate the influence of factors including a minimum weight of resection on HRQoL outcomes following breast reduction surgery.

4.2 Participants and methods

4.2.1 Surgical cohort

The participants in the surgical cohort are introduced and described in detail in Chapter 2. However, a point of difference in this study was that participants self-completed the conditionspecific BREAST-Q Reduction module patient-reported outcome measure at two timepoints, preoperatively and 12 months post-operatively. In addition, the number of participants within the surgical cohort described in this chapter was lower as the BREAST-Q patient-reported outcome measure was not developed until after the commencement of this study and was introduced into the study in March of 2010.

This study therefore includes women aged 18 years and above with symptomatic breast hypertrophy who were eligible for bilateral breast reduction surgery between March 2010 and February 2016. A detailed description of participant recruitment and data collected for surgical participants is detailed in Chapter 2 section 2.2.1. In brief, age, height, weight, bra cup size and smoking status were recorded at the time of enrolment. Intraoperatively, the surgical technique and weight of tissue resected from each breast was documented. Post-operatively, a "comprehensive complications assessment" checklist was completed by one of the doctors in the treating team (Appendix F).

Pre-operative height and weight measurements were used to calculate body surface area (BSA) for individual surgical participants using the DuBois and DuBois formula: BSA $(m^2) = 0.007184 \text{ x}$ height $(cm)^{0.725} \text{ x}$ weight $(kg)^{0.425}$ and the patients were subsequently divided into two groups depending on whether or not they met Schnur's 22nd percentile criteria based on BSA and required volume of tissue resected at surgery (Schnur et al., 1991).

4.2.2 Breast hypertrophy control cohort

The participants in the breast hypertrophy control cohort are introduced and described in detail in Chapter 2 section 2.2.1. In brief, eligible participants were mailed study information and questionnaires and consenting participants were required to self-complete and return the BREAST-Q questionnaire and provide information for relevant variables such as age, height, weight, bra cup size and health status at a baseline time-point and 12 months following.

4.2.3 The BREAST-Q

The BREAST-Q condition-specific patient-reported outcome measure is described in detail in Chapter 3. In this study, the BREAST-Q Reduction module was used to evaluate breast-related satisfaction and HRQoL in women with breast hypertrophy who underwent breast reduction surgery, in a control group of women with breast hypertrophy, and a comparison made to normative population BREAST-Q data generated in Chapter 3. The pre-operative version of the BREAST-Q comprises the following scales: Satisfaction with Breasts, Psychosocial Wellbeing, Sexual Wellbeing and Physical Wellbeing (Appendix H). Additionally, the post-operative version includes: Satisfaction with Outcome (n = 8 items), Satisfaction with Nipple (n = 5 items), Satisfaction with Information (n = 12 items), Satisfaction with Surgeon (n = 12 items), Satisfaction with Medical Staff (n = 7 items) and Satisfaction with Office Staff (n = 7 items) (Pusic et al., 2009) (Appendix I). Questionnaire responses were transformed using the Q-Score software (New York, NY; <u>http://qportfolio.org/score-breast-q-reduction-mastopexy</u>) to generate a score between 0 and 100, where a score of 0 is the minimum score and 100 is the maximum score and indicates the highest level of satisfaction or wellbeing.

4.2.4 Minimal important difference estimates

Distribution-based methods were used to estimate a clinically meaningful change, or MID, in BREAST-Q scores. Distribution-based approaches rely on statistical methods to express an effect in terms of providing a measure of sample variability to estimate the MID. Four alternate distribution-based approaches were used to estimate a clinically meaningful change in BREAST-Q scores including: one-half a standard deviation; effect size; standardised response mean; and standard error of measurement. These approaches were described in detail in Chapter 2 section 2.2.3.

4.2.5 Statistical analysis

Statistical analyses were performed using SPSS v25.0 statistical software (IBM Corp., Armonk, N.Y.). Descriptive statistics including mean, standard deviation and 95% confidence intervals (CI) were computed for continuous variables. To assess the significance of changes in HRQoL, mean pre-operative and 12 months post-operative BREAST-Q scores were compared using a paired ttest for dependent samples. Change effect size statistics were computed as the mean difference between pre-operative and post-operative BREAST-Q scores divided by the standard deviation (SD) of the baseline score. Change effect sizes were interpreted according to the thresholds established by Cohen as follows: <0.20, trivial; 0.20 to 0.50, small; >0.50 to 0.80, moderate; and >0.80, large (Cohen, 1988). Comparisons were made between groups using a two-sample *t*-test for continuous data and a difference in means with 95% confidence intervals. A chi-square or Fisher's exact test were used to compare categorical variables between groups. BREAST-Q scores from the surgical cohort were compared to Australian population normative data using an independent samples t-test. The Pearson correlation coefficient (r) was used to investigate the strength of association between two continuous variables. Multiple linear regression analysis was used to explore the relationship between BREAST-Q scores and independent demographic variables. Statistical significance was established at a *p*-value of less than 0.05.

4.3 Results

4.3.1 Study participants

4.3.1.1 Surgical cohort

One hundred and sixty-eight eligible participants who underwent reduction mammaplasty between March 2010 and February 2016 completed the BREAST-Q questionnaire pre-operatively and 156 completed the questionnaire post-operatively. Of these, 132 participants (76%) completed the BREAST-Q Reduction module at both time points. Participant demographics from the surgical cohort are summarised in Table 4.1. The median age was 42 years (range, 18 to 72 years) with a mean \pm SD body mass index (BMI) of 32.1 \pm 5.7 kg/m². The mean total weight of breast tissue resected at surgery was 1298.7 \pm 824.7 grams. The inferior pedicle was used in the majority (78%) of surgeries, with the superomedial in 19%.

4.3.1.2 Breast hypertrophy control cohort

Study questionnaires were initially posted to 350 women; 160 respondents (46%) completed and returned the questionnaires at baseline and of these, 124 responded 12 months later. Twenty-two of those contacted to participate in the study underwent breast reduction surgery during the study timeframe and were therefore ineligible to participate. Participant demographics for the hypertrophy control cohort are summarised in Table 4.1. The mean participant age was 45.6 years (range, 20 to 79 years) and the mean BMI was $31.6 \pm 6.4 \text{ kg/m}^2$. The most common bra cup sizes were F (23%) and G (28%).

There were no statistically significant differences in baseline characteristics of participants who completed BREAST-Q questionnaires at both study timepoints including age, BMI, obesity status, or smoking history between the surgical group and hypertrophy control group (Table 4.1).

Variable	Surgical cohort	Hypertrophy control cohort	P value of
	(<i>n</i> = 132)	(<i>n</i> = 124)	difference ⁺
Median (range) age (years)	42 (18 – 72)	47 (20 – 79)	
Mean (± SD) age (years)	42.1 ± 13.6	45.3 ± 13.1	0.058
Age group (years):			
18–24	17 (13)	12 (9.7)	
25–34	24 (18)	15 (12.1)	
35–44	42 (32)	26 (21.0)	
45–54	25 (19)	43 (34.7)	
55–64	17 (13)	21 (16.9)	
65+	6 (5)	7 (5.6)	
Mean (± SD) BMI (kg/m²)	32.1 ± 5.7	32.1 ± 6.0	0.960
Obesity status:			0.519
non-obese (BMI <30 kg/m²)	47 (36)	49 (39.5)	
obese (BMI ≥30 kg/m²)	85 (64)	74 (59.7)	
Missing	0 (0)	1 (0)	
Smoking Status:			0.686
Non-smoker	77 (58)	79 (63.7)	
Current	21 (16)	14 (11.3)	
Ex-smoker <12 months	8 (6)	6 (4.8)	
Ex-smoker >12 months	26 (20)	25 (20.2)	
Bra cup size:			
C	1 (0.76)	0 (0)	
D	5 (3.8)	4 (3.2)	
DD	31 (23.5)	13 (10.5)	
E	29 (22.0)	19 (15.3)	
F	27 (20.0)	27 (21.8)	
G	23 (17)	37 (29.8)	
Н	7 (5)	15 (12.1)	
>H	8 (6)	4 (3.2)	
Missing	0 (0)	5 (4)	

Table 4.1 Baseline characteristics of participants.

Values are numbers (percentages) unless stated otherwise.

+ Using independent samples *t*-test or chi-square test as appropriate.

4.3.2 BREAST-Q scores before and after breast reduction surgery

Summary statistics for the BREAST-Q scores measured in the surgical participants are presented in Table 4.3. Statistically significant improvements were found across all BREAST-Q scales measured pre- and post-operatively; Satisfaction with Breasts, Psychosocial Wellbeing, Sexual Wellbeing, and Physical Wellbeing, when compared to pre-operative scores (p < 0.001) (Table 4.2). Satisfaction with Breasts improved with a mean ± SD change of 51.4 ± 20.3; Psychosocial Wellbeing improved with a mean change of 36.9 ± 21.7; Sexual Wellbeing improved with a mean change of 31.4 ± 25.0; and Physical Wellbeing improved with a mean change of 32.7 ± 17.9. Change effect size calculations show a very large effect in all BREAST-Q scales when comparing pre-operative and post-operative scores: Satisfaction with breasts (effect size 3.6; 95% Cl, 3.2 to 3.9); Psychosocial Wellbeing (effect size 2.2; 95% Cl, 1.9 to 2.5); Sexual Wellbeing (effect size 1.6; 95% Cl, 1.3 to 1.9); and Physical Wellbeing (effect size 2.3; 95% Cl, 2.0 to 2.7).

		Mea	n (SD)	Maan ahanga		
Scale	n	Pre-operative	Post-operative		<i>p-</i> value [‡]	
	11	score	score	(95% CI)		
Satisfaction with Breasts	132	21.9 (10.9)	73.3 (17.3)	51.4 (47.9 - 54.9)	<0.001	
Psychosocial Wellbeing	132	32.4 (12.8)	69.3 (19.8)	36.9 (33.1 - 40.6)	<0.001	
Sexual Wellbeing	111	30.4 (17.6)	61.9 (22.1)	31.4 (26.7 - 36.1)	<0.001	
Physical Wellbeing	132	44.0 (15.5)	76.6 (12.1)	32.7 (29.6 - 35.7)	<0.001	

Table 4.2 BREAST-Q scores before and after breast reduction	surgery
---	---------

[‡] Paired-samples *t*-test.

Patient photos before and 12-months after breast reduction surgery demonstrate BREAST-Q scores at both time points, with the magnitude of change represented with by the percentage increase in scores following surgery (Figure 4.1 and Figure 4.2).



BREAST-Q scale	Pre-operative	12m post-operative	% change
Satisfaction with breasts	23	90	291%
Psychosocial Wellbeing	43	77	79%
Sexual Wellbeing	46	100	204%
Physical Wellbeing	56	79	41%

Figure 4.1 Pre- and post-operative photos of a patient who underwent an inferior pedicle Wise pattern breast reduction with corresponding BREAST-Q scores



BREAST-Q scale	Pre-operative	12m post-operative	% change
Satisfaction with breasts	29	57	97%
Psychosocial Wellbeing	30	70	133%
Sexual Wellbeing	40	56	40%
Physical Wellbeing	30	61	103%

Figure 4.2 Pre- and post-operative photos of a patient who underwent a Hall-Findlay pattern breast reduction with corresponding BREAST-Q scores

An independent *t*-test was conducted to compare BREAST-Q scores between socio-demographic categorical variables. Figure 4.3 demonstrates that participants with a higher BMI of 30 kg/m² or greater (n = 85, 64%) had a significantly greater mean change in Satisfaction with Breasts (p = 0.027), Psychosocial Wellbeing (p = 0.031), and Physical Wellbeing (p = 0.003) than those with a body mass index less than 30 kg/m² (n = 47, 36%) when comparing pre-operative to 12 month post-operative scores. Whilst participants with a higher BMI were also found to have a greater improvement following surgery in 'Sexual Wellbeing', this difference did not reach statistical significance (p = 0.101). Participants aged 40 years and above had a greater mean change in Satisfaction with Breasts (p = 0.050) and Psychosocial Wellbeing (p = 0.022) than those less than 40 years of age (Figure 4.3).



Figure 4.3 Mean change in BREAST-Q scores following surgery.

Error bars represent 95% confidence intervals. Orange line indicates mean score for individual BREAST-Q scales.

Mean change in BREAST-Q scores was not significantly different between patients who had a post-operative surgical complication and those who did not (Table 4.3).

Soolo	Surgical co	mplication	Difference in	n voluo	
Scale	Yes (<i>n</i> = 45)	No (<i>n</i> = 87)	means (95% CI)	<i>p</i> -value	
Satisfaction with Breasts	52.1 (21.3)	51.0 (19.8)	-1.1 (-8.5 to 6.3)	0.77	
Psychosocial Wellbeing	34.1 (19.7)	38.3 (22.7)	4.2 (-3.6 to 12.1)	0.29	
Sexual Wellbeing	30.0 (24.8)	32.3 (25.2)	2.3 (-7.5 to 12.1)	0.64	
Physical Wellbeing	31.3 (17.7)	33.3 (18.1)	2.0 (-4.5 to 8.5)	0.55	

Table 4.3 Significance of surgical complications on mean change (SD) in BREAST-Q scores

4.3.3 BREAST-Q scores in women with breast hypertrophy awaiting surgery

Mean BREAST-Q scores for women in the breast hypertrophy control group who did not undergo breast reduction surgery during the 12-month study follow-up are described in Table 4.4. BREAST-Q scores were lower when comparing 12 months to baseline scores across all scales, with scores for Sexual Wellbeing reaching statistical significance using a paired samples *t*-test (p = 0.01).

Mean (SD) Mean change 12m post-Scale **Baseline** *p*-value (95% CI) n score baseline score -1.8(-3.9-0.2)Satisfaction with Breasts 124 22.1 (12.1) 20.3 (12.0) 0.08 Psychosocial Wellbeing -1.2(-3.2-0.8)124 29.0 (13.8) 27.8 (14.8) 0.23 -3.7(-6.5-0.9)Sexual Wellbeing 110 25.1 (17.2) 21.4 (16.1) 0.01 Physical Wellbeing 124 42.1 (14.9) 41.2 (16.1) -0.9 (-3.2 - 1.4) 0.44

Table 4.4 BREAST-Q scores in the breast hypertrophy control group

Mean BREAST-Q scores for women in the breast hypertrophy control group remained significantly lower than post-operative scores for women who underwent surgery across all scales: Satisfaction with breasts (mean difference 53.0-points; 95% CI, 49.4 to 56.7, p < 0.001); Psychosocial Wellbeing (mean difference 41.5-points; 95% CI, 37.1 to 45.8, p < 0.001); Sexual Wellbeing (mean difference 39.3-points; 95% CI, 34.0 to 44.5, p < 0.001); and Physical Wellbeing (mean difference 35.4-points; 95% CI, 31.9 to 38.9, p < 0.001).

4.3.4 Distribution-based estimates of MID

Table 4.5 presents the range of estimated minimal important difference (MID) values for each scale of the Reduction module of the BREAST-Q. MID estimates varied by scale, ranging from 6-points to 9-points based on the one-half standard deviation approach, and 5-points to 8-points using the one-third standardised response mean. MID estimates were lower using the standard error of

measurement analyses based on the reliability of each subscale and ranged from 2.9-points to 6.4-points.

	Distr	ibution-ba	sed approa	MID estimate		
Scale	0.5 SD	0.2 SRM	0.3 SRM	SEM	Based on 0.5 SD	Based on 0.3 SRM
Satisfaction with Breasts	5.5	4.1	6.1	3.3	6	6
Psychosocial Wellbeing	6.4	4.3	6.5	2.9	6	7
Sexual Wellbeing	8.8	5.0	7.5	4.7	9	8
Physical Wellbeing	7.8	3.6	5.4	6.4	8	5

Table 4.5 BREAST-Q Reduction module minimal important difference estimates

The improvement in BREAST-Q scores in participants who underwent breast reduction surgery was well in excess of all estimates of minimal important difference across all four scales. This finding is shown in Figure 4.4 by plotting the mean change from baseline to 12-months in both surgical and hypertrophy control groups in comparison to the one-half standard deviation estimate of minimal important difference.



Figure 4.4 Mean change in BREAST-Q scores from baseline to 12-months for surgical and breast hypertrophy control groups in comparison to MID estimate.

Error bars represent 95% confidence intervals.

4.3.5 Comparison of BREAST-Q scores between women with breast hypertrophy and women in the general Australian population

Normative BREAST-Q scores generated from Australian women (Chapter 3) were compared to scores from patients in the surgical cohort with symptomatic breast hypertrophy who proceeded to reduction mammaplasty. Figure 4.5 demonstrates mean BREAST-Q scores for surgical patients before and 12 months after surgery in comparison to norms for Australia and the United States. In comparison to normative values, women with breast hypertrophy who were awaiting surgery had significantly lower BREAST-Q scores pre-operatively across all scales (p < 0.001). Post-operatively, mean scores significantly improved to levels at least that of the normative Australian population.





Error bars represent 95% confidence intervals.

Mean BREAST-Q scores from "satisfaction" scales that are captured in the post-operative version are presented in Figure 4.6. High levels of satisfaction were observed across all scales included in the post-operative version. As an example, the mean score for overall 'Satisfaction with Outcome' was 87-points at 12-months following surgery.



Figure 4.6 Post-operative only BREAST-Q Reduction Module satisfaction scale scores. Error bars represent 95% confidence intervals.

Results of the Pearson correlation indicated that there was a strong, positive association between Satisfaction with Outcome and Satisfaction with Breasts (r = 0.7; p < 0.001), Satisfaction with Surgeon (r = 0.7; p < 0.001) and Satisfaction with Information (r = 0.6; p < 0.001) (Table 4.6). Weak positive correlations were observed between Satisfaction with Outcome and the remaining BREAST-Q scales.

Table 4.6 Correlation between Satisfaction with Outcome and post-operative BREAST-Q scale

Scale	Pearson's r	<i>p</i> value
Satisfaction with Breast	0.7	< 0.001
Psychosocial Wellbeing	0.3	0.001
Sexual Wellbeing	0.3	0.001
Physical Wellbeing	0.4	< 0.001
Satisfaction with Information	0.6	< 0.001
Satisfaction with Nipples	0.4	< 0.001
Satisfaction with Surgeon	0.7	< 0.001
Satisfaction with Medical Staff	0.3	< 0.001
Satisfaction with Office Staff	0.2	0.013

When considering 12-month outcomes, no differences in 'Satisfaction with Outcome' BREAST-Q scores were found between patients who experienced a complication (mean 85.1, SD 16.6) and those who did not (mean 87.6, SD 18.0). Participants aged 40 years and older (mean 90.0, SD 13.6) had significantly higher mean Satisfaction with Outcome scores than younger participants (mean 82.5, SD 21.3; p = 0.015). Patients with a higher body mass index of 30 kg/m² or greater were found to have significantly higher outcome scores (Table 4.7). Scores were not significantly different between participants who met traditional insurance criteria requirements (Schnur sliding scale and the 500g minimum rule) and those who did not (Table 4.7).

		Satisfaction with Outcome			
		n	Mean	SD	<i>p</i> value
BMI (kg/m ²)	<30	47	82.8	15.6	0.046
	≥30	85	88.9	18.2	0.040
Eligible by Schnur Sliding Scale 22 percentile criterion? ^{$+$}	Yes	81	88.1	14.7	0.204
	No	49	85.8	21.2	0.394
Eligible by 500 g per breast Minimum Rule? [‡]	Yes	73	88.2	14.8	0.204
	No	57	85.3	20.4	0.324

Table 4.7 Validity of traditional insurance selection criteria for coverage of breast reduction surgery and 'Satisfaction with Outcome' BREAST-Q Scores

+ Source: Schnur et al., 1991.

‡ Source: Seitchik, 1995.

4.4 Discussion

Health-related quality of life (HRQoL) outcome studies reported from the patient's perspective provide an important measure of the impact of a health condition and the success of surgical interventions (Brazier et al., 2017). Research studies using generic HRQoL instruments such as the Short Form-36 (SF-36) have highlighted that breast reduction surgery is of significant benefit to women with breast hypertrophy, providing relief of symptoms and improved quality of life, often to a level greater than that of the general population (Behmand et al., 2000, Blomqvist and Brandberg, 2004, Collins et al., 2002, Kececi et al., 2015, Klassen et al., 1996, Miller et al., 2005, O'Blenes et al., 2006, Saariniemi et al., 2008a, Shakespeare and Cole, 1997). However, it has been found that whilst generic instruments play an important complementary role in patient-reported outcome studies, they may not be sensitive or responsive enough to detect changes as a result of surgery or to capture all aspects of outcome after breast surgery (Pusic et al., 2007a). The development of validated condition-specific patient-reported outcome instruments such as the BREAST-Q have facilitated outcome studies exploring the unique outcomes of breast surgery from the patients' perspective (Pusic et al., 2009).

Previous outcome studies have utilised the BREAST-Q Reduction module to explore patient satisfaction and health-related quality of life in women with breast hypertrophy. In the largest retrospective study to date, Cogliandro et al. utilised the BREAST-Q to report high levels of patient satisfaction and wellbeing following breast reduction surgery (Cogliandro et al., 2017a). However, retrospective studies such as this are limited by their inability to provide pre-operative health-related quality of life levels and therefore assume that all respondents had an approximately equal baseline level of satisfaction and wellbeing before surgery (Andrade et al., 2018, Carty et al., 2012, Derby et al., 2016, Gonzalez et al., 2012, Krucoff et al., 2019, Menéndez-Cardo et al., 2017). Furthermore, the broad range of time since surgery could have affected patient responses in these studies. Whilst several prospective studies demonstrate a statistically significant improvement in BREAST-Q scores following reduction mammaplasty, most are limited by a relatively small sample size and in some studies there was minimal overlap in patients who completed the BREAST-Q at each study time point (Cohen et al., 2016a, Coriddi et al., 2013).

Although no known magnitude has previously been determined to establish a minimal clinically important difference for the change in BREAST-Q Reduction module scores following surgical intervention, a rule of thumb of a 10-point change on 100-point guality of life scales (King, 1996) or one-half a standard deviation has been suggested as a default value for patient-perceived important change on health-related quality of life measures (Cano et al., 2014, Norman et al., 2003). Whereas minimal important difference estimates have been described for the augmentation module (Cano et al., 2014) and the reconstruction module (Voineskos et al., 2020) of the BREAST-Q, estimates for the reduction module remain undetermined. This study is the first to describe a minimal important difference estimate for the reduction module of the BREAST-Q, enhancing the interpretability of scores for both patients and clinicians as to what represents a meaningful change. Accordingly, results from this study demonstrate that reduction mammaplasty is of significant health benefit to women with symptomatic breast hypertrophy and provides a clinically important improvement in all areas of patient satisfaction and health-related quality of life. Furthermore, this study provides evidence that the considerable health benefits of reduction mammaplasty are comprehensive and the marked improvement in satisfaction and guality of life is experienced by patients regardless of characteristics including body mass index, age or a minimum resection. Additionally, patients with a higher body mass index and older women who may be ineligible for surgery according to restrictions in some jurisdictions were found to have an even greater level of improvement in satisfaction with breasts, physical wellbeing and psychosocial wellbeing than their respective comparison groups. The surgical cohort were not biased by restrictions that previous studies have reported based on body mass index or resection weight and therefore includes a broad spectrum across these variables (Andrade et al., 2018, Cabral et al., 2017a, Eggert et al., 2009, Freire et al., 2004, Lewin et al., 2019, Mello et al., 2010, Miller et al., 2005, Thoma et al., 2013, Thoma et al., 2007). No differences were found in improvement in

110

quality of life between those patients meeting traditional insurance cut-offs including Schnur sliding scale and the 500-gram per breast minimum rule and those who did not. This study therefore supports previous findings that there is no evidence or rationale to justify any policy that restricts funding for reduction mammaplasty based on arbitrary cut-offs for body mass index or a minimum weight of reduction (Chao et al., 2002, Collins et al., 2002, Kerrigan et al., 2002, Nguyen et al., 2008, Spector et al., 2008, Strong and Hall-Findlay, 2015, Thoma et al., 2007, Wagner and Alfonso, 2005).

Despite over one-third of women experiencing a surgical complication, there were no differences in either the change in scores in all four BREAST-Q scales or with the overall satisfaction with outcome of surgery between those patients who experienced a post-operative complication and those who did not. This finding supports the results described in chapter 2 using the SF-36 instrument and highlights that the overall health benefits and improvement in quality of life gained by women following breast reduction surgery appear to compensate for the negative impact of post-operative complications following surgery in the majority of cases.

This study found that women reported high scores for satisfaction with outcome following surgery. This BREAST-Q scale is important as it represents the overall level of patient satisfaction with the results of surgery. Post-operative Satisfaction with Outcome was found to be greater in participants with a higher BMI and in older women than their respective comparison groups. In contrast, eligibility by traditional insurance requirements including the Schnur sliding scale or 500-gram minimum rule was not found to be predictive of satisfaction with outcome. These findings strongly refute the validity of traditional selection criteria for coverage by insurance providers or restrictions on access to surgery based on body mass index and a minimum resection weight.

Overall patient Satisfaction with Outcome following surgery was found to be strongly correlated with Satisfaction with Surgeon, Satisfaction with Breasts and Satisfaction with Information scales. Conversely, only weak associations were observed with remaining BREAST-Q scales. These findings highlight the central importance of the doctor-patient relationship and managing patient expectations (Yip et al., 2015). Our findings confirm those of Coriddi and colleagues who reported that satisfaction with outcome was strongly correlated with satisfaction with breast appearance (Coriddi et al., 2013).

The comparison of generated norms to BREAST-Q data from participants in the surgical cohort confirmed that breast hypertrophy represents a significant health impairment to women, with preoperative scores significantly lower in all areas of satisfaction and health-related quality of life. At one year following reduction mammaplasty, mean scores significantly increased across all BREAST-Q scales to levels at least that of the norm. This finding demonstrates the long-term health benefits and success of breast reduction surgery in bringing satisfaction and quality of life to

111

levels of the general female population. This study facilitated the comparison of normative BREAST-Q data to a prospective study of women undergoing reduction surgery and therefore addresses previous gaps in the literature (Mundy et al., 2017c).

A potential limitation of this study is the administration of the BREAST-Q questionnaire to the surgical cohort at a post-operative time point of 12 months. Whilst this has affirmed the long-term benefits of breast reduction surgery on patient satisfaction and HRQoL, for comparative purposes the addition of a shorter post-operative time point may also have been informative. Secondly, estimation of the minimal important difference was limited to the use of distribution-based approaches in the absence of a clinical anchor; existing studies have shown that using a combination of distribution-based and anchor-based approaches in a process of triangulation provides the most accurate estimates (Revicki et al., 2008).

Strengths of this study include the prospective design and relatively large sample size for the surgical cohort. The post-operative outcomes described were at 12 months, beyond the convalescence period following surgery, and therefore represent an appropriate measure of long-term outcomes for comparison to population norms. This study is the first to describe a minimal important difference estimate for the reduction module of the BREAST-Q, providing a valuable threshold as to what represents a clinically meaningful change. Furthermore, the generation of corresponding population-specific normative BREAST-Q data enabled an accurate comparison for the interpretation of health-related quality of life between surgical patients and population norms.

4.5 Conclusion

This study demonstrates that breast reduction surgery significantly improves the breast-related satisfaction and health-related quality of life in women. Women with breast hypertrophy presenting for reduction mammaplasty have a significant health burden and reduced quality of life, which is significantly improved following surgery to levels of the general population. This study provides strong evidence to support the health benefits and clinical effectiveness of breast reduction surgery and establishes that indications for surgery should not be restricted to women who meet traditional insurance coverage based on body mass index or a minimum resection.

5. LONG-TERM OUTCOMES OF BREAST REDUCTION SURGERY

5.1 Introduction

Previous studies have described outcomes following breast reduction surgery using validated patient-reported outcome measures (PROMs) to assess health-related quality of life (HRQoL); however, the majority of these have been limited to a relatively short-term follow-up of up to 12-months after surgery. Whilst findings from these studies are important, there remains a need to evaluate patient-reported outcomes beyond the short-term to understand the full health benefits and effectiveness of surgery.

Of the existing studies evaluating patient satisfaction and health-related quality of life outcomes in the longer-term following surgery, few have been of prospective design (Blomqvist and Brandberg, 2004, O'Blenes et al., 2006, Saariniemi et al., 2011b); and retrospective studies remain the primary source of research output in the literature (Bai et al., 2019, Carty et al., 2012, Cogliandro et al., 2017a, Gonzalez et al., 2012, Hermans et al., 2005, Janik et al., 2019, Krucoff et al., 2019, Menéndez-Cardo et al., 2017). However, these studies are limited by their design and by the absence of baseline assessments of patient satisfaction and quality of life for which to compare long-term outcomes following surgery. Consequently, these studies are unable to assess whether or not levels of health-related quality of life in patients who underwent surgery several years prior remain stable or return to baseline levels in the longer-term. Additionally, these studies assume that all women presenting for breast reduction surgery have equal levels of wellbeing and health-related quality of life.

Finally, the prospective long-term assessment of patient-reported outcomes following breast reduction surgery using validated condition-specific outcome measures has yet to be explored in the literature. The use of the BREAST-Q patient-reported outcome instrument (Pusic et al., 2009) in this study provides valuable information for evaluating and gaining a better understanding the long-term outcomes specifically related to breast hypertrophy and breast reduction surgery.

This study aims to build on the 12-month patient-reported outcomes from the original prospective cohort study (described in Chapter 2 and Chapter 4) by evaluating the long-term health benefits and patient satisfaction up to 12 years following breast reduction surgery. To ensure a consistent and comprehensive comparison of long-term outcomes after surgery, a combination of both generic and condition-specific validated patient-reported outcome measures were administered as per the original study. The goals of this study were therefore to evaluate the long-term outcomes of breast reduction surgery, and to compare these levels of health-related quality of life several years after surgery to country-specific population reference data. It is anticipated that information

113

from the present study will inform clinicians, patients, and health care funders and providers about the expected long-term quality of life benefits of surgery and further support the clinical effectiveness of surgery for women with symptomatic breast hypertrophy.

5.2 Participants and methods

5.2.1 Participant recruitment

A long-term follow-up study was conducted within the Department of Plastic and Reconstructive Surgery at Flinders Medical Centre between March and August of 2019. Eligible participants were identified from the study database of women who underwent bilateral breast reduction surgery and consented to participate in the original prospective cohort study with 12-month follow-up detailed in Chapter 2 and Chapter 4. Inclusion criteria included women aged 18 years and above who consented to participate in the original study and underwent breast reduction surgery. Exclusion criteria were: participants who did not proceed to bilateral breast reduction surgery; participants who had withdrawn from the original study; participants who were identified from the electronic hospital system as deceased; and participants who were identified from the electronic hospital system as having had a breast cancer diagnosis. Ethics and governance approval were obtained for this study from the local Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC approval number 299.18) (Appendix N).

For this long-term follow-up study, demographic information for potential participants was accessed from the original study database and checked for updated information and eligibility through the electronic hospital system. A letter of invitation (Appendix O), participant information sheet (Appendix P) and study questionnaires were mailed to eligible participants. Participants who were willing to complete and return study questionnaires by mail consented to participate in this study. A follow-up phone call was made to non-responders in an attempt to ascertain their reasons for non-response or to obtain permission to mail out a subsequent study questionnaire pack if required.

5.2.2 Patient-reported outcome measures

The validated patient-reported outcome measures selected for this study were the same questionnaires introduced and discussed in detail in Chapter 2 and Chapter 4. In brief, the Short Form-36 version 2 (SF-36v2) (Appendix D) and BREAST-Q Reduction module post-operative version (Appendix I) were used to capture patient-reported satisfaction and health-related quality of life. The BREAST-Q and SF-36 instruments provide a transformed score on a scale between 0 and 100, with higher scores representing greater satisfaction or quality of life. The Multidimensional Body-Self Relations Questionnaire (MBSRQ) (Appendix E) was used for the assessment of self-attitudinal aspects of the body-image and weight-related constructs. MBSRQ scales were calculated on a scale between one and five, with a higher score on orientation scales

indicating a greater emphasis on one's looks or appearance and higher scores on evaluation scales indicating a person feels more satisfied in relation to the construct (i.e. appearance, fitness or health) (Cash, 2000).

Comparisons were made between long-term follow-up outcome scores and normative values for each of the validated patient-reported outcome instruments used in this study. For the BREAST-Q, general Australian female population reference values were generated and were described previously in Chapter 3. Normative data from the general female Australian population for the SF-36v2 were obtained from the 2008 South Australian Health Omnibus Survey dataset (Marin et al., 2009, Population Research and Outcome Studies Unit, 2008), with scores for each subscale weighted to correspond to the age distribution of the study participants. MBSRQ normative scores for females have been previously published by the developers of the instrument (Cash, 2000).

5.2.3 Adjunct questionnaires

A supplementary customised questionnaire was included to assess long-term patient satisfaction and outcomes following breast reduction surgery and to obtain relevant participant sociodemographic information. Participants were asked a series of questions including: (1) whether they would choose to have breast reduction surgery again if they had their time over; (2) whether or not surgery has alleviated the symptoms due to having large breasts; (3) whether they would recommend surgery to a relative or friend; (4) whether any further surgery was performed on their breasts since the original breast reduction surgery; (5) whether they experienced any new chronic/long-term health issues or a worsening of an existing health issue since surgery; and (6) current height, weight and bra cup size.

A standard issue hospital patient health questionnaire was administered to capture comprehensive information of self-reported health issues and medical co-morbidities (Appendix Q).

5.2.4 Statistical analysis

Descriptive statistics including mean, standard deviation and 95% confidence interval were computed for continuous socio-demographic variables and health-related quality of life outcome scores. Categorical variables were presented as counts and percentages. Participant characteristics were analysed using an independent *t*-test for continuous variables or a χ^2 test for categorical variables. Linear mixed-model analysis was used to assess the significance of changes in HRQoL scores for the surgical cohort over time (pre-operatively, 3-, 6-, 12-months post-operatively, and long-term post-operative follow-up), where the dependent variable was each individual health-related quality of life scale and time was the fixed effect. An independent *t*-test was used to compare long-term outcome scores between groups, or when comparing long-term outcomes with normative population reference scores. An analysis of covariance (ANCOVA) was used to assess long-term patient-reported outcome scores with participant characteristics

(categorical) as a fixed-factor and baseline scores as a covariate.

Not all participants in the original study who were assessed following surgery agreed to participate in this long-term follow-up study. To evaluate selection bias, a χ^2 test and independent *t*-test were used in a comparison of participant characteristics of responders and non-responders.

Statistical analyses were performed using SPSS v25.0 statistical software (IBM Corp., Armonk, N.Y.) and a *p*-value of less than 0.05 was considered statistically significant.

5.3 Results

5.3.1 Study participants

Participant recruitment and study completion rates are summarised in Figure 5.1. In summary, a total of 231 participants who remained enrolled at the completion of the original prospective study (Chapter 2) were screened for eligibility. Of these, 14 potential participants were excluded: 4 had a breast cancer diagnosis; 3 were deceased; 2 had undergone other types of breast surgery; and 5 were omitted due to ongoing psychological issues (1 hospitalised in a drug dependency facility and 4 with recent hospitalisations for psychiatric admissions). Therefore, a total of 217 eligible participants were invited to participate in this long-term follow-up postal questionnaire survey. Of these, 103 participants completed and returned study questionnaires and were included in the long-term follow-up cohort, yielding a total response rate of 47.5%.



Figure 5.1 Study flow-chart

5.3.2 Participant characteristics

Participant demographics for those in the surgical cohort who consented to the long-term follow-up study are summarised in Table 5.1. Median age at follow-up was 48 years (range, 22 to 81 years) with a median follow-up time since surgery of 6.0 years (range, 3 to 12 years). The mean body mass index at follow-up was $32.0 \pm 6.2 \text{ kg/m}^2$, with the majority of participants (52%) classified as obese (\geq 30 kg/m²). The proportion of participants reporting the presence of medical co-morbidities at the long-term follow-up are presented in Table 5.1 and were as follows: diabetes (8%), hypertension (12%), asthma (22%), chronic pain (20%) and mental health problems including anxiety (19%) and depression (20%).

Characteristic	Value (%)
Study number, <i>n</i>	103
Age at surgery, years	43.3 ± 13.6
Median (range)	42 (18 to 73)
Current age, years	, , , , , , , , , , , , , , , , , , ,
Mean ± SD	49.7 ± 13.6
Median (range)	48 (22 to 81)
Time since surgery, years	
Mean ± SD	6.8 ± 2.5
Median (range)	6.0 (3 to 12)
BMI, kg/m ²	, , , , , , , , , , , , , , , , , , ,
Mean ± SD	32.0 ± 6.2
Obesity status	
Non-obese (<30 kg/m²)	40 (39)
Obese (≥30 kg/m²)	53 (52)
Missing	10 (10)
Bra cup size	
A	0 (0)
В	3 (3)
С	28 (28)
D	31 (30)
DD	18 (18)
>DD	16 (16)
Missing	7 (7)
Co-morbidities ($n = 93$)	
Diabetes	
No	85 (83)
Yes	8 (8)
Hypertension	
No	81 (79)
Yes	12 (12)
Asthma	
No	70 (68)
Yes	23 (22)
Chronic pain	
No	72 (70)
Yes	21 (20)
Anxiety	
No	73 (71)
Yes	20 (19)
Depression	
No	72 (70)
Yes	21 (20)

Values are numbers (percentages) unless stated otherwise.

5.3.3 Non-responders

Participant characteristics were compared between those who elected to participate by returning study questionnaires and those who either did not respond or declined to participate in the long-term follow-up study. Demographic and clinical characteristics were similar between responders and non-responders, with no significant differences observed in relevant variables including age at surgery, body mass index, and amount of breast tissue resected at surgery, or current age at long-term follow-up. Table 5.2 provides a summary of these findings.

	Mea	an (SD)	Difforence in	<i>p</i> - value⁺	
	Responders (<i>n</i> = 103)	Non-responders (<i>n</i> = 114)	means (95% CI)		
Age at surgery, y	43.3 (13.6)	41.2 (12.5)	2.1 (-1.4 – 5.6)	0.25	
BMI, kg/m²	32.2 (5.9)	33.6 (5.6)	1.33 (-0.2 – 2.9)	0.09	
Resection weight, g	1259 (693)	1430 (821)	169 (-43.8 – 383.4)	0.12	
Age (long-term), y	49.7 (13.6)	48.3 (13.3)	1.43 (-2.2 – 5.0)	0.44	

Table 5.2 Comparison of participant characteristics between respor	nders and non-responders
--	--------------------------

[†] Using an independent samples *t*-test.

5.3.4 Long-term patient-reported outcome scores

5.3.4.1 Short Form-36

Summary statistics for SF-36 subscale and summary scores measured in a long-term outcome assessment for women who underwent bilateral breast reduction surgery are presented in Table 5.3. A comparison between SF-36 scores up to 12 months post-operative from the original study and long-term post-operative scores showed that scores remained stable and significantly higher than baseline over time, with no significant differences found in all eight SF-36 subscales or physical and mental summary scales (Figure 5.2). When compared with previously published age-adjusted normative data for the female Australian population (Marin et al., 2009), long-term SF-36 scores for women who underwent surgery several years prior remained comparable and no significant difference was found across seven of the eight SF-36 scales or physical and mental summary scales (p > 0.05). In contrast, mean scores for 'bodily pain' at the long-term review were slightly lower than normative values, with a mean difference of 5.7 points (95% CI, 1.24 to 10.2; p = 0.02).

	Normative [‡]	Pre-operative	Post-operative			
			3 months	6 months	12 months	Long-term
SF-36 scale	(<i>n</i> = 1551)	(<i>n</i> = 209)	(<i>n</i> = 190)	(<i>n</i> = 181)	(<i>n</i> = 191)	(<i>n</i> = 103)
Physical function	84.2 (19.1)	61.0 (25.4)	80.1 (22.3)	80.8 (24.2)	83.4 (22.1)	82.6 (21.5)
Role physical	82.0 (24.8)	56.0 (28.3)	79.5 (23.7)	81.1 (25.4)	81.3 (25.1)	81.6 (24.7)
Bodily pain	73.0 (21.4)	38.5 (21.5)	67.4 (24.1)	67.6 (27.6)	71.6 (25.9)	67.4 (24.7)*
General health	70.2 (22.2)	57.9 (21.7)	69.1 (19.8)	69.5 (19.8)	70.4 (19.1)	68.1 (21.8)
Vitality	57.3 (20.9)	39.7 (20.2)	57.7 (20.4)	58.6 (19.1)	58.9 (19.7)	57.4 (19.4)
Social function	82.6 (24.3)	57.1 (27.9)	78.8 (25.9)	79.4 (25.9)	81.4 (23.4)	82.4 (23.4)
Role emotional	88.3 (20.3)	61.7 (28.9)	80.1 (25.1)	82.3 (23.1)	84.6 (22.2)	85.2 (22.9)
Mental health	77.0 (18.2)	59.8 (20.2)	73.7 (18.7)	73.8 (18.4)	74.3 (18.6)	73.6 (18.8)
Physical Component Score	49.7 (9.7)	39.7 (9.7)	48.9 (9.2)	49.0 (10.3)	49.9 (9.9)	48.9 (9.6)
Mental Component Score	47.6 (11.4)	37.0 (13.2)	45.4 (11.9)	45.7 (11.5)	46.2 (11.6)	46.7 (11.5)

Table 5.3 Mean (SD) SF-36 scores for surgical participants over time in comparison to normative data

‡ Age-standardised normative data for females from the South Australian Health Omnibus Survey (SAHOS) 2008 (Marin et al., 2009).

* denotes significant difference between normative and long-term SF-36 scores (p < 0.05).



Figure 5.2 Mean SF-36 subscale scores over time compared to female general population reference values

5.3.4.2 BREAST-Q

Mean BREAST-Q scores obtained pre-operatively and 12 months post-operatively (Chapter 4) were compared to long-term outcome scores and are presented in Table 5.4. As described in the previous study, statistically significant improvements were observed when comparing baseline to 12 months following surgery across all four BREAST-Q scales measured pre- and post-operatively (p < 0.001). In this long-term follow-up study, mean BREAST-Q scores were found to remain stable between 12 months post-operative and the long-term assessment, with no significant differences observed across all scales: satisfaction with breasts (mean difference, -3.1; 95% CI, -7.5 to 1.3; p = 0.27); psychosocial wellbeing (mean difference, 1.2; 95% CI, -3.8 to 6.2; p = 0.61); sexual wellbeing (mean difference, 3.8; 95% CI, -2.0 to 9.6; p = 0.19); and physical wellbeing (mean difference, -0.3; 95% CI, -3.6 to 3.0; p = 0.72) (Table 5.4 and Figure 5.3). Mean BREAST-Q scores at long-term follow-up remained significantly higher than baseline levels in all four scales (p < 0.001). In comparison to the normative data generated from women within the Australian general population (Chapter 3), long-term outcome scores remained stable and at levels above population norms (Figure 5.3).

	Normative ⁺	Pre-operative	Post-operative		
			12 months	Long-term	
Scale	(<i>n</i> = 513)	(<i>n</i> = 132)	(<i>n</i> = 132)	(<i>n</i> = 103)	
Satisfaction with Breasts	51.9 (17.3)	21.9 (10.9)	73.3 (17.3)	70.2 (16.4)	
Psychosocial Wellbeing	55.0 (21.8)	32.4 (12.8)	69.3 (19.8)	70.5 (18.3)	
Sexual Wellbeing	48.5 (24.0)	29.3 (18.2)	60.6 (23.3)	64.4 (20.9)	
Physical Wellbeing	71.2 (13.1)	44.0 (15.5)	76.6 (12.1)	76.3 (13.6)	

Table 5.4 Summary of mean (SD) BREAST-Q scores in surgical participants over time

⁺ Normative BREAST-Q scores derived from the Australian general population (Chapter 3).



Figure 5.3 Mean BREAST-Q scores for surgical participants over time in comparison to population norms

Error bars represent 95% confidence intervals.

Table 5.5 presents a comparison of scores reported exclusively in the post-operative version of the BREAST-Q Reduction module between surgical participants at 12 months following surgery and long-term follow-up. Mean scores remained stable in the long-term across all six BREAST-Q scales measured post-operatively only, with no significant differences found in satisfaction with outcome, satisfaction with information, satisfaction with nipple, or satisfaction with surgeon, satisfaction with medical staff and satisfaction with office staff scales.

Table 5.5 Summary	of post-operative	BREAST-Q scale sc	ores in surgical	participants
· · · · · · · · · · · · · · · · · · ·				P

	Mear	n (SD)	Difforance in		
Scale	12 months post-operative	Long-term post-operative	means (95% CI)	<i>p</i> -value	
Satisfaction with outcome	86.7 (17.5)	87.1 (15.8)	0.4 (-3.9 – 4.7)	0.27	
Satisfaction with information	79.8 (18.6)	81.8 (17.4)	2.0 (-2.7 – 6.7)	0.17	
Satisfaction with nipple	82.3 (22.3)	82.5 (23.2)	0.2 (-5.6 – 6.1)	0.96	
Satisfaction with surgeon	88.1 (18.4)	87.6 (19.3)	-0.5 (-5.4 – 4.4)	0.64	
Satisfaction with medical staff	95.6 (12.0)	94.7 (12.5)	-0.9 (-4.1 – 2.3)	0.86	
Satisfaction with office staff	92.3 (16.4)	94.0 (12.8)	1.7 (-2.2 – 5.6)	0.60	
5.3.4.3 MBSRQ

Mean MBSRQ scores for women who underwent bilateral breast reduction surgery are summarised in Table 5.6. When compared with scores obtained at the final 12-months post-operatively assessment in the original study (Chapter 2), scores remained stable and were not significantly different in the longer-term follow-up across all MBSRQ scales with the exception of 'appearance orientation' and 'fitness orientation'. Mean scores for these two scales were significantly lower at long-term follow-up, with a mean difference of 0.18 points (95% CI, 0.04 to 0.32; p = 0.03) and 0.17 points (95% CI, 0.01 to 0.34; p = 0.048), respectively. This finding demonstrates women placed less importance on their appearance and fitness at the long-term follow-up when compared to 12 months post-operatively.

When compared to scores reported pre-operatively, scores remained significantly higher at the long-term follow-up for 'appearance evaluation' (mean difference 0.85; 95%Cl, 0.68 to 1.02, p < 0.001), 'health evaluation' (mean difference 0.20; 95%Cl, 0.06 to 0.34, p = 0.005) and 'body areas satisfaction score' (mean difference 0.51; 95%Cl, 0.37 to 0.64, p < 0.001). In contrast, scores were significantly lower for 'appearance orientation' (mean difference -0.33; 95%Cl, -0.48 to -0.18, p < 0.001), 'health orientation' (mean difference -0.25; 95%Cl, -0.35 to -0.14, p < 0.001), and 'self-classified weight' (mean difference -0.19; 95%Cl, -0.33 to 0.05, p = 0.08). This finding demonstrates that even several years after breast reduction surgery women continued to be more satisfied with their appearance, reported being in better health and were more satisfied with their body than those who did not undergo surgery despite placing less importance on their appearance and health than they did before surgery. The results for the MBSRQ scores at each timepoint are presented in Figure 5.4.

	Pre-operative				
Scale	(<i>n</i> = 209)	3 months (<i>n</i> = 190)	6 months (<i>n</i> = 181)	12 months (<i>n</i> = 191)	Long-term (<i>n</i> = 103)
Appearance evaluation	1.96 (0.64)	2.81 (0.70)*	2.83 (0.79)	2.87 (0.77)	2.84 (0.81) [‡]
Appearance orientation	3.57 (0.75)	3.53 (0.60)	3.53 (0.61)	3.50 (0.60)	3.32 (0.59) ^{†,‡}
Fitness evaluation	3.10 (0.78)	3.31 (0.79)*	3.30 (0.86)	3.35 (0.80)	3.23 (0.91)
Fitness orientation	2.94 (0.64)	3.03 (0.61)	3.03 (0.65)	3.06 (0.68)	2.89 (0.70) [†]
Health evaluation	2.99 (0.72)	3.29 (0.78)*	3.34 (0.73)	3.32 (0.75)	3.36 (0.82) [‡]
Health orientation	3.62 (0.62)	3.44 (0.59)*	3.54 (0.56)	3.50 (0.60)	3.44 (0.65) [‡]
Illness orientation	3.04 (0.78)	3.14 (0.75)	3.14 (0.80)	3.13 (0.77)	3.14 (0.81)
Body Areas Satisfaction Score	2.47 (0.56)	2.93 (0.61) [*]	2.99 (0.63)	3.00 (0.62)	2.97 (0.74) [‡]
Overweight preoccupation	3.31 (0.83)	3.17 (0.85)	3.16 (0.91)	3.23 (0.80)	3.13 (0.85)
Self-Classified Weight	4.14 (0.71)	4.00 (0.64)*	3.98 (0.64)	4.00 (0.62)	3.94 (0.81) [‡]

Table 5.6 Comparison of mean (SD) MBSRQ scores over time for surgical participants

* denotes significant difference (p < 0.05) between pre-operative & 3-month post-operative scores using linear mixed-model analysis.

[†] denotes significant difference (p < 0.05) between 12-month post-operative and long-term post-operative scores

⁺ denotes significant difference (*p* < 0.05) between pre-operative & long-term post-operative scores.



Figure 5.4 Mean MBSRQ scores pre-operatively versus scores at 12 months post-operative and long-term follow-up

5.3.4.4 Adjunct questionnaire

The majority of women (97/103, 94.3%) who underwent breast reduction surgery stated they would have their surgery again, if they had their time over; one participant replied that they would not have surgery again, and two (2/103, 1.9%) responded they were unsure. Three participants in the long-term follow-up study failed to answer the adjunct questionnaire. When asked whether or not breast reduction surgery alleviated the symptoms due to having large breasts, a total of 92 (89.3%) agreed; two participants (1.9%) disagreed; four (3.9%) were unsure; and five (4.9%) did not respond to this question. Most women (96/103, 93.2%) who underwent surgery would recommend it to a relative or friend; two (1.9%) were unsure; five (4.9%) did not respond; with no (0%) participants responding they would not recommend surgery. A total of ten participants (9.7%) stated they had further surgery on their breasts since the original surgery and of these 9 underwent revisional procedures related to the original breast reduction surgery and 1 participant had surgery for the removal of a benign breast lump; 80 (78%) had no further breast surgery; and 13 participants (12.6%) failed to respond to this question. Twenty participants (19%) stated they had experienced a new chronic or long-term health issue, or a worsening of an existing health issue in the time since their breast reduction surgery; 68 (66%) did not; and eight (7.8%) were unsure.

5.3.4.5 Other clinical outcomes

When comparing participant characteristics at baseline prior to surgery (Chapter 2) versus at longterm follow-up, there were no significant differences found in average weight of participants (mean 84.2 versus 84.6 kg, p = 0.66) or corresponding body mass index (32.0 versus 32.02 kg/m², p =0.949). In contrast, a chi-square test showed that bra cup size remained significantly smaller at the long-term assessment (χ^2 (5) = 29.9, p < 0.001) when compared to preoperatively. Frequency distributions for bra cup sizes at baseline in comparison to those reported at the long-term followup assessment are presented in Figure 5.5.



Figure 5.5 Bra cup size frequency distribution at pre-operative compared to long-term follow-up

5.3.5 Factors influencing long-term quality of life

Participants were categorised into the two groups based on BMI at the pre-operative assessment. A one-way ANCOVA was conducted to compare long-term BREAST-Q scores between non-obese (BMI <30 kg/m²) and obese participants (BMI \geq 30 kg/m²) whilst controlling for baseline BREAST-Q scores. There were no significant differences found in long-term BREAST-Q scores between BMI groups in all scales measured both pre- and post-operatively: satisfaction with breasts, psychosocial wellbeing, sexual wellbeing and physical wellbeing. Summary statistics and mean differences between BMI groups are summarised in Table 5.7.

Table 5.7	' Summary	of adjusted	long-term	BREAST-Q	scores by	BMI category
-----------	-----------	-------------	-----------	-----------------	-----------	---------------------

	Mean	ı (SD)	Difforence in	
Scale	Non-obese (<i>n</i> = 38)	Obese (<i>n</i> = 63)	means (95% CI)	<i>p</i> -value⁺
Satisfaction with Breasts	71.1 (15.5)	69.1 (15.2)	-1.9 (-9.5 to 5.6)	0.61
Psychosocial Wellbeing	70.1 (18.3)	71.7 (18.0)	1.5 (-7.6 to 10.6)	0.74
Sexual Wellbeing	68.6 (19.1)	67.1 (18.9)	-1.5 (-12.0 to 8.9)	0.77
Physical Wellbeing	76.2 (12.2)	76.4 (12.0)	0.2 (-5.8 to 6.2)	0.95

Non-obese <30 kg/m², obese \geq 30 kg/m².

⁺ Using an ANCOVA to compare BREAST-Q scale scores between groups after adjustment for preoperative BREAST-Q scores.

Post-operative BREAST-Q scales including satisfaction with outcome, satisfaction with nipple, satisfaction with information, satisfaction with surgeon, satisfaction with medical staff and satisfaction with office staff are not included in the pre-operative module and therefore do not have baseline values to be used as a covariate in the ANCOVA analysis. Therefore, an independent samples *t*-test was conducted to compare long-term BREAST-Q scores between obese and non-obese participants at baseline. There were no statistically significant differences between the two groups in all BREAST-Q scales. These findings are summarised in Table 5.8.

	Mean	ı (SD)	Difforence in		
Scale	Non-obese (<i>n</i> = 38)	Obese (<i>n</i> = 63)	means (95% CI)	<i>p</i> -value⁺	
Satisfaction with outcome	85.3 (14.6)	88.2 (16.5)	2.9 (-3.6 to 9.3)	0.38	
Satisfaction with information	80.2 (16.1)	82.8 (18.2)	2.6 (-4.5 to 9.7)	0.47	
Satisfaction with nipple	85.0 (21.6)	81.0 (24.2)	-4.0 (-13.5 to 5.4)	0.40	
Satisfaction with surgeon	86.7 (20.5)	88.2 (18.7)	1.4 (-6.5 to 9.4)	0.72	
Satisfaction with medical staff	94.8 (12.6)	94.7 (12.5)	-0.1 (-5.3 to 5.0)	0.95	
Satisfaction with office staff	95.2 (12.1)	93.2 (13.2)	-2.0 (-7.2 to 3.3)	0.45	

Table 5.8 Summary of unadjusted long-term BREAST-Q scores by BMI category

Non-obese <30 kg/m², obese \geq 30 kg/m².

+ Using an independent *t*-test to compare BREAST-Q scale scores between groups.

Patient-reported long-term SF-36 subscale and summary component scores were also compared between the two BMI groups using a one-way ANCOVA. While participants with a higher BMI reported lower scores for several subscales, in particular 'role physical', 'bodily pain' and 'general health' compared to participants with a lower BMI, this difference of 5.3-, 7.9- and 6.9-points, respectively, did not reach statistical significance. There were no statistically significant differences were found in all eight SF-36 subscales or physical and mental health summary measures between these groups whilst controlling for baseline SF-36 scores (p > 0.05). Table 5.9 provides a summary of the results.

	Mean	(SD)	Difference in	
SF-36 scale	Non-obese (<i>n</i> = 38)	Obese (<i>n</i> = 63)	means (95% CI)	<i>p</i> -value [†]
Physical function	83.5 (19.8)	81.8 (19.5)	-1.7 (-10.0 to 6.6)	0.68
Role physical	85.1 (24.3)	79.8 (24.4)	-5.3 (-15.6 to 5.0)	0.31
Bodily pain	72.1 (24.3)	64.2 (23.2)	-7.9 (-18.2 to 2.3)	0.13
General health	72.2 (21.2)	65.3 (21.0)	-6.9 (-15.8 to 1.9)	0.12
Vitality	57.7 (18.9)	56.9 (18.9)	-0.8 (-8.7 to 7.0)	0.83
Social function	82.6 (22.2)	82.1 (22.0)	-0.5 (-9.8 to 8.7)	0.90
Role emotional	86.3 (22.5)	84.3 (22.6)	-2.0 (-11.4 to 7.4)	0.68
Mental health	73.3 (18.8)	73.5 (18.8)	0.2 (-7.6 to 8.0)	0.96
Physical Component Score	50.6 (8.9)	48.1 (9.0)	-2.5 (-6.4 to 1.3)	0.19
Mental Component Score	46.0 (11.3)	46.9 (11.4)	0.9 (-3.9 to 5.7)	0.71

Table 5.9 Summary of adjusted long-term SF-36 scores by BMI category

Non-obese <30 kg/m², obese \geq 30 kg/m².

⁺ Using an ANCOVA to compare SF-36 scale scores between groups after adjustment for preoperative scores.

The relationship between age and long-term quality of life outcome scores were explored using a one-way ANCOVA, controlling for pre-operative baseline scores. Participants were categorised into two groups according to their age at surgery of less than 40 years and 40 years and above. While older women reported higher scores for 'satisfaction with breasts' and 'psychosocial wellbeing', this difference of 5.3- and 6.0-points did not reach statistical significance (Table 5.10). In summary, there were no statistically significant differences found in long-term outcome scores between these age groups in all dimensions of the BREAST-Q (Table 5.10).

	Mear	n (SD)	SD) Difference in		
Scale	< 40 years (<i>n</i> = 31)	≥ 40 years (<i>n</i> = 43)	means (95% CI)	<i>p</i> -value⁺	
Satisfaction with Breasts	66.7 (15.0)	72.0 (13.5)	5.3 (-1.9 to 12.5)	0.15	
Psychosocial Wellbeing	67.7 (17.2)	73.6 (17.4)	6.0 (-2.2 to 14.2)	0.15	
Sexual Wellbeing	67.4 (18.9)	67.9 (18.5)	0.5 (-9.3 to 10.3)	0.91	
Physical Wellbeing	74.8 (11.8)	77.4 (11.5)	2.6 (-2.9 to 8.2)	0.35	

Table 5.10 Summary of adjusted long-term BREAST-Q scores by age category

 Using an ANCOVA to compare BREAST-Q scale scores between groups after adjustment for preoperative scores.

An independent samples *t*-test was conducted to compare long-term post-operative only BREAST-Q scores between older and younger participants. Older women age 40 years and above were found to have significantly higher scores for 'satisfaction with outcome' when compared to younger women with an average difference of 6.7-points (p = 0.034). There were no statistically significant differences in long-term BREAST-Q scores between the two groups in the remaining postoperative scales. These findings are summarised in Table 5.11.

	Меа	an (SD)	Difference in	
Scale	Age <40 years	Age ≥40 years	means (95% CI)	<i>p</i> -value
	(<i>n</i> = 42)	(<i>n</i> = 59)		
Satisfaction with outcome	83.2 (17.3)	89.9 (14.2)	6.7 (0.5 to 13.0)	0.034*
Satisfaction with information	79.3 (17.1)	83.6 (17.6)	4.3 (-2.6 to 11.3)	0.220
Satisfaction with nipple	77.9 (25.3)	85.8 (21.2)	7.9 (-1.4 to 17.1)	0.094
Satisfaction with surgeon	84.5 (20.4)	89.9 (18.3)	5.5 (-2.3 to 13.2)	0.164
Satisfaction with medical staff	93.3 (13.9)	95.7 (11.3)	2.4 (-2.6 to 7.4)	0.351
Satisfaction with office staff	92.4 (15.3)	95.2 (10.6)	2.8 (-2.3 to 8.0)	0.277

A one-way ANCOVA was conducted to compare long-term outcome scores using the SF-36 questionnaire between younger and older women, adjusting for baseline values before surgery. Statistically significant differences were observed in 'physical functioning' and 'role physical' scales. Mean scores were 11.2- and 10.3-points lower, respectively, in older women aged 40 years and above at surgery compared to younger women aged less than 40 years. When compared to previously published normative data stratified by age group it was found that this finding was consistent to that of the general population, with older women reporting significantly lower scores for 'physical functioning' and 'role physical' than younger women, with a 14.3 and 9.0-point difference, respectively. There were no statistically significant differences found between age groups in all other dimensions of the SF-36 (Table 5.12).

	Mear	ı (SD)	Difforence in	
SF-36 scale	<40 years (<i>n</i> = 40)	≥40 years (<i>n</i> = 60)	means (95% CI)	<i>p</i> -value [†]
Physical function	89.1 (18.6)	77.9 (18.4)	-11.2 (-18.9 to -3.5)	0.005*
Role physical	87.9 (23.8)	77.6 (24.0)	-10.3 (-20.1 to -0.6)	0.038*
Bodily pain	70.4 (24.6)	65.0 (24.0)	-5.4 (-15.4 to 4.5)	0.281
General health	66.6 (21.0)	68.7 (21.2)	2.1 (-6.5 to 10.7)	0.626
Vitality	59.7 (19.0)	55.6 (18.7)	-4.2 (-11.8 to 3.5)	0.283
Social function	85.0 (22.8)	80.5 (22.2)	-4.5 (-13.7 to 4.7)	0.332
Role emotional	87.4 (22.5)	83.4 (22.7)	-3.9 (-13.1 to 5.2)	0.395
Mental health	75.6 (19.0)	72.1 (18.7)	-3.5 (-11.2 to 4.2)	0.365
Physical Component Score	51.0 (8.6)	47.7 (8.6)	-3.3 (-6.9 to 0.3)	0.069
Mental Component Score	47.1 (11.4)	46.2 (11.4)	-0.9 (-5.7 to 3.8)	0.703

Table 5.12 Summary of adjusted long-term SF-36 scores by age category

⁺ Using an ANCOVA to compare SF-36 scale scores between groups after adjustment for preoperative scores.

* Denotes statistical significance at p < 0.05 level.

5.4 Discussion

Patient-reported outcome measures utilised in long-term clinical outcome studies can provide valuable insight into the effectiveness and stability of medical or surgical interventions over time. Findings from this study using a combination of validated generic and condition-specific patient-reported outcome measures demonstrate that the positive health benefits and improved health-related quality of life are sustained for up to 12 years following breast reduction surgery. Furthermore, the comparison of long-term patient-reported outcomes with pre-operative scores demonstrated that health-related quality of life did not deteriorate to levels observed at baseline over time. This is a novel finding as existing outcome studies reporting a long-term follow-up have primarily been retrospective and were therefore unable to address this fundamental question (Bai et al., 2019, Carty et al., 2012, Cogliandro et al., 2017a, Gonzalez et al., 2012, Krucoff et al., 2019, Makki and Ghanem, 1998). In addition, this is the first study to use the BREAST-Q Reduction module to prospectively evaluate patient-reported outcomes beyond 1 year and compare patients' results to normative values; providing valuable insight as to whether patient-reported outcomes specific to breast reduction surgery change over time and in the longer-term.

This study found that years after undergoing breast reduction surgery patients continue to report a greater level of satisfaction and quality of life than levels observed prior to surgery, and in comparison to women within the general population across all BREAST-Q scales. Furthermore, the finding of a high level of long-term satisfaction with outcome is significant as this BREAST-Q scale represents a patient's overall appraisal of the results after surgery. In support of this finding, the majority of women reported a high level of long-term satisfaction and responded in an adjunct questionnaire that they would have surgery again or recommend surgery to a relative or friend; and almost 90 percent of respondents reported that breast reduction surgery had successfully alleviated the symptoms of having large breasts. Finally, this study demonstrates that the significant and long-term improvement in health-related quality of life following surgery is experienced by patients' regardless of patient factors including body mass index and age. In addition, consistent with the finding at the 12-month follow-up in the original study, older women continued to report a higher level of satisfaction with the overall outcome of surgery in the longer-term when compared to younger women.

While existing long-term outcome studies described in the literature using the BREAST-Q Reduction module have provided promising preliminary results, they have all been retrospective and interpretation is therefore limited as there are no baseline values for comparison. Accordingly, it is difficult to make meaningful comparisons to the findings of this prospective longitudinal study. Menéndez-Cardo et al. conducted a retrospective study using the post-operative version of the BREAST-Q to compare outcomes between vertical and inverted-T techniques; they reported a high level of long-term satisfaction with outcome, although scores were lower than those found in this

132

present study and the absence of any prior timepoint makes further interpretation of these findings difficult (Menéndez-Cardo et al., 2017). In 2019, a retrospective review describing the long-term follow-up of 37 young reduction mammaplasty patients less than 25 years of age using the BREAST-Q was conducted (Krucoff et al., 2019). The authors found that decades after surgery these patients reported high levels of breast-related quality of life when compared to normative population. Unfortunately, the analysis from this study focused on only four of the post-operative scales and therefore did not present information for long-term satisfaction with outcome for comparison; this was also the case in the retrospective study by Cogliandro and colleagues (Cogliandro et al., 2017a). In their 10 year retrospective analysis using the BREAST-Q, Gonzalez and colleagues did assess satisfaction with outcome and found that over 95% of patients were satisfied in the long-term following surgery; however, participants' responses for satisfaction with outcome were incorrectly reported as Likert item responses instead of being transformed to a 0 to 100 scale (Gonzalez et al., 2012). Nevertheless, the results of these aforementioned studies in conjunction with results of the current study provide evidence to demonstrate that breast reduction surgery provides a long-term improvement in quality of life outcomes for women with breast hypertrophy.

In this study, health-related quality of life outcomes as measured by the generic SF-36 instrument were consistent with findings from the BREAST-Q and were found to be stable in the longer-term up to 12 years after surgery and remained significantly higher than baseline levels in all eight subscales and summary measures. Furthermore, scores were equivalent to those from women within the general population in all quality of life scales with the exception of the 'bodily pain'. However, it is likely that this finding is due to ageing and is impacted by life events other than the surgery; with one-fifth of women reporting they had experienced either a new chronic health condition or worsening of an existing health issue since their surgery. These findings are consistent with a previously published study that demonstrated quality of life scores improve following surgery and are stable over time in a shorter-term follow-up. Blomqvist and Brandberg published a prospective study of 39 patients at 1 and 3 years post-operative using the SF-36 and demonstrated improved quality of life was stable at 3 years after surgery with only minor nonsignificant reductions found between 1 and 3 year assessments (Blomqvist and Brandberg, 2004). Notably, the authors also used a supplementary questionnaire assessing pain in six different body locations and found that whilst the reduction of bodily pain was still significantly lower when compared to baseline levels, there was a minor increase in levels of bodily pain between 1 and 3 years assessment, but this did not reach statistical significance. The finding of improved quality of life across all SF-36 scales in the longer-term is also consistent with a study exploring quality of life outcomes with a median follow-up time of 19.1 months following reduction mammoplasty in adolescents and young women aged 12 to 21 years (Nuzzi et al., 2019).

A number of existing studies have demonstrated that women with breast hypertrophy often display

a high level of dissatisfaction with their overall body image (Collins et al., 2002, Cunningham et al., 2005, Kerrigan et al., 2001, Sarwer et al., 1998, Thoma et al., 2013, Thoma et al., 2005). The results of the present study are consistent with this finding. In Chapter 2, patients who underwent breast reduction surgery were found to self-report improved levels of body image and overall evaluation of their appearance and health over a 12-month period. However, to the authors knowledge, no existing studies have assessed body image in women undergoing breast reduction surgery beyond this timeframe. Accordingly, this study demonstrates that the improved levels of body image satisfaction and evaluation of overall health and appearance continue well beyond this timeframe and are sustained for many years after breast reduction surgery. In addition, the finding that women continue to be less focused on their appearance and are more content with most areas of their body when compared to levels observed pre-operatively further demonstrates the ongoing health benefits of surgery.

Findings of this study have demonstrated the long-term health benefits in terms of improved health-related quality of life following breast reduction surgery. However, other factors including the potential for weight loss in the years following surgery and whether or not reduced breast size is sustained in the longer-term are also important outcomes. This study found that women reported wearing significantly smaller bra cup sizes in the long-term than at the pre-operative assessment. In contrast, no significant differences were found in body weight and corresponding BMI in the long-term follow-up after surgery when compared to baseline values. Therefore, although surgery did not result in any long-term weight loss, there was also no increase in average body weight or BMI in the longer-term. Weight gain is influenced by many external factors including lifestyle, diet, physical activity and hormonal changes in aging such as pregnancy and menopausal status; and in Australian women has been shown to steadily increase in younger (average 649g per year) and middle-aged women over time (average 492g per year) (Adamson L. 2007, Australian Longitudinal Study on Women's Health, 2005, Australian Longitudinal Study on Women's Health, 2018, Australian Longitudinal Study on Women's Health, 2019). Accordingly, weight gain in women who underwent breast reduction surgery several years prior was less than the rate observed in the general population over time. These findings are supported by existing studies that demonstrated the majority of patients tend to return to their original body weight up to 5 years following surgery (Bayramicli et al., 2017) or show a tendency towards a slight weight gain but at a slower rate than the general population (Pauzenberger et al., 2014). Whilst other studies have provided evidence contradictory to this, their follow-up was considerably shorter-term at less than two years following surgery (O'Blenes et al., 2006, Singh et al., 2010). Finally, given that this present study found patient satisfaction and quality of life were stable and did not deteriorate in the longer-term following surgery, it further demonstrates that these two factors are often independent of one another.

5.4.1 Strengths and limitations

A strength of this study was the prospective design and long-term follow-up of surgical participants previously enrolled in the original study with a 12-month follow-up (described in Chapter 2 and Chapter 4). Therefore, eligible participants in the present study had baseline preoperative and 12-month follow-up data for comparison to the longer-term outcomes. In addition, this study employed a series of validated generic and condition-specific patient-reported outcome questionnaires and the first study to prospectively assess the long-term outcomes of breast reduction surgery using the BREAST-Q instrument. These factors overcome limitations previously described in the literature describing long-term outcomes based on retrospectively designed studies or prospective studies with a small sample size. Finally, the comparison of long-term patient-reported outcomes to population-specific normative data, including generated population reference data for the BREAST-Q Reduction module described previously (Chapter 3), provides further insight into the expected patient satisfaction and sustained improvement in health-related quality of life following breast reduction surgery when compared to women in the general population.

This long-term outcome study has several potential weaknesses that should be noted. Achieving satisfactory response rates when administering questionnaires by post can present a challenge, with non-response reducing sample size and may introduce bias (Edwards et al., 2002). The follow-up of participants in the present study is incomplete; of the 217 participants from the original study who were eligible for participation in this study, 114 (52.5%) did not participate in the study, reducing study power and raising the possibility of bias. However, socio-demographic and clinical characteristics were found to be similar between responders and non-responders which suggests this long-term cohort provides an accurate representation of the original surgical cohort. Furthermore, this study remained adequately powered with 103 respondents to detect clinically meaningful changes in patient-reported outcomes and is in keeping with response rates reported by other long-term mail-out survey studies in the absence of ongoing contact with participants from their 12-month post-operative follow-up. Finally, the present study reports patient experiences and outcome data from one state, potentially limiting the generalisability of the findings.

5.5 Conclusion

This study assessed the long-term outcomes and patient-reported quality of life in women who underwent bilateral breast reduction surgery for symptomatic breast hypertrophy. This study demonstrated that patients report a high level of satisfaction with the outcome of surgery and experience the ongoing health benefits and improvement in quality of life for many years after surgery. In the long term, the significant improvement in patient satisfaction and health-related quality of life observed in the shorter-term following surgery is sustained and does not deteriorate to levels reported preoperatively over time. In a climate of increasingly constrained healthcare budgets, this study provides evidence as to the value of surgery in this population which continues

well beyond the short-term gains previously reported.

6. RELATIONSHIP BETWEEN BREAST AND BODY SIZE AND THE IMPROVEMENT IN HEALTH-RELATED QUALITY OF LIFE FOLLOWING BREAST REDUCTION SURGERY

A version of this chapter including the comparison of three-dimensional laser scanning to water displacement for the measurement of breast volume has been published as an original article in the Australasian Journal of Plastic Surgery; attached as Appendix R (Crittenden et al., 2018a).

6.1 Introduction

Measurement of breast volume plays an important role in breast reduction, reconstruction, developmental asymmetry and augmentation. In women with breast hypertrophy it is important for pre-operative planning, in intraoperative decision-making regarding the amount of tissue to be taken from each breast to achieve symmetry, and where removal of a minimum amount of breast tissue is required to justify surgery (Boukovalas et al., 2019, Descamps et al., 2008, Dvoracek et al., 2019, Hernanz et al., 2014, Kececi and Sir, 2014, Klassen et al., 1996, Kocak et al., 2011, Koltz et al., 2013, Nguyen et al., 2008, O'Blenes et al., 2006, Seitchik, 1995, Sommer et al., 2002, Wamalwa et al., 2018, Wampler et al., 2019). Accurate estimation of breast volume and predicted tissue resection weight also promotes improved counsel to the patient and provides a valuable guide to training surgeons.

A variety of techniques are described in the literature to fulfil the need for an accurate and objective measurement of breast volume in the clinical setting. These techniques include magnetic resonance imaging (MRI), mammography, plaster casting, Grossman-Roudner plastic cups, water displacement, anthropometric measurement and three-dimensional surface imaging (Bulstrode et al., 2001, Howes et al., 2017, Kovacs et al., 2007, Kwong et al., 2020, Lee et al., 2016, Longo et al., 20 al., 2013, Losken et al., 2005, Muslu et al., 2019, Qiao et al., 1997, 1986, Sigurdson and Kirkland, 2006, Smith et al., 1986, Wesselius et al., 2018). In a situation where a woman is undergoing post-mastectomy breast reconstruction, the gold standard for measuring the breast tissue volume for reconstruction is the Archimedes method of water displacement of the mastectomy specimen (Yip et al., 2012). When measuring breast tissue volume in the intact state, such as in candidates for breast reduction surgery, MRI is reported as being the most accurate (Choppin et al., 2016). However, MRI is expensive and not always accessible. Three-dimensional (3D) laser scanning has also been demonstrated as a valid method of breast volume measurement (Howes et al., 2017, Kwong et al., 2020, Lee et al., 2016, O'Connell et al., 2015, Yip et al., 2012), but some centres lack access to this technology. Anthropometric measurements based on defined anatomic points from skeletal and soft tissue landmarks provide a fast and useful tool to assess breast aesthetics and predict breast volume (Brown et al., 2012, Longo et al., 2013, Muslu et al., 2019, Qiao et al., 1997, Sigurdson and Kirkland, 2006). Water displacement of the intact breast has

137

been used in some centres since 1970, but has not previously been validated against more recent methods (Bouman, 1970).

Research studies reporting on the relationship between breast size and body size and shape in women undergoing breast reduction surgery are limited. Whilst some studies have attempted to explore this relationship, many have relied on indirect proxy measures for body size including the use of BMI (Benditte-Klepetko et al., 2007), or have used substitute measures to estimate breast size including tissue resection weight (Findikcioglu et al., 2013) and bra cup size (Brown et al., 2012). Gaining a better understanding of whether factors such as the degree of breast hypertrophy or breast size as a proportion of body size have an impact on the improvement in health-related quality of life (HRQoL) following surgery is relevant both at a patient and clinician level but also for healthcare funders and the wider community worldwide.

The original aims of the surgical study described in Chapter 2 was not only to investigate the health benefits of breast reduction surgery but to also assess the effect of body shape or 'disproportionality' on HRQoL outcomes following surgery. In planning this study, it was hypothesised by the team of investigators that women with disproportionately large breasts in relation to their total body size would have a greater improvement in HRQoL following surgery than women with large breasts and a larger body size. In order to address this hypothesis, the original study was designed to include a comprehensive body shape assessment pre-operatively and at 12 months following breast reduction surgery with breast and body size measured using a series of approaches including water displacement, 3D laser scanning and anthropometric measurements. In this way, if it was determined there was a relationship between disproportionality and the level of improvement in HRQoL then this information could be used to generate a prioritisation system that was evidence-based in order to allow for fairer waiting lists within the public hospital system. Additionally, if this hypothesis was proven correct a further aim was to determine if anthropometric measurements or water displacement would be a simple and practical method for general practitioners to triage patients and determine eligibility for publicly funded surgery. Therefore, the primary objectives of this chapter were to evaluate the relationship of disproportionality on healthrelated quality of life outcomes following surgery; to investigate breast and body shape in women undergoing breast reduction surgery; and to compare and assess the validity of a series of techniques for measuring breast and body size in women with breast hypertrophy.

6.2 Participants and methods

6.2.1 Surgical participants

The participants in the surgical cohort are introduced and described in detail in Chapter 2 (Section 2.2.1). However, a point of difference in this study was that surgical participants underwent a comprehensive body shape assessment at two of the four timepoints; pre-operatively and at 12

138

months following surgery. Breast size was measured with 3 different methods: anthropometric measurements, three-dimensional (3D) laser scanning, and water displacement. Body size was measured with anthropometry, 3D laser scanning and through the conventional calculation of body mass index. A professor of anthropology (Professor Maciej Henneberg, University of Adelaide), certified anthropometrist (Daisy Veitch, SHARP Dummies Pty Ltd, Adelaide) and a public health physician (Professor David Ben-Tovim, Flinders University) were all involved in the original research design of this study with the plastic surgery team at Flinders Medical Centre.

6.2.2 Anthropometric body shape assessment

6.2.2.1 Surgical participants

A series of direct anthropometric body measurements were taken from surgical participants using a measuring tape and anthropometer with sliding calliper (GPM, Switzerland) as per International Standards Organisation (ISO) document 8559 (ISO 8559 (part 1), 2017). A standard set of direct measurements were performed by a criterion anthropometrist to assess the composition of the body and included: height, weight and body mass index; body circumferences (chest, bust, underbust, intermammary fold, waist and hips); and point-to-point/width measurements including bust width, nipple-to-nipple distance, shoulder and hip width (Appendix S). Suprasternal notch-to-nipple distance was recorded in the preoperative consultation with the plastic surgeon.

A formula based on anthropometric measurements for the estimation of total breast volume was proposed by a Professor of Anthropological and Comparative Anatomy at the University of Adelaide, Professor Maciej Henneberg. This proposed formula was based on three anthropometric measurements (bust girth, under bust girth and bust width) and a geometric volume formula: Total breast volume = $(0.6667 \pi ((bust girth - under bust girth)/\pi)^{*}(bust width/2)^{2})/500000$

6.2.2.2 Normative anthropometry data

Normative anthropometric body data from 521 Australian female volunteers within the general population obtained from the 2009 Sharp Dummies Australian Body Sizing Survey was kindly provided by Daisy Veitch (SHARP Dummies Pty Ltd, Adelaide, Australia). Participants were represented across a broad range of variables including age, body weight and geographical location. Participant characteristics including age and weight were recorded. Anthropometric body measurements were performed by the same criterion anthropometrist employed in the assessment of surgical participants in the present study. One-dimensional anthropometric measurements included height, acromial height (shoulder to ground), waist height, bust width (nipple to nipple), bust girth, waist girth, hip girth, bi-acromial (shoulder) width, and hip width.

6.2.3 Measuring breast and body volume using a 3D laser scanner

Three-dimensional (3D) laser body scanning was performed using a Cyberware WBX scanner (Cyberware, Monterey, USA) and CySlice software (Headus Pty Ltd, Perth, Australia). The

Cyberware WBX scanner has four sets of laser heads and cameras that move vertically to scan the surface anatomy of subjects in the upright position. Breast and body volume was measured according to the published protocol described previously at our institution (Veitch et al., 2012). The accuracy of the Cyberware WBX 3D scanner for the measurement of breast volume has previously been established at our institution by comparison to water displacement of mastectomy specimens (Yip et al., 2012) and by comparison to magnetic resonance imaging (MRI) (Howes et al., 2017). Prior to scanning, the assessor placed 'landmarks' on subjects using palpation to demarcate selected anatomical areas and the breast perimeter. Figure 6.1 illustrates the application of landmarks on a large-breasted woman to section the breast from the torso for volume analysis.





(a) photography showing landmark placement following determination of breast borders by palpation, (b) scan image with outline of breast and torso boundary (c) CySlice image to section the breast portion of the scan from the torso. Source: (Crittenden et al., 2018a).

6.2.4 Water displacement for measuring breast volume

Water displacement measurements based on Archimedes' principle was performed on each breast to determine the volume, as described by Schultz et al. (Schultz et al., 1986). A large calibrated container was filled with warm water and the patient asked to individually immerse each breast into the container, using a specially modified surgical bed and ensuring that the superior and lateral edges of the breast were at water level (Figure 6.2). Individual breast volumes were then determined by the amount of water displaced.





6.2.5 Statistical analysis

Descriptive statistics including mean, standard deviation and 95% confidence interval were computed for continuous variables. Measurements undertaken before and after surgery were compared using a paired samples *t*-test. An independent samples *t*-test was undertaken when comparing continuous variables between two different groups. A one-way analysis of variance (ANOVA) was used when comparing the mean measurements of more than two groups, with a post-hoc Tukey test to determine which of the groups differ. Pearson correlation coefficients (r)were calculated to assess the strength of linear association between two variables, including when comparing the two measurement techniques. Correlation coefficients less than 0.29 were described as weak, between 0.30 to 0.49 as moderate, and greater than or equal to 0.5 as strong (Cohen, 1988). The strength of any correlation between two measurement techniques was determined with a Bland-Altman analysis (Bland and Altman, 1995). This analysis plots the mean breast volume measurements for both techniques against their differences. If the two methods are comparable and in agreement then the differences should be small, and the mean of the differences should be close to zero. The 95% limits of agreement were formed using the mean difference in volumes ±1.96 standard deviation (SD) of the differences in volumes. Linear regression analysis was performed to assess proportional bias between the two measurement techniques. All analyses were performed using SPSS v25.0 statistical software (IBM Corp., Armonk, New York). All tests were two-sided and a p-value of less than 0.05 was regarded as statistically significant.

6.3 Results

6.3.1 Participants

A total of 251 participants were assessed pre-operatively and 190 participants at 12-months postoperatively. Clinical and demographic information for surgical participants was summarised previously in Chapter 2 (Section 2.3.1).

6.3.2 Missing data

The total numbers of surgical participants assessed by each measurement approach varied due to several factors. Firstly, water displacement for the measurement of breast volume was discontinued prior to study completion and therefore there were a lower number of participants with data for this technique. Secondly, on occasion participants had limited time and therefore not all assessments at each study timepoint may have been completed.

6.3.3 Anthropometric body shape assessment before and after breast reduction surgery

A total of 251 participants had anthropometric measurements at the pre-operative assessment and 190 participants at 12-months post-operatively. Table 6.1 summarises the comparison of these measurements before and after surgery. Bust girth, breast point-to-point (distance between breast apices) and nipple-to-nipple distance were found to significantly decrease following surgery. In contrast, no significant differences were found when comparing pre- and post-operative measurements for body weight, height, BMI, waist girth, under-bust girth, shoulder width and hip girth. Pre-operatively, mean sternal notch-to-nipple distance for the left breast was 32.4 cm (SD 4.4 cm, range 24 to 54 cm) and 32.7 cm (SD 4.5 cm, range 24 to 54 cm) for the right breast. When estimating breast volume using the proposed formula based on anthropometric measurements, preoperative mean total breast volume was 4267 mL (SD 1594 mL). At 12 months post-operatively mean total breast volume was significantly lower and estimated at 2906 mL (SD 846 mL).

When comparing the mean age of surgical participants (mean 41.9 years, SD 13.3) to women in the normal population sample (mean 41.7 years, SD 10.9; range 18 to 57 years), there were no significant differences found between the two groups (mean difference 0.16 years, 95% CI: -1.6 to 1.9; p = 0.858). Table 6.1 presents a comparison of anthropometric measurements between surgical participants and women within the general Australian population sample. In comparison to both baseline and post-operative measurements from surgical participants, women within the normal population sample had significantly lower measurements for body weight, BMI, bust girth, waist girth, hip girth, and nipple-to-nipple distance. In contrast, data were similar between these groups for shoulder width and height (stature).

	Surgica	l group	Mean difference (95% CI)	Normative Mean difference		Mean difference (95% CI)	
	Pre-op (<i>n</i> = 251) Mean (SD)	12m post-op (<i>n</i> = 190) Mean (SD)	pre-op vs 12m post-op; <i>p</i> -value⁺	(<i>n</i> = 521) Mean (SD)	pre-op vs normal; <i>p</i> -value [‡]	12m post-op vs normal; <i>p</i> -value [‡]	
Weight, kg	86.4 (17.5)	86.4 (17.0)	-0.50 (-1.3 to 0.27); 0.200	66.0 (9.3)	20.4 (18.5 to 22.3); <0.001	20.4 (18.4 to 22.4); <0.001	
Height, mm	1619.3 (66.5)	1620.8 (65.5)	-2.1 (-3.2 to -0.91); 0.201	1627.3 (65.0)	-8.1 (-18.0 to 1.9); 0.111	-6.6 (-17.4 to 4.3); 0.234	
BMI, kg/m²	32.7 (5.9)	32.8 (5.8)	-0.14 (-0.5 to 0.18); 0.375	25.0 (3.6)	7.9 (7.2 to 8.6); <0.001	7.8 (7.1 to 8.5); <0.001	
Bust girth, mm	1161.5 (125.2)	1113.9 (111.2)	47.6 (40.0 to 55.2); <0.001	947.8 (82.3)	216.3 (201.4 to 231.1); <0.001	166.2 (151.1 to 181.3); <0.001	
Waist girth, mm	979.3 (129.3)	982.1 (130.2)	-2.8 (-10.2 to 4.2); 0.460	806.9 (89.1)	174.5 (158.3 to 190.7); <0.001	175.2 (158.3 to 192.1); <0.001	
Hip girth, mm	1171.8 (128.9)	1178.5 (128.5)	-6.6 (-13.4 to 0.15); 0.055	1040.4 (76.2)	131.9 (117.3 to 146.5); <0.001	138.1 (122.6 to 153.5); <0.001	
Shoulder width	366.0 (23.0)	366.4 (22.9)	-0.35 (-3.4 to 2.8); 0.825	368.5 (19.3)	-1.3 (-4.4 to 1.8); 0.410	-2.1 (-5.4 to 1.3); 0.230	
Nipple-to-nipple, mm	240.8 (37.2)	214.6 (22.2)	26.9 (21.7 to 32.1); <0.001	189.5 (22.2)	51.5 (47.2 to 55.8); <0.001	25.2 (21.5 to 28.9); <0.001	
Breast point-to-point	235.5 (29.1)	218.3 (20.8)	17.2 (13.6 to 20.8); <0.001	-	-	-	
Under-bust girth	938.1 (113.4)	933.2 (105.4)	4.9 (-1.8 to 11.5); 0.148	-	-	-	

Table 6.1 Anthropometric measurements for surgical participants in comparison to normative general population sample

[†] Paired samples *t*-test

[‡] Independent samples *t*-test

6.3.4 3D laser scanner

A total of 217 participants had 3D laser scans at baseline and 189 participants at 12 months postoperative. Descriptive statistics and mean comparisons before and after surgery for breast and body volumes measured by the 3D laser scanner are presented in Table 6.2. Breast and body volume measurements were compared for surgical participants with both a pre-operative and 12month post-operative assessment using a paired samples *t*-test. A statistically significant difference was found when comparing individual breast volumes at baseline to 12-months postoperative measurements. Pre-operatively, mean total breast volume measured by 3D laser scanner was 3339 mL (range 1472 to 9622 mL). At 12 months post-operatively, mean total breast volume was significantly lower at 2193 mL (range 963 to 4392 mL). In contrast, no significant difference was found in total body volumes. This finding was consistent with the body mass index calculated from height and weight measurements at pre- and 12 months post-operative timepoints.

20	Mean	i (SD)	Difference in means	
volume (mL)	Pre-operative (<i>n</i> = 162)	Post-operative (<i>n</i> = 162)	(95% CI)	<i>p</i> -value
Left breast	1697.9 (569.2)	1137.8 (318.6)	560.2 (494.0 to 626.3)	<0.001
Right breast	1641.3 (559.5)	1055.9 (290.7)	585.5 (514.1 to 656.8)	<0.001
Total breast	3339.2 (1115.1)	2193.6 (589.5)	1145.6 (1010.7 to 1280.5)	<0.001
Total body	89394.0 (17936.6)	89864.2 (17885.0)	-470.2 (-1149.3 to 508.8)	0.344

Table 6.2 Breast and bod	y volumes measured b	y 3D scanner	pre- and 12 months	post-operative
--------------------------	----------------------	--------------	--------------------	----------------

When comparing the change in breast volumes measured by the 3D laser scan to the weight of tissue resected from each breast at surgery, there was a strong positive correlation found (r = 0.851, p < 0.001). This finding is presented in Figure 6.3 below.



Figure 6.3 Relationship between the change in breast volume and weight of tissue resection at surgery

6.3.4.1 Relationship between breast volume and bra cup size

The relationship between pre-operative breast volume measured using the 3D laser scanner and patient-reported bra cup size was also assessed. While there was a trend of increasing breast volume with increasing bra cup size, there was considerable variability within cup size categories (Figure 6.4). Furthermore, correlation analysis revealed only a weak positive association between the two variables (r = 0.338, n = 217, p < 0.001). As an example, participants who self-reported a bra cup size of DD cup were found to have a mean breast volume of 3116 mL (SD 1303 mL) and ranged between 1601 mL and 9622 mL. A one-way ANOVA was conducted to compare mean breast volumes between the seven bra cup categories. However, post-hoc comparisons using the Tukey test indicated that only the mean breast volumes for I/J bra cup size were significantly different from; C/D cup size (p < 0.001), DD cup (p < 0.001), E/EE cup (p = 0.001), F/FF cup (p = 0.002), and G/GG cup (p = 0.004). Breast volumes for all other bra cup categories did not differ significantly from one another.



Figure 6.4 Distribution of measured breast volume by bra cup size

6.3.4.2 Relationship between breast volume and anthropometric variables

Correlation analysis was conducted to evaluate the relationship between pre-operative breast volume estimated using 3D laser scanner with anthropometric measurements (n = 217). Table 6.3 provides a summary of these findings and demonstrates that pre-operative breast volume was most strongly correlated with the amount of tissue resected at surgery, indicating that women with larger breasts had more breast tissue excised. Pre-operative breast volume also demonstrated a large positive correlation with sternal notch-to-nipple distance, bust girth, breast point-to-point distance, total body volume, weight and BMI. The association with these latter three variables indicates that heavier women or those with a larger body size tend to have larger breasts. All other variables showed a medium correlation with breast volume with the exception of height, which was found to have a weak association.

Table 6.3 Relationship between pre-operative breast volume and anthropometric and 3D scan volume variables

	Pearson's r	<i>p</i> -value
	correlation to	
	preoperative	
	breast volume	
3D scan total body volume	0.614	<0.001
Tissue resection weight	0.843	<0.001
Unisex chest girth	0.542	<0.001
Bust girth	0.687	<0.001
Under-bust girth	0.575	<0.001
Intermammary fold girth	0.505	<0.001
Nipple-to-nipple distance	0.580	<0.001
Breast point-to-point distance	0.612	<0.001
Suprasternal notch-to-nipple distance	0.723	<0.001
Bra cup size	0.320	<0.001
BMI	0.612	<0.001
Weight	0.613	<0.001
Height	0.153	0.025
Hip girth	0.563	<0.001
Waist girth	0.542	<0.001
Shoulder width	0.345	<0.001
Hip width	0.620	<0.001

6.3.4.3 Breast asymmetry

Individual breast volumes estimated by 3D laser scanning were used to evaluate the degree of breast asymmetry pre-operatively within the surgical cohort. Figure 6.5 presents the distribution and degree of breast asymmetry when comparing left breast to right breast volume at the baseline assessment. The median difference in volume between breasts was found to be 166 mL (mean 149.7 mL, SD 115.2 mL) and ranged from 1 mL to 532 mL, with a difference of greater than 200 mL found in 62 participants (28.6%). When assessing the degree of asymmetry between breasts as a proportion of total breast volume, the median degree of breast asymmetry was found to be 3.97% and ranged from 0% to 17.7%. The majority of participants (30.1%) had between 5% and 10%; and 13 participants (6.3%) were found to have greater than 10% difference in volume between breasts.



Figure 6.5 Frequency distribution of the degree of breast asymmetry at baseline assessment

6.3.4.4 Breast size as a proportion of body size

Using 3D scan breast volume measurements, the extent of 'disproportionality' was assessed using the total breast volume as a proportion of the total body volume at the pre-operative assessment. The frequency distribution is presented in Figure 6.6. The median ratio of breast to body volume was 3.74% and ranged from 1.49% to 7.88%. Breast volume as a proportion of total body volume was less than 5% for the majority of participants (197/217, 90.8%).



Figure 6.6 Frequency distribution of proportion of breast to body size

The relationship between the degree of disproportionality and the level of improvement in healthrelated quality of life following surgery as measured by the SF-36 physical summary scores (SF-36 PCS) was investigated using Pearson correlation analysis. It was found that there was no significant association between the two variables (r = 0.137, p = 0.09) (Figure 6.7).



Figure 6.7 Relationship between disproportionality and change in SF-36 PCS score

The relationship between degree of disproportionality and the change in SF-36 mental summary scores (SF-36 MCS) was also evaluated using Pearson correlation analysis. There was no significant association found between the two variables (r = 0.10, p = 0.212) (Figure 6.8).



Figure 6.8 Relationship between disproportionality and change in SF-36 MCS score

6.3.4.5 Relationship between body volume and BMI

When comparing the total body volume measurement by 3D laser scanner to the calculated BMI from height and weight measurements at the baseline assessment, there was found to be a large positive correlation (r = 0.915, p < 0.001). This suggests a strong relationship between these two variables describing body size (Figure 6.9).



Figure 6.9 Relationship between total body volume and BMI

6.3.5 Water displacement

A total of 120 participants had water displacement measurements at the pre-operative assessment and 64 at the 12 months post-operative assessment. Mean (SD) breast volumes at baseline were found to be 1795.8 mL (839.2 mL) and 1791.5 mL (845.8 mL) for left and right breasts, respectively, with a mean total breast volume of 3587.3 mL (1667.2 mL). Post-operatively mean volumes were 929.5 mL (413.7 mL) and 916.9 mL (411.2 mL) for left and right breast volumes, respectively, with a mean total volume of 1846.4 mL (813.6 mL). Using a paired samples *t*-test, breast volumes were compared for those participants who completed both a pre- and postoperative assessment and these findings are presented in Table 6.4. Individual and total breast volumes were found to be significantly reduced at 12 months following surgery when compared to baseline.

	Mean (SD) breast volume (mL)		Difference in means	
	Pre-operative (<i>n</i> = 62)	Post-operative (<i>n</i> = 62)	(95% CI)	<i>p</i> - value⁺
Left breast	1716.2 (776.4)	915.7 (401.8)	800.5 (647.5 to 953.5)	<0.001
Right breast	1712.9 (772.0)	907.6 (407.7)	805.2 (646.2 to 964.3)	<0.001
Total breast	3429.1 (1519.5)	1823.3 (798.7)	1605.7 (1304.9 to 1906.5)	<0.001

Table 6.4 Breast volumes for surgical participants pre- and post-operatively

⁺ Paired samples *t*-test.

6.3.6 Comparison of breast volume measurement using 3D scan and water displacement

In comparing the individual breast volumes obtained from 3D laser scanning and water displacement, a Pearson correlation demonstrated a strong, positive linear association between the two different methods (r = 0.89, n = 322, p < 0.001), with higher volume estimates by 3D scanner measurement associated with higher volume estimates by water displacement (Figure 6.10).



Figure 6.10 Pearson correlation of breast volume measurement – water displacement versus 3D laser scan

A Bland-Altman analysis was subsequently conducted to evaluate the level of agreement between the two methods. Figure 6.11 presents the results of the Bland-Altman plot showing the differences between the two techniques plotted against the averages of the two techniques. A mean difference of -21.7 mL with a standard deviation of 399 mL was found between the two techniques. While most values were within the 95% limits of agreement (±1.96 x SD) on the plot (a lower limit of -804 mL and an upper limit of 760 mL), the majority of the data points are widely spread across this range and do not lie close to zero, which would have been the case if there was good agreement between the measured values. This analysis demonstrated there was a large difference in measurement of breast volume using the two methods.

Linear regression analysis confirmed a degree of proportional bias, meaning that one method gave values that were consistently higher or lower than those of the other method. In this instance, the breast volumes obtained from water displacement tended to be larger than those from the 3D scanner, and the difference between the two methods increased as the mean volume increased. This is evident from the plot where there was a discernible trend for more points to be above the line of mean difference in volume as the mean volume increased (Figure 6.11).



Figure 6.11 Bland-Altman plot of breast volume measured by water displacement and 3D scanning techniques.

The analyses therefore suggest that despite a strong linear correlation between the two methods of breast volume measurement, the measures have low agreement on actual values. The water displacement values were consistently larger than the 3D scan values. The greatest agreement in breast volume measurement between the two methods (within 10%) was seen for volumes between 1000 and 2000 mL (Figure 6.12).



Figure 6.12 The level of agreement between the two methods. Shaded area highlights the range of breast volumes with the highest level of agreement

6.3.7 Comparison of breast volume measurement using 3D scan and anthropometric formula

In comparing the total breast volumes obtained from 3D laser scanning and anthropometric formula, a Pearson correlation demonstrated a strong, positive linear association between the two different methods (r = 0.72, n = 414, p < 0.001), with higher volume estimates by 3D scanner measurement associated with higher volume estimates by the predictive formula (Figure 6.13).



Figure 6.13 Pearson correlation of breast volume measurement – anthropometric formula versus 3D laser scan

A Bland-Altman analysis was subsequently conducted to evaluate the level of agreement between the two methods. Figure 6.14 presents the results of the Bland-Altman plot showing the differences between the two techniques plotted against the averages of the two techniques. A mean difference of 757 mL with a standard deviation of 993 mL was found between the two techniques. While most values were within the 95% limits of agreement (±1.96 x SD) on the plot (a lower limit of -1189 mL and an upper limit of 2703 mL), the majority of the data points are widely spread across this broad range and do not lie close to zero, which would have been the case if there was good agreement between the measured values. Furthermore, the limits of agreement of breast volume, with total breast volumes estimated by the anthropometric formula consistently larger than the 3D scan values. Therefore, despite a relatively strong linear correlation between the two methods of breast volume measurement, the measures have low agreement on actual values and the proposed anthropometric formula is therefore an unsuitable substitute for 3D laser scanning.



Figure 6.14 Bland-Altman plot of breast volume measured by anthropometric formula and 3D scanning techniques

6.4 Discussion

A large number of techniques for the objective measurement of breast volume have been described in the literature. However, these techniques often vary in accuracy and reliability and some are limited in their application to large-breasted patients. Many studies specifically on women with breast hypertrophy aim to develop a formula for estimating breast reduction weight in reduction mammoplasty rather than the assessment of breast volumes both before and after surgery (Appel et al., 2010, Boukovalas et al., 2019, Chan et al., 2019, Descamps et al., 2008, Dvoracek et al., 2019, Eder et al., 2007, Hernanz et al., 2014, Kececi and Sir, 2014, Kocak et al., 2011, Sommer et al., 2002, Wampler et al., 2019). This is most likely driven by restrictions placed on breast reduction surgery by healthcare providers or insurance companies which often require surgeons to predict a minimum weight of resected breast tissue. Although a formula to predict tissue resection weight is clinically useful, there remains a need for the accurate determination of absolute breast volumes both in planning breast surgery and in assessing outcomes following surgery.

In this study a series of approaches including anthropometry, 3D laser scanning and water displacement were used to measure breast and body size in women with breast hypertrophy who underwent breast reduction surgery. These measurements were invaluable in order to investigate the relationship between factors including degree of breast hypertrophy, breast asymmetry and

disproportionality on quality of life outcomes following surgery; and overcomes limitations associated with existing studies that have used indirect measures such as bra cup size or resection weight as a proxy to describe breast size. While it was hypothesised that factors including disproportionality may influence the health benefits of surgery and be used to establish an evidence-based system to prioritise patients presenting for surgery, results from this study demonstrate that there was no significant relationship between the ratio of disproportionality and the improvement in the physical or mental summary measures of the SF-36. This finding supports the conclusions described in previous chapters that the improvement in quality of life is independent of these factors and that restrictions on access to surgery based on variables including BMI, a minimum weight of resection or by the degree of disproportionality are not supported by evidence.

Findings from this study indicate that the majority of women with breast hypertrophy who presented for breast reduction surgery had a degree of breast asymmetry. Almost one-third of participants in this study were found to have a difference of greater than 200 mL between their breasts. This finding was consistent with a previous study by Tenna and colleagues who reported a difference of greater than 200g between breasts in 20% of their patients with breast hypertrophy (Tenna et al., 2012). Another important finding from the present study was that breast volume measured using 3D laser scanning was only weakly correlated with bra cup size in women with large breasts. This finding is consistent with another study which used anthropometric measurements to predict breast volume formula in women with breast hypertrophy (Sigurdson and Kirkland, 2006) and highlights that bra cup size is a poor indicator of breast volume for use in outcome studies exploring the relationship of breast volume to other clinical and outcome variables. Additionally, the results of this study demonstrate that breast size was found to be strongly correlated with body size and body mass index. In other words, women with a larger BMI and body volume tend to have larger breasts. This finding further demonstrates that BMI is an inappropriate exclusion criteria as women with a higher BMI are likely to have a larger breast volume and supports previous literature (Benditte-Klepetko et al., 2007, Brown et al., 2012). Finally, measurements using anthropometry and 3D scanning demonstrated that whilst breast size was found to significantly decrease following surgery, body size and composition measurements did not differ. This result was confirmed by the finding that BMI and body weight remained similar at pre- and post-operative timepoints.

The most accurate metric of true breast tissue volume is water displacement of mastectomy specimens, however, this is not applicable other than in post-mastectomy reconstruction scenarios (Kayar et al., 2011, Losken et al., 2005, Yip et al., 2012). Additionally, volume measurement of the intact breast has proven to be more variable due to the challenge of correctly differentiating breast tissue from the chest wall. Within the Plastic Surgery Unit at Flinders Medical Centre, 3D laser scanning has previously been validated as an accurate and reliable method of breast volume measurement and demonstrated equivalence in comparison to mastectomy volume (Yip et al.,

2012) and in comparison to breast volume measured by MRI (Howes et al., 2017). The present study further enhances the understanding of measuring breast volume using 3D laser scanning with a comparison to direct measurement of the intact breast by water displacement, and with a comparison to breast volumes measured by anthropometry in women with breast hypertrophy. The measurement of breast volumes by both 3D laser scan and water displacement showed a close association between the two techniques. However, it was found the water displacement method had a clear tendency to estimate a larger volume than the 3D scan. A possible explanation for the difference between techniques is that measurement of the intact breast volume using water displacement requires the patient to be in a prone position, which may change the form of the breast and add axillary or abdominal fat to the breast measurement. In contrast, patients are in the upright position for measurement with the 3D scanner. Another source of difference is that 3D scanning technology allows for a curved cut plane at the back of the breast where it can digitally follow the curved chest wall, while water displacement has by definition a flat cut plane at the water level. The water displacement technique is also highly dependent on the positioning of the patient. Alternative devices for the water displacement technique have been described to address the disadvantages of the immersion technique used in this study (Tezel and Numanoglu, 2000). This study therefore supports previous findings that, although water displacement is easily accessible and a low-cost option, the technique has relatively low accuracy for measurement of the intact breast due to poor reproducibility (Bulstrode et al., 2001, Henseler et al., 2011).

When comparing breast volume measurement by 3D laser scanning to anthropometric measurements for use in a predictive formula, it was found that the two methods had low agreement on values with the proposed formula resulting in significantly larger estimates of breast volume. This is likely explained by the limitations of the proposed formula which is based on geometric shape calculations when applied to large, ptotic breasts. This is consistent with the finding that whilst linear anthropometric measurements for use in predictive formula are fast and the costs are negligible, they are often less accurate and have been shown to be limited to estimates of breast volumes within a specific range (Longo et al., 2013, Qiao et al., 1997, Sigurdson and Kirkland, 2006). Therefore, the selection of the most appropriate formula and anthropometric measurements is crucial to their application in the clinical setting.

Strengths of this study include the relatively large sample size and that the data was collected prospectively. Furthermore, the manual positioning of landmarks prior to 3D scanning using palpation to accurately identify the margins of the breast base and control for potential location error, rather than relying on marking these points later in the scan image. Finally, while existing studies have relied on proxy measurements for breast and body volume such as tissue resection weight and BMI, the direct and accurate measurement of breast and body size in the present study provides a more comprehensive assessment to further enhance the assessment of outcomes

following surgery.

This study has several limitations that should be noted. Firstly, not all participants had breast volume measurements using all techniques at both the pre-operative and 12 months post-operative timepoints. It was decided that additional measurement by water displacement was too onerous for some patients given the commitments associated with voluntarily participating in the study. As a result, both of these techniques were performed on discrete subsets of patients at each timepoint. However, the majority of participants had an anthropometric body shape assessment and 3D scanning both pre- and post-operatively. Lastly, while a series of anthropometric measurements were taken to assess breast and body size, anatomic measurements specific to alternate published predictive anthropometric formulae to estimate breast volume were not performed. This limited the ability to explore the agreement between breast volume estimations using these predictive formulae and 3D scanning in the present study population.

6.5 Conclusion

This study measured breast and body size using water displacement, anthropometry and 3D laser scanning in women with large breasts undergoing breast reduction surgery. The degree of disproportionality in women with breast hypertrophy was not found to influence the improvement in health-related quality of life following surgery. Furthermore, breast size was strongly associated with body size and BMI, indicating that larger women tend to have a greater breast volume. Therefore, restrictive criteria or prioritised waiting lists based on BMI or disproportionality are not appropriate based on this evidence. Significant correlations were found between breast size and anthropometric variables, in particular sternal notch-to-nipple distance. In contrast, breast volume was found to be weakly associated with bra cup size, highlighting that cup size is not an accurate indicator of breast volume in clinical outcome studies. While this study found a strong association between the measurement of breast volumes using water displacement and 3D laser scanning techniques and anthropometric measurements and 3D scanning in women with breast hypertrophy, the methods had low agreement on actual values. While water displacement and anthropometry may be more convenient and accessible in clinical practice, 3D scanning remains preferable as it has been proven to be a more accurate and reliable technique for the determination of intact breast volume.
7. ECONOMIC EVALUATION OF BREAST REDUCTION SURGERY IN AUSTRALIA

7.1 Introduction

Measurement of health-related quality of life using validated outcome measures is increasingly being employed to measure the burden of disease and the impact of treatment and provide comparative data between different surgical and medical interventions. Traditionally, health economists attempt to guide priority-setting and resource allocation decisions by healthcare funders by measuring and valuing both the quality and quantity of health, typically measured in quality-adjusted life years (QALYs). The estimated QALY gain is often calculated from health utility scores, which provide a single measure of quality of life for a given health state, using the area under the curve approach (Matthews et al., 1990).

A cost-utility analysis is considered a subset of cost-effectiveness analysis and compares net costs against net health outcomes as measured by the QALY. As a cost-utility analysis provides a consistent unit of measure (incremental cost per QALY gained), direct comparisons can be made between funding options, and therefore this analysis is preferred by the Australian Government's Pharmaceutical Benefits Advisory Committee (PBAC) (Pharmaceutical Benefits Advisory Committee, 2016, Taylor and Jan, 2017) and Medical Services Advisory Committee (MSAC) (Medical Services Advisory Committee, 2016). The incremental cost-effectiveness ratio (ICER) threshold incorporates a society's willingness-to-pay in terms of the incremental cost for an incremental QALY gain. In Australia, while there is no explicit value, research studies have suggested that an acceptable threshold is AUD\$50,000 per QALY gained (George et al., 2001, Henry et al., 2005). This implicit threshold was further supported when comparing ICERs in paired submissions to the PBAC in Australia versus NICE in the United Kingdom, the decision-making outcomes were found to be largely consistent in being above or below their corresponding threshold (Wang et al., 2018). More recently, Edney and colleagues proposed a lower estimate of AUD\$28,033 per QALY gained for the ICER as a reference to inform value-based decision making in Australian health care system (Edney et al., 2018).

In an era of tight healthcare budgets, it has become increasingly important to quantify the value of surgical interventions for the treatment of non-life-threatening conditions such as breast hypertrophy. Despite published evidence on the health gains of breast reduction surgery, there is a lack of high-quality research internationally regarding the cost-effectiveness of surgery. In addition, there are no published studies from the Australian healthcare (Medicare) perspective.

The primary objective of this study was to conduct an economic evaluation of bilateral breast reduction surgery in the Australian public health system. This study describes findings of a

prospective outcome study with a 12-month follow-up and includes the comparison of surgical participants to a reference population of women with breast hypertrophy who did not undergo surgery. The results of this study, in terms of incremental costs per QALY gained, will help determine whether breast reduction surgery is justifiable for government funding within the Australian healthcare sector.

7.2 Participants and methods

7.2.1 Participants

The data for this health economic evaluation were obtained from the prospective cohort study described previously in Chapter 2, comprising a surgical cohort and a breast hypertrophy control cohort. All participants were referred to the Department of Plastic and Reconstructive Surgery, Flinders Medical Centre, Adelaide, South Australia. In summary, women who underwent bilateral breast reduction surgery completed assessments pre-operatively and at 3, 6 and 12 months following surgery. A control group of women with breast hypertrophy who were actively seeking breast reduction surgery but were not expected to undergo surgery within 12 months were recruited for comparison and completed study questionnaires via postal survey. Ethics approval was obtained for this study from the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC reference numbers 118.056 and 73.17) (Appendix B and Appendix C).

7.2.2 Healthcare costs

Direct hospital costs for the surgical intervention were determined from the Australian (Medicare) healthcare perspective. Flinders Medical Centre is a tertiary care teaching hospital with 593 beds capacity and is predominantly funded through the Australian Medicare Scheme. Individual patient costs were obtained from the Power Performance Management (PPM) system (PowerHealth Pty Ltd, Adelaide, Australia) using surgical procedure codes (International Statistical Classification of Diseases and Related Health Problems, Australian Modification, ICD-10-AM codes)(IHPA, 2018) including the hospital costs of the original surgical admission and hospital costs for all readmissions and outpatient appointments within 12 months of surgery. Cost data for hospital admission was captured at the individual patient level and included theatre costs, specialist costs, critical care, imaging, pathology, pharmacy, allied health, ward nursing, ward medical, ward supply and non-clinical salaries. All hospital costs within 12 months of surgery for outpatient clinical appointments, return to theatre admissions and revisional procedures were included in the analyses to capture all direct hospital costs associated with surgical intervention. Costs were inflation-adjusted to 2016 Australian dollars using the relevant consumer price index (CPI), as this was the final year of recruitment in the study (Australian Bureau of Statistics, 2016b). Indirect costs such as those incurred by the participant including productivity costs due to possible absence from work were not included.

7.2.3 Health-related quality of life

Health-related quality of life (HRQoL) was assessed using the Short Form-36 (SF-36) version 2 instrument (Ware and Sherbourne, 1992). Surgical participants competed the SF-36 questionnaire pre-operatively and again at 3, 6, and 12 months post-operatively. Participants in the breast hypertrophy control group who were awaiting consultation for surgery completed the questionnaire at baseline and at 12 months following. This was described in further detail in Chapter 2.

Responses to individual items from the SF-36 questionnaire can be used in economic evaluations by transformation of responses into a single SF-6D utility score. In this study, the preferencebased scoring algorithm developed by Brazier and colleagues was used to calculate individual SF-6D health utility scores for participants at each study timepoint (Brazier et al., 2002). This original algorithm was derived from a general population sample from the United Kingdom (UK). An algorithm for Australian preference weights for the SF-6D utility has since been developed by Norman et al. and was also used to convert SF-36 responses to individual utility scores in this study (Norman et al., 2014). The algorithm calculates a SF-6D index score based on responses to 11 of the 36 items from 6 dimensions of the SF-36 questionnaire including Physical Functioning (items 3a, 3b, and 3j); Role Limitation due to physical problems (item 4c) and Role Limitation due to emotional problems (item 5b); Social Functioning (item 10); both bodily pain items (items 7 and 8); Mental Health (items 9b and 9f); and Vitality (item 9e) (Appendix D). The SF-6D defines utility values for 18,000 possible health states of the SF-36 on a scale, with a score of one representing full health and a health state equivalent to being dead valued at zero. Negative values indicate a poor health state considered worse than death.

For comparison purposes, population normative SF-36v2 scores were obtained from the 2008 South Australian Health Omnibus Survey (SAHOS), with de-identified raw data kindly provided by David Banham and Professor Robert Goldney, Department of Psychiatry, University of Adelaide. Normative SF-6D scores were then derived from the SF-36 responses from female respondents using the United Kingdom and Australian preference weights described previously. Normative SF-6D population data was age-standardised to correspond to the age distribution of the surgical participants.

7.2.4 Cost-utility analysis

Effectiveness of the surgical intervention was measured in terms of quality-adjusted life years (QALYs) gained. The QALY gain over the 12 month period was calculated at an individual patient level using SF-6D utility values and the area under the curve method (Drummond, 2005). A cost per QALY was calculated by dividing the mean costs by the mean number of QALYs gained to provide an estimate of cost-utility. Both unadjusted and adjusted QALYs (adjusted for any differences in SF-6D utility scores at baseline) were calculated. As there is no SF-6D data for the control group at the 3- and 6-months follow-up periods, only baseline and 12 months SF-6D data

were used in calculating the QALYs for the base case analysis. QALYs based on all four timepoints in the intervention groups were also calculated and examined within a sensitivity analysis (see Section 7.2.5). For the base case, 12-month QALYs were extrapolated to a 10-year time horizon as HRQoL gains were shown to be maintained in this longer time period as described in Chapter 5. As per PBAC and MSAC guidelines, a discount rate of 5% per year was applied to QALY gains over any period beyond 1 year (Medical Services Advisory Committee, 2016, Pharmaceutical Benefits Advisory Committee, 2016). Calculations for discounting involve multiplying the value of costs and benefits for each year in the future using the formula: $(1/(1 + D)^y)$, where the discount rate is denoted by *D* and *y* is the number of years (Drummond, 2005). However, whilst the health gains from treatment were expected to last several years, the direct costs of surgery in this study were expected to be confined to the 12-month study period and were therefore not discounted.

The economic evaluation used in this study was a cost-utility analysis (CUA), with primary outcomes expressed in terms of QALYs. Secondary outcomes of the CUA were expressed as incremental costs per QALYs gained in order to determine the incremental cost effectiveness ratio (ICER). The ICERs were calculated as the differences in costs between the intervention and the control group divided by the difference in QALY gain: $ICER=C_a - C_b/E_a - E_b$, where C_a is the cost of the intervention, C_b is the cost of the control, E_a is the effectiveness of the intervention, and E_b is the effectiveness of the control. In this study, cost-effectiveness at willingness to pay thresholds of AUD\$28,033 per QALY gained (Edney et al., 2018) and AUD\$50,000 per QALY gained (Harris et al., 2008) were used for decision-making. Bootstrapping was used to account for uncertainty due to sampling variation in the ICER; and 5,000 paired estimates of mean differences in costs and outcomes (QALYs) were derived. The bootstrapped pairs were summarized in cost-effectiveness planes (CEPs) (Black, 1990). Figure 7.1 illustrates the decision rules for the four-quadrants of the incremental cost-effectiveness plane. In interpreting the results of the cost-effectiveness planes, observations that fall within the 'north-west' quadrant indicate the intervention is less effective and more costly than the comparator and therefore rejected; and in the 'south-east' quadrant the intervention is more effective and less costly than the comparator and therefore has an acceptable cost-effectiveness profile. Observations that fall within the 'north-east' quadrant are more effective and more costly than the comparator; and in the 'south-west' quadrant the intervention is considered to be less effective and less costly than the comparator, with both quadrants indicating the intervention is potentially acceptable. The probability of the intervention being more costeffective compared to the control at varying willingness-to-pay thresholds was presented using cost-effectiveness acceptability curves (CEACs) (Fenwick et al., 2001).







Due to the presence of missing data on costs and outcomes, multiple imputation was used to account for missing values. Data was deemed to be missing at random (MAR) and imputed values were generated for cost and outcome variables using the multiple imputation with Predictive Mean Matching (PMM) algorithm, a method that matches the missing value to the observed value with the closest predicted estimate (Little, 1988, Rubin, 1986). The PMM method was used as the normality assumption for the SF-6D was violated, that is, the SF-6D scores were not normally distributed. A total of 50 multiple-imputed complete datasets were generated and the following variables were used to predict the missing values for the surgical cohort: age, height, weight, body mass index, smoking, bra cup size, and length of hospital stay. For the control cohort the variables used to predict missing values included age, height, weight, body mass index, smoking and bra cup size. The multiple imputed data is presented as the base case, with the complete case analysis explored in a sensitivity analysis.

7.2.5 Sensitivity analysis

Sensitivity analyses were performed to test the robustness of the base case results. The first sensitivity analysis focused on assessing the impact of missing cost and outcome data by

164

comparing results of the economic evaluation based on complete cases with those estimated using multiple imputation. The second sensitivity analysis explored using QALY estimates based on the four study timepoints in surgical intervention group in comparison to results based on baseline and 12-month follow-up only. The third sensitivity analysis considered 12-month QALYs extrapolated 40 years (average life years remaining) under the assumption that the HRQoL gain was maintained until the end of the statistical life expectancy for the study cohort with a mean age of 43 years; for females in Australia this was 84.6 years (Australian Institute of Health and Welfare, 2019a). Although this is not strictly accurate as HRQoL is known to deteriorate over time, this approach is widely used for the calculation of QALYs gained following medical interventions. Finally, all sensitivity analyses explored results using UK versus Australian preference-based weights for transforming the SF-36 responses to health utilities, and cost-effectiveness was determined at two different willingness-to-pay thresholds of \$28,033 and \$50,000 per QALY gained.

7.2.6 Statistical analysis

Statistical analyses were performed using SPSS v25.0 statistical software for Windows (IBM Corp., Armonk, N.Y.) and Stata statistical software version 16 (StataCorp, College Station, TX). Continuous variables were presented as mean values with standard deviations, and differences between groups as mean differences with 95% confidence intervals. Comparisons of participant socio-demographic variables were made between groups using an independent samples *t*-test. Categorical variables were expressed as frequencies (percentage) and were compared using chi-square (χ^2) statistics or Fisher's exact test as appropriate.

7.3 Results

A total of 251 participants were enrolled in the study and underwent bilateral breast reduction surgery. Of these, 209 (83.3%) completed at least one post-operative assessment; 191 (91%) at 3 months, 183 (88%) at 6 months and 193 (92%) at 12 months. Missing data were due to participants repeatedly not attending appointments or choosing to not complete and return the study questionnaires. Twenty-three participants formally withdrew from the study following surgical intervention and were therefore excluded from the analysis. A total of 159 (63.3%) participants completed all four study timepoints. Of these, 141 had no missing responses to items from the SF-36 questionnaire and therefore SF-6D utility scores were generated. This was the study group used for the complete case cost-utility analysis.

In the breast hypertrophy control group, study questionnaires were initially posted to 350 women on the waiting list; 160 (46%) completed and returned the questionnaires at baseline, and of these 124 responded again 12 months later. Of these, SF-6D utility scores were able to be generated from 119 participants in the control hypertrophy group and this was the study group used for the

165

complete case analysis. Baseline demographic characteristics from the surgical and hypertrophy control groups were similar between the two groups and are summarised in Table 7.1.

	Surgical intervention	Control hypertrophy	
Characteristic	(<i>n</i> = 209)	(<i>n</i> = 124)	<i>p</i> -value⁺
Age mean (SD), years	42.6 (13.4)	45.3 (13.1)	0.079
BMI, mean (SD), kg/m²	32.7 (6.0)	32.1 (6.0)	0.468
Obesity status:			
Non-obese <30 kg/m ²	71 (34)	48 (39)	0.326
Obese ≥30 kg/m²	138 (66)	74 (61)	
Smoking status:			0.243
Non-smoker	108 (52)	78 (63)	
Current smoker	35 (17)	14 (11)	
Reformed < 1 year	15 (7)	5 (5)	
Reformed ≥ 1 year	47 (23)	25 (20)	

Table 7.1	Baseline	characteristics	of	participants
-----------	----------	-----------------	----	--------------

Values are numbers (percentages) unless stated otherwise.

+ Using an independent samples *t*-test, X^2 or Fisher's exact test as appropriate.

7.3.1 Healthcare costs

Direct healthcare costs were calculated at the individual patient level for the original surgical admission and included relevant hospital costs for all readmissions and outpatient appointments within 12 months of surgery. A summary of the mean costs (in 2016 Australian dollars) for the surgical intervention across the 12-month period are shown in Table 7.2. The greatest cost was for the hospitalisation episode (AUD\$9464), with outpatient clinic appointments providing an average additional cost of AUD\$740 to the cost of the surgery. Any required return to theatre episodes and/or revisional procedures were included in the total direct hospital costs. A comparison of costing data from the National Hospital Cost Data Collection (NHCDC) Australian Public Hospitals Cost Report 2015 to 2016, Round 20, for AR-DRG version 8.0 reported the mean cost of surgery in an Australian public hospital to be AUD\$9199.96, which is consistent with our findings (IHPA, 2018). The breast hypertrophy control group did not incur any direct hospital costs as they did not undergo any intervention during the study timeframe.

Table 7.2 Costs per participant fo	r intervention and control g	groups over 12 months (\$AUD)
------------------------------------	------------------------------	-------------------------------

Costs	Intervention	Control	Between-group difference
	Mean (SD)	Mean (SD)	(95% bootstrapped CI)
Base-case analysis (imputed)	<i>n</i> = 209	<i>n</i> = 124	
Direct cost – hospital stay	9464 (3621)	0.00 (0.00)	9464 (9244 to 9685)
Outpatient clinic consultation	250 (105)	0.00 (0.00)	250 (234 to 266)
Plastic surgery outpatient treatment	490 (320)	0.00 (0.00)	490 (458 to 522)
Total cost	10,204 (3719)	0.00 (0.00)	10,204 (9981 to 10,427)
Sensitivity analysis (complete cases)	<i>n</i> = 196	<i>n</i> = 124	
Direct cost – hospital stay	9370 (3624)	0.00 (0.00)	9370 (8859 to 9880)
Outpatient clinic consultation	250 (102)	0.00 (0.00)	250 (236 to 265)
Plastic surgery outpatient treatment	492 (311)	0.00 (0.00)	492 (448 to 536)
Total cost	10,112 (3728)	0.00 (0.00)	10,112 (9863 to 10,360)

7.3.2 Outcomes

Health-related quality of life outcomes as assessed by the SF-6D and scored using both the Australian and UK scoring algorithms are presented in Figure 7.2. Utility scores in the surgical intervention group significantly increased from baseline to 3 months and remained stable at 6 and 12 months (p < 0.001). In contrast, health-related quality of life decreased in the control group from baseline to 12 months.

Australian female normative SF-6D values were derived from SF-36 data using the two different algorithms for comparison to study participants, resulting in an age-adjusted mean score of 0.65 (SD 0.04) (Australian weights) and 0.76 (SD 0.01) (Brazier's UK weights) from 1708 respondents. In comparison to population norms, participants with breast hypertrophy had significantly lower utility scores (p < 0.001). In the group who underwent surgical intervention, utility scores significantly increased (mean change 0.32 points using Australian weights and 0.15 points using UK weights) to levels equivalent to those of the normal population (Figure 7.2).

Within the 12-month time frame of the study, the QALY gains were higher for the surgical intervention group (mean 0.469, SD 0.223) than for the control (mean 0.283, SD 0.235), with a mean difference of 0.186 (95% CI, 0.163 to 0.210) when using Australian SF-6D scoring weights. Similarly, QALY gains in the intervention group (mean 0.670, SD 0.106) were higher than in the control group (mean 0.590, SD 0.100) with a mean difference of 0.080 (95% CI, 0.074 to 0.085) using Brazier's UK scoring algorithm.

Table 7.3 summarises SF-6D utility scores and QALYs gained for the surgical intervention and control groups. For the base case analysis, the adjusted QALY gains over a 10-year time horizon were higher for the surgical intervention group (mean difference of 1.519; 95% CI, 1.362 to 1.675) for the SF-6D Australian valuation and for the SF-6D UK valuation (mean difference of 0.690; 95% CI, 0.618 to 0.761). A similar pattern was observed with the unadjusted analyses and complete case analyses.



a. SF-6D scores based on Australian preference-based scoring weights



b. SF-6D scores based on UK preference-based scoring weights

Figure 7.2 Mean SF-6D utility scores for surgical intervention and control cohorts in comparison to age-adjusted normative data

	Australian Scoring Algorithm		UK Scoring Algorithm			
Variable	Intervention	Control	Difference	Intervention	Control	Difference
	Mean (SD)	Mean (SD)	(Bootstrapped 95% CI)	Mean (SD)	Mean (SD)	(Bootstrapped 95% CI)
Base case analysis (imputed cases)	n = 209	<i>n</i> = 124		n = 209	<i>n</i> = 124	
SF-6D scores						
Baseline	0.313 (0.263)	0.296 (0.267)	0.017 (-0.011, 0.045)	0.596 (0.116)	0.595 (0.116)	0.001 (-0.012, 0.015)
3 months	0.575 (0.287)	-		0.719 (0.141)	-	
6 months	0.616 (0.294)	-		0.736 (0.154)	-	
12 months	0.626 (0.277)	0.270 (0.257)	0.356 (0.322, 0.390)	0.744 (0.139)	0.585 (0.110)	0.159 (0.147, 0.170)
Unadjusted QALYs	4.094 (1.941)	2.467 (2.050)	1.627 (1.425, 1.828)	5.841 (0.922)	5.145 (0.868)	0.697 (0.648, 0.745)
Adjusted ^a QALYs			1.519 (1.362, 1.675)			0.690 (0.618, 0.761)
Length of inpatient stay, days	2.39 (0.88)	-	-	-	-	-
Sensitivity analysis (complete cases) ^b	<i>n</i> = 141	<i>n</i> = 119		<i>n</i> = 141	<i>n</i> = 119	
SF-6D scores						
Baseline	0.314 (0.259)	0.295 (0.262)	0.017 (-0.011, 0.045)	0.596 (0.114)	0.595 (0.114)	0.001 (-0.012, 0.015)
3 months	0.570 (0.270)	-		0.718 (0.133)	-	
6 months	0.612 (0.280)	-		0.734 (0.143)	-	
12 months	0.628 (0.265)	0.270 (0.256)	0.356 (0.322, 0.390)	0.744 (0.135)	0.585 (0.109)	0.159 (0.147, 0.170)
Unadjusted QALYs	4.152 (1.927)	2.447 (2.060)	1.705 (1.307, 2.102)	5.859 (0.922)	5.132 (0.081)	0.727 (0.533, 0.920)
Adjusted ^a QALYs			1.491 (1.020, 1.961)			0.683 (0.477, 0.889)
Length of inpatient stay, days	2.43 (0.87)	-	-	-	-	-
Sensitivity analysis (imputed cases – S	F-6D values at	all 4 timepoints	s used for QALY calcula	tion in the interv	vention)	
Unadjusted QALYs	4.973 (1.883)	2.467 (2.050)	2.505 (2.216, 2.794)	6.245 (0.948)	5.145 (0.868)	1.101 (0.996, 1.204)
Adjusted ^a QALYs			2.419 (2.201, 2.637)			1.095 (0.991, 1.199)
Sensitivity analysis (imputed cases - Q	ALYs gained ov	ver mean lifetin	ne remaining years)			
Unadjusted QALYs	8.523 (4.041)	5.137 (4.269)	3.387 (2.967, 3.807)	12.16 (1.920)	10.71 (1.807)	1.451 (1.350, 1.551)
Adjusted ^a QALYs			3.162 (2.836, 3.488)			1.436 (1.288, 1.584)

Table 7.3 Utility scores and QALYs gained for the intervention and control groups

^a Predicted scores were adjusted for SF-6D utility score at baseline ^b Study participants with complete information at baseline and 12-month outcomes

7.3.3 Cost-utility analysis

7.3.3.1 Base case analysis

As the surgical intervention group was more expensive than the control group by \$10,204 per patient (95% CI: \$9,981 to \$10,427), the intervention was associated with an incremental cost-effectiveness ratio (ICER) of \$6,719 (95% CI: \$6,142 to \$6,919) per QALY gained (based on the Australian SF-6D valuation) and of \$14,795 (95% CI: \$13,811 to \$15,215) per QALY gained (based on the UK SF-6D valuation).

Figure 7.3 presents the cost-effectiveness planes for the base case analysis in the present study based on UK and Australian preference weights. All (100%) of the bootstrapped paired estimates of mean differences in costs and outcomes appear within the north-east quadrant, indicating that the intervention provides better health outcomes but at a higher cost than the control comparator. The tight grouping of data points indicates a low level of variability in the data and therefore a relatively high degree of certainty in the cost-effectiveness results.

The cost-effectiveness acceptability curves (Figure 7.4) show the probability of the intervention being cost-effective at different willingness to pay thresholds. When the Australian SF-6D valuation is used, the intervention had a 94% probability of being cost-effective at the revised willingness to pay threshold of \$28,033 per QALY gained, and 100% probability of being the cost-effective option at a willingness to pay threshold of \$50,000 per QALY gained which is the implicit criterion used for assessing the cost-effectiveness of new pharmaceuticals and medical services in Australia (Harris et al., 2008). When the UK SF-6D valuation is used, the intervention again had a 74% probability of being cost-effective at a willingness to pay threshold of \$28,033 per QALY gained, and 98% probability of being the cost-effective option at a willingness to pay threshold of \$50,000 per QALY gained.



a. Cost-effectiveness plane for the difference in QALYs calculated using Australian weights over 10 years



b. Cost-effectiveness plane for the difference in QALYs calculated using UK weights over 10 years

Figure 7.3 Cost-Effectiveness Planes



a. Differences in QALYs gained based on SF-6D utility scores calculated using Australian weights over 10 years



b. Differences in QALYs gained based on SF-6D utility scores calculated using UK weights over 10 years

Figure 7.4 Cost-Effectiveness Acceptability Curves

7.3.3.2 Sensitivity analysis

Sensitivity analyses were performed and found the results of the base case to be robust. Results of the sensitivity analyses are summarised in Table 7.3. In the base case analysis, multiple imputation was utilised to estimate missing data on costs (13 observations or 6.2%) and SF-6D utility scores at 12-months (16 observations or 7.7%). In the first sensitivity analysis, using the complete case dataset for analysis did not have an effect on the QALY gains or the incremental effectiveness over the 10-year time horizon (Table 7.3). The adjusted QALY gains over the 10-year time period were still higher for the surgical intervention group in comparison to the control group with a mean difference of 1.491 (95% CI: 1.020 to 1.961) for the SF-6D Australian valuation and a difference of 0.683 (95% CI: 0.477 to 0.889) for the SF-6D UK valuation. A similar pattern was observed with the unadjusted analyses. The surgical intervention group was more expensive than the control group by \$10,112 per patient (95% CI: \$9,863 to \$10,360). The resultant ICERs were \$6,784 per QALY gained (95% CI: \$6,416 to \$6,997) based on the Australian SF-6D valuation and \$14,808 per QALY gained (95% CI: \$13,681 to \$15,964) based on the UK SF-6D valuation.

The second sensitivity analysis considered using imputed dataset of SF-6D values collected at 4 time points to calculate QALY gains over 10 years. The adjusted QALYs gains over the 10-year period were much higher for the surgical intervention group than the control group with a mean difference of 2.419 (95% CI: 2.201 to 2.637) for the SF-6D Australian valuation and a mean difference of 1.095 (95% CI: 0.991 to 1.199) for the SF-6D UK valuation. A similar pattern was observed with the unadjusted or complete-cases analyses. The resultant ICERs were \$4218 (95% CI: \$3980 to \$4455) per QALY gained (based on the Australian SF-6D valuation) and \$9319 (95% CI: \$8615 to \$9756) per QALY gained (based on the UK SF-6D valuation).

The final sensitivity analysis extrapolated QALY gains to estimated lifetime years remaining. The adjusted QALY gains were higher for the surgical intervention group than the control group with a mean difference of 3.162 (95% CI: 2.836 to 3.488) for the SF-6D Australian valuation and a difference of 1.436 (95% CI: 1.288 to 1.584) for the SF-6D UK valuation. The intervention was associated with an incremental cost-effectiveness ratio (ICER) of \$3,227 per QALY gained (95% CI: \$2,950 to \$3,323) based on the Australian SF-6D valuation and \$7,106 per QALY gained (95% CI: \$6,633 to \$7,308) based on the UK SF-6D valuation.

Results of each of the sensitivity analyses did not change the final interpretation of the study as the impact of the intervention still exceeded the control and all ICERs for the intervention were considerably lower than the recommended willingness-to-pay thresholds of \$28,033 per QALY gained or \$50,000 per QALY gained.

All cost-utility analyses were conducted using the two different scoring algorithms for the calculation of SF-6D utility scores based on UK and Australian population weights. A scatterplot

comparing SF-6D utility values calculated using the two algorithms is presented in Figure 7.5, with a line indicating perfect equivalence. It was apparent that substantial differences existed in the health utility values derived using each scoring approach, particularly in the lower scoring range. As an example, the Australian algorithm estimated health states for some participants to be valued below zero, meaning they were considered to represent a health state worse than death. In contrast, Brazier's UK algorithm did not estimate any health states to be below zero, and the minimum value was found to be with 0.3. This finding is consistent with that reported by the developers of the Australian algorithm based on normative data (Norman et al., 2014).



Figure 7.5 Comparison of health state using different algorithms

7.4 Discussion

The findings of this study indicate that women with breast hypertrophy who were awaiting surgery in both surgical and control hypertrophy groups reported a significantly lower health status in comparison to the female general population. Following surgery health utility scores significantly improved to levels equivalent to the general female population, demonstrating the effectiveness of surgery. The mean improvement in SF-6D health utility of 0.32-points using the Australian valuation or 0.15-points using the UK valuation following surgery is well in excess of the minimal important difference reported by Walters and Brazier who found that the SF-6D MID for 11 disease groups ranged from 0.011 to 0.097 with a mean of 0.041 (Walters and Brazier, 2005).

The analysis for this study focused on conducting an economic evaluation of breast reduction surgery to determine the likelihood of the intervention being regarded as cost-effective within the Australian public healthcare setting. Results from this study indicate that the cost-effectiveness of surgery is favourable within an Australian context. The finding that that the QALY gain following breast reduction surgery is considerable and at a reasonable cost to the health system, with a mean cost per QALY gained estimated at \$6719 Australian dollars per QALY gained based on the Australian SF-6D valuation. Accordingly, results from the base-case analysis for this study show that the probability of breast reduction surgery being cost-effective was 100% at a willingness-to-pay \$50,000 per QALY gained and 94% at the lower threshold of \$28,033 per QALY gained. Therefore, under these reference criteria, the findings of our study demonstrate that surgery is justified for inclusion within the Australian healthcare system.

The time horizon of 10 years used in this analysis may be considered conservative as it is likely that the positive effects of breast reduction surgery continue over a longer period of time. Results from the long-term outcome study described in Chapter 5 support that the benefits of surgery were maintained and did not deteriorate over this follow-up. This finding provides reliable data to support that the cost-utility of surgery would be even more favourable given that the health benefits of surgery are likely to continue well beyond 10 years. This was confirmed with the results of the sensitivity analysis which extrapolated QALY gains over the average number of lifetime years remaining. Furthermore, the validity of the results was further strengthened by additional sensitivity analyses which demonstrated that changes in QALY calculations or choice of scoring algorithm did not change the main conclusions of the study.

Research assessing the cost-utility or cost-effectiveness of breast reduction surgery is limited and has only been described in a few countries worldwide including the United Kingdom (Taylor et al., 2004), Finland (Saariniemi et al., 2012, Tykka et al., 2010), Canada (Thoma et al., 2014) and Brazil (Araujo et al., 2014). Araujo and colleagues evaluated the cost-utility of reduction mammaplasty in the Brazilian public health system and reported a gain of 0.74 QALYs using the SF-6D from SF-36 data (Araujo et al., 2014). However, the average direct cost of surgery was low at R\$391.47 (approximately £104 or AUD\$130), corresponding to a cost per QALY of R\$536.26 (approximately £142 or AUD\$178). This cost-utility ratio reflects the extremely low compensation paid by funders to public hospitals and medical staff in Brazil. Unfortunately, further comparisons are limited as this study did not extrapolate QALY gains beyond the 6-month follow-up nor report any sensitivity analyses. In the United Kingdom, the cost per QALY for reduction mammaplasty was found to be between £4733 and £5729 (approximately AUD\$9427 and AUD\$11,411) with a three-year time horizon and was deemed to be cost-effective when compared to other interventions approved by NICE. However, HRQoL data using the SF-36 instrument was obtained from a Swedish study of 49 women, whilst the cost data was the average cost taken from the United Kingdom Department of Health Schedule of Reference Costs (Taylor et al., 2004). In addition, this study did not report

the number of QALYs gained or perform any sensitivity analyses. In Finland, the average direct hospital costs for reduction mammoplasty was approximately €3601 (approximately AUD\$6408) and the mean cost per QALY gained was €1180 (approximately AUD\$2100) (Saariniemi et al., 2012). In another Finnish study, the mean direct costs were found to be comparable at \in 3383 (approximately AUD\$6020), however, the cost per QALY of €3638 (approximately AUD\$6474) was considerably higher using the same 15D utility instrument (Tykka et al., 2010). Despite these differences, both of these studies found surgery to be cost-effective and to compare favourably with a number of other surgical procedures. Finally, Thoma and colleagues conducted a randomised-controlled trial to compare the cost-effectiveness of vertical scar reduction in comparison to inverted T-shaped reduction mammoplasty using the Health Utilities Index Mark 3 (Thoma et al., 2014). The authors found that inverted T-shaped reduction dominated vertical scar reduction from the Ministry of Health perspective by being slightly less costly (\$3090.06 versus \$3106.58 in 2012 Canadian dollars) and slightly more effective (0.87 quality-adjusted life years versus 0.86 quality-adjusted life years). In summary, whilst the comparison of findings from other economic evaluations may be problematic due to the variation between analyses with regard to factors including HRQoL instrument, health care system, currency values, perspective, time horizon and discount rate used, results from this study are consistent with previously published findings in supporting the cost-effectiveness of surgery.

In comparison to medical interventions for other chronic health conditions, findings from the present study provide evidence to demonstrate that the incremental cost per QALY gain following breast reduction surgery compares favourably. Assessing the effectiveness of surgery based on cost per QALY estimates enables other interventions with different outcome measures to be compared from within the Australian healthcare perspective to be compared to a societal willingness-to-pay threshold. Segal and colleagues investigated the cost-effectiveness of total hip replacement and total knee replacement for osteoarthritis in Australia and found surgery to be costeffective at an estimated cost per QALY of AUD\$7500 and AUD\$10,000, respectively (Segal et al., 2004). James and colleagues evaluated the cost-utility of three types of bariatric surgery in comparison with usual care for the treatment of obesity in Australia and reported an ICER of between AUD\$22,645 and AUD\$27,253 (James et al., 2017). Foteff and colleagues evaluated the cost-utility of cochlear implantation in comparison to bilateral hearing aids in Australian adults and found an ICER of AUD\$11,160 per QALY (Foteff et al., 2016). In 2014, Abell and Vote performed a comparative cost-effectiveness analysis of laser-assisted cataract surgery and conventional cataract surgery and found that whereas laser surgery was not cost-effective at AUD\$92,862 per QALY, conventional cataract surgery was at AUD\$4,378 per QALY gained (Abell and Vote, 2014). In summary, the cost per QALY gains following breast reduction compare favourably to other commonly implemented and widely accepted medical interventions within the Australian public healthcare setting.

7.4.1 Strengths and limitations

This study reports findings from the most comprehensive economic evaluation to date of the costeffectiveness of breast reduction surgery. A major strength of this study was the prospective collection of patient-reported HRQoL outcome data at multiple time points for estimation of the cost-utility. This enabled the calculation of the QALY gain from baseline to the 12-month postoperative time-point based on responses from the patients themselves rather than being reliant on estimates by clinicians and health economists. In addition, this study captured costing data at an individual patient level, accounting for factors that may influence total costs including variations in length of stay, in-hospital complications and return to theatre episodes. Furthermore, costing data included additional costs accrued beyond the initial hospital admission including all hospital outpatient clinic appointments and any revisional procedures over the 12-month follow-up period. Whilst the compliance rate for this study was relatively high, a sensitivity analysis was performed to compare the complete case dataset to an imputed dataset. Results were similar, indicating that the impact of multiple imputation did not lead to biased results.

In estimating the SF-6D health utility score, a population-specific algorithm was used that was developed based on data from the Australian population (Norman et al., 2014) alongside the original algorithm by Brazier was based on data from the UK population (Brazier et al., 2002). While it may be hypothesized that the use of the updated scoring weights best represents the preferences of the Australian population, it is likely results of an economic evaluation would be sensitive to the choice of algorithm. Therefore, a strength of this study was that all analyses were conducted to explore the sensitivity of results using the two algorithms, with both revealing a beneficial cost per QALY below the incremental cost-effectiveness ratio.

This study had several potential limitations when assessing the economic value of surgery. While we included direct hospital costs for each surgical participant, our analysis did not account for indirect costs to the participant and their families such as loss of work due to periods of absence. In addition, the economic evaluation was not a randomised clinical trial and did not include indirect costing data for the control breast hypertrophy group such as time off work, GP visits, specialist consultations, physiotherapy and other associated expenses outside of the hospital setting, and/or hospitalisations attributable to the condition. However, it is assumed to be likely that the indirect costs at baseline would have been similar in both groups with symptoms of breast hypertrophy and would not significantly alter the conclusions from this study. Whilst HRQoL was prospectively captured at baseline and at 12 months follow-up, data was not collected at 3- and 6-months in the hypertrophy control cohort; therefore, the differences between surgical intervention and control groups used baseline and 12-month follow-up data points only. However, a sensitivity analysis explored the use of the four study timepoints for estimation of QALYs and found that this did not alter the conclusions of the study. In fact, using the four study timepoints resulted in an even more favourable cost-utility ratio given the marked improvement observed between baseline and 3

months post-operative utility scores which results in a greater area under the curve estimate for QALY calculations.

7.5 Conclusion

The findings of this cost-utility analysis demonstrate that bilateral breast reduction surgery for symptomatic breast hypertrophy is likely to be cost-effective in the Australian healthcare setting with incremental cost per QALY gained considerably lower than the implicit cost-effectiveness threshold. This conclusion was further supported by the finding that the cost-effectiveness acceptability curves showed that the intervention had a high probability of being cost-effectiveness over a range of willingness-to-pay values. This study provides clinical and economic evidence that surgery is justifiable for government funding within the Australian healthcare setting.

8. THESIS CONCLUSIONS

8.1 Summary of thesis

The main objective of this thesis was to comprehensively evaluate the health burden of breast hypertrophy and the clinical- and cost-effectiveness of breast reduction surgery in improving health-related quality of life in Australian women. To investigate this objective a comprehensive literature review was conducted (Chapter 1) and subsequently five studies were conducted: a longitudinal prospective cohort study using validated generic patient-reported outcome measures comprised of a surgical group and a non-surgical control group (Chapter 2); a normative study to derive population reference data for the BREAST-Q Reduction module (Chapter 3); a prospective cohort study using the condition-specific BREAST-Q outcome measure (Chapter 4); a long-term follow-up study to determine whether or not surgery delivers long-term health benefits (Chapter 5); and an economic evaluation study to evaluate the cost-effectiveness of surgery from the Australian healthcare perspective (Chapter 7). In addition, the objective measurement of breast and body volume was undertaken (Chapter 6) to enable the comprehensive comparison between physical assessment and patient-reported outcomes. A summary as to how these studies formed a cohesive body of work to address the research objectives is addressed in this chapter. Implications for practice and healthcare policy are also discussed along with future directions for further research studies in the field of in plastic and reconstructive surgery.

A prospective cohort study was conducted to evaluate the health burden of breast hypertrophy and the comparative effectiveness of breast reduction surgery in improving health-related quality of life. Participants in this study were women who underwent breast reduction surgery and were followed up for 12 months; and a comparison control cohort comprised women with breast hypertrophy who did not undergo surgery. The primary outcome measure was health-related quality of life measured pre-operatively and up to 12 months following using the Short Form-36 (SF-36) and the Multidimensional Body-Self Relations Questionnaire (MBSRQ). Secondary outcome measures included post-surgical complications. This study demonstrated that women with breast hypertrophy had significantly lower scores compared with population norms across all SF-36 scales. In the surgical cohort, SF-36 scores improved to such an extent within 3 months of surgery that the health deficits were eliminated and quality of life scores were 'normalised' to levels equivalent to that of the general population across all eight dimensions and summary measures of the SF-36. This improvement was found to be sustained at 6- and 12-months following surgery. In contrast, SF-36 scores for the breast hypertrophy controls remained at baseline across 12 months. When assessing the relationship between participant or clinical characteristics and patient-reported outcomes, the improvement in health-related quality of life was independent of traditionally used criteria based on breast resection weight and body mass index and demonstrates the health benefits of surgery regardless of these factors. Women who underwent surgery were also found to

180

report significantly improved MBSRQ scores for how they evaluated their overall health, fitness, appearance and body satisfaction. Finally, the marked improvement in the SF-36 physical summary score was proven to be comparable to other widely accepted and approved surgical procedures including total joint replacement, spinal fusion, bariatric surgery, hernia repair, and coronary artery bypass grafting which are not the subject of restrictive access policies that are commonly enforced for breast reduction surgery. In addition, the improvement in the mental summary score following surgery was found to exceed that of all other interventions cited. In summary, this study provided evidence as to the comparative effectiveness of breast reduction surgery in relieving the functional symptoms and impaired quality of life associated with breast hypertrophy.

In Chapter 3, normative data were derived for the BREAST-Q Reduction module from women within the Australian general population, describing only the second set of normative values for this instrument reported worldwide. This study was undertaken to provide population-specific reference data for comparison to BREAST-Q scores from the surgical and breast hypertrophy control cohort (described in Chapter 4). Within the normative study, participants with a higher BMI and those with a chronic health condition were found to have significantly lower scores across all four BREAST-Q scales when compared to their respective comparison groups. Participants aged 40 years and above were found to have significantly higher scores for satisfaction with breasts, psychosocial wellbeing and physical wellbeing than their younger counterparts. Women with a larger bra cup size were found to have significantly lower scores for physical wellbeing than those with a smaller cup size. In a comparison between derived Australian normative data and published United States norms, this study demonstrated significant differences existed in all four BREAST-Q domains of satisfaction and wellbeing and highlighted the importance of using population-specific normative data wherever possible for the accurate interpretation of health-related quality of life data in research studies and in the clinical setting.

Findings of the prospective outcome study using validated generic measures such as the SF-36 were complemented with the administration of the condition-specific BREAST-Q to participants within the surgical and breast hypertrophy control groups pre-operatively and at 12 months follow-up. This study found that women with breast hypertrophy had significantly lower health-related quality of life scores across all BREAST-Q scales when compared to normative data. Following surgery, scores were found to significantly increase to levels equivalent to that of the general population. This is in contrast to BREAST-Q scores for women in the breast hypertrophy control group, with scores remaining at baseline levels. When assessing the relationship between participant characteristics and improvement in BREAST-Q scores, participants with a higher BMI were found to have a significantly greater improvement following surgery than those with a lower BMI in satisfaction with breasts, psychosocial wellbeing and physical wellbeing. In contrast, the level of improvement in all four BREAST-Q scales was found to be similar between patients who

had a post-operative surgical complication and those who did not. Overall, a high level of patient satisfaction was found following surgery, and satisfaction with outcome was not found to be affected by the presence of surgical complications. Furthermore, this study demonstrated that surgery was of health benefit to all women, including those who do not meet traditional insurance criteria requirements such as the 500-gram minimum weight of resection or using Schnur's sliding scale.

A long-term prospective outcome study was undertaken to evaluate the health benefits and patient satisfaction several years following breast reduction surgery. Participants who underwent breast reduction surgery and were followed up in the original 12-month outcome study self-completed the same series of validated patient-reported outcome measures up to 12 years following surgery. This study demonstrated that the majority of women continue to report a high level of satisfaction with the outcome of surgery and experience the ongoing health benefits and improvement in quality of life for many years after surgery; with no significant differences found when comparing long-term outcome scores with those reported at 12-months post-operatively. Furthermore, the comparison of long-term outcome data with population-specific norms for the SF-36 and BREAST-Q in this study confirmed that health-related quality of life remains at levels of women in the general population and did not decline towards levels reported pre-operatively over time; highlighting the health benefits of surgery continue for many years. The significant long-term improvement in health-related quality of life and relief of painful symptoms promotes the provision of insurance coverage for surgery as a functional rather than cosmetic procedure.

The objective measurement of breast and body size was conducted before and after breast reduction surgery using a series of techniques including water displacement, 3D laser scanning and anthropometry. This enabled the investigation of the relationship between breast and body size, and the comparison between physical assessment and health-related quality of life outcomes following surgery. This was an area of research that warranted further investigation, with a limited number of existing studies relying on proxy measurements for breast and body size such as tissue resection weight, bra cup size and BMI. An important finding was that the level of disproportionality in women with breast hypertrophy did not influence the improvement in healthrelated quality of life following surgery. In addition, breast size was found to be strongly correlated with body size and BMI, indicating that larger women tend to have a greater breast volume and provides evidence to further support that restrictive criteria should not be based on these factors. In contrast, breast volume was found to be weakly associated with bra cup size, highlighting that cup size is not an accurate indicator of breast volume in clinical outcome studies. This study also assessed the validity of measuring breast tissue volume using water displacement of the intact breast and anthropometry in comparison to 3D laser scanning in women with breast hypertrophy. While a strong association was found between the measurement of breast volumes using water displacement and 3D laser scanning techniques and also between anthropometric measurements

and 3D laser scanning in women with breast hypertrophy, the methods had low agreement on actual values and 3D scanning was proven to be a more accurate and reliable method for the determination of breast volume.

An economic evaluation of bilateral breast reduction surgery within the Australian public healthcare system was undertaken. SF-36 responses from the original prospective outcome study comprising a surgical cohort and a breast hypertrophy control cohort were transformed to a single SF-6D health utility score using preference-based scoring weights from Australia and the United Kingdom. Effectiveness of the surgical intervention was measured in terms of quality-adjusted life years (QALYs) gained. The findings of this cost-utility analysis demonstrate that bilateral breast reduction surgery for symptomatic breast hypertrophy is likely to be considered cost-effective in the Australian healthcare setting with incremental cost per QALY gained of AUD\$6719, which is considerably lower than the recommended cost-effectiveness threshold of AUD\$28,033 per QALY gained.

8.2 Limitations

Further to the limitations highlighted in individual study chapters, this body of research has several overall potential limitations that should be noted. Firstly, the prospective cohort studies described in this thesis were conducted at a single institution, potentially limiting the generalisability of these findings. Secondly, it is important to acknowledge the possibility of reporting bias when using patient-reported outcome measures in this type of research. This is a valid concern for patients seeking surgical treatment who may feel their responses could determine their access to surgery. Whilst participants in this study were informed that their responses to the health-related quality of life questionnaires would not be used to determine their access to surgery, it is possible that participants still overestimated their symptoms. However, the potential for bias in the surgical cohort was considered to be low as the pre-operative questionnaires were completed after participants had already been assessed by a plastic surgeon and were scheduled for surgical admission. It was also conceivable that when completing the study questionnaires post-operatively participants may have responded more favourably and felt an expectation to please the surgical team. In order to minimise the potential influence of the clinician or treating team on patient responses, study questionnaires were mailed to participants and these were self-completed independent of the hospital setting. Whilst it is acknowledged that the potential for bias is an inevitable accompaniment of this type of research design and may affect the strength of the conclusions, there is also no solution to this problem and efforts were made to limit this influence wherever possible when undertaking this research.

8.3 Original contribution to knowledge

The studies that comprise this thesis have provided a significant original contribution to knowledge

in the practice of evidence-based medicine and in the field of patient-reported outcomes following breast reduction surgery in a number of ways. Firstly, this body of research details the first studies to evaluate the clinical effectiveness of breast reduction surgery for the treatment of symptomatic breast hypertrophy in Australian women. In addition, the use of a combination of validated generic and condition-specific patient-reported outcome measures enabled a comprehensive assessment of health-related quality of life outcomes. Normative population data and the recruitment of a nonsurgical control group of women with breast hypertrophy who were referred for, but did not undergo, breast reduction surgery provided appropriate comparison groups and further strengthened the interpretation of outcomes.

This study reported the first general population reference values for the BREAST-Q Reduction module specific to Australia. The generation of normative data for the BREAST-Q from women within the Australian general population provided valuable insight into breast-related satisfaction and wellbeing in Australian women who were not pursuing breast surgery. This study established a valuable clinical reference point or benchmark for the interpretation of health-related quality of life scores; in turn providing new clinical context to better understand the health burden of breast hypertrophy or macromastia and the health benefits of breast reduction surgery. Prior to this normative data was limited to the United States population and findings from this study demonstrated that potentially significant differences existed when comparing health-related quality of life between populations. When assessing the relationship between participant sociodemographic characteristics and normative BREAST-Q scores it was found that some of these variables, in particular age, had a significant influence on scores. This finding highlighted the importance that the population sample in this study were representative of the Australian population across gender, age and geographical variables and therefore overcame limitations associated with the United States sample, which was found to be potentially biased by not being representative of the age distribution of the United States general population. Finally, given that the BREAST-Q is the most widely used patient-reported outcome measure in breast surgery, this study provides a valuable contribution to the literature as it presents normative data from a second, diverse population sample for use in both existing and future clinical or research studies.

The prospective design of the research studies in this thesis overcome the limitations associated with many existing outcome studies which are based on retrospective review. This is particularly important in patient-reported outcome studies evaluating quality of life, as retrospective studies lack baseline data and therefore make the considerable assumption that health-related quality of life is identical in all individuals when they present for surgery. In addition, retrospective studies are inflexible in the collection of clinical outcome data such as surgical complications as these studies are entirely dependent on the standard of medical record keeping or predetermined database fields collected at the time of surgery and are often limited to the early post-operative period. The complication data included in this thesis was recorded prospectively over multiple

post-operative timepoints and is the first study to report the prospective assessment of surgical complications following breast reduction surgery using a standardised grading system. This standardised reporting of surgical complications overcomes comparison difficulties that currently exist in the literature, with the sizeable discrepancies in the reporting of incidences and risk factors for complications likely to be attributable to inconsistencies in the collection and in the classification of complications data. In addition, this body of work included studies with a relatively large sample size to generate adequate power for further subgroup analyses; overcoming limitations associated with many existing prospective outcome studies have been limited by small sample sizes. Finally, another strength of this study was that post-operative outcomes were not limited to a single follow-up timepoint and were not restricted to the early outcomes during the convalescence period following surgery; outcomes in this study were assessed at multiple timepoints over a 12-month period and again in a long-term follow-up study.

The study evaluating the long-term outcomes of breast reduction surgery provides a significant contribution to knowledge with the finding that health-related quality of life remains stable up to 12 years after surgery and does not trend or decline to baseline levels. In addition, this study detailed a comparison of long-term outcome scores to population reference data and demonstrated that the health-related quality of life remained comparable over time. Whilst a small number of existing studies have explored long-term outcomes of surgery, the majority of these have been based on retrospective review and are limited by the lack of baseline data; and are therefore unable to address the important issue as to whether health-related quality of life returns to baseline values over time. In a climate of increasingly tight healthcare budgets, this study provides evidence as to the value in surgery which continues well beyond the short- to medium-term gains previously reported.

Research gaps have remained in this field due to inconsistencies reported in the literature regarding the relationship between participant characteristics and clinical factors and the outcomes of surgery. The surgical study cohort described in this thesis were not biased by restrictive access policies that have been reported in previous studies based on a minimum weight of resection or BMI and therefore includes a broad range across these variables. This was particularly important as it enabled the accurate assessment of these factors as potential predictors of the change in health-related quality of life and outcomes of surgery and therefore overcomes limitations associated with existing studies. Importantly, this study established that the improvement in health-related quality of life following surgery is independent of commonly used criteria including those based on breast size, BMI or a minimum weight of resection; highlighting that the validity of such criteria is poorly supported by evidence, potentially ruling out access to surgery for many women who would greatly benefit. This finding is not only relevant within the Australian public healthcare system where access to surgery is inconsistent and ultimately reliant on state and local policies, some of which include eligibility criteria based on BMI, but is also relevant on a global

scale. This body of work provides strong evidence-base to refute the validity of restrictions enforced by policymakers and healthcare funders in many countries and jurisdictions worldwide.

Finally, existing studies in the literature describing the cost-utility and cost-effectiveness of breast reduction surgery are scarce; and there are currently no studies from the Australian perspective. This thesis details a thorough economic evaluation and provides strong economic evidence that breast reduction surgery is a cost-effective intervention and is therefore justifiable for inclusion within the Australian public healthcare sector.

8.4 Future directions

Within the field of plastic and reconstructive surgery there are further areas where research is warranted to provide high-quality evidence to better inform and support clinical practice and healthcare policy. An example of this is abdominoplasty with surgical repair of the rectus diastasis following pregnancy. Similar to breast reduction, this surgery has commonly been subjected to ongoing restrictions and is often considered a cosmetic procedure rather than a functional operation by many policymakers and healthcare funders. In 2016, the item number for abdominoplasty (for repair of the rectus diastasis) was removed from the Australian Medicare Benefits Schedule (MBS) and is therefore this procedure is currently excluded within the Australian public hospital system. Therefore, there are no MBS-funded treatment options for patients with pregnancy-acquired rectus diastasis and who are nonresponsive to conservative treatments such as physiotherapy, exercise, lifestyle changes and painkillers. Preliminary studies suggest that abdominoplasty in women with pregnancy-acquired rectus diastasis reduces urinary incontinence symptoms, back pain and improves health-related quality of life. However, these existing studies have limitations including the lack of a control group and a risk of bias and unfortunately do not currently provide the high-quality evidence required to provide justification for inclusion within the healthcare system. Following on from the present study of outcomes following breast reduction surgery, conducting a well-designed prospective study using validated patient-reported outcome measures to compare outcomes in a surgical cohort in comparison to an appropriate control group to evaluate both the clinical-effectiveness and cost-effectiveness of abdominoplasty surgery for rectus diastasis following pregnancy within the Australian healthcare setting is therefore warranted. Furthermore, an investigation of the participant and clinical variables such as those included in this thesis and their relationship on the outcomes following surgery would offer a significant contribution. This study would be useful both at a clinical and patient level, but also to inform healthcare funders and policymakers to support the provision of funding and reinstate access to surgery in the future. Finally, the examination and investigation into biases in surgical and health care rationing more generally, and the relationship between degree of need to access healthcare and restrictions preventing that access, would be an important direction for future research studies.

186

8.5 Conclusion

Breast hypertrophy is a chronic health problem and the combination of pain and functional deficits has a significant impact on health-related quality of life in affected women. Breast reduction surgery is an effective intervention for providing symptomatic relief and improving health-related quality of life in Australian women to levels equivalent to that of the general population. The health benefits were found to be sustained over time and remain for many years following surgery. The improvement in health-related quality of life following surgery was comparable to other widely accepted and approved surgical interventions and was found to be independent of traditionally used restrictive criteria based on body mass index or a minimum weight of resection and demonstrates that the health benefits of surgery regardless of these factors. A comprehensive economic evaluation found that surgery is cost-effective and justifiable for inclusion within the Australian healthcare setting.

The collective findings from this body of research provide strong evidence as to the clinical- and cost-effectiveness of bilateral breast reduction surgery for the treatment of symptomatic breast hypertrophy in Australian women. The results from this research provide a contribution to knowledge and it is anticipated that this thesis will be a definitive piece of work to inform clinicians and the wider community, and to guide the Australian healthcare system in offering breast reduction services in the future.

REFERENCES

ABELL, R. G. & VOTE, B. J. 2014. Cost-Effectiveness of Femtosecond Laser-Assisted Cataract Surgery versus Phacoemulsification Cataract Surgery. *Ophthalmology*, 121, 10-16.

ACT GOVERNMENT. 2016. Waiting Time and Elective Surgery Access Policy. Available: <u>https://www.act.gov.au/search?query=access+to+elective+surgery&collection=act-gov&profile=www-act</u> [Accessed January 25, 2018].

- ADAMSON L, B. W., BYLES J, CHOJENTA C, DOBSON A, FITZGERALD D, HOCKEY R, LOXTON D, POWERS J, SPALLEK M, WATERS B & WATSON M 2007. Women's weight: Findings from the Australian Longitudinal Study on Women's Health. Report prepared for the Australian Government Department of Health and Ageing. Canberra, ACT Australia.
- AMERICAN SOCIETY OF ANESTHESIOLOGISTS. 2014. ASA Physical Status Classification System. Available: <u>https://www.asahq.org/standards-and-guidelines/asa-physical-statusclassification-system</u> [Accessed 6 January 2020].
- AMERICAN SOCIETY OF PLASTIC SURGEONS. 2016. 2016 Plastic Surgery Statistics Report. Available: <u>https://www.plasticsurgery.org/documents/News/Statistics/2016/plastic-surgery-statistics-full-report-2016.pdf</u> [Accessed January 8, 2018].
- ANDERSON, R. C., CUNNINGHAM, B., TAFESSE, E. & LENDERKING, W. R. 2006. Validation of the breast evaluation questionnaire for use with breast surgery patients. *Plast Reconstr Surg*, 118, 597-602.
- ANDRADE, A. C., VEIGA, D. F., AGUIAR, I. C., JULIANO, Y., SABINO-NETO, M. & FERREIRA,
 L. M. 2018. Outcomes analysis of breast reduction in Brazilian women using the BREAST-Q(R) questionnaire: a cross-sectional controlled study. *Clinics (Sao Paulo)*, 73, e313.
- ANTONY, Á. K., YEGIYANTS, S. S., DANIELSON, K. K., WÍSEL, S., MORRIS, D., DÓLEZAL, R. F. & COHEN, M. N. 2013. A matched cohort study of superomedial pedicle vertical scar breast reduction (100 breasts) and traditional inferior pedicle Wise-pattern reduction (100 breasts): an outcomes study over 3 years. *Plast Reconstr Surg*, 132, 1068-76.
- APPEL, J. Z., WENDEL, J. J., ZELLNER, E. G., HAGAN, K. F., SHACK, R. B. & CORLEW, D. S. 2010. Association between preoperative measurements and resection weight in patients undergoing reduction mammaplasty. *Ann Plast Surg*, 64, 512-5.
- ARAUJO, C. D., VEIGA, D. F., HOCHMAN, B. S., ABLA, L. E., OLIVEIRA, A. C., NOVO, N. F., VEIGA-FILHO, J. & FERREIRA, L. M. 2014. Cost-utility of reduction mammaplasty assessed for the Brazilian public health system. *Aesthet Surg J*, 34, 1198-204.
- ATTERHEM, H., HOLMNER, S. & JANSON, P. E. 1998. Reduction mammaplasty: symptoms, complications, and late results. A retrospective study on 242 patients. *Scand J Plast Reconstr Surg Hand Surg*, 32, 281-6.
- AUSTRALIAN BUREAU OF STATISTICS 1997. National Health Survey: SF-36 Population Norms, Australia. Canberra: Australian Bureau of Statistics.
- AUSTRALIAN BUREAU OF STATISTICS. 2016a. 2071.0 Census of Population and Housing: Reflecting Australia - Stories from the Census, 2016, 'Snapshot of Australia, datacube: excel spreadsheet, viewed 21 May 2018. Available: http://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/2071.02016?OpenDocument.
- AUSTRALIAN BUREAU OF STATISTICS. 2016b. 6401.0 Consumer Price Index, Australia, Mar 2016. Time Series Spreadsheet, excel spreadsheet. Available: <u>https://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/6401.0Mar%202016?OpenDoc</u> ument [Accessed 10 July 2019].
- AUSTRALIAN BUREAU OF STATISTICS. 2016c. Census of Population and Housing: Socio-Economic Indexes for Areas (SEIFA), Australia. 'Table 2: Postal Area (POA) Index of Relative Socio-economic Disadvantage, SEIFA 2016', Data cube: Excel spreadsheet, cat. no. 2033.0.55.00, viewed 7 May 2018. Available: <u>http://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/2033.0.55.0012016?OpenDocu</u> ment.
- AUSTRALIAN BUREAU OF STATISTICS. 2018. 4364.0.55.001 National Health Survey: First Results, 2017-18 Available:

https://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/4364.0.55.0012017-

18?OpenDocument#Publications.

- AUSTRALIAN INSTITUTE OF HEALTH AND WELFARE. 2008. Elective surgery in Australia: new measures of access. Available: <u>https://www.aihw.gov.au/reports/hospitals/elective-surgery-australia/contents/table-of-contents</u> [Accessed 28 January 2018].
- AUSTRALIAN INSTITUTE OF HEALTH AND WELFARE. 2019a. *Life expectancy and potentially avoidable deaths in 2015–2017.* [Online]. Canberra:AIHW. Available: <u>https://www.aihw.gov.au/reports/life-expectancy-deaths/life-expectancy-avoidable-deaths-2015-2017/contents/summary [Accessed 05 March 2020].</u>
- AUSTRALIAN INSTITUTE OF HEALTH AND WELFARE. 2019b. Procedures data cubes. Available: <u>https://www.aihw.gov.au/reports/hospitals/procedures-data-cubes/contents/data-cubes</u> [Accessed 6 May 2019].
- AUSTRALIAN LONGITUDINAL STUDY ON WOMEN'S HEALTH. 2005. Australian women and their weight - a growing problem. Available: <u>https://www.alswh.org.au/images/content/pdf/achievement_reports/achievements-</u> weight.pdf [Accessed 3 February 2020].
- AUSTRALIAN LONGITUDINAL STUDY ON WOMEN'S HEALTH. 2018. The obesity epidemic in Australia. Available: <u>http://www.alswh.org.au/images/content/pdf/submissions/Obesity-</u> <u>Inquiry-Submission-Australian-Longitudinal-Study-on-Womens-Health_2018.pdf</u> [Accessed 3 February 2020].
- AUSTRALIAN LONGITUDINAL STUDY ON WOMEN'S HEALTH. 2019. Weight and Weight Gain Policy Brief. Available:

https://www.alswh.org.au/images/content/pdf/major_reports/Weight%20and%20Weight%20 Gain%20Policy%20Brief%20_ALSWH%202019.pdf [Accessed 3 February 2020].

- BADHIWALA, J. H., WITIW, C. D., NASSIRI, F., AKBAR, M. A., JAJA, B., WILSON, J. R. & FEHLINGS, M. G. 2018. Minimum Clinically Important Difference in SF-36 Scores for Use in Degenerative Cervical Myelopathy. Spine (Phila Pa 1976), 43, E1260-E1266.
- BAI, J., ROSEN, C. M., NGAAGE, L. M., MCNICHOLS, C. H. L., DIACONU, S. C., IHENATU, C., ELEGBEDE, A., BELLAVANCE, E. C., SLEZAK, S. S. & RASKO, Y. M. 2019. Longevity of Outcomes Following Reduction Mammoplasty. *Eplasty*, 19, e18.
- BALDWIN, C. J., KELLY, E. J. & BATCHELOR, A. G. 2010. The variation in breast density and its relationship to delayed wound healing: a prospective study of 40 reduction mammoplasties. *J Plast Reconstr Aesthet Surg*, 63, 663-5.
- BALTODANO, P. A., REINHARDT, M. E., ATA, A., SIMJEE, U. F., ROTH, M. Z. & PATEL, A. 2017. The Baltodano Breast Reduction Score: A Nationwide, Multi-Institutional, Validated Approach to Reducing Surgical-Site Morbidity. *Plast Reconstr Surg*, 140, 258e-264e.
- BARCHA, C. P. M. S. P. H. & RANZER, M. J. M. D. 2018. Impact of Chronic Steroid Use on Plastic Surgery Outcomes: Analysis of 94,140 Cases. *Plastic & Reconstructive Surgery*, 142, 770e-779e.
- BARTSCH, R. H., WEISS, G., KÄSTENBAUER, T., PATOCKA, K., DEUTINGER, M., KRAPOHL,
 B. D. & BENDITTE-KLEPETKO, H. C. 2007. Crucial aspects of smoking in wound healing after breast reduction surgery. *J Plast Reconstr Aesthet Surg*, 60, 1045-9.
- BAUERMEISTER, A. J., GILL, K., ZURIARRAIN, A., EARLE, S. A. & NEWMAN, M. I. 2019. "Reduction mammaplasty with superomedial pedicle technique: A literature review and retrospective analysis of 938 consecutive breast reductions". *J Plast Reconstr Aesthet Surg*, 72, 410-418.
- BAYRAMICLI, M., SIRINOGLU, H. & YALCIN, D. 2017. Outcome After Breast Reduction Considering Body Mass Index and Resection Amount. *Aesthet Surg J*, 37, 1103-1110.
- BEHMAND, R. A., TANG, D. H. & SMITH, D. J. 2000. Outcomes in breast reduction surgery. Ann Plast Surg, 45, 575-80.
- BENDITTE-KLEPETKO, H., LEISSER, V., PATERNOSTRO-SLUGA, T., RAKOS, M., TRATTNIG, S., HELBICH, T., SCHEMPER, M. & DEUTINGER, M. 2007. Hypertrophy of the breast: a problem of beauty or health? *J Womens Health (Larchmt)*, 16, 1062-9.
- BERALDO, F. N., VEIGA, D. F., VEIGA-FILHO, J., GARCIA, E. S., VILAS-BÔAS, G. S., JULIANO, Y., SABINO-NETO, M. & FERREIRA, L. M. 2016. Sexual Function and Depression Outcomes Among Breast Hypertrophy Patients Undergoing Reduction Mammaplasty: A Randomized Controlled Trial. *Ann Plast Surg*, 76, 379-82.
- BERBEROGLU, O., TEMEL, M. & TURKMEN, A. 2015. Effects of Reduction Mammaplasty

Operations on the Spinal Column: Clinical and Radiological Response. *Aesthetic Plast Surg*, 39, 514-22.

- BIKHCHANDANI, J., VARMA, S. K. & HENDERSON, H. P. 2007. Is it justified to refuse breast reduction to smokers? *J Plast Reconstr Aesthet Surg*, 60, 1050-4.
- BLACK, W. C. 1990. The CE plane: a graphic representation of cost-effectiveness. *Med Decis Making*, 10, 212-4.
- BLAND, J. M. & ALTMAN, D. G. 1995. Comparing methods of measurement: why plotting difference against standard method is misleading. *Lancet*, 346, 1085-7.
- BLOMQVIST, L. 1996. Reduction mammaplasty: analysis of patients' weight, resection weights, and late complications. *Scand J Plast Reconstr Surg Hand Surg*, 30, 207-10.
- BLOMQVIST, L. & BRANDBERG, Y. 2004. Three-year follow-up on clinical symptoms and healthrelated quality of life after reduction mammaplasty. *Plast Reconstr Surg*, 114, 49-54.
- BLOMQVIST, L., ERIKSSON, A. & BRANDBERG, Y. 2000. Reduction mammaplasty provides long-term improvement in health status and quality of life. *Plast Reconstr Surg*, 106, 991-7.
- BOSCHERT, M. T., BARONE, C. M. & PUCKETT, C. L. 1996. Outcome analysis of reduction mammaplasty. *Plast Reconstr Surg*, 98, 451-4.
- BOUKOVALAS, S., PADILLA, P. L., SPRATT, H., TRAN, J. P., LI, R. T., BOSON, A. L., HOWLAND, N. & PHILLIPS, L. G. 2019. Redefining the Role of Resection Weight Prediction in Reduction Mammaplasty and Breaking the "One-Scale-Fits-All" Paradigm. *Plast Reconstr Surg*, 144, 18e-27e.
- BOUMAN, F. G. 1970. Volumetric measurement of the human breast and breast tissue before and during mammaplasty. *Br J Plast Surg*, 23, 263-4.
- BOZOLA, A. R. 1990. Breast reduction with short L scar. Plast Reconstr Surg, 85, 728-38.
- BRAIG, D., EISENHARDT, S. U., STARK, G. B. & PENNA, V. 2016. Impact of increasing age on breast reduction surgery: A single centre analysis. J Plast Reconstr Aesthet Surg, 69, 482-6.
- BRAZIER, J., RATCLIFFE, J., SALOMAN, J. & TSUCHIYA, A. 2017. *Measuring and valuing health benefits for economic evaluation*, OXFORD university press.
- BRAZIER, J., ROBERTS, J. & DEVERILL, M. 2002. The estimation of a preference-based measure of health from the SF-36. *J Health Econ*, 21, 271-92.
- BREUNING, E. E., OIKONOMOU, D., SINGH, P., RAI, J. K. & MENDONCA, D. A. 2010. Cosmetic surgery in the NHS: Applying local and national guidelines. *J Plast Reconstr Aesthet Surg*, 63, 1437-42.
- BROWN, J. R., HOLTON, L. H., 3RD, CHUNG, T. L. & SLEZAK, S. 2008. Breast-feeding, selfexam, and exercise practices before and after reduction mammoplasty. *Ann Plast Surg*, 61, 375-9.
- BROWN, N., WHITE, J., MILLIGAN, A., RISIUS, D., AYRES, B., HEDGER, W. & SCURR, J. 2012. The relationship between breast size and anthropometric characteristics. *Am J Hum Biol*, 24, 158-64.
- BROWN, T. A., CASH, T. F. & MIKULKA, P. J. 1990. Attitudinal body-image assessment: factor analysis of the Body-Self Relations Questionnaire. *J Pers Assess*, 55, 135-44.
- BULSTRODE, N., BELLAMY, E. & SHROTRIA, S. 2001. Breast volume assessment: comparing five different techniques. *Breast*, 10, 117-23.
- CABRAL, I. V., DA SILVA GARCIA, E., SOBRINHO, R. N., PINTO, N. L. L., JULIANO, Y., VEIGA-FILHO, J., FERREIRA, L. M. & VEIGA, D. F. 2017a. Use of the BREAST-Q Survey in the Prospective Evaluation of Reduction Mammaplasty Outcomes. *Aesthetic Plast Surg*.
- CABRAL, I. V., GARCIA, E. D., SOBRINHO, R. N., PINTO, N. L., JULIANO, Y., VEIGA-FILHO, J., FERREIRA, L. M. & VEIGA, D. F. 2017b. Increased Capacity for Work and Productivity After Breast Reduction. *Aesthetic surgery journal*, 37, 57-62.
- CANO, S. J., KLASSEN, A. & PUSIC, A. L. 2009. The science behind quality-of-life measurement: a primer for plastic surgeons. *Plast Reconstr Surg*, 123, 98e-106e.
- CANO, S. J., KLASSEN, A. F., SCOTT, A., ALDERMAN, A. & PUSIC, A. L. 2014. Interpreting clinical differences in BREAST-Q scores: minimal important difference. *Plast Reconstr Surg*, 134, 173e-175e.
- CARTY, M. J., DUCLOS, A., GU, X., ELELE, N. & ORGILL, D. 2012. Patient satisfaction and surgeon experience: a follow-up to the reduction mammaplasty learning curve study. *Eplasty*, 12, e22.

- CASH, T. F. 2000. The Multidimensional Body-Self Relations Questionnaire users' manual. Available: <u>http://www.body-images.com/assessments/mbsrq.html</u>.
- CASH, T. F. & PRUZINSKY, T. E. 1990. Body images: Development, deviance, and change, Guilford Press.
- CEROVAC, S., ALI, F. S., BLIZARD, R., LLOYD, G. & BUTLER, P. E. 2005. Psychosexual function in women who have undergone reduction mammaplasty. *Plast Reconstr Surg*, 116, 1306-13.
- CHADBOURNE, E. B., ZHANG, S., GORDON, M. J., RO, E. Y., ROSS, S. D., SCHNUR, P. L. & SCHNEIDER-REDDEN, P. R. 2001. Clinical outcomes in reduction mammaplasty: a systematic review and meta-analysis of published studies. *Mayo Clin Proc*, 76, 503-10.
- CHAHRAOUI, K., DANINO, A., BENONY, H., FRACHEBOIS, C., CLERC, A. S. & MALKA, G. 2006. Anxiety and subjective quality of life preoperatively and 4 months after reduction mammaplasty. *J Psychosom Res*, 61, 801-6.
- CHALKIDOU, K., TUNIS, S., LOPERT, R., ROCHAIX, L., SAWICKI, P. T., NASSER, M. & XERRI, B. 2009. Comparative Effectiveness Research and Evidence-Based Health Policy: Experience from Four Countries. *The Milbank Quarterly*, 87, 339-367.
- CHAN, L. K., WITHEY, S. & BUTLER, P. E. 2006. Smoking and wound healing problems in reduction mammaplasty: is the introduction of urine nicotine testing justified? *Ann Plast Surg*, 56, 111-5.
- CHAN, M., LONIE, S., MACKAY, S. & MACGILL, K. 2019. Reduction Mammaplasty: What Cup Size Will I Be? *Plastic and Reconstructive Surgery – Global Open,* 7, e2273.
- CHANG, D. W., REECE, G. P., WANG, B., ROBB, G. L., MILLER, M. J., EVANS, G. R., LANGSTEIN, H. N. & KROLL, S. S. 2000. Effect of smoking on complications in patients undergoing free TRAM flap breast reconstruction. *Plast Reconstr Surg*, 105, 2374-80.
- CHAO, J. D., MEMMEL, H. C., REDDING, J. F., EGAN, L., ODOM, L. C. & CASAS, L. A. 2002. Reduction mammaplasty is a functional operation, improving quality of life in symptomatic women: a prospective, single-center breast reduction outcome study. *Plast Reconstr Surg*, 110, 1644-52; discussion 1653-4.
- CHEN, C. L., SHORE, A. D., JOHNS, R., CLARK, J. M., MANAHAN, M. & MAKARY, M. A. 2011. The impact of obesity on breast surgery complications. *Plast Reconstr Surg*, 128, 395e-402e.
- CHOPPIN, S. B., WHEAT, J. S., GEE, M. & GOYAL, A. 2016. The accuracy of breast volume measurement methods: A systematic review. *Breast*, 28, 121-9.
- CHUN, Y. S., SCHWARTZ, M. A., GU, X., LIPSITZ, S. R. & CARTY, M. J. 2012. Body mass index as a predictor of postoperative complications in reduction mammaplasty. *Plast Reconstr Surg*, 129, 228e-233e.
- CLAVIEN, P. A., BARKUN, J., DE OLIVEIRA, M. L., VAUTHEY, J. N., DINDO, D., SCHULICK, R.
 D., DE SANTIBANES, E., PEKOLJ, J., SLANKAMENAC, K., BASSI, C., GRAF, R.,
 VONLANTHEN, R., PADBURY, R., CAMERON, J. L. & MAKUUCHI, M. 2009. The Clavien-Dindo classification of surgical complications: five-year experience. *Ann Surg*, 250, 187-96.
- COGLIANDRO, A., BARONE, M., CASSOTTA, G., TENNA, S., CAGLI, B. & PERSICHETTI, P. 2017a. Patient Satisfaction and Clinical Outcomes Following 414 Breast Reductions: Application of BREAST-Q. *Aesthetic Plast Surg*.
- COGLIANDRO, A., BARONE, M. & PERSICHETTI, P. 2017b. Prospective Analysis of Satisfaction by Means of BREASTQ in 156 Patients Who Underwent Breast Reduction. *Plast Reconstr Surg*, 140, 752e-753e.
- COHEN, J. 1988. *Statistical power analysis for the behavioral sciences,* Hillsdale, N.J., Hillsdale, N.J., L. Erlbaum Associates.
- COHEN, W. A., HOMEL, P. & PATEL, N. P. 2016a. Does Time Affect Patient Satisfaction and Health-Related Quality of Life After Reduction Mammoplasty? *Eplasty*, 16, e7.
- COHEN, W. A., MUNDY, L. R., BALLARD, T. N., KLASSEN, A., CANO, S. J., BROWNE, J. & PUSIC, A. L. 2016b. The BREAST-Q in surgical research: A review of the literature 2009-2015. *J Plast Reconstr Aesthet Surg*, 69, 149-62.
- COLE, R. P. & SHAKESPEARE, V. 1998. Measuring patient-based outcomes in a plastic surgery service: breast reduction surgical patients. *Br J Plast Surg*, 51, 79-80.
- COLLINS, E. D., KERRIGAN, C. L., KIM, M., LOWERY, J. C., STRIPLIN, D. T., CUNNINGHAM, B. & WILKINS, E. G. 2002. The effectiveness of surgical and nonsurgical interventions in

relieving the symptoms of macromastia. *Plast Reconstr Surg*, 109, 1556-66.

- COLTMAN, C. E., STEELE, J. R. & MCGHEE, D. E. 2019a. Does breast size affect how women participate in physical activity? *J Sci Med Sport*, 22, 324-329.
- COLTMAN, C. E., STEELE, J. R. & MCGHEE, D. E. 2019b. Effect of Breast Size on Upper Torso Musculoskeletal Structure and Function: A Cross-Sectional Study. *Plast Reconstr Surg*, 143, 686-695.
- CORIDDI, M., NADEAU, M., TAGHIZADEH, M. & TAYLOR, A. 2013. Analysis of satisfaction and well-being following breast reduction using a validated survey instrument: the BREAST-Q. *Plast Reconstr Surg*, 132, 285-90.
- CORREA, M. P. D., DORNELAS, M. T., DE CARVALHO, E. N., BARRA, A., VENTURELLI, E. P., JR., CORREA, L. D. & CHAOUBAH, A. 2018. Assessment of quality of life in patients who underwent breast reduction using BREAST-Q. *J Plast Reconstr Aesthet Surg*, 71, 929-931.
- CORRIVEAU, S. & JACOBS, J. S. 1990. Macromastia in adolescence. *Clin Plast Surg*, 17, 151-60. CRITTENDEN, T., VEITCH, D. E., HENNEBERG, M., BURFORD, K., VAN ESSEN, P., DEUT, K., ZEITOUNEH, K., LOMAX, E., GRIFFIN, P. A. & DEAN, N. R. 2018a. Measuring breast volume in hypertrophy: laser scanning or water displacement? *Australasian Journal of*
 - Plastic Surgery, 1, 33-40.
- CRITTENDEN, T., WATSON, D., RATCLIFFE, J. & DEAN, N. 2018b. Normative data for the BREAST-Q Reduction module in an Australian population. *Qual. Life Res.*, 27, S56-S57.
- CRITTENDEN, T., WATSON, D., RATCLIFFE, J. & DEAN, N. 2019a. Breast reduction surgery improves health-related quality of life in women with breast hypertrophy: a comparison to other surgical interventions. *Quality of Life Research*, 28, S115-S115.
- CRITTENDEN, T., WATSON, D. I., RATCLIFFE, J., GRIFFIN, P. A., DEAN, N. R. & AFESA RESEARCH GROUP 2020. Does breast reduction surgery improve health-related quality of life? A prospective cohort study in Australian women. *BMJ Open*, 10, e031804.
- CRITTENDEN, T. A., WATSON, D. I., RATCLIFFE, J., GRIFFIN, P. A. & DEAN, N. R. 2019b. Outcomes of Breast Reduction Surgery Using the BREAST-Q: A Prospective Study and Comparison with Normative Data. *Plast Reconstr Surg*, 144, 1034-1044.
- CROSBY, R. D., KOLOTKIN, R. L. & WILLIAMS, G. R. 2003. Defining clinically meaningful change in health-related quality of life. *Journal of clinical epidemiology*, 56, 395-407.
- CRUZ-KORCHIN, N., KORCHIN, L., GONZALEZ-KEELAN, C., CLIMENT, C. & MORALES, I. 2002. Macromastia: how much of it is fat? *Plast Reconstr Surg*, 109, 64-8.
- CUNNINGHAM, B. L., GEAR, A. J., KERRIGAN, C. L. & COLLINS, E. D. 2005. Analysis of breast reduction complications derived from the BRAVO study. *Plast Reconstr Surg*, 115, 1597-604.
- CURTIS, A. J., RUSSELL, C. O. H., STOELWINDER, J. U. & MCNEIL, J. J. 2010. Waiting lists and elective surgery: ordering the queue. *The Medical journal of Australia*, 192, 217.
- DABBAH, A., LEHMAN, J. A., PARKER, M. G., TANTRI, D. & WAGNER, D. S. 1995. Reduction mammaplasty: an outcome analysis. *Ann Plast Surg*, 35, 337-41.
- DAKIN, H., DEVLIN, N., FENG, Y., RICE, N., O'NEILL, P. & PARKIN, D. 2015. The influence of cost-effectiveness and other factors on nice decisions. *Health Economics (United Kingdom)*, 24, 1256-1271.
- DAKIN, H., GRAY, A., FITZPATRICK, R., MACLENNAN, G., MURRAY, D. & THE, K. A. T. T. G. 2012. Rationing of total knee replacement: a cost-effectiveness analysis on a large trial data set. *BMJ Open*, *2*, e000332.
- DANCEY, A., KHAN, M., DAWSON, J. & PEART, F. 2008. Gigantomastia a classification and review of the literature. *Journal of Plastic, Reconstructive & Aesthetic Surgery,* 61, 493-502.
- DAVIS, G. M., RINGLER, S. L., SHORT, K., SHERRICK, D. & BENGTSON, B. P. 1995. Reduction mammaplasty: long-term efficacy, morbidity, and patient satisfaction. *Plast Reconstr Surg*, 96, 1106-10.
- DERBY, B. M., GROTTING, J. C. & REDDEN, D. T. 2016. Vertical Sculpted Pillar Reduction Mammaplasty in 317 Patients: Technique, Complications, and BREAST-Q Outcomes. *Aesthet Surg J*, 36, 417-30.
- DESCAMPS, M. J., LANDAU, A. G., LAZARUS, D. & HUDSON, D. A. 2008. A formula determining resection weights for reduction mammaplasty. *Plast Reconstr Surg*, 121, 397-400.
- DEVLIN, N., & APPLEBY, J. 2010. Getting the most out of PROMS. Putting health outcomes at the heart of NHS decision making, London, The King's Fund.

- DEVLIN, N. & PARKIN, D. 2004. Does NICE have a cost-effectiveness threshold and what other factors influence its decisions? A binary choice analysis. *Health Econ*, 13, 437-52.
- DEX, E. A., ASPLUND, O., ARDEHALI, B. & ECCLES, S. J. 2008. A method to select patients for vertical scar or inverted-T pattern breast reduction. *J Plast Reconstr Aesthet Surg*, 61, 1294-302.
- DINDO, D., DEMARTINES, N. & CLAVIEN, P. A. 2004. Classification of surgical complications: a new proposal with evaluation in a cohort of 6336 patients and results of a survey. *Ann Surg*, 240, 205-13.
- DRUMMOND, M. 2001. Introducing economic and quality of life measurements into clinical studies. *Ann Med*, 33, 344-9.
- DRUMMOND, M. F. 2005. *Methods for the economic evaluation of health care programmes,* Oxford, Oxford : Oxford University Press.
- DUCIC, I., IORIO, M. L. & AL-ATTAR, A. 2010. Chronic headaches/migraines: extending indications for breast reduction. *Plast Reconstr Surg*, 125, 44-9.
- DVORACEK, L. A., GUSENOFF, J. A., PETER RUBIN, J. & MANDERS, E. K. 2019. Quick Calculation of Breast Resection Mass Using the Schnur Scale. *Ann Plast Surg*, 82, 316-319.
- EDER, M., GRABHORN, A., WALDENFELS, F., SCHUSTER, T., PAPADOPULOS, N. A., MACHENS, H. G. & KOVACS, L. 2013. Prediction of breast resection weight in reduction mammaplasty based on 3-dimensional surface imaging. *Surg Innov,* 20, 356-64.
- EDER, M., PAPADOPULOS, N. A. & KOVACS, L. 2007. Breast volume determination in breast hypertrophy. *Plast Reconstr Surg*, 120, 356-7.
- EDNEY, L. C., HAJI ALI AFZALI, H., CHENG, T. C. & KARNON, J. 2018. Estimating the Reference Incremental Cost-Effectiveness Ratio for the Australian Health System. *Pharmacoeconomics*, 36, 239-252.
- EDWARDS, P., ROBERTS, I., CLARKE, M., DIGUISEPPI, C., PRATAP, S., WENTZ, R. & KWAN, I. 2002. Increasing response rates to postal questionnaires: systematic review. *BMJ*, 324, 1183.
- EGGERT, E., SCHUSS, R. & EDSANDER-NORD, A. 2009. Clinical outcome, quality of life, patients' satisfaction, and aesthetic results, after reduction mammaplasty. *Scand J Plast Reconstr Surg Hand Surg*, 43, 201-6.
- EVANS, J. G. 1997. The rationing debate: Rationing health care by age: The case against. *BMJ*, 314, 822.
- FARIA, F. S., GUTHRIE, E., BRADBURY, E. & BRAIN, A. N. 1999. Psychosocial outcome and patient satisfaction following breast reduction surgery. *Br J Plast Surg*, 52, 448-52.
- FAULCONBRIDGE, L. F., WADDEN, T. A., THOMAS, J. G., JONES-CORNEILLE, L. R., SARWER, D. B. & FABRICATORE, A. N. 2013. Changes in depression and quality of life in obese individuals with binge eating disorder: bariatric surgery versus lifestyle modification. *Surg Obes Relat Dis*, 9, 790-6.
- FENWICK, E., CLAXTON, K. & SCULPHER, M. 2001. Representing uncertainty: the role of costeffectiveness acceptability curves. *Health Economics*, 10, 779-787.
- FINDIKCIOGLU, K., FINDIKCIOGLU, F., BULAM, H., SEZGIN, B. & OZMEN, S. 2013. The impact of breast reduction surgery on the vertebral column. *Ann Plast Surg*, 70, 639-42.
- FISCHER, J. P., CLEVELAND, E. C., SHANG, E. K., NELSON, J. A. & SERLETTI, J. M. 2014. Complications following reduction mammaplasty: a review of 3538 cases from the 2005-2010 NSQIP data sets. *Aesthet Surg J*, 34, 66-73.
- FOREMAN, K. B., DIBBLE, L. E., DROGE, J., CARSON, R. & ROCKWELL, W. B. 2009. The impact of breast reduction surgery on low-back compressive forces and function in individuals with macromastia. *Plast Reconstr Surg*, 124, 1393-9.
- FOTEFF, C., KENNEDY, S., MILTON, A. H., DEGER, M., PAYK, F. & SANDERSON, G. 2016. Cost–Utility Analysis of Cochlear Implantation in Australian Adults. *Otology & Neurotology*, 37, 454-461.
- FREIRE, M., NETO, M. S., GARCIA, E. B., QUARESMA, M. R. & FERREIRA, L. M. 2004. Quality of life after reduction mammaplasty. *Scand J Plast Reconstr Surg Hand Surg*, 38, 335-9.
- FREIRE, M., NETO, M. S., GARCIA, E. B., QUARESMA, M. R. & FERREIRA, L. M. 2007. Functional capacity and postural pain outcomes after reduction mammaplasty. *Plast Reconstr Surg*, 119, 1149-56; discussion 1157-8.

- FRENDL, D. M. & WARE, J. E., JR. 2014. Patient-reported functional health and well-being outcomes with drug therapy: a systematic review of randomized trials using the SF-36 health survey. *Med Care*, 52, 439-45.
- FREY, J. D., KOLTZ, P. F., BELL, D. E. & LANGSTEIN, H. N. 2014. The complex insurance reimbursement landscape in reduction mammaplasty: how does the American plastic surgeon navigate it? *Ann Plast Surg*, 72, 23-9.
- FRIELING, M. A., DAVIS, W. R. & CHIANG, G. 2013. The SF-36v2 and SF-12v2 health surveys in New Zealand: norms, scoring coefficients and cross-country comparisons. *Aust N Z J Public Health*, 37, 24-31.
- FUZESI, S., CANO, S. J., KLASSEN, A. F., ATISHA, D. & PUSIC, A. L. 2017. Validation of the electronic version of the BREAST-Q in the army of women study. *The Breast*, 33, 44-49.
- GAMBOA-BOBADILLA, G. M. & KILLINGSWORTH, C. 2007. Large-volume reduction mammaplasty: the effect of body mass index on postoperative complications. *Ann Plast Surg*, 58, 246-9.
- GANDEK, B., WARE, J. E., JR., AARONSON, N. K., ALONSO, J., APOLONE, G., BJORNER, J., BRAZIER, J., BULLINGER, M., FUKUHARA, S., KAASA, S., LEPLEGE, A. & SULLIVAN, M. 1998. Tests of data quality, scaling assumptions, and reliability of the SF-36 in eleven countries: results from the IQOLA Project. International Quality of Life Assessment. *J Clin Epidemiol,* 51, 1149-58.
- GAWANDE, A. 2015. *Being Mortal : Illness, Medicine, and What Matters in the End,* London, Profile Books Ltd.
- GEORGE, B., HARRIS, A. & MITCHELL, A. 2001. Cost-effectiveness analysis and the consistency of decision making: evidence from pharmaceutical reimbursement in australia (1991 to 1996). *Pharmacoeconomics*, 19, 1103-9.
- GODWIN, Y., BARRON, E. J., EDMUNDS, M. C., MEYER, M., BARDSLEY, A., LOGAN, A. M., O'NEILL, T. J. & WOOD, S. H. 2014. A comparison of the patient and surgeon opinion on the long-term aesthetic outcome of reduction mammaplasty: have we improved over 15 years? J Plast Reconstr Aesthet Surg, 67, 932-8.
- GODWIN, Y., WOOD, S. H. & O'NEILL, T. J. 1998. A comparison of the patient and surgeon opinion on the long-term aesthetic outcome of reduction mammaplasty. *Br J Plast Surg*, 51, 444-9.
- GOES, J. C. 1996. Periareolar mammaplasty: double skin technique with application of polyglactine or mixed mesh. *Plast Reconstr Surg*, 97, 959-68.
- GONZALEZ, F., WALTON, R. L., SHAFER, B., MATORY, W. E., JR. & BORAH, G. L. 1993. Reduction mammaplasty improves symptoms of macromastia. *Plast Reconstr Surg*, 91, 1270-6.
- GONZALEZ, M. A., GLICKMAN, L. T., ALADEGBAMI, B. & SIMPSON, R. L. 2012. Quality of life after breast reduction surgery: a 10-year retrospective analysis using the Breast Q questionnaire: does breast size matter? *Ann Plast Surg*, 69, 361-3.
- GOVERNMENT OF SOUTH AUSTRALIA SA HEALTH. 2018. *Restricted Elective Surgery Policy Directive*. [Online]. Available: <u>https://www.sahealth.sa.gov.au/wps/wcm/connect/3d568a77-a030-437a-957e-</u> 9b60bb143341/Directive Restricted+Elective+Surgery 03072018.pdf?MOD=AJPERES&C

<u>ACHEID=ROOTWORKSPACE-3d568a77-a030-437a-957e-9b60bb143341-mhygh9U</u> [Accessed September 12, 2018].

- GOVERNMENT OF WESTERN AUSTRALIA. 2017. *Elective Surgery Access and Waiting List Management Policy.* [Online]. Available: <u>https://ww2.health.wa.gov.au/About-us/Policy-frameworks/Clinical-Services-Planning-and-Programs/Mandatory-requirements/Elective-Services/Elective-Surgery-Access-and-Waiting-List-Management-Policy</u> [Accessed January 25, 2018].
- GRAVANTE, G., ARACO, A., SORGE, R., CARUSO, R., NICOLI, F., ARACO, F., DELOGU, D. & CERVELLI, V. 2008. Postoperative wound infections after breast reductions: the role of smoking and the amount of tissue removed. *Aesthetic Plast Surg*, 32, 25-31.
- GRECO, R. & NOONE, B. 2017. Evidence-Based Medicine: Reduction Mammaplasty. *Plast Reconstr Surg*, 139, 230e-239e.
- GUEMES, A., PEREZ, E., SOUSA, R., GIL, I., VALCARRERES, M. P., CARRERA, P., GRACIA, M. D., ARTIGAS, L. & GASCON, A. 2016. Quality of Life and Alleviation of Symptoms After

Breast Reduction for Macromastia in Obese Patients: Is Surgery Worth It? *Aesthetic Plast Surg*, 40, 62-70.

- GULCELIK, M. A., DOGAN, L., CAMLIBEL, M., KARAMAN, N., KURU, B., ALAGOL, H. & OZASLAN, C. 2011. Early complications of a reduction mammoplasty technique in the treatment of macromastia with or without breast cancer. *Clin Breast Cancer*, 11, 395-9.
- GURUNLUOGLU, R., SACAK, B. & ARTON, J. 2013. Outcomes analysis of patients undergoing autoaugmentation after breast implant removal. *Plast Reconstr Surg*, 132, 304-15.
- GUST, M. J., SMETONA, J. T., PERSING, J. S., HANWRIGHT, P. J., FINE, N. A. & KIM, J. Y. 2013. The impact of body mass index on reduction mammaplasty: a multicenter analysis of 2492 patients. *Aesthet Surg J*, 33, 1140-7.
- GUTHRIE, E., BRADBURY, E., DAVENPORT, P. & SOUZA FARIA, F. 1998. Psychosocial status of women requesting breast reduction surgery as compared with a control group of large-breasted women. *J Psychosom Res*, 45, 331-9.
- GUYATT, G. H., OSOBA, D., WU, A. W., WYRWICH, K. W., NORMAN, G. R. & CLINICAL SIGNIFICANCE CONSENSUS MEETING, G. 2002. Methods to explain the clinical significance of health status measures. *Mayo Clin Proc*, 77, 371-83.
- HALL-FINDLAY, E. J. 2004. Vertical breast reduction. Semin Plast Surg, 18, 211-24.
- HAMMOND, D. C. 1999. Short scar periareolar inferior pedicle reduction (SPAIR) mammaplasty. *Plast Reconstr Surg,* 103, 890-901; discussion 902.
- HANSSON, E., MANJER, J., BORREN, J., LEVIN, M., MULDER, L. & RINGBERG, A. 2014. A feasible computer-based evaluation tool for reduction mammaplasty patients: indications for operation and monitoring of guidelines. *J Plast Reconstr Aesthet Surg*, 67, 927-31.
- HANWRIGHT, P. J., HIRSCH, E. M., SETH, A. K., CHOW, G., SMETONA, J., MCNICHOLS, C., GAIDO, J. A., FINE, N. A., BILIMORIA, K. Y. & KIM, J. Y. 2013. A multi-institutional perspective of complication rates for elective nonreconstructive breast surgery: an analysis of NSQIP data from 2006 to 2010. *Aesthet Surg J*, 33, 378-86.
- HARRIS, A. H., HILL, S. R., CHIN, G., LI, J. J. & WALKOM, E. 2008. The role of value for money in public insurance coverage decisions for drugs in Australia: a retrospective analysis 1994-2004. *Med Decis Making*, 28, 713-22.
- HAWTHORNE, G., OSBORNE, R. H., TAYLOR, A. & SANSONI, J. 2007. The SF36 Version 2: critical analyses of population weights, scoring algorithms and population norms. *Qual Life Res*, 16, 661-73.
- HENRY, D. A., HILL, S. R. & HARRIS, A. 2005. Drug prices and value for money: the Australian Pharmaceutical Benefits Scheme. *JAMA*, 294, 2630-2.
- HENRY, S. L., CRAWFORD, J. L. & PUCKETT, C. L. 2009. Risk factors and complications in reduction mammaplasty: novel associations and preoperative assessment. *Plast Reconstr Surg*, 124, 1040-6.
- HENSELER, H., KHAMBAY, B. S., BOWMAN, A., SMITH, J., PAUL SIEBERT, J., OEHLER, S., JU, X., AYOUB, A. & RAY, A. K. 2011. Investigation into accuracy and reproducibility of a 3D breast imaging system using multiple stereo cameras. *J Plast Reconstr Aesthet Surg*, 64, 577-82.
- HERMANS, B. J., BOECKX, W. D., DE LORENZI, F. & VAN DER HULST, R. R. 2005. Quality of life after breast reduction. *Ann Plast Surg*, 55, 227-31.
- HERNANZ, F., FIDALGO, M., MUNOZ, P., NORIEGA, M. G. & GOMEZ-FLEITAS, M. 2016. Impact of reduction mammoplasty on the quality of life of obese patients suffering from symptomatic macromastia: A descriptive cohort study. *J Plast Reconstr Aesthet Surg*, 69, e168-73.
- HERNANZ, F., MUÑOZ, P., FIDALGO, M. & GÓMEZ-FLEITAS, M. 2014. Breast reduction surgery—an easy formula to estimate the resection weight to be removed. *European Journal of Plastic Surgery*, 37, 373-380.
- HILLAM, J. S., BORSTING, E. A., CHIM, J. H. & THALLER, S. R. 2017. Smoking as a risk factor for breast reduction: An analysis of 13,503 cases. *J Plast Reconstr Aesthet Surg*.
- HJERMSTAD, M. J., FAYERS, P. M., BJORDAL, K. & KAASA, S. 1998. Using reference data on quality of life--the importance of adjusting for age and gender, exemplified by the EORTC QLQ-C30 (+3). *Eur J Cancer*, 34, 1381-9.
- HOPMAN, W. M., TOWHEED, T., ANASTASSIADES, T., TENENHOUSE, A., POLIQUIN, S., BERGER, C., JOSEPH, L., BROWN, J. P., MURRAY, T. M., ADACHI, J. D., HANLEY, D.
A. & PAPADIMITROPOULOS, E. 2000. Canadian normative data for the SF-36 health survey. Canadian Multicentre Osteoporosis Study Research Group. *CMAJ*, 163, 265-71.

- HOWES, B. H., WATSON, D. I., FOSH, B., YIP, J. M., KLEINIG, P. & DEAN, N. R. 2017. Magnetic Resonance Imaging Versus 3-Dimensional Laser Scanning for Breast Volume Assessment After Breast Reconstruction. *Ann Plast Surg*, 78, 455-459.
- HUNTER-SMITH, D. J., SMOLL, N. R., MARNE, B., MAUNG, H. & FINDLAY, M. W. 2012. Comparing breast-reduction techniques: time-to-event analysis and recommendations. *Aesthetic Plast Surg*, 36, 600-6.
- IACOBUCCI, G. 2018. NHS proposes to stop funding 17 "unnecessary" procedures. *BMJ*, 362, k2903.
- IHPA 2018. National Hospital Cost Data Collection, Public Hospitals Cost Report, Round 20 (Financial year 2015-16), Appendix (actual) for AR-DRG version 8.0.
- ISO 8559 (PART 1) 2017. Size designation of clothes Part 1: Anthropometric definitions for body measurement.
- IWUAGWU, O. C., STANLEY, P. W., PLATT, A. J., DREW, P. J. & WALKER, L. G. 2006a. Effects of bilateral breast reduction on anxiety and depression: results of a prospective randomised trial. *Scand J Plast Reconstr Surg Hand Surg*, 40, 19-23.
- IWUAGWU, O. C., WALKER, L. G., STANLEY, P. W., HART, N. B., PLATT, A. J. & DREW, P. J. 2006b. Randomized clinical trial examining psychosocial and quality of life benefits of bilateral breast reduction surgery. *Br J Surg*, 93, 291-4.
- JAESCHKE, R., SINGER, J. & GUYATT, G. H. 1989. Measurement of health status. Ascertaining the minimal clinically important difference. *Control Clin Trials*, 10, 407-15.
- JAMES, R., SALTON, R. I., BYRNES, J. M. & SCUFFHAM, P. A. 2017. Cost-utility analysis for bariatric surgery compared with usual care for the treatment of obesity in Australia. *Surg Obes Relat Dis,* 13, 2012-2020.
- JANIK, P. E., CHARYTONOWICZ, D., MISZCZYK, J. & CHARYTONOWICZ, M. 2019. Female Sexual Function and Sexual Well-being Before and After Breast Reduction: A Pilot Crosssectional Study and Review of Literature. *Ann Plast Surg*, 82, 609-613.
- JAYADEVAPPA, R., COOK, R. & CHHATRE, S. 2017. Minimal important difference to infer changes in health-related quality of life-a systematic review. *J Clin Epidemiol*, 89, 188-198.
- JENKINSON, C., STEWART-BROWN, S., PETERSEN, S. & PAICE, C. 1999. Assessment of the SF-36 version 2 in the United Kingdom. *Journal of Epidemiology and Community Health*, 53, 46-50.
- KARAMANOS, E., WEI, B., SIDDIQUI, A. & RUBINFELD, I. 2015. Tobacco use and body mass index as predictors of outcomes in patients undergoing breast reduction mammoplasty. *Annals of Plastic Surgery*, 75, 383-387.
- KATCH, V. L., CAMPAIGNE, B., FREEDSON, P., SADY, S., KATCH, F. I. & BEHNKE, A. R. 1980. Contribution of breast volume and weight to body fat distribution in females. *Am J Phys Anthropol,* 53, 93-100.
- KAYAR, R., CIVELEK, S., COBANOGLU, M., GUNGOR, O., CATAL, H. & EMIROGLU, M. 2011. Five methods of breast volume measurement: a comparative study of measurements of specimen volume in 30 mastectomy cases. *Breast Cancer (Auckl)*, 5, 43-52.
- KAZIS, L. E., ANDERSON, J. J. & MEENAN, R. F. 1989. Effect sizes for interpreting changes in health status. *Med Care*, 27, S178-89.
- KECECI, Y. & SIR, E. 2014. Prediction of resection weight in reduction mammaplasty based on anthropometric measurements. *Breast Care (Basel)*, 9, 41-5.
- KECECI, Y., SIR, E. & GUNGOR, M. 2015. Patient-reported quality-of-life outcomes of breast reduction evaluated with generic questionnaires and the breast reduction assessed severity scale. *Aesthet Surg J*, 35, 48-54.
- KECECI, Y., SIR, E. & ZENGEL, B. 2013. Validation of the Turkish version of the Breast Reduction Assessed Severity Scale. *Aesthet Surg J*, 33, 66-74.
- KEMALOGLU, C. A. & OZOCAK, H. 2018. Comparative Outcomes of Inferior Pedicle and Superomedial Pedicle Technique With Wise Pattern Reduction in Gigantomastic Patients. *Ann Plast Surg*, 80, 217-222.
- KERRIGAN, C. L. 2013. Commentary on: Validation of the Turkish version of the Breast Reduction Assessed Severity Scale. *Aesthet Surg J*, 33, 75-6.
- KERRIGAN, C. L., COLLINS, E. D., KIM, H. M., SCHNUR, P. L., WILKINS, E., CUNNINGHAM, B.

& LOWERY, J. 2002. Reduction mammaplasty: defining medical necessity. *Med Decis Making*, 22, 208-17.

- KERRIGAN, Č. L., COLLINS, E. D., KNEELAND, T. S., VOIGTLAENDER, D., MONCUR, M. M., MATHENEY, T. H., GROVE, M. R. & TOSTESON, A. N. 2000. Measuring health state preferences in women with breast hypertrophy. *Plast Reconstr Surg*, 106, 280-8.
- KERRIGAN, C. L., COLLINS, E. D., STRIPLIN, D., KIM, H. M., WILKINS, E., CUNNINGHAM, B. & LOWERY, J. 2001. The health burden of breast hypertrophy. *Plast Reconstr Surg*, 108, 1591-9.
- KHAN, H. A. & BAYAT, A. 2008. A geometric method for nipple localization. *Can J Plast Surg,* 16, 45-7.
- KING, M. T. 1996. The interpretation of scores from the EORTC quality of life questionnaire QLQ-C30. *Qual Life Res,* 5, 555-67.
- KING, M. T. 2011. A point of minimal important difference (MID): a critique of terminology and methods. *Expert Rev Pharmacoecon Outcomes Res*, 11, 171-84.
- KLASSEN, A., FITZPATRICK, R., JENKINSON, C. & GOODACRE, T. 1996. Should breast reduction surgery be rationed? A comparison of the health status of patients before and after treatment: postal questionnaire survey. *BMJ*, 313, 454-7.
- KLASSEN, A. F., PUSIĆ, A. L., SCOTT, A., KLOK, J. & CANO, S. J. 2009. Satisfaction and quality of life in women who undergo breast surgery: a qualitative study. *BMC Womens Health*, 9, 11.
- KOCAK, E., CARRUTHERS, K. H. & MCMAHAN, J. D. 2011. A reliable method for the preoperative estimation of tissue to be removed during reduction mammaplasty. *Plast Reconstr Surg*, 127, 1059-64.
- KOLTZ, P. F., FREY, J. D. & LANGSTEIN, H. N. 2013. Insurance coverage and reduction mammaplasty: a systematic review of current health care policies. *Plast Reconstr Surg*, 132, 692e-693e.
- KOVACS, L., EDER, M., HOLLWECK, R., ZIMMERMANN, A., SETTLES, M., SCHNEIDER, A., ENDLICH, M., MUELLER, A., SCHWENZER-ZIMMERER, K., PAPADOPULOS, N. A. & BIEMER, E. 2007. Comparison between breast volume measurement using 3D surface imaging and classical techniques. *Breast*, 16, 137-45.
- KRUCOFF, K. B., CARLSON, A. R., SHAMMAS, R. L., MUNDY, L. R., LEE, H.-J. & GEORGIADE, G. S. 2019. Breast-Related Quality of Life in Young Reduction Mammaplasty Patients: A Long-Term Follow-Up Using the BREAST-Q. *Plastic and Reconstructive Surgery*, 144.
- KWONG, J. W., TIJERINA, J. D., CHOI, S., LUAN, A., FENG, C. L., NGUYEN, D. H., LEE, G. K. & NAZERALI, R. S. 2020. Assessing the Accuracy of a 3-Dimensional Surface Imaging System in Breast Volume Estimation. *Ann Plast Surg*.
- LAHIRI, A., DUFF, C. G., BROWN, T. L. & GRIFFITHS, R. W. 2006. Anthropometric measurements and their value in predicting complications following reduction mammaplasty and abdominoplasty. *Ann Plast Surg*, 56, 248-50.
- LASSUS, C. 1970. A technique for breast reduction. Int Surg, 53, 69-72.
- LEE, W. Y., KIM, M. J., LEW, D. H., SONG, S. Y. & LEE, D. W. 2016. Three-Dimensional Surface Imaging is an Effective Tool for Measuring Breast Volume: A Validation Study. *Arch Plast Surg*, 43, 430-7.
- LEJOUR, M., ABBOUD, M., DECLETY, A. & KERTESZ, P. 1990. [Reduction of mammaplasty scars: from a short inframammary scar to a vertical scar]. *Ann Chir Plast Esthet*, 35, 369-79.
- LETTERMAN, G. & SCHURTER, M. 1980. The effects of mammary hypertrophy on the skeletal system. *Ann Plast Surg*, 5, 425-31.
- LEWIN, R., ELANDER, A., LUNDBERG, J., HANSSON, E., THORARINSSON, A., CLAUDELIN, M., BLADH, H. & LIDÉN, M. 2018. Validation of the breast evaluation questionnaire for breast hypertrophy and breast reduction. *Journal of Plastic Surgery and Hand Surgery*, 1-8.
- LEWIN, R., GORANSSON, M., ELANDER, A., THORARINSSON, A., LUNDBERG, J. & LIDEN, M. 2014. Risk factors for complications after breast reduction surgery. *J Plast Surg Hand Surg*, 48, 10-4.
- LEWIN, R., LIDEN, M., LUNDBERG, J., HANSSON, E., SELVAGGI, G., THORARINSSON, A. & ELANDER, A. 2019. Prospective Evaluation of Health After Breast Reduction Surgery Using the Breast-Q, Short-Form 36, Breast-Related Symptoms Questionnaire, and Modified

Breast Evaluation Questionnaire. Ann Plast Surg.

- LIANG, M. H., FOSSEL, A. H. & LARSON, M. G. 1990. Comparisons of five health status instruments for orthopedic evaluation. *Med Care*, 28, 632-42.
- LITTLE, R. J. A. 1988. Missing-Data Adjustments in Large Surveys. *Journal of Business & Economic Statistics*, 6, 287-296.
- LONGO, B., FARCOMENI, A., FERRI, G., CAMPANALE, A., SOROTOS, M. & SANTANELLI, F. 2013. The BREAST-V: A Unifying Predictive Formula for Volume Assessment in Small, Medium, and Large Breasts. *Plastic and Reconstructive Surgery*, 132, 1e-7e.
- LONIE, S., SACHS, R., SHEN, A., HUNTER-SMITH, D. J., ROZEN, W. M. & SEIFMAN, M. 2019. A systematic review of patient reported outcome measures for women with macromastia who have undergone breast reduction surgery. *Gland Surgery*, 8, 431-440.
- LOSKEN, A., HART, A. M., BROECKER, J. S., STYBLO, T. M. & CARLSON, G. W. 2017. Oncoplastic Breast Reduction Technique and Outcomes: An Evolution over 20 Years. *Plast Reconstr Surg*, 139, 824e-833e.
- LOSKEN, A., SEIFY, H., DENSON, D. D., PAREDES, A. A., JR. & CARLSON, G. W. 2005. Validating three-dimensional imaging of the breast. *Ann Plast Surg*, 54, 471-6; discussion 477-8.
- LYDICK, E. & EPSTEIN, R. S. 1993. Interpretation of quality of life changes. *Qual Life Res,* 2, 221-6.
- MAGLINTE, G. A., HAYS, R. D. & KAPLAN, R. M. 2012. US general population norms for telephone administration of the SF-36v2. *Journal of Clinical Epidemiology*, 65, 497-502.
- MAKKI, A. S. & GHANEM, A. A. 1998. Long-term results and patient satisfaction with reduction mammaplasty. *Ann Plast Surg*, 41, 370-7.
- MANAHAN, M. A., BURETTA, K. J., CHANG, D., MITHANI, S. K., MALLALIEU, J. & SHERMAK, M. A. 2015. An outcomes analysis of 2142 breast reduction procedures. *Annals of Plastic Surgery*, 74, 289-292.
- MANASSA, E. H., HERTL, C. H. & OLBRISCH, R. R. 2003. Wound healing problems in smokers and nonsmokers after 132 abdominoplasties. *Plast Reconstr Surg*, 111, 2082-7; discussion 2088-9.
- MARIN, T., TAYLOR, A. & GILL, T. 2009. A population profile of quality of life in South Australia population norms for 2008 using the SF36 Version 2. Population Research and Outcome Studies Unit. South Australian Department of Health, Adelaide.
- MATTHEWS, J. N., ALTMAN, D. G., CAMPBELL, M. J. & ROYSTON, P. 1990. Analysis of serial measurements in medical research. *BMJ*, 300, 230-5.
- MCCAFFREY, N., KAAMBWA, B., CURROW, D. C. & RATCLIFFE, J. 2016. Health-related quality of life measured using the EQ-5D-5L: South Australian population norms. *Health Qual Life Outcomes*, 14, 133.
- MCCULLEY, S. J. & HUDSON, D. A. 2001. Short-scar breast reduction: why all the fuss? *Plast Reconstr Surg*, 107, 965-9.
- MCNUTT, R. A. 2004. Shared medical decision making: Problems, process, progress. *JAMA*, 292, 2516-2518.
- MEDICAL SERVICES ADVISORY COMMITTEE. 2016. Technical Guidelines for preparing assessment reports for the Medical Services Advisory Committee – Medical Service Type: Therapeutic (Version 2.0) Available: <u>http://www.msac.gov.au/internet/msac/publishing.nsf/Content/assessment-groups</u>

[Accessed 6 January 2020].

- MELLO, A. A., DOMINGOS, N. A. & MIYAZAKI, M. C. 2010. Improvement in quality of life and selfesteem after breast reduction surgery. *Aesthetic Plast Surg*, 34, 59-64.
- MENÉNDEZ-CARDO, A., GUILLEN-GRIMA, F. & HONTANILLA, B. 2017. Analysis of satisfaction after breast reduction comparing vertical scar versus inverted T-shaped technique using the Breast-Q questionnaire. Is patient satisfaction influenced by the amount of tissue removed? *J Plast Surg Hand Surg*, 1-6.
- MERCIECA-BEBBER, R., COSTA, D. S., NORMAN, R., JANDA, M., SMITH, D. P., GRIMISON, P., GAMPER, E. M. & KING, M. T. 2019. The EORTC Quality of Life Questionnaire for cancer patients (QLQ-C30): Australian general population reference values. *Med J Aust*, 210, 499-506.
- MILLER, A. P., ZACHER, J. B., BERGGREN, R. B., FALCONE, R. E. & MONK, J. 1995. Breast

reduction for symptomatic macromastia: can objective predictors for operative success be identified? *Plast Reconstr Surg*, 95, 77-83.

- MILLER, B. J., MORRIS, S. F., SIGURDSON, L. L., BENDOR-SAMUEL, R. L., BRENNAN, M., DAVIS, G. & PALETZ, J. L. 2005. Prospective study of outcomes after reduction mammaplasty. *Plast Reconstr Surg*, 115, 1025-31.
- MITTMANN, N., TRÁKAS, K., RISEBROUGH, N. & LIU, B. A. 1999. Utility scores for chronic conditions in a community-dwelling population. *Pharmacoeconomics*, 15, 369-76.
- MULLER-NORDHORN, J., ROLL, S. & WILLICH, S. N. 2004. Comparison of the short form (SF)-12 health status instrument with the SF-36 in patients with coronary heart disease. *Heart*, 90, 523-7.
- MUNDY, L. R., HOMA, K., KLASSEN, A. F., PUSIC, A. L. & KERRIGAN, C. L. 2017a. Breast Cancer and Reconstruction: Normative Data for Interpreting the BREAST-Q. *Plast Reconstr Surg*, 139, 1046e-1055e.
- MUNDY, L. R., HOMA, K., KLASSEN, A. F., PUSIC, A. L. & KERRIGAN, C. L. 2017b. Normative Data for Interpreting the BREAST-Q: Augmentation. *Plast Reconstr Surg*, 139, 846-853.
- MUNDY, L. R., HOMA, K., KLASSEN, A. F., PUSIC, A. L. & KERRIGAN, C. L. 2017c. Understanding the Health Burden of Macromastia: Normative Data for the BREAST-Q Reduction Module. *Plast Reconstr Surg*, 139, 846e-853e.
- MUSLU, U., DEMIR, E., OZDEMIR, F., OZKOCAK, V. & YILDIRIM, E. 2019. Investigation of the Anthropometric Changes in Breast Volume and Measurements After Breast Reduction. *Cureus*, 11, e4312.
- MUSORO, Z. J., HAMEL, J. F., EDIEBAH, D. E., COCKS, K., KING, M. T., GROENVOLD, M., SPRANGERS, M. A. G., BRANDBERG, Y., VELIKOVA, G., MARINGWA, J., FLECHTNER, H. H., BOTTOMLEY, A., COENS, C. & GROUP, E. Q. O. L. 2018. Establishing anchorbased minimally important differences (MID) with the EORTC quality-of-life measures: a meta-analysis protocol. *BMJ Open*, 8, e019117.
- MYUNG, Y. & HEO, C. Y. 2017. Relationship Between Obesity and Surgical Complications After Reduction Mammaplasty: A Systematic Literature Review and Meta-Analysis. *Aesthet Surg J*, 37, 308-315.
- NATIONAL INSTITUTE FOR HEALTH AND CLINICAL EXCELLENCE. 2014a. Developing NICE guidelines: the manual. Available: <u>https://www.nice.org.uk/process/pmg20/resources/developing-nice-guidelines-the-manual-</u> pdf-72286708700869 [Accessed 7 January, 2020].
- NATIONAL INSTITUTE FOR HEALTH AND CLINICAL EXCELLENCE. 2014b. Osteoarthritis: Care and Management in Adults. Available: <u>https://www.nice.org.uk/guidance/cg177</u> [Accessed 27 January 2018].
- NELSON, J. A., FISCHER, J. P., CHUNG, C. U., WEST, A., TUGGLE, C. T., SERLETTI, J. M. & KOVACH, S. J. 2014a. Obesity and early complications following reduction mammaplasty: an analysis of 4545 patients from the 2005-2011 NSQIP datasets. *J Plast Surg Hand Surg*, 48, 334-9.
- NELSON, J. A., FISCHER, J. P., WINK, J. D. & KOVACH, S. J., 3RD 2014b. A population-level analysis of bilateral breast reduction: does age affect early complications? *Aesthet Surg J*, 34, 409-16.
- NETSCHER, D. T., MEADE, R. A., GOODMAN, C. M., BREHM, B. J., FRIEDMAN, J. D. & THORNBY, J. 2000. Physical and psychosocial symptoms among 88 volunteer subjects compared with patients seeking plastic surgery procedures to the breast. *Plast Reconstr Surg*, 105, 2366-73.
- NGUYEN, J. T., WHEATLEY, M. J., SCHNUR, P. L., NGUYEN, T. A. & WINN, S. R. 2008. Reduction mammaplasty: a review of managed care medical policy coverage criteria. *Plast Reconstr Surg*, 121, 1092-100.
- NHS ENGLAND. 2018. Evidence-Based Interventions: Response to the public consultation and next steps. Available: <u>https://www.england.nhs.uk/wp-content/uploads/2018/11/ebi-consultation-response-document.pdf</u> [Accessed 4 January, 2019].
- NOLTE, S., LIEGL, G., PETERSEN, M. A., AARONSON, N. K., COSTANTINI, A., FAYERS, P. M., GROENVOLD, M., HOLZNER, B., JOHNSON, C. D., KEMMLER, G., TOMASZEWSKI, K.
 A., WALDMANN, A., YOUNG, T. E. & ROSE, M. 2019. General population normative data for the EORTC QLQ-C30 health-related quality of life questionnaire based on 15,386

persons across 13 European countries, Canada and the Unites States. *European Journal* of Cancer, 107, 153-163.

- NORMAN, G. R., SLOAN, J. A. & WYRWICH, K. W. 2003. Interpretation of changes in healthrelated quality of life: the remarkable universality of half a standard deviation. *Med Care*, 41, 582-92.
- NORMAN, R., CHURCH, J., VAN DEN BERG, B. & GOODALL, S. 2013. Australian health-related quality of life population norms derived from the SF-6D. *Australian and New Zealand Journal of Public Health*, 37, 17-23.
- NORMAN, R., VINEY, R., BRAZIER, J., BURGESS, L., CRONIN, P., KING, M., RATCLIFFE, J. & STREET, D. 2014. Valuing SF-6D Health States Using a Discrete Choice Experiment. *Medical Decision Making*, 34, 773-786.
- NSW GOVERNMENT DEPARTMENT OF HEALTH. 2012. *Waiting Time and Elective Surgery Policy* [Online]. Available: http://www1.health.nsw.gov.au/pds/ActivePDSDocuments/PD2012_011.pdf [Accessed]

January 25, 2018]. NUZZI, C. L., FIRRIOLO, M. J., PIKE, M. C., DIVASTA, D. A. & LABOW, I. B. 2019. Complications

- NUZZI, C. L., FIRRIOLO, M. J., PIKE, M. C., DIVASTA, D. A. & LABOW, I. B. 2019. Complications and Quality of Life following Reduction Mammaplasty in Adolescents and Young Women. *Plastic and Reconstructive Surgery*, 144, 572-581.
- O'BLENES, C. A. E., DELBRIDGE, C. L., MILLER, B. J., PANTELIS, A. & MORRIS, S. F. 2006. Prospective study of outcomes after reduction mammaplasty: Long-term follow-up. *Plast Reconstr Surg*, 117, 351-358.
- O'CONNELL, R. L., STEVENS, R. J., HARRIS, P. A. & RUSBY, J. E. 2015. Review of threedimensional (3D) surface imaging for oncoplastic, reconstructive and aesthetic breast surgery. *Breast*, 24, 331-42.

O'GRADY, K. F., THOMA, A. & DAL CIN, A. 2005. A comparison of complication rates in large and small inferior pedicle reduction mammaplasty. *Plast Reconstr Surg*, 115, 736-42.

- OGUNLEYE, A. A., LEROUX, O., MORRISON, N. & PREMINGER, A. B. 2017. Complications After Reduction Mammaplasty: A Comparison of Wise Pattern/Inferior Pedicle and Vertical Scar/Superomedial Pedicle. *Ann Plast Surg*.
- OKORO, S. A., BARONE, C., BOHNENBLUST, M. & WANG, H. T. 2008. Breast reduction trend among plastic surgeons: a national survey. *Plast Reconstr Surg*, 122, 1312-20.
- OSOBA, D., RODRIGUES, G., MYLES, J., ZEE, B. & PATER, J. 1998. Interpreting the significance of changes in health-related quality-of-life scores. *J Clin Oncol*, 16, 139-44.
- OUSMEN, A., TOURAINE, C., DELIU, N., COTTONE, F., BONNETAIN, F., EFFICACE, F., BREDART, A., MOLLEVI, C. & ANOTA, A. 2018. Distribution- and anchor-based methods to determine the minimally important difference on patient-reported outcome questionnaires in oncology: a structured review. *Health Qual Life Outcomes*, 16, 228.
- PARIKH, R. P., SHARMA, K., QURESHI, A. A., FRANCO, M. J. & MYCKATYN, T. M. 2018. Quality of Surgical Outcomes Reporting in Plastic Surgery: A 15-Year Analysis of Complication Data. *Plast Reconstr Surg*, 141, 1332-1340.
- PAUZENBERGER, R., BAUER, T., KIRCHEBNER, M., PIKULA, R., VERSTAPPEN, R., MUHLBACHER, J., DJEDOVIC, G., PIERER, G. & RIEGER, U. M. 2014. Reduction mammaplasty: outcome analysis based on bodyweight. *Aesthetic Plast Surg*, 38, 120-128.
- PEREZ-PANZANO, E., GASCON-CATALAN, A., SOUSA-DOMINGUEZ, R., CARRERA-LASFUENTES, P., GARCIA-CAMPAYO, J. & GUEMES-SANCHEZ, A. 2017. Reduction mammaplasty improves levels of anxiety, depression and body image satisfaction in patients with symptomatic macromastia in the short and long term. *J Psychosom Obstet Gynaecol*, 1-11.
- PEREZ-PANZANO, E., GUEMES-SANCHEZ, A. & GASCON-CATALAN, A. 2016. Quality of Life Following Symptomatic Macromastia Surgery: Short- and Long-term Evaluation. *Breast J*, 22, 397-406.
- PHARMACEUTICAL BENEFITS ADVISORY COMMITTEE. 2016. Guidelines for preparing submissions to the Pharmaceutical Benefits Advisory Committee Version 5.0. September 2016. Available: <u>https://pbac.pbs.gov.au/</u> [Accessed 3 December 2019].
- PIVEC, R., ISSA, K., GIVEN, K., HARWIN, S. F., GREENE, K. A., HITT, K. D., SHI, S. & MONT,
 M. A. 2015. A prospective, longitudinal study of patient satisfaction following total knee arthroplasty using the Short-Form 36 (SF-36) survey stratified by various demographic and

comorbid factors. J Arthroplasty, 30, 374-8.

- PLATT, A. J., MOHAN, D. & BAGULEY, P. 2003. The effect of body mass index and wound irrigation on outcome after bilateral breast reduction. *Ann Plast Surg*, 51, 552-5.
- POLLY, W. D., GLASSMAN, D. S., SCHWENDER, D. J., SHAFFREY, I. C., BRANCH, K. C., BURKUS, F. J. & GORNET, F. M. 2007. SF-36 PCS Benefit-Cost Ratio of Lumbar Fusion Comparison to Other Surgical Interventions: A Thought Experiment. *Spine*, 32, S20-S26.
- POPULATION RESEARCH AND OUTCOME STUDIES UNIT 2008. Brief Report, The Short Form 36 Version 2 Main differences and Revisions: 2008 Update. SA Department of Health.
- PUSIC, A. L., CHEN, C. M., CANO, S., KLASSEN, A., MCCARTHY, C., COLLINS, E. D. & CORDEIRO, P. G. 2007a. Measuring quality of life in cosmetic and reconstructive breast surgery: a systematic review of patient-reported outcomes instruments. *Plast Reconstr Surg*, 120, 823-37; discussion 838-9.
- PUSIC, A. L., KLASSEN, A., CANO, S. J. & KERRIGAN, C. L. 2007b. Validation of the breast evaluation questionnaire. *Plast Reconstr Surg*, 120, 352-3.
- PUSIC, A. L., KLASSEN, A. F., SCOTT, A. M., KLOK, J. A., CORDEIRO, P. G. & CANO, S. J. 2009. Development of a new patient-reported outcome measure for breast surgery: the BREAST-Q. *Plast Reconstr Surg*, 124, 345-53.
- PUSIC, A. L., LEMAINE, V., KLASSEN, A. F., SCOTT, A. M. & CANO, S. J. 2011. Patient-reported outcome measures in plastic surgery: use and interpretation in evidence-based medicine. *Plast Reconstr Surg*, 127, 1361-7.
- QIAO, Q., ZHOU, G. & LING, Y. 1997. Breast volume measurement in young Chinese women and clinical applications. *Aesthetic Plast Surg*, 21, 362-8.
- RAWLINS, M. & DILLON, A. 2005. NICE discrimination. J Med Ethics, 31, 683-4; discussion 685-8.
- RAWLINS, M. D. & CULYER, A. J. 2004. National Institute for Clinical Excellence and its value judgments. *BMJ*, 329, 224-227.
- REAVEY, P. L., KLASSEN, A. F., CANO, S. J., MCCARTHY, C., SCOTT, A., RUBIN, J. P., SHERMAK, M. & PUSIC, A. L. 2011. Measuring quality of life and patient satisfaction after body contouring: a systematic review of patient-reported outcome measures. *Aesthet Surg J*, 31, 807-13.
- REVICKI, D., HAYS, R. D., CELLA, D. & SLOAN, J. 2008. Recommended methods for determining responsiveness and minimally important differences for patient-reported outcomes. *Journal of clinical epidemiology*, 61, 102-109.
- ROBBINS, T. H. 1977. A reduction mammaplasty with the areola-nipple based on an inferior dermal pedicle. *Plast Reconstr Surg*, 59, 64-7.
- ROEHL, K., CRAIG, E. S., GÓMEZ, V. & PHILLIPS, L. G. 2008. Breast reduction: safe in the morbidly obese? *Plast Reconstr Surg*, 122, 370-8.
- ROGMARK, P., PETERSSON, U., BRINGMAN, S., EZRA, E., OSTERBERG, J. & MONTGOMERY, A. 2016. Quality of Life and Surgical Outcome 1 Year After Open and Laparoscopic Incisional Hernia Repair: PROLOVE: A Randomized Controlled Trial. *Ann Surg*, 263, 244-50.
- ROHRICH, R. J., GOSMAN, A. A., BROWN, S. A., TONADAPU, P. & FOSTER, B. 2004. Current preferences for breast reduction techniques: a survey of board-certified plastic surgeons 2002. *Plastic and reconstructive surgery*, 114, 1724.
- ROMEO, M., CUCCIA, G., ZIRILLI, A., WEILER-MITHOFF, E. & STAGNO D'ALCONTRES, F. 2010. Reduction mammaplasty and related impact on psychosexual function. *J Plast Reconstr Aesthet Surg*, 63, 2112-6.
- ROSENBERG, M. 1965. *Society and the adolescent self-image,* Princeton, N.J., Princeton, N.J., Princeton University Press.
- ROSER, K., MADER, L., BAENZIGER, J., SOMMER, G., KUEHNI, C. E. & MICHEL, G. 2019. Health-related quality of life in Switzerland: normative data for the SF-36v2 questionnaire. *Qual Life Res*, 28, 1963-1977.
- ROYAL COLLEGE OF SURGEONS OF ENGLAND. 2012. Access all Ages. Available: <u>https://www.rcseng.ac.uk/-/media/files/rcs/about-rcs/...relations.../access all ages-(1).pdf</u> [Accessed 28 January 2018].
- ROYAL COLLEGE OF SURGEONS OF ENGLAND. 2014. *Is Access to Surgery a Postcode Lottery*? [Online]. Available: <u>https://www.rcseng.ac.uk/-/media/files/rcs/news-and-events/is-</u> <u>access-to-surgery-a-postcode-lottery.pdf?la=en</u> [Accessed 27 January, 2018].

- ROYAL COLLEGE OF SURGEONS OF ENGLAND. 2016. Smokers and overweight patients: Soft targets for NHS savings? Available: <u>https://www.rcseng.ac.uk/-/media/files/rcs/library-and-publications/non-journal-publications/smokers-and-overweight-patients--soft-targets-for-nhs-savings.pdf</u> [Accessed 18 January 2018].
- RUBIN, D. B. 1986. Statistical Matching Using File Concatenation with Adjusted Weights and Multiple Imputations. *Journal of Business & Economic Statistics*, 4, 87-94.
- RUSSELL, C., ROBERTS, M., WILLIAMSON, T. G., MCKERCHER, J., JOLLY, S. E. & MCNEIL, J. 2003. Clinical categorization for elective surgery in Victoria. *ANZ Journal of Surgery*, 73, 839-842.
- SAARINIEMI, K., LUUKKALA, T. & KUOKKANEN, H. 2011a. The outcome of reduction mammaplasty is affected more by psychosocial factors than by changes in breast dimensions. *Scand J Surg*, 100, 105-9.
- SAARINIEMI, K. M., JOUKAMAA, M., RAITASALO, R. & KUOKKANEN, H. O. 2009. Breast reduction alleviates depression and anxiety and restores self-esteem: a prospective randomised clinical trial. *Scand J Plast Reconstr Surg Hand Surg*, 43, 320-4.
- SAARINIEMI, K. M., KERANEN, U. H., SALMINEN-PELTOLA, P. K. & KUOKKANEN, H. O. 2008a. Reduction mammaplasty is effective treatment according to two quality of life instruments. A prospective randomised clinical trial. *J Plast Reconstr Aesthet Surg*, 61, 1472-8.
- SAARINIEMI, K. M., KUOKKANEN, H. O., RASANEN, P., SINTONEN, H. & TUKIAINEN, E. J. 2012. The cost utility of reduction mammaplasty at medium-term follow-up: a prospective study. *J Plast Reconstr Aesthet Surg*, 65, 17-21.
- SAARINIEMI, K. M., KUOKKANEN, H. O. & TUKIAINEN, E. J. 2011b. The outcome of reduction mammaplasty remains stable at 2-5 years' follow-up: a prospective study. *J Plast Reconstr Aesthet Surg,* 64, 573-6.
- SAARINIEMI, K. M., SINTONEN, H. & KUOKKANEN, H. O. 2008b. The improvement in quality of life after breast reduction is comparable to that after major joint replacement. *Scand J Plast Reconstr Surg Hand Surg*, 42, 194-8.
- SABINO NETO, M., DEMATTÊ, M. F., FREIRE, M., GARCIA, E. B., QUARESMA, M. & FERREIRA, L. M. 2008. Self-esteem and functional capacity outcomes following reduction mammaplasty. *Aesthet Surg J*, 28, 417-20.
- SAHIN, I., ISKENDER, S., OZTURK, S., BALABAN, B. & ISIK, S. 2013. Evaluation of breast reduction surgery effect on body posture and gait pattern using three-dimensional gait analysis. *Aesthetic Plast Surg*, 37, 549-53.
- SAMSA, G., EDELMAN, D., ROTHMAN, M. L., WILLIAMS, G. R., LIPSCOMB, J. & MATCHAR, D. 1999. Determining clinically important differences in health status measures: a general approach with illustration to the Health Utilities Index Mark II. *Pharmacoeconomics*, 15, 141-55.
- SARWER, D. B., BARTLETT, S. P., BUCKY, L. P., LAROSSA, D., LOW, D. W., PERTSCHUK, M. J., WADDEN, T. A. & WHITAKER, L. A. 1998. Bigger is not always better: body image dissatisfaction in breast reduction and breast augmentation patients. *Plast Reconstr Surg*, 101, 1956-61; discussion 1962-3.
- SCHNUR, P. L. 1999. Reduction mammaplasty-the schnur sliding scale revisited. *Ann Plast Surg,* 42, 107-8.

SCHNUR, P. L., HOEHN, J. G., ILSTRUP, D. M., CAHOY, M. J. & CHU, C. P. 1991. Reduction mammaplasty: cosmetic or reconstructive procedure? *Ann Plast Surg*, 27, 232-7.

SCHNUR, P. L., SCHNUR, D. P., PETTY, P. M., HANSON, T. J. & WEAVER, A. L. 1997. Reduction mammaplasty: an outcome study. *Plast Reconstr Surg*, 100, 875-83.

SCHULTZ, R. C., DOLEZAL, R. F. & NOLAN, J. 1986. Further applications of Archimedes' principle in the correction of asymmetrical breasts. *Ann Plast Surg*, 16, 98-101.

SCHUMACHER, H. H. 2005. Breast reduction and smoking. Ann Plast Surg, 54, 117-9.

- SCHWARTZ, C. E. & SPRANGERS, M. A. G. 1999. Methodological approaches for assessing response shift in longitudinal health-related quality-of-life research. *Social Science & Medicine*, 48, 1531-1548.
- SCOTT, G. R., CARSON, C. L. & BORAH, G. L. 2005. Maximizing outcomes in breast reduction surgery: a review of 518 consecutive patients. *Plast Reconstr Surg*, 116, 1633-9; discussion 1640-1.

- SEGAL, L., DAY, S. E., CHAPMAN, A. B. & OSBORNE, R. H. 2004. Can we reduce disease burden from osteoarthritis? *Med J Aust*, 180, S11-7.
- SEITCHIK, M. W. 1995. Reduction mammaplasty: criteria for insurance coverage. *Plast Reconstr Surg*, 95, 1029-32.
- SELBY, J. V., BEAL, A. C. & FRANK, L. 2012. The Patient-Centered Outcomes Research Institute (PCORI) national priorities for research and initial research agenda. *Jama*, 307, 1583-1584.
- SETALA, L., PAPP, A., JOUKAINEN, S., MARTIKAINEN, R., BERG, L., MUSTONEN, P. & HARMA, M. 2009. Obesity and complications in breast reduction surgery: are restrictions justified? *J Plast Reconstr Aesthet Surg*, 62, 195-9.
- SHAH, R., AL-AJAM, Y., STOTT, D. & KANG, N. 2011. Obesity in mammaplasty: a study of complications following breast reduction. *J Plast Reconstr Aesthet Surg*, 64, 508-14.
- SHAKESPEARE, V. & COLE, R. P. 1997. Measuring patient-based outcomes in a plastic surgery service: breast reduction surgical patients. *Br J Plast Surg*, 50, 242-8.
- SHAKESPEARE, V. & POSTLE, K. 1999. A qualitative study of patients' views on the effects of breast-reduction surgery: a 2-year follow-up survey. *Br J Plast Surg*, 52, 198-204.
- SHERMAK, M. A., CHANG, D., BURETTA, K., MITHANI, S., MALLALIEU, J. & MANAHAN, M. 2011. Increasing age impairs outcomes in breast reduction surgery. *Plast Reconstr Surg*, 128, 1182-7.
- SIGURDSON, L., KIRKLAND, S. A. & MYKHALOVSKIY, E. 2007a. Validation of a questionnaire for measuring morbidity in breast hypertrophy. *Plast Reconstr Surg*, 120, 1108-14.
- SIGURDSON, L., MYKHALOVSKIY, E., KIRKLAND, S. A. & PALLEN, A. 2007b. Symptoms and Related Severity Experienced by Women with Breast Hypertrophy. *Plastic and Reconstructive Surgery*, 119, 481-486.
- SIGURDSON, L. J. & KIRKLAND, S. A. 2006. Breast volume determination in breast hypertrophy: an accurate method using two anthropomorphic measurements. *Plast Reconstr Surg*, 118, 313-20.
- SIMPSON, A. M., DONATO, D. P., KWOK, A. C. & AGARWAL, J. P. 2019. Predictors of complications following breast reduction surgery: A National Surgical Quality Improvement Program study of 16,812 cases. J Plast Reconstr Aesthet Surg, 72, 43-51.
- SINGER, P. 2009. Why We Must Ration Health Care. *The New York Times*, July 19, 2009, p.MM38.
- SINGH, K. A. & LOSKEN, A. 2012. Additional benefits of reduction mammaplasty: a systematic review of the literature. *Plast Reconstr Surg*, 129, 562-70.
- SINGH, K. A., PINELL, X. A. & LOSKEN, A. 2010. Is reduction mammaplasty a stimulus for weight loss and improved quality of life? *Ann Plast Surg*, 64, 585-7.
- SINTONEN, H. 2001. The 15D instrument of health-related quality of life: properties and applications. *Ann Med*, 33, 328-36.
- SMITH, D. J., JR., PALIN, W. E., JR., KATCH, V. L. & BENNETT, J. E. 1986. Breast volume and anthropomorphic measurements: normal values. *Plast Reconstr Surg*, 78, 331-5.
- SOMMER, N. Z., ZOOK, E. G. & VERHULST, S. J. 2002. The prediction of breast reduction weight. *Plast Reconstr Surg*, 109, 506-11.
- SOX, H. C. 2010. Comparative effectiveness research: a progress report. *Annals of internal medicine*, 153, 469.
- SPECTOR, J. A. & KARP, N. S. 2007. Reduction mammaplasty: a significant improvement at any size. *Plast Reconstr Surg*, 120, 845-50.
- SPECTOR, J. A., SINGH, S. P. & KARP, N. S. 2008. Outcomes after breast reduction: does size really matter? *Ann Plast Surg*, 60, 505-9.
- SPRANGERS, M. A. G. 1996. Response-shift bias: a challenge to the assessment of patients' quality of life in cancer clinical trials. *Cancer Treatment Reviews*, 22, 55-62.
- SRINIVÁSAIÁH, N., IWUCHUKWU, O. C., STANLEY, P. R., HART, N. B., PLATT, A. J. & DREW,
 P. J. 2014. Risk factors for complications following breast reduction: results from a randomized control trial. *Breast J*, 20, 274-8.
- STARLEY, I. F., BRYDEN, D. C., TAGARI, S., MOHAMMED, P. & JONES, B. P. 1998. An investigation into changes in lung function and the subjective medical benefits from breast reduction surgery. *Br J Plast Surg*, 51, 531-4.
- STEVENS, W. G., GEAR, A. J., STOKER, D. A., HIRSCH, E. M., COHEN, R., SPRING, M., VATH, S. D., SCHANTZ, S. A. & HECK, R. T. 2008. Outpatient reduction mammaplasty: an

eleven-year experience. Aesthet Surg J, 28, 171-9.

- STICKLES, B., PHILLIPS, L., BROX, W. T., OWENS, B. & LANZER, W. L. 2001. Defining the relationship between obesity and total joint arthroplasty. *Obes Res*, 9, 219-23.
- STRONG, B. & HALL-FINDLAY, E. J. 2015. How Does Volume of Resection Relate to Symptom Relief for Reduction Mammaplasty Patients? *Ann Plast Surg*, 75, 376-82.
- TADIPARTHI, S. & LIEW, S. H. 2008. Use of patient body mass index as a rationing tool in breast reduction surgery. *Plast Reconstr Surg*, 122, 35e-36e.
- TAFT, C., KARLSSON, J. & SULLIVAN, M. 2004. Performance of the Swedish SF-36 version 2.0. *Quality of Life Research*, 13, 251-256.
- TARONE, R. E., LIPWORTH, L., YOUNG, V. L. & MCLAUGHLIN, J. K. 2004. Breast reduction surgery and breast cancer risk: does reduction mammaplasty have a role in primary prevention strategies for women at high risk of breast cancer? *Plast Reconstr Surg*, 113, 2104-10; discussion 2111-2.
- TASMANIAN GOVERNMENT DEPARTMENT OF HEALTH AND HUMAN SERVICES. 2015. Improving Time to Treatment - Tasmania's Elective Surgery Access Policy. [Online]. Available:

http://www.dhhs.tas.gov.au/ data/assets/pdf file/0020/60176/Improving Time to Treatm ent - Tasmania s Elective Surgery Access - DHHS Wide - THOs - Policy -200911201 - UNDER REVIEW.pdf [Accessed January 25, 2018].

- TAYLOR, A. J., TATE, D., BRANDBERG, Y. & BLOMQVIST, L. 2004. Cost-effectiveness of reduction mammaplasty. Int J Technol Assess Health Care, 20, 269-73.
- TAYLOR, C. & JAN, S. 2017. Economic evaluation of medicines. Aust Prescr, 40, 76-78.
- TENNA, S., COGLIANDRO, A., CAGLI, B., BARONE, M., DELLE FEMMINE, P. & PERSICHETTI, P. 2012. Breast hypertrophy and asymmetry: a retrospective study on a sample of 344 consecutive patients. *Acta Chir Plast*, 54, 9-12.
- TEZEL, E. & NUMANOGLU, A. 2000. Practical do-it-yourself device for accurate volume measurement of breast. *Plast Reconstr Surg*, 105, 1019-23.
- THEOCHARIDIS, V., KATSAROS, I., SGOUROMALLIS, E., SERIFIS, N., BOIKOU, V., TASIGIORGOS, S., KOKOSIS, G. & ECONOMOPOULOS, K. P. 2018. Current evidence on the role of smoking in plastic surgery elective procedures: A systematic review and meta-analysis. *J Plast Reconstr Aesthet Surg*, 71, 624-636.
- THOMA, A., IGNACY, T. A., DUKU, E. K., PATTERSON, R. S., CIN, A. D., LEVIS, C. M. & GOLDSMITH, C. H. 2013. Randomized controlled trial comparing health-related quality of life in patients undergoing vertical scar versus inverted T-shaped reduction mammaplasty. *Plast Reconstr Surg*, 132, 48e-60e.
- THOMA, A., KAUR, M. N., TSOI, B., ZIOLKOWSKI, N., DUKU, E. & GOLDSMITH, C. H. 2014. Cost-effectiveness analysis parallel to a randomized controlled trial comparing vertical scar reduction and inverted T-shaped reduction mammaplasty. *Plastic and Reconstructive Surgery*, 134, 1093-1107.
- THOMA, A., SPRAGUE, S., VELTRI, K., DUKU, E. & FURLONG, W. 2005. Methodology and measurement properties of health-related quality of life instruments: a prospective study of patients undergoing breast reduction surgery. *Health Qual Life Outcomes*, 3, 44.
- THOMA, A., SPRAGUE, S., VELTRI, K., DUKU, E. & FURLONG, W. 2007. A prospective study of patients undergoing breast reduction surgery: health-related quality of life and clinical outcomes. *Plast Reconstr Surg*, 120, 13-26.
- TYKKA, E., RASANEN, P., TUKIAINEN, E., ASKO-SELJAVAARA, S., HEIKKILA, A., SINTONEN, H. & ROINE, R. P. 2010. Cost-utility of breast reduction surgery--a prospective study. *J Plast Reconstr Aesthet Surg*, 63, 87-92.
- UNITED STATES CENSUS BUREAU. 2016. Age and Sex Composition in the United States: 2016. Available: <u>https://www.census.gov/data/tables/2016/demo/age-and-sex/2016-age-sex-</u> composition.html [Accessed 12 November 2019].
- VALTONEN, J. P., SETALA, L. P., MUSTONEN, P. K. & BLOM, M. 2014. Can the efficacy of reduction mammoplasty be predicted? The applicability and predictive value of breast-related symptoms questionnaire in measuring breast-related symptoms pre- and postoperatively. *J Plast Reconstr Aesthet Surg*, 67, 676-81.
- VANDEPUT, J. J. & NELISSEN, M. 2002. Considerations on anthropometric measurements of the female breast. *Aesthetic Plast Surg*, 26, 348-55.

- VEITCH, D., BURFORD, K., DENCH, P., DEAN, N. & GRIFFIN, P. 2012. Measurement of breast volume using body scan technology(computer-aided anthropometry). *Work*, 41 Suppl 1, 4038-45.
- VICTORIAN GOVERNMENT DEPARTMENT OF HUMAN SERVICES. 2015. Elective surgery access policy. [Online]. Available: https://www2.bealth.vic.gov.au/about/publications/policiesandquidelines/elective-surgery

https://www2.health.vic.gov.au/about/publications/policiesandguidelines/elective-surgeryaccess-policy-2015 [Accessed 25 January, 2018].

- VOINESKOS, S. H., KLASSEN, A. F., CANO, S. J., PUSIC, A. L. & GIBBONS, C. J. 2020. Giving Meaning to Differences in BREAST-Q Scores: Minimal Important Difference for Breast Reconstruction Patients. *Plast Reconstr Surg*, 145, 11e-20e.
- VOINESKOS, S. H., NELSON, J. A., KLASSEN, A. F. & PUSIC, A. L. 2018. Measuring Patient-Reported Outcomes: Key Metrics in Reconstructive Surgery. *Annu Rev Med*, 69, 467-479.
- WAGNER, D. S. & ALFONSO, D. R. 2005. The influence of obesity and volume of resection on success in reduction mammaplasty: an outcomes study. *Plast Reconstr Surg.* 115, 1034-8.
- WALTERS, S. J. & BRAZIER, J. E. 2005. Comparison of the minimally important difference for two health state utility measures: EQ-5D and SF-6D. *Qual Life Res*, 14, 1523-32.
- WAMALWA, A. O., WANJERI, J. K., NANGOLE, F. W. & KHAINGA, S. O. 2018. Anthropometrics in preoperative prediction of resection weight in reduction mammaplasty: Is it really accurate? *J Plast Reconstr Aesthet Surg*, 71, e15-e16.
- WAMPLER, A., POWELSON, I. & MATTHEW, M. K. 2019. Comparative Accuracy of Clinical and Formula-Based Estimation of Resection Weight in Reduction Mammoplasty and Implications for Insurance Preapproval and Payment. *Ann Plast Surg*, 82, S259-S263.
- WANG, C., LUAN, J., CHENG, H., CHEN, L., LI, Z., PÁNAYI, A. C. & LIU, C. 2019. Menstrual Cycle-Related Fluctuations in Breast Volume Measured Using Three-Dimensional Imaging: Implications for Volumetric Evaluation in Breast Augmentation. *Aesthetic Plast Surg*, 43, 1-6.
- WANG, S., GUM, D. & MERLIN, T. 2018. Comparing the ICERs in Medicine Reimbursement Submissions to NICE and PBAC—Does the Presence of an Explicit Threshold Affect the ICER Proposed? *Value in Health*, 21, 938-943.
- WARE, J., KOSINSKI, M. A. & DEWEY, J. 2000. *How to Score Version 2 of the SF-36*® *Health Survey*.
- WARE, J. E., JR., KOSINSKI, M., BAYLISS, M. S., MCHORNEY, C. A., ROGERS, W. H. & RACZEK, A. 1995. Comparison of methods for the scoring and statistical analysis of SF-36 health profile and summary measures: summary of results from the Medical Outcomes Study. *Med Care*, 33, AS264-79.
- WARE, J. E., JR. & SHERBOURNE, C. D. 1992. The MOS 36-item short-form health survey (SF-36). I. Conceptual framework and item selection. *Med Care*, 30, 473-83.
- WARE, J. E., KELLER, S. D. & KOSINSKI, M. 1994. *SF-36 physical and mental health summary scales*, Health Assessment Lab Boston.
- WATSON, E. K., FIRMAN, D. W., BAADE, P. D. & RING, I. 1996. Telephone administration of the SF-36 health survey: validation studies and population norms for adults in Queensland. *Aust N Z J Public Health*, 20, 359-63.
- WEBB, M. L., CERRATO, F., ROSEN, H., DIVASTA, A. D., GREENE, A. K. & LABOW, B. I. 2012. The effect of obesity on early outcomes in adolescents undergoing reduction mammaplasty. *Ann Plast Surg*, 68, 257-60.
- WESSELIUS, T. S., VERHULST, A. C., VREEKEN, R. D., XI, T., MAAL, T. J. J. & ULRICH, D. J. O. 2018. Accuracy of Three Software Applications for Breast Volume Calculations from Three-Dimensional Surface Images. *Plast Reconstr Surg*, 142, 858-865.
- WHO 1947. Constitution of the World Health Organization. Chron World Health Organ, 1, 29-43.
- WINTER, R., HAUG, I., LEBO, P., GROHMANN, M., REISCHIES, F. M. J., CAMBIASO-DANIEL, J., TUCA, A., RIENMULLER, T., FRIEDL, H., SPENDEL, S., FORBES, A. A., WURZER, P. & KAMOLZ, L. P. 2017. Standardizing the complication rate after breast reduction using the Clavien-Dindo classification. *Surgery*, 161, 1430-1435.
- WINTER, R., REISCHIES, F. M. J., TUCA, A., WURZER, P., SCHUBERT, C., WOLFSBERGER, C. H., RIENMUELLER, T., FRIEDL, H., SLJIVICH, M., LUMENTA, D. B. & KAMOLZ, L. P. 2019. BMI and specimen weight: impact on personalized risk profiling for optimized informed consent in breast reduction surgery? *Sci Rep*, 9, 12690.

- WISE, R. J. 1956. A preliminary report on a method of planning the mammaplasty. *Plast Reconstr Surg* (1946), 17, 367-75.
- WOLFSWINKEL, É. M., LEMAINE, V., WEATHERS, W. M., CHIKE-OBI, C. J., XUE, A. S. & HELLER, L. 2013. Hyperplastic breast anomalies in the female adolescent breast. *Semin Plast Surg*, 27, 49-55.
- WORLD HEALTH ORGANIZATION. 2018. *Obesity and overweight* [Online]. Available: <u>https://www.who.int/news-room/fact-sheets/detail/obesity-and-overweight</u> [Accessed 10 December 2019].
- WRAIGHT, W. M., TAY, S. K., NDUKA, C. & PEREIRA, J. A. 2007. Bilateral breast reduction surgery in England: a postcode lottery. *J Plast Reconstr Aesthet Surg*, 60, 1039-44.
- WYRWICH, K. W., BULLINGER, M., AARONSON, N., HAYS, R. D., PATRICK, D. L., SYMONDS, T. & CLINICAL SIGNIFICANCE CONSENSUS MEETING, G. 2005. Estimating clinically significant differences in quality of life outcomes. *Qual Life Res,* 14, 285-95.
- WYRWICH, K. W., NIENABER, N. A., TIERNEY, W. M. & WOLINSKY, F. D. 1999a. Linking clinical relevance and statistical significance in evaluating intra-individual changes in health-related quality of life. *Med Care*, 37, 469-78.
- WYRWICH, K. W., TIERNEY, W. M. & WOLINSKY, F. D. 1999b. Further evidence supporting an SEM-based criterion for identifying meaningful intra-individual changes in health-related quality of life. *Journal of clinical epidemiology*, 52, 861-873.
- YIP, J. M., MOURATOVA, N., JEFFERY, R. M., VEITCH, D. E., WOODMAN, R. J. & DEAN, N. R. 2012. Accurate assessment of breast volume: a study comparing the volumetric gold standard (direct water displacement measurement of mastectomy specimen) with a 3D laser scanning technique. *Ann Plast Surg*, 68, 135-41.
- YIP, J. M., WATSON, D. I., TIGGEMANN, M., HSIA, S., SMALLMAN, A. E. & DEAN, N. R. 2015. Determinants of breast reconstruction outcome: How important is volume symmetry? *J Plast Reconstr Aesthet Surg*, 68, 679-85.
- YOUNG, Z. T., CLOSE, M. & HERRERA, F. A. 2019. National Surgical Quality Improvement Program Analysis of 9110 Reduction Mammaplasty Patients: Identifying Risk Factors Associated With Complications in Patients Older Than 60 Years. *Ann Plast Surg*, 82, S446-S449.
- ZHANG, M. X., CHEN, C. Y., FANG, Q. Q., XU, J. H., WANG, X. F., SHI, B. H., WU, L. H. & TAN,
 W. Q. 2016. Risk Factors for Complications after Reduction Mammoplasty: A Meta-Analysis. *PLoS One*, 11, e0167746.
- ZUBOWSKI, R., ZINS, J. E., FORAY-KAPLON, A., YETMAN, R. J., LUCAS, A. R., PAPAY, F. A., HEIL, D. & HUTTON, D. 2000. Relationship of obesity and specimen weight to complications in reduction mammaplasty. *Plast Reconstr Surg*, 106, 998-1003.

APPENDICES

Open access

BMJ Open Does breast reduction surgery improve health-related quality of life? A prospective cohort study in Australian women

Tamara Crittenden ^{1,2} David I Watson,² Julie Ratcliffe ³, ³ Philip A Griffin,¹ Nicola R Dean ³, ^{1,2} on behalf of the AFESA Research Group

To cite: Crittenden T, Watson DI, Ratcliffe J, *et al.* Does breast reduction surgery improve health-related quality of life? A prospective cohort study in Australian women. *BMJ Open* 2020;10:e031804. doi:10.1136/ bmjopen-2019-031804

► Prepublication history and additional material for this paper are available online. To view these files, please visit the journal online (http://dx.doi. org/10.1136/bmjopen-2019-031804).

Received 20 May 2019 Revised 11 December 2019 Accepted 18 December 2019

Check for updates

© Author(s) (or their employer(s)) 2020. Re-use permitted under CC BY-NC. No commercial re-use. See rights and permissions. Published by BMJ.

¹Department of Plastic and Reconstructive Surgery, Flinders Medical Centre, Bedford Park, South Australia, Australia ²Discipline of Surgery, College of Medicine and Public Health, Flinders University, Bedford Park, South Australia, Australia ³Caring Futures Institute, College of Nursing and Health Sciences, Flinders University, Adelaide, South Australia, Australia

Correspondence to Dr Nicola R Dean;

nicola.dean@sa.gov.au

ABSTRACT

Objectives To assess the health burden of breast hypertrophy and the comparative effectiveness of breast reduction surgery in improving health-related quality of life.

Design Prospective cohort study.

Setting A major public tertiary care hospital in Australia. Participants Women with symptomatic breast hypertrophy who underwent breast reduction surgery were followed for 12 months. A comparison control cohort comprised women with breast hypertrophy who did not undergo surgery.

Interventions Bilateral breast reduction surgery for women in the surgical cohort.

Main outcome measures The primary outcome measure was health-related quality of life measured preoperatively and at 3, 6 and 12 months postoperatively using the Short Form-36 (SF-36) questionnaire. Secondary outcome measures included post-surgical complications Results 209 patients in the surgical cohort completed questionnaires before and after surgery. 124 patients in the control hypertrophy cohort completed baseline and 12-month follow-up questionnaires. At baseline, both groups had significantly lower scores compared with population norms across all scales (p<0.001). In the surgical cohort significant improvements were seen across all eight SF-36 scales (p<0.001) following surgery. Within 3 months of surgery scores were equivalent to those of the normal population and this improvement was sustained at 12 months. SF-36 physical and mental component scores both significantly improved following surgery, with a mean change of 10.2 and 9.2 points, respectively (p<0.001). In contrast, SF-36 scores for breast hypertrophy controls remained at baseline across 12 months. The improvement in quality of life was independent of breast resection weight and body mass index.

Conclusion Breast reduction significantly improved quality of life in women with breast hypertrophy. This increase was most pronounced within 3 months of surgery and sustained at 12-month follow-up. This improvement in quality of life is comparable to other widely accepted surgical procedures. Furthermore, women benefit from surgery regardless of factors including body mass index and resection weight.

Strengths and limitations of this study

- This large prospective longitudinal study reports 12-month follow-up using a validated patientreported outcome measure for health-related quality of life assessment.
- The completion rate of the study was 83% for participants who underwent surgery.
- Comparisons were made with a control cohort of women with breast hypertrophy not undergoing surgery, and also to a normative female reference
- population.
- This was a non-randomised study design.

INTRODUCTION

Breast reduction surgery is a common plastic surgery procedure and it has previously been shown to be effective for relieving pain and functional problems associated with breast hypertrophy,^{1–5} whereas conservative approaches to treatment such as physiotherapy, hormonal therapy and weight loss have much less impact.⁶⁷ However, despite clear published evidence to the contrary, breast reduction surgery is often regarded more as a cosmetic rather than a functional procedure by the general public and many medical professionals.^{1 & 9} This is in spite of the finding that breast hypertrophy is a chronic health problem and relief of physical symptoms is the primary motivator for most women who are pursuing breast reduction surgery.¹⁰

The increasing demand for breast reduction surgery and increasing pressure to constrain healthcare spending have led to lengthy waiting times and restrictions placed on surgery in numerous countries and jurisdictions worldwide.^{411–15} While 'rationing' of healthcare is an essential process in public healthcare systems globally, it has the potential to threaten equity of access to surgical

BMJ

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

1

Original research

Open access

treatment. Within the Australian public hospital system, access to breast reduction surgery for patients is ultimately reliant on state and local policies.¹⁶⁻²¹ Similarly, in the UK, reports on the rationing of surgery by the National Health Service (NHS) on the basis of geograph-^{22 23} In ical location have resulted in a 'postcode lottery'.² 2018, reports from the NHS England 'Evidence-Based Interventions Programme' proposed to restrict funding for procedures it considers 'unnecessary', to save money and eliminate unwarranted clinical variation.²⁴ The inclusion of breast reduction surgery as a 'procedure of limited effectiveness' implies that it is a marginal and low priority procedure in comparison to other medical interventions.²⁵ However, labelling breast reduction surgery an 'ineffective' and 'unnecessary' procedure might be misleading and inaccurate, with little evidence to support this claim. Furthermore, restrictive access policies are in place in both public and private sectors in many countries and jurisdictions worldwide; often these restrictions are based on body mass index (BMI) or a minimum weight of breast resection at surgery.^{4 8 9 11–15 22 23 25} The validity of such criteria might not be evidence-based, resulting in women with a medical need for surgery being denied access to it.

The primary aim of this study was to longitudinally assess health-related quality of life (HRQoL) in women with breast hypertrophy before and after breast reduction surgery, and to compare these outcomes to control groups of women with breast hypertrophy not undergoing surgery, and also to a normative female reference population. The Short Form-36 (SF-36) is a well-established indicator of patient-reported outcome for evaluating the burden of disease states and the outcomes of medical interventions and was therefore chosen as the primary outcome measure for this study. Second, this study aimed to assess the impact of patient demographics and surgical characteristics including, but not limited to, those commonly used as selection criteria for access to surgery and insurance coverage on preoperative HRQoL scores and the long-term improvement in HRQoL following surgery.

METHODS

Design and participants

A prospective cohort study was performed at Flinders Medical Centre in Adelaide, Australia. All women aged 18 years and over with symptomatic breast hypertrophy who were assessed for bilateral breast reduction surgery between April 2007 and February 2018 were informed of the study. Patients who underwent breast reduction surgery comprised the surgery cohort. Patients who were referred for surgery and were placed on the waiting list but were not expected to undergo surgery within 12 months comprised the controls.

All participants who consented to the study were asked to complete the SF-36 questionnaire at set time points. For the surgical patients this was preoperatively and 3, 6

Statistical analyses were performed using SPSS V.25.0 statistical software (IBM Corp). Descriptive statistics including mean, SD and 95% CI were computed for continuous variables. Comparisons between groups were

and 12 months postoperatively. For the control patients, the questionnaire was completed at baseline and again 12 months after enrolment. Data including age, height, weight, bra cup size, health status and smoking status were determined for all patients at baseline and again at follow-up. Women who were unable to complete written questionnaires or were enrolled in the control group and had breast reduction surgery within 12 months of enrolment, or who did not return study questionnaires, were excluded from the study.

Outcome measures

The SF-36 V.2 was used to measure the general HRQoL.²⁶ This contains 36 items which assess health across eight subscales. Questionnaire responses were transformed as per the SF-36 V.2 scoring manual to provide the eight subscales, each with a score between 0 and 100, with higher scores indicating better health.² The subscales were converted into two summary scores: Physical Component Summary (PCS) score and Mental Component Summary (MCS) score using norm-based methods and scoring coefficients from the Australian population.²⁸ For comparison purposes, general female population normative scores were obtained from the 2008 South Australian Health Omnibus Survey and scores weighted to correspond to the age distribution of the study participants.²

Sample size was determined a priori and a minimum sample size of 98 patients per group was calculated to give 80% power at a two-sided significance level of 5% to detect a mean difference of 10-points with an estimated SD of 25-points in the SF-36 questionnaire score.

Study-specific questionnaires, which asked about time off work and consumption and expenditure on medications, were administered at the baseline and 12-month postoperative time points. Participants in the surgical cohort were asked postoperatively whether they would have the surgery again if they had their time over. Additional data were collected pertaining to the surgical technique used, and the weight of breast tissue removed. Hospital records were used to determine the length of hospital stay, number of outpatient clinic appointments relating to the surgery and complications leading to re-hospitalisation, or a further operative procedure within the 12 months follow-up period. A comprehensive complications checklist was completed prospectively during the study by the treating doctor. Three-dimensional laser body scanning was performed preoperatively and at 12 months postoperatively using a Cyberware WBX scanner (Cyberware) and Cyslice software (Headus Pty Ltd). Breast and body volume were measured from the scan according to a protocol described previously.³

Statistical analysis

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

Open access

made using t-tests for continuous data and χ^2 tests for categorical data, with Fisher's exact test as appropriate. Linear mixed models were used to assess the significance of changes in SF-36 subscale scores over multiple time points. For each SF-36 scale an improvement score was calculated using the score obtained at the last available assessment, with a higher score representing a greater improvement from baseline. Pearson correlation coefficients were calculated to assess the linear association between SF-36 scores and baseline participant and clinical characteristics; variables that showed a significant association were entered into the regression model. Candidate variables included age, BMI, preoperative breast volume, bra cup size, tissue resection weight (grams), breast asymmetry and ratio of breast to body volume. Variables were continuous except for bra cup size which was categorised into six groups as follows: D, DD, E, F, G and ≥H cup. Multiple linear regression was used to assess whether any of the collected sociodemographic or clinical variables were associated with first, SF-36 PCS score at baseline, and second, with the change in SF-36 PCS scores from baseline to 12 months after surgery. Statistical significance was accepted at a p value of less than 0.05.

Patient and public involvement

At the design stage of the study two group meetings were held with women with breast hypertrophy to discuss their perspective on the condition, deliver education material and discuss this study. In addition, one consumer was more extensively involved with the design of the study and trialling different methods of breast volume measurement. Study results will be disseminated to the public through presentations and local health newsletter.

Table 1 Baseline characteristics of	of participants		
Characteristic	Surgical cohort (n=209)	Hypertrophy control cohort (n=124)	P value of difference*
Mean (SD; range) age (years)	42.6 (13.4; 18 to 72)	45.3 (13.1; 20 to 79)	0.079
Age group (years):			
18–24	24 (12)	12 (10)	
25–34	38 (18)	15 (12)	
35–44	64 (31)	26 (21)	
45–54	41 (20)	43 (34)	
55–64	31 (15)	21 (18)	
≥65	11 (5)	7 (6)	
Mean (SD) BMI (kg/m ²)	32.7 (6.0)	32.2 (6.1)	0.468
Obesity status:			
Non-obese (<30)	71 (34)	48 (39)	0.326
Obese (≥30)	138 (66)	74 (61)	
Missing	0 (0)	2 (0)	
Smoking status:			
Non-smoker	108 (52)	78 (63)	0.243
Current smoker	35 (17)	14 (11)	
Ex-smoker <12 months	15 (7)	5 (5)	
Ex-smoker >12 months	47 (23)	25 (20)	
Missing	4 (0)	2 (0)	
Bra cup size:			
≤D	13 (6)	4 (3)	
DD	43 (21)	13 (11)	
E	50 (24)	19 (15)	
F	46 (22)	27 (22)	
G	35 (17)	37 (30)	
≥H	19 (10)	19 (15)	
Missing	3 (0)	5 (0)	

Values are numbers (percentages) unless stated otherwise.

*Using independent samples t-test or χ^2 test as appropriate.

BMI, body mass index.

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

3

Open access

RESULTS

Surgical cohort

Of 251 participants who completed a baseline assessment and underwent bilateral breast reduction surgery, 209 (83.3%) completed at least one postoperative follow-up assessment and were included in the study group for analysis. Missing data were due to participants repeatedly not attending appointments or choosing to not complete and return the study questionnaires at some time points. Twenty-three participants formally withdrew from the study following surgical intervention. Baseline characteristics were compared between participants who were lost to follow-up and those who completed at least one postoperative assessment. No difference was observed for age, BMI, tissue weight resected or preoperative SF-36 scales and summary scores except for the mental health scale, where nonrespondents had a lower mean score of 6.8 points less than responders (p=0.034).

Participant demographics for the surgical cohort are summarised in table 1. Preoperatively, mean total breast volume measured by 3D laser scanner was 3391 mL (range 1472–9622 mL). At 12 months postoperatively, mean total breast volume was 2184 mL (range 963 to 4392 mL). The mean total weight of breast tissue resected at surgery was 1338 g±817 g. An inferior pedicle breast reduction technique was the most commonly used approach (161/209, 77%), followed by a superior pedicle technique (35/209, 17%). The average hospital stay was 2.3 days. Fifty-nine patients (28%) experienced at least one surgical complication. Eight patients (3.8%) had subsequent procedures for revision of surgical scars or to correct 'dog-ears'.

The majority of participants (204/209, 97.6%) responded in the postoperative questionnaire that they would have the surgery again, while others were either unsure (4/209, 1.9%) or would not have surgery again (1/209, 0.5%). Following surgery, participants on average spent less money on medications and treatments (AU\$26.41 vs AU\$5.73 per month, p<0.001) and took fewer days off work (4.5 days vs 0.1 days in the previous 6-month period, p=0.009) when compared with before surgery. Using bivariate analysis, obesity was not associated with an increased incidence of surgical complications (p=0.323), with the incidence of complications in non-obese participants (17/71, 24%) and obese participants (42/138, 30%). Furthermore, there were no differences in the incidence of major complications based on obesity status.

The SF-36 was completed preoperatively and at least once postoperatively by 209 surgical participants; 191 (91%) completed the postoperative questionnaires at 3 months, 183 (88%) at 6 months and 193 (92%) at 12 months. When compared with previously published age-adjusted normative data for the female Australian population,²⁹ mean baseline SF-36 scores for the surgical cohort were significantly lower across all scales (p<0.001) (table 2). A comparison of mean preoperative and

3-month postoperative SF-36 scores showed that scores were significantly higher across all eight SF-36 subscales (p<0.001) (table 2) such that they reached the level of the normative population (figure 1). Mean SF-36 PCS and MCS scores significantly improved following surgery, increasing by 10.2 (95% CI; 8.2 to 12.1) and 9.2 (95% CI; 6.9 to 11.6) points, respectively (p<0.001) (figure 2 and online supplementary table S1). The mean change in SF-36 PCS and MCS scores was in excess of the developerrecommended 3-point minimal important difference (MID) threshold.^{32 33} SF-36 scores were stable at 6 and 12 months post-surgery and linear mixed-model analysis showed no significant difference from those at 3 months post-surgery. The mean change in SF-36 scores from baseline to 12 months following surgery was in excess of MID threshold estimates based on a rule of thumb 10-point change on 100-point quality of life scales³⁴ or 0.5 SD default value for patient-perceived important change³ in all eight SF-36 subscales (figure 2). SF-36 scores for obese women improved equally, if not greater than their non-obese counterparts following surgery, reaching statistical significance for the physical functioning subscale (table 3).

Breast hypertrophy control cohort

Study questionnaires were initially posted to 350 women with breast hypertrophy who were not scheduled for surgery; 160 (46%) completed and returned the questionnaires at baseline, and of these 124 responded again 12 months later. Twenty-four of those contacted to participate in the study underwent breast reduction surgery during the study time frame and were therefore excluded. Participant demographics for the hypertrophy control cohort are summarised in table 1. No significant differences were observed when comparing spending on medications and number of days off work between baseline and 12 months following enrolment, with both remaining significantly higher than postoperative surgical participants (p<0.001).

Mean baseline SF-36 scores for women in the breast hypertrophy control group were significantly lower than the normative population across all dimensions (table 2). At 12 months post-baseline, SF-36 scores showed no significant improvement and remained significantly lower than population norms (table 2) and postoperative scores for women in the surgical cohort (figure 2). Mean SF-36 PCS and MCS summary scores for women in the breast hypertrophy control group were significantly lower than those who underwent breast reduction surgery, with a mean difference of 10.6 (95% CI; 8.3 to 12.8) and 11.1 points (95% CI; 8.2 to 13.9), respectively (p<0.001) (table 2).

$\label{eq:comparing} \begin{array}{c} \text{Comparing the improvement in HRQoL with other surgical} \\ \text{interventions} \end{array}$

The improvement in SF-36 physical and mental summary scores in women who underwent surgery in our study was compared with existing studies which describe 12-month postoperative outcomes from other surgical interventions

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

б

Table 2	Mean (95% CI) SF-36	scores for participants	in the surgical cohort, h	nypertrophy control coh	ort and normative fema	le population	
	Normative*	Hypertrophy control	cohort	Surgical cohort			
SF-36 scale	(n=1551)	Baseline (n=160)	12 months (n=124)	Preoperative (n=209)	3 months postoperative (n=190	6 months) postoperative (n=181	12 months) postoperative (n=191
ΡF	84.2 (83.2 to 85.2)	64.7 (60.9 to 68.6)	61.1 (56.7 to 65.7)	61.0 (57.6 to 64.5)	80.1 (76.9 to 83.3)	80.8 (77.3 to 84.4)	83.4 (80.2 to 86.5)
ЧH	82.0 (80.7 to 83.3)	58.3 (53.8 to 62.8)	58.1 (53.0 to 63.0)	56.0 (52.2 to 59.9)	79.5 (76.1 to 82.9)	81.1 (77.4 to 84.8)	81.3 (77.7 to 84.9)
ВР	73.0 (71.9 to 74.1)	39.8 (36.3 to 40.4)	37.9 (34.2 to 41.7)	38.5 (35.6 to 41.5)	67.4 (63.9 to 70.8)	67.6 (63.6 to 71.7)	71.6 (67.9 to 75.2)
ВH	70.2 (69.1 to 71.4)	49.7 (46.3 to 53.1)	49.8 (46.0 to 54.0)	57.9 (55.0 to 60.9)	69.1 (66.3 to 72.0)	69.5 (66.6 to 72.4)	70.4 (67.7 to 73.1)
Ţ	57.3 (56.2 to 58.3)	36.7 (33.6 to 39.8)	35.1 (31.3 to 38.8)	39.7 (37.0 to 42.5)	57.7 (54.8 to 60.6)	58.6 (55.8 to 61.3)	58.9 (56.1 to 61.7)
SF	82.6 (81.3 to 83.8)	55.2 (50.6 to 59.8)	55.1 (50.1 to 59.9)	57.1 (53.3 to 60.9)	78.8 (75.0 to 82.5)	79.4 (75.6 to 83.2)	81.4 (78.1 to 84.7)
RE	88.3 (87.2 to 89.3)	62.8 (58.0 to 67.5)	60.2 (55.4 to 65.7)	61.7 (57.8 to 65.7)	80.1 (76.5 to 83.7)	82.3 (78.9 to 85.7)	84.6 (81.4 to 87.7)
НΜ	77.0 (76.1 to 78.0)	58.8 (55.2 to 62.5)	56.1 (52.4 to 59.9)	59.8 (57.0 to 62.5)	73.7 (71.0 to 76.4)	73.8 (71.1 to 76.5)	74.3 (71.6 to 76.9)
PCS	49.7 (49.2 to 50.2)	39.6 (38.1 to 41.1)	39.3 (37.5 to 41.1)	39.7 (38.4 to 41.0)	48.9 (47.6 to 50.3)	49.0 (47.5 to 50.5)	49.9 (48.4 to 51.3)
MCS	47.6 (47.0 to 48.2)	36.2 (33.8 to 38.6)	35.1 (32.7 to 37.6)	37.0 (35.2 to 38.8)	45.4 (43.6 to 47.1)	45.7 (44.0 to 47.4)	46.2 (44.5 to 47.9)
*Source: BP, Bodil physical;	age-standardised normal ly pain; GH, General healt SF-36, Short Form-36; S	ive data from the South A h; MCS, Mental Compone F, Social function; VT, Vita	Australian female populat ent Summary; MH, Menta ality.	ion. ²⁹ al health; PCS, Physical C	omponent Summary; PF,	Physical function; RE, Rc	ole emotional; RP, Role

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

(table 4). Breast reduction surgery provided a greater gain in SF-36 PCS scores than a coronary artery bypass graft and hernia repair and the improvement was similar to that experienced by patients undergoing total knee replacement surgery. The improvement in SF-36 MCS scores following breast reduction surgery exceeded that of all other surgical procedures.

The impact of participant characteristics on HRQoL and benefit of surgical intervention

There was a significant positive correlation between baseline BMI and the total amount of breast tissue resected at surgery. That is, as the BMI increased there was an associated increase in the amount of breast tissue removed (Pearson's r=0.641, p<0.001). When exploring baseline SF-36 PCS scores, a significant negative correlation was found between SF-36 PCS scores and age (r=-0.13), BMI (r=-0.30), tissue resection weight (r=-0.26), degree of breast hypertrophy (r=-0.28) and ratio of breast to body volume (r=-0.19). Multivariate linear regression of candidate variables against baseline SF-36 PCS scores found BMI to be the only variable significantly related to preoperative SF-36 PCS scores ($R^2=0.16$, p<0.001). Multivariate regression analysis was also used to analyse predictors of the change in SF-36 PCS score following surgery and showed that improvement in SF-36 PCS scores was not significantly associated with any of these factors.

DISCUSSION

Principal findings

Findings from this study demonstrate that women with symptomatic breast hypertrophy have impaired quality of life compared with those in the general population. At baseline, participants in both the surgical and control breast hypertrophy groups scored significantly lower than the female general population in all SF-36 subscales, with pain being the most prominent. Surgical participants quality of life improved following breast reduction to such an extent that the health deficits were eliminated at 3 months following surgery and quality of life was 'normalised' to levels equivalent to that of the general population across all dimensions. This normalisation effect was stable across 12 months follow-up. The SF-36 health gain ranged from 14.5 to 33.1 points, and this exceeded the minimally important difference threshold estimates of one-half a SD approach³⁵ or a rule-of-thumb of a 10-point change on 100-point subscales,³⁴ supporting the contention that breast reduction surgery provides a clinically relevant health benefit.

Secondary aims of this study were to investigate factors that have the potential to influence the level of improvement in quality of life following surgery: BMI, degree of hypertrophy, bra cup size, age, preoperative breast symmetry and weight of tissue resection at surgery. Several of these factors are frequently used to restrict access to breast reduction surgery, none of which are based on high-quality evidence. In our study the improvement in

6



Figure 1 Comparison of mean preoperative and postoperative Short Form-36 scores with age-standardised female population norms (South Australian Health Omnibus Survey).²⁹

HRQoL was independent of these factors, suggesting that all women with symptomatic breast hypertrophy can benefit from this surgery regardless of commonly scrutinised factors. This is of clinical relevance as it highlights that women with a higher BMI or those with a lower weight of resection benefit equally and should not be discriminated against based on criteria-based restrictions. Furthermore, there was no increase in the complication rate in the obese participants.

Comparison with other studies

The finding that women with symptomatic breast hypertrophy have a considerable health deficit and impaired quality of life compared with women in the general



Figure 2 Mean change in Short Form-36 (SF-36) scores from baseline to 12 months for surgical and breast hypertrophy control groups. Error bars represent 95% CI. BP, bodily pain; GH, general health; MCS, mental component summary; MH, mental health; PCS, physical component summary; PF, physical function; RE, role emotional; RP, role physical; SF, social function, VT, vitality.

population is supported by existing studies within the literature.^{4 14 36–39} These studies also report that surgical intervention provides symptomatic relief and improves HRQoL to levels of the general population. Our findings support those of Blomqvist *et al* and demonstrate that the improvement in quality of life is stable for up to 1 year after surgery, enabling women to return to levels of HRQoL that are similar to the normal population.¹

Our study demonstrated that symptom relief and improvement in HRQoL are not impacted by BMI or the removal of a minimum weight of resection. This finding is consistent with existing studies using the SF-36; however, two of these studies were potentially biased due to the BMI restrictions on their study populations.^{6 40 41} Our study also supports previous findings of no significant difference in the complication rate based on obesity status.^{41–43} In spite of these findings access restrictions for breast reduction surgery are in place in many countries, despite a lack of supporting evidence.

The intervention effect of breast reduction surgery in our study was well in excess of the minimal clinically important difference for SF-36 PCS and MCS scores, which has been recommended by the developers as a 3-point change.^{32 33} The improvements in the SF-36 PCS score at 1 year following surgery were comparable to those of other widely accepted surgical interventions such as total hip and total knee replacement,⁴⁴ spinal fusion,⁴⁵ bariatric surgery⁴⁶ and coronary artery bypass graft surgery.⁴⁷ The improvements in the mental component score following breast reduction surgery actually exceeded those of all other interventions cited. Breast reduction surgery is a relatively inexpensive procedure, and the improvement in HRQoL provides evidence as to the comparative effectiveness of this intervention in

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

6

Open access

Table 3 Comparison of mean change (95% CI) in SF-36 scores following surgery in non-obese and obese participants							
SF-36 subscale	Non-obese (n=71)	Obese (n=138)	Difference in means (95% CI)	P value of difference*			
Physical function	16.1 (11.2 to 22.1)	23.4 (19.5 to 27.3)	6.8 (0.03 to 13.5)	0.050			
Role physical	19.4 (12.4 to 26.3)	25.9 (21.2 to 30.5)	6.5 (-1.7 to 14.7)	0.121			
Bodily pain	28.6 (22.8 to 34.5)	32.3 (27.8 to 36.9)	3.7 (–3.9 to 11.4)	0.337			
General health	10.2 (6.0 to 14.3)	12.2 (8.4 to 16.0)	2.0 (-4.1 to 8.2)	0.516			
Vitality	18.9 (14.8 to 23.1)	18.3 (14.2 to 22.4)	-0.7 (-7.2 to 5.9)	0.842			
Social function	23.6 (17.8 to 29.4)	21.9 (16.6 to 27.2)	-1.7 (-10.2 to 6.8)	0.701			
Role emotional	18.9 (12.8 to 25.0)	22.5 (17.2 to 27.8)	3.6 (–5.0 to 12.2)	0.409			
Mental health	14.9 (11.4 to 18.5)	13.0 (9.2 to 16.9)	-1.9 (-7.9 to 4.1)	0.532			

Obesity status: non-obese (<30 kg/m²), obese (\geq 30 kg/m²).

*Using an independent t-test.

SF-36, Short Form-36.

relieving the health burden and the functional symptoms of breast hypertrophy.

Strengths and limitations of this study

A potential limitation of our study was that the participant response rate for the breast hypertrophy control cohort was relatively low at 46%, which may be due to the recruitment process via postal questionnaire. Furthermore, while the total follow-up period for this cohort was 12 months, the intermediate time points of 3 and 6 months that were collected in the surgical cohort were not included in this cohort, although the consistency of outcomes at baseline and 12 months suggest that 3 and 6 month outcomes are likely to have been similar.

The strengths of our study were the prospective design, the relatively large sample size and the inclusion of a non-surgical control sample of women with breast hypertrophy who were recruited from the same waiting list as those in the surgical cohort. In addition, the postoperative outcomes described in this study included multiple time points over a 12-month period. In addition, our surgical cohort was not biased by restrictions that have been reported in previous studies based on a minimum weight of resection or BMI and therefore includes a broad spectrum across these variables. This is particularly important as it enables the accurate assessment of these factors as potential predictors of the change in HRQoL and outcomes of surgery and overcomes these limitations.

CONCLUSIONS AND POLICY IMPLICATIONS

Breast hypertrophy is a painful condition which is effectively treated by breast reduction surgery. The marked improvement in quality of life following breast reduction surgery is comparable to other widely accepted and approved surgical interventions. This study highlights that the improvement in quality of life following surgery is independent of traditionally used criteria based on BMI or a minimum weight of resection and demonstrates the health benefits of surgery regardless of these factors. This confirms the clinical effectiveness of breast reduction surgery and supports the contention that there is no justification for excluding women based on criteria such as BMI or the extent of breast hypertrophy.

Acknowledgements We thank the participants for their involvement in the study, especially Diane Barker for her valuable contribution; Andrea Smallman, Suzanne Patten and Richard Woodman for statistical advice; and Anne Taylor, Tiffany Gill and

Table 4 Mean improvement in SF-36 PCS and MCS scores between surgical interventions								
Reference	Surgical intervention	Preop PCS	Postop PCS	ΔPCS	Preop MCS	Postop MCS	∆MCS	N
This study	Bilateral breast reduction	39.7	49.9	10.2	37.0	46.2	9.2	191
Pivec et al ⁴⁴	Total knee replacement	33.0	47.8	14.8	52.9	55.9	3.0	281
Stickles et al ⁴⁸	Total hip replacement	28.0	41.2	13.2	51.2	53.9	2.7	551
Muller-Nordhorn <i>et al</i> ⁴⁷	Coronary artery bypass grafting	36.0	43.0	7.3	45.0	50.0	4.3	412
Polly et al ⁴⁵	Lumbar fusion (spine)	26.6	40.0	13.4	n/a	n/a	n/a	1826
Rogmark <i>et al</i> ⁴⁹	Incisional hernia repair	41.6	49.5	8.1	50.2	52.3	1.7	124
Faulconbridge et al ⁴⁶	Bariatric surgery	37.7	46.4	8.7	43.1	45.5	2.4	36

Δ, mean change in SF-36 score from preoperative to 12 months postoperative; MCS, Mental Component Summary; N, number of participants; n/a, not applicable; PCS, Physical Component Summary; SF-36, Short Form-36.

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

Open access

Jodie Avery for providing normative Short Form-36 data and scoring algorithms TC is the recipient of an Australian Government Research Training Programme PhD Scholarship.

Collaborators The following are members of the Anthropometry for Elective Surgery Assessment (AFESA) Research Group: Tamara Crittenden, Nicola Dean (Flinders Medical Centre, Flinders University); Philip Griffin, Karen Burford, Kaveeta Deut, Elizabeth Lomax, Phillipa van Essen (Flinders Medical Centre); Maciej Henneberg (University of Adelaide) and Daisy Veitch (SHARP Dummies, Flinders University).

Contributors NRD, TC, PAG, JR, DIW, DV and MH contributed to study design. TC, KB, EL, PE and KD contributed to data collection. TC was responsible for statistical analyses and original preparation of the paper. TC, JR, DIW and NRD drafted the manuscript. All authors approved the final manuscript and contributed intellectually important content. NRD attests that all listed authors meet authorship criteria and that no others meeting the criteria have been omitted. NRD is the guarantor of the paper and takes full responsibility for the work and/or the conduct of the study, had access to the data and controlled the decision to publish.

Funding The authors have not declared a specific grant for this research from any funding agency in the public, commercial or not-for-profit sectors.

Competing interests None declared.

Patient consent for publication Not required.

Ethics approval This study was approved by the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC reference numbers 118.056 and 73.17).

Provenance and peer review Not commissioned; externally peer reviewed. Data availability statement No additional data available

Open access This is an open access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited, appropriate credit is given, any changes made indicated, and the use is non-commercial. See: http://creativecommons.org/licenses/by-nc/4.0/.

ORCID iDs

Tamara Crittenden http://orcid.org/0000-0002-6484-3977 Julie Ratcliffe http://orcid.org/0000-0001-7365-1988 Nicola R Dean http://orcid.org/0000-0001-7084-2359

REFERENCES

8

- Blomqvist L, Eriksson A, Brandberg Y. Reduction mammaplasty provides long-term improvement in health status and quality of life
- *Plast Reconstr Surg* 2000;106:991–7. Freire M, Neto MS, Garcia EB, *et al.* Functional capacity and postural pain outcomes after reduction mammaplasty. Plast Reconstr S 2007;119:1149-56. discussion 57-8. Mello AA, Domingos NAM, Miyazaki MC. Improvement in quality of
- 3 life and self-esteem after breast reduction surgery. *Aesthetic Pl. Surg* 2010;34:59–64.
- Klassen A, Fitzpatrick R, Jenkinson C, et al. Should breast reduction surgery be rationed? A comparison of the health status of patients 4 before and after treatment: postal questionnaire survey. BMJ 1996;313:454-7.
- 5 Thoma A. Sprague S, Veltri K, et al. A prospective study of patients undergoing breast reduction surgery: health-related quality of life and
- clinical outcomes. *Plast Reconstr Surg* 2007;120:13–26. Collins ED, Kerrigan CL, Kim M, *et al.* The effectiveness of surgical and nonsurgical interventions in relieving the symptoms of
- macromastia. *Plast Reconstr Surg* 2002;109:1556–66. Iwuagwu OC, Walker LG, Stanley PW, *et al.* Randomized clinical trial 7 examining psychosocial and quality of life benefits of bilateral breast reduction surgery. *Br J Surg* 2006;93:291–4. Frey JD, Koltz PF, Bell DE, *et al.* The complex insurance reimbursement landscape in reduction mammaplasty: how does the
- 8
- American plastic surgeon navigate it? Ann Plast Surg 2014;72:23-9. Taylor AJ, Tate D, Brandberg Y, et al. Cost-Effectiveness of reduction mammaplasty. Int J Technol Assess Health Care 2004;20:269–73. Davis GM, Ringler SL, Short K, et al. Reduction mammaplasty: long-term efficacy, morbidity, and patient satisfaction. Plast Reconstr Surg 10 1995:96:1106-10.

- Breuning EE, Oikonomou D, Singh P, et al. Cosmetic surgery in the NHS: applying local and national guidelines. J Plast Reconstr Aesthet ra 2010:63:1437-42.
- Koltz PF, Frey JD, Langstein HN. Insurance coverage and reduction mammaplasty: a systematic review of current health care policies. Plast Reconstr Surg 2013;132:692e–3.
- 13 Nguyen JT, Wheatley MJ, Schnur PL, et al. Reduction mammaplasty: a review of managed care medical policy coverage criteria. Plast Reconstr Surg 2008;121:1092–100. 14 O'Blenes CAE, Delbridge CL, Miller BJ, et al. Prospective study of
- outcomes after reduction mammaplasty: long-term follow-up. *Reconstr Surg* 2006;117:351–8.
- Seitchik MW. Reduction mammaplasty: criteria for insurance coverage. *Plast Reconstr Surg* 1995;95:1029–32.
 ACT Government. Waiting time and elective surgery access policy.
- 2016 January 25, 2018. Available: https://www.act.gov.au/search? guery=access+to+elective+surgery&collection=act-gov&profile=
- NSW Government Department of Health, Waiting time and elective 17 surgery policy, 2012. Available: http://www1.health.nsw.gov.au/pds/ ActivePDSDocuments/PD2012_011.pdf
- Activer/DSDocuments/PD2012_0112_01 Government of South Australia SA Health. Restricted elective surgery policy directive, 2018. Available: https://www.sahealth.sa.gov.au/ wps/wcm/connect/3d568a77-a030-437a-957e-9b60bb143341/ Directive_Restricted+Elective+Surgery_D3072018.pdf?MOD= AJPERES&CACHEID=ROOTWORKSPACE-3d568a77-a030-437a-957e-9b60bb143341-mhygh9U
- 95 /e-9b60bb143341-mhygh9U Tasmanian Government Department of Health and Human Services. Improving Time to Treatment Tasmania's Elective Surgery Access Policy, 2015. Available: http://www.dhhs.tas.gov.au/__data/assets/ pdf_file/0020/60176/Improving_Time_to_Treatment__Tasmania_ s_Elective_Surgery_Access___DHHS_Wide___THOs__Policy___ 200911201_-_UNDER_REVIEW.pdf Victorian Government Department of Human Services. Elective surgery access policy 2015. Available: https://www.dhs.tas.gov.au/__data/setvices.telective surgery_access_policy_2015. 19
- 20 surgery access policy, 2015. Available: https://www2.health.vic. gov.au/about/publications/policiesandguidelines/elective-surgeryccess-policy-2015
- Government of Western Australia. Elective surgery access and waiting list management policy, 2017. Available: https://ww2. health.wa.gov.au/About-us/Policy-frameworks/Clinical-Services-Planning-and-Programs/Mandatory-requirements/Elective-Services/Elective-Surgery-Access-and-Waiting-List-Management-Policy
- Royal College of Surgeons of England. Is access to surgery a 22 Postcode lottery? 2014. Available: https://www.rcseng.ac.uk/-/ media/files/rcs/news-and-events/is-access-to-surgery-a-postcodelottery.pdf?la=en
- Wraight WM, Tay SKL, Nduka C, et al. Bilateral breast reduction 23 surgery in England: a postcode lottery. J Plast Reconstr Aesthet Surg 2007;60:1039–44. Iacobucci G. NHS proposes to stop funding 17 "unnecessary
- 24 procedures. *BMJ* 2018;362:k2903. NHS England. Evidence-Based interventions: response to the public
- 25 nhs.uk/wp-content/uploads/2018/11/ebi-consultation-responsedocument.pdf Ware JE, Sherbourne CD. The mos 36-item short-form health survey
- 26 (SF-36). I. conceptual framework and item selection. *Med C* 1992;30:473–83.
- 27 Ware J, Kosinski MA, Dewey J. How to score version 2 of the SF-
- Wate J, Rosinski MA, Dewey J. How to score version 2 of the S 36® health survey, 2000. Hawthorne G, Osborne RH, Taylor A, et al. The SF36 version 2: critical analyses of population weights, scoring algorithms and population norms. Qual Life Res 2007;16:661–73. 28
- Marin T, Taylor A, Gill T. A population profile of quality of life in South Australia – population norms for 2008 using the SF36 version 2 Population research and outcome studies unit. South Australian Adelaide: Department of Health. 2009.
- Veitch D, Burford K, Dench P, et al. Measurement of breast volume 30 using body scan technology(computer-aided anthropometry). Work 2012;41:4038–45. Yip JM, Mouratova N, Jeffery RM, *et al.* Accurate assessment of
- breast volume: a study comparing the volumetric gold standard (direct water displacement measurement of mastectomy specimen) with a 3D laser scanning technique. *Ann Plast Surg* 2012;68:135–41. Frendl DM, Ware JE. Patient-Reported functional health and
- well-being outcomes with drug therapy; a systematic review of randomized trials using the SF-36 health survey. *Med Care* 2014:52:439-45
- Ware JE, Kosinski M, Bayliss MS, et al. Comparison of methods 33 for the scoring and statistical analysis of SF-36 health profile and

Crittenden T. et al. BMJ Open 2020:10:e031804. doi:10.1136/bmiopen-2019-031804

6

Open access

6

summary measures: summary of results from the medical outcomes study. *Med Care* 1995;33:AS264–79.

- 34
- King MT. The interpretation of scores from the EORTC quality of life questionnaire QLQ-C30. *Qual Life Res* 1996;5:555–67. Norman GR, Sloan JA, Wynwich KW. Interpretation of changes in health-related quality of life: the remarkable universality of half a standard deviation. *Med Care* 2003;41:582–92. Miller BJ, Morris SF, Sigurdson LL, *et al.* Prospective study of eutocomes after reduction mammalexty. *Diate Beconstr. Surg* 35
- 36 outcomes after reduction mammaplasty. *Plast Reconstr Surg* 2005;115:1025–31.
- Shakespeare V, Cole RP. Measuring patient-based outcomes in a plastic surgery service: breast reduction surgical patients. *Br J Plast Surg* 1997;50:242–8. 37
- Kececi Y, Sir E, Gungor M. Patient-Reported quality-of-life outcomes of breast reduction evaluated with generic questionnaires and 38 the breast reduction assessed severity scale. Aesthet Surg J 2015;35:48–54.
- 2015;35:48–54. Blomqvist L, Brandberg Y. Three-Year follow-up on clinical symptoms and health-related quality of life after reduction mammaplasty. *Plast Reconstr Surg* 2004;114:49–54. Eggent E, Schuss R, Edsander-Nord A. Clinical outcome, 39
- 40 quality of life, patients' satisfaction, and aesthetic results, after reduction mammaplasty. Scand J Plast Reconstr Surg Hand Surg 2009:43:201-6.
- Güemes A, Pérez E, Sousa R, et al. Quality of life and alleviation of symptoms after breast reduction for Macromastia in obese patients: 41 is surgery worth it? Aesthetic Plast Surg 2016;40:62-70.

- 42 Wagner DS, Alfonso DR. The influence of obesity and volume of resection on success in reduction mammaplasty: an outcomes study.
- Plast Reconstr Surg 2005;115:1034–8.
 43 Cunningham BL, Gear AJL, Kerrigan CL, et al. Analysis of breast Cunningnam BL, Gear ADL, Kerrigan CL, et al. Analysis of breast reduction complications derived from the BRAVO study. *Plast Reconstr Surg* 2005;115:1597–604.
 Pivec R, Issa K, Given K, et al. A prospective, longitudinal study of patient satisfaction following total knee arthroplasty using the https://www.commune.com/study/stud
- short-form 36 (SF-36) survey stratified by various demographic and comorbid factors. *J Arthroplasty* 2015;30:374–8. Polly DW, Glassman SD, Schwender JD, *et al.* Sf-36 PCS
- 45 benefit-cost ratio of lumbar fusion comparison to other surgical interventions: a thought experiment. Spine 2007;32:S20-6.
- Faulconbridge LF, Wadden TA, Thomas JG, et al. Changes in depression and quality of life in obese individuals with binge eating 46 disorder: bariatric surgery versus lifestyle modification. Surg Obe Relat Dis 2013;9:790–6.
- Relat Dis 2013;9:790–6. Müller-Nordhorn J, Roll S, Willich SN. Comparison of the short form (SF)-12 health status instrument with the SF-36 in patients with coronary heart disease. *Heart* 2004;90:523–7. Stickles B, Phillips L, Brox WT, *et al.* Defining the relationship 47
- 48 between obesity and total joint arthroplasty. Obes Res
- 2001;9:219–23. Rogmark P, Petersson U, Bringman S, *et al.* Quality of life and 49 surgical outcome 1 year after open and laparoscopic incisional hernia repair: PROLOVE; a randomized controlled trial. Ann Surg 2016;263:244-50

Crittenden T, et al. BMJ Open 2020;10:e031804. doi:10.1136/bmjopen-2019-031804

BMJ Open: first published as 10.1136/bmjopen-2019-031804 on 17 February 2020. Downloaded from http://bmjopen.bmj.com/ on February 20, 2020 at Medical Library Flinders Univ of South
Australia. Protected by copyright.

Content removed due to privacy reasons.

Appendix C. Ethics approval for study 73.17

Content removed for privacy reasons.

Appendix D. Short Form-36 (SF-36)

Appendix E. Multidimensional Body-Self Relations Questionnaire (MBSRQ)

Appendix F. Surgical Complications Checklist

BILATERAL BREAST REDUCTION RESEARCH Complications Checklist

DOCTORS: PLEASE TICK ANY RELEVANT COMPLICATIONS AND TREATMENTS REQUIRED

Research Code: B					
Date:					
Timeframe post BBR surgery (circle): 4 wks	3 m	6m	12m	18m	Other:
Date:	•	•			

		1	1
On the day of this assessment has the patient any of the following (please tick):	Y/N	Treatment(s)	Comments
		(as per code list below)	
2= Incomplete wound healing (wound not yet epithelised completely)			
3= Scar issues eg hypertrophic/hypersensitive			
7= Asymmetry (noticable by patient AND Dr)			
8= Altered nipple sensation (total numbness or changes to the extent that it is causing patient distress)			
9= Skin necrosis due to loss of blood supply (greater than 2cm squared> otherwise tick "2"))			
13= Dog ears requiring further surgery (**only tick once listed for surgery)			
		1	1
Since last assessed, has this patient had any of the following:	Y/N	Treatment(s)	Comments
1= Wound infection requiring antibiotics			
4= Thrombo-embolic event eg PE, DVT			
5= Breast Haematoma - palpable mass requiring aspiration or surgery (do not count simple brusing)			
6= Chest infection/pneumonia within the first 4/52 post surgery			
10= Reaction to anaesthetic			
11= Constipation due to analgesic use			
···			

Treatment codes for complications: 0= Not applicable/no treatment 1 = Antibiotics 2= Return to theatre (eg skin graft, washout, debridement, evacuation of haematoma) 3= Dressings 4= Other- comment 5= Revisional surgery (eg dog ears, scar revision, re-reduction) ** only code once listed for surgery
Appendix G. Ethics approval for study 7848

Appendix H. Pre-operative BREAST-Q Reduction module

Appendix I. Post-operative BREAST-Q Reduction module

Appendix J. Data Usage Agreement

Item responses	Ν	Mean	SD	Min	Max	% floor	% ceiling	Cronbach's alpha
Satisfaction with Breasts	513	51.9	17.3	0	100	0.2%	4.1%	0.92
1. With your breasts in mind, in the past 2 weeks, how <u>satisfied or dissatisfied</u> have you been with:								
a. How your breasts look in clothes?	513	2.96	0.80	1	4	5.3%	24.8%	
b. How your breast size matches the rest of your body?	513	2.96	0.87	1	4	7.0%	28.3%	
c. The size of your breasts?	513	2.86	0.88	1	4	7.4%	25.2%	
d. The shape of your breasts when you are wearing a bra?	513	3.05	0.78	1	4	3.9%	29.6%	
e. How equal in size your breasts are to each other?	513	3.12	0.81	1	4	4.5%	34.9%	
f. How comfortably your bras fit?	513	2.81	0.84	1	4	6.8%	20.9%	
g. The shape of your breasts when you are not wearing a bra?	513	2.43	0.95	1	4	18.9%	13.5%	
h. How you look in the mirror clothed?	513	2.86	0.83	1	4	7.0%	20.9%	
i. How your breasts sit/hang on your chest?	513	2.56	0.91	1	4	13.8%	15.2%	
j. How normal your breasts look?	513	3.01	0.79	1	4	5.5%	25.9%	
k. How you look in the mirror unclothed?	513	2.29	0.94	1	4	23.0%	10.5%	
Psychosocial Wellbeing	513	55.0	21.8	0	100	1.9%	6.2%	0.97
2. With your breasts in mind, in the past 2 weeks, <u>how often</u> have you felt:								
a. Confident in a social setting?	513	3.62	1.06	1	5	4.1%	20.7%	
b. Of equal worth to other women?	513	3.66	1.19	1	5	6.6%	28.9%	
c. Good about yourself?	513	3.51	1.12	1	5	5.5%	19.3%	
d. Self-assured?	513	3.41	1.13	1	5	7.0%	16.6%	
e. Confident in your clothes?	513	3.39	1.19	1	5	8.0%	18.3%	
f. Accepting of your body?	513	3.3	1.22	1	5	9.9%	17.4%	
g. That your appearance matches who you are inside?	513	3.28	1.27	1	5	12.3%	17.5%	
h. Confident about your body?	513	3.08	1.25	1	5	14.6%	12.7%	
i. Attractive?	513	2.91	1.18	1	5	15.6%	9.0%	

Appendix K. Descriptive statistics and internal consistency of normative BREAST-Q scores

Item responses	Ν	Mean	SD	Min	Max	% floor	% ceiling	Cronbach's alpha
Sexual Wellbeing	473	48.5	24	0	100	8.2%	4.2%	0.92
3. Thinking of your sexuality, <u>how often</u> do you generally feel:								
a. Comfortable/at ease during sexual activity?	405	3.34	1.25	1	5	11.4%	18.3%	
b. Confident sexually?	435	3.23	1.24	1	5	12.2%	15.9%	
c. Satisfied with your sex life?	410	3.26	1.28	1	5	12.2%	18.1%	
d. Sexually attractive in your clothes?	464	2.93	1.22	1	5	15.9%	9.3%	
e. Sexy when unclothed?	458	2.5	1.31	1	5	31.4%	8.1%	
Physical Wellbeing	513	71.2	13.1	0	100	0.2%	4.1%	0.89
4. In the past 2 weeks, <u>how often</u> have you experienced:								
a. Headaches?	513	2.12	1.00	1	5	32.2%	2.0%	
b. Pain in your breast area?	513	1.42	0.75	1	5	71.0%	0.6%	
c. Lack of energy?	513	2.72	1.08	1	5	11.7%	7.2%	
d. Difficulty doing vigorous physical activities (e.g. running or exercising)?	513	2.64	1.33	1	5	25.2%	11.3%	
e. Feeling physically unbalanced?	513	2.24	1.19	1	5	36.5%	4.3%	
f. Shoulder pain?	513	1.89	1.11	1	5	52.4%	2.5%	
g. Difficulty sleeping because of discomfort in your breast area?	513	1.3	0.71	1	5	80.7%	1.0%	
h. Neck pain?	513	1.99	1.11	1	5	45.2%	2.9%	
i. Painful gouges or grooves in your shoulders from your bra straps?	513	1.73	1.08	1	5	61.2%	3.1%	
j. Feeling physically uncomfortable?	513	2.14	1.08	1	5	34.9%	3.1%	
k. Rashes under your breasts?	513	1.58	0.97	1	5	67.1%	2.0%	
I. Back pain?	513	2.26	1.24	1	5	37.2%	5.5%	
m. Arm pain?	513	1.49	0.87	1	5	69.8%	1.6%	
n. Pain, numbness or tingling in your hands because of your breast size?	513	1.2	0.61	1	5	88.1%	0.4%	

Sociodemographic		Australi	а	United States			
variable	n	Mean	SD	n	Mean	SD	
Satisfaction with Breasts							
Age							
18–24	62	49.8	13.7	6	63.8	22.7	
25–34	99	50.2	16.1	93	57.1	16.2	
35–44	88	48.8	18.4	171	55.9	16	
45–54	87	49.7	16.1	241	56.7	13.6	
55–64	75	56.6	18.1	402	56.8	17.6	
65+	102	56.1	18.6	289	56.5	16.6	
BMI							
< 30 kg/m ²	356	54.0	16.7	912	58.7	16.4	
≥ 30 kg/m²	157	47.4	17.8	287	50	14	
Bra cup size							
	283	52.8	17 0	719	59.3	16 1	
≥ D cup	230	50.9	17.6	487	52.7	15.7	
Chronic health condition?							
Yes	148	48.6	17.0	595	56.4	16.5	
No	365	53.3	17.3	606	56.9	16.1	
Employment status							
Full-time	119	52.9	17.1	511	57.1	15.8	
Part-time	143	50.8	17.8	168	56.5	17.2	
Voluntary work	3	58.7	11.0	33	59.6	14.2	
Homemaker	84	51.0	19.2	85	54.6	15.4	
Student	26	49.7	10.6	16	60.4	15.8	
Retired	85	56.1	17.3	326	56.8	16.9	
Unable to work or disabled	21	48.0	18.0	12	50.9	18.5	
Unemployed	24	45.8	13.0	19	50.4	15.3	
Other	8	55.5	12.6	32	55.8	15.9	
All participants	513	51.9	17.3	1206	56.7	16.3	

Appendix L. BREAST-Q scores for each of the four scales stratified by socio-demographic characteristics for Australian and United States populations

Sociodemographic	1	Australi	а	United States			
variable	n Me		SD	n	Mean	SD	
Psychosocial Wellbeing							
Age							
18–24	62	50.0	19.8	6	61.0	27.5	
25–34	99	51.9	19.9	93	62.7	18.6	
35–44	88	49.1	21.6	171	61.5	21.2	
45–54	87	52.0	22.3	241	67.2	17.0	
55–64	75	59.9	21.3	402	68.9	19.3	
65+	102	64.8	21.6	289	70.7	17.8	
BMI							
< 30 kg/m²	356	57.7	21.3	912	70.4	17.8	
≥ 30 kg/m²	157	48.6	21.7	287	58.1	19.4	
Bra cup size							
< D cup	283	55.8	21.8	719	69.9	18.3	
≥ D cup	230	53.9	21.9	487	63.8	19.5	
Chronic health condition?							
Yes	148	49.4	23.0	596	66.8	19.2	
No	365	57.2	21.0	606	68.1	18.8	
Employment status							
Full-time	119	55.0	23.2	511	66.6	19.1	
Part-time	143	54.8	20.0	168	68.0	18.3	
Voluntary work	3	61.3	27.7	33	70.5	19.6	
Homemaker	84	50.8	22.7	85	61.5	20.7	
Student	26	51.8	17.4	16	61.1	21.5	
Retired	85	64.6	19.6	326	70.4	18.1	
Unable to work or disabled	21	45.4	24.1	12	54.8	11.3	
Unemployed	24	45.5	23.5	19	61.0	21.5	
Other	8	60.8	13.5	32	72.2	18.1	
All participants	513	55.0	21.8	1206	67.5	19.0	

Sociodemographic		Australi	а	United States			
variable	n	Mean	SD	n	Mean	SD	
Sexual Wellbeing							
Age							
18–24	56	50.0	24.1	5	53.6	13.3	
25–34	98	50.9	23.1	88	55.7	16.7	
35–44	84	43.7	23.6	157	54.9	19.9	
45–54	83	47.5	23.4	226	56.9	18.2	
55–64	68	50.5	24.8	345	55.8	20.0	
65+	84	49.0	25.0	200	52.5	20.3	
BMI							
< 30 kg/m²	336	51.4	23.4	778	57.5	19.0	
≥ 30 kg/m²	137	41.4	23.9	240	47.8	18.9	
Bra cup size							
< D cup	258	50.1	24.2	602	57.2	19.2	
≥ D cup	215	46.7	23.6	423	52.4	19.2	
Chronic health condition?							
Yes	131	39.6	25.0	492	54.1	20.3	
No	342	51.9	22.7	529	56.3	18.5	
Employment status							
Full-time	118	48.9	23.1	461	55.9	18.4	
Part-time	135	50.8	23.2	151	58.0	18.0	
Voluntary work	3	49.7	18.3	29	51.5	16.1	
Homemaker	83	48.3	24.5	80	53.6	22.0	
Student	19	44.2	24.1	14	54.5	18.5	
Retired	69	50.5	25.1	233	54.2	21.1	
Unable to work or disabled	20	33.0	25.0	12	39.7	19.3	
Unemployed	20	44.7	25.4	16	51.1	24.2	
Other	6	50.0	22.6	25	55.5	18.2	
All participants	473	48.5	24.0	1025	55.2	19.4	

Sociodemographic		Australi	a	United States			
variable	n	Mean	SD	n	Mean	SD	
Physical Wellbeing							
Age							
18–24	62	70.0	12.6	6	68.7	13.9	
25–34	99	68.9	13.7	93	74.7	11.3	
35–44	88	70.3	12.8	171	75.4	10.7	
45–54	87	70.5	12.8	241	75.5	10.1	
55–64	75	74.4	12.7	402	75.5	11.4	
65+	102	73.2	13.0	289	76.9	10.7	
BMI							
< 30 kg/m ²	356	73.0	12.9	912	77.6	10.2	
≥ 30 kg/m²	157	67.2	12.5	287	69.8	11.0	
Bra cup size							
< D cup	283	72.9	13.0	719	78.1	9.8	
≥ D cup	230	69.2	12.9	487	72.2	11.4	
Chronic health condition?							
Yes	148	66.6	11.7	596	73.9	10.8	
No	365	73.1	13.1	606	77.5	10.7	
Employment status							
Full-time	119	71.2	14.1	511	75.8	10.8	
Part-time	143	71.2	13.2	168	75.7	9.8	
Voluntary work	3	67.7	4.2	33	80.2	9.7	
Homemaker	84	71.9	13.4	85	74.9	11.1	
Student	26	71.2	11.0	16	77.7	12	
Retired	85	73.6	11.1	326	76.4	10.2	
Unable to work or disabled	21	63.9	12.1	12	62.7	12.5	
Unemployed	24	67.6	14.1	19	68.7	20.6	
Other	8	70.4	14.4	32	74.3	11	
All participants	513	71.2	13.1	1206	75.7	10.9	

Appendix M. Plastic and Reconstructive Surgery journal published article

Crittenden TA, Watson DI, Ratcliffe J, Griffin PA, Dean NR. Outcomes of Breast Reduction Surgery Using the BREAST-Q: A Prospective Study and Comparison with Normative Data. *Plast Reconstr Surg*. 2019;144(5):1034-1044.

Removed due to copyright restriction. Can be accessed at:

https://journals.lww.com/plasreconsurg/Abstract/2019/11000/Outcomes_of_Breast_Reduction_Sur gery_Using_the.3.aspx

Appendix N. Ethics approval for study 299.18

6

6

MR288A-5

Patient Health Questionnaire



Government of South Australia SA Health

Verify name, DOB and address with patient / family

Information	
Date form completed:	
Who completed the form:	Designed to be 1
Height: Weight:	Patient Labei
Body Mass Index (BMI):	

Answer all questions and tick	boxes on all of the 4 pages	Please tick
Anaesthesia		
Have you had an anaesthetic before?		Yes N
If yes, list most recent and/or major operation(s):		
Have you or any blood relative ever had any problems with	anaesthetics?	
Details:		Yes N
How far can you walk without stopping?	What stops you from walking further?	
\Box round the house only \Box 1–2 blocks	Shortness of breath	Chest Pain
regular walking and exercise	Leg Pain Other:	chestrum
Do you have: 🗌 Loose Teeth 🗌 Chipped Teeth 🗌 Caj	oped / Crowned Teeth?	Yes N
Do you have dentures? Upper: Partial Full Lo	wer: Partial Full	Yes N
Do you have any neck stiffness? or Difficulty opening	g your mouth?	Yes N
Heartburn Reflux Indigestion Hiatus he	rnia 🗌 Stomach Ulcers	Yes N
Gastric band Weight loss surgery Date:		Yes N
Have you had steroids such as Prednisolone, Cortisone or Hyd last 3 months?	rocortisone in a tablet or through your vein in the	Yes N
When? What f	or?	
In the last two weeks have you had: Cold Flu	Sore Throat Chest Infection	Yes N
Females, is there any possibility you may be pregnant?		Yes N
Females, are you breastfeeding?		Yes N
Do you have any questions or concerns about the anaesthetic	, procedure or operation that you would like	
to discuss? For example: needle phobia, anxiety. Details:		Yes N
Allergies / Sensitivity		
Do you have a known allergy / sensitivity to latex or rubber?	? What reaction?	Yes N
Do you have any allergies / sensitivities to medications, foods,	or skin sensitivities?	
Please list:		Yes N
Alcohol, Tobacco and Drugs		
Do you drink alcohol? How many days a week?	How many glasses a day on average?	Yes N
Do you smoke cigarettes? If yes, how many per day?		Yes N
Thinking of guitting?		

Answer all questions and tick boxes on all of the 4 pages	Please tick	
Have you ever smoked in the past? If yes, how many cigarettes/day?	Yes No	
Por now many years? When did you quit?	0.442.6653	
Do you use recreational drugs? For example Marijuana, Ecstasy, Cocaine	Yes No	
Cardiac		
Do you have high blood pressure?		
Heart failure AF (Atrial fibrillation) Rheumatic Fever Heart Murmur		
Chest Pain or Angina Chest Tightness Palpitations		
How often? What triggers it?		
A heart attack? When?		
Have you ever had heart surgery? Details:		
Do you have: Pacemaker Implantable Defibrillator Heart Stent/s When inserted?		
Type: (Bring information card if available)		
Any heart investigations (ie: Echocardiogram, Angiogram) in the last 3 years?	Yes No	
What? Where? When?		0
Do you attend a heart clinic or see a Cardiologist? Where?	Yes No	
Who? Last appointment date:		
Respiratory		
Do you have troublesome shortness of breath? When?	Yes No	
Chronic bronchitis Emphysema Tuberculosis Persistent cough COPD	Yes No	
Do you have Asthma? How often? What triggers it?	Yes No	
Do you use a puffer, inhaler, nebulizer (eg. Ventolin)?		
What? How often?		
Have you ever been in hospital for your Asthma? Where? When?	Yes No	
Snore loudly? Have you been observed to stop breathing while asleep?	Yes No	
Fall asleep easily during the day while sitting/driving?		
Sleep Apnoea Use CPAP Sleep studies Where? When? Bring CPAP machine into hospital when you are admitted When?	Yes No	
Neurological		
Have you had? Strokes Mini Strokes (TIAs) Blackouts Funny Turns Seizures	Yes No	
How often? When was your last one?		
Muscular Dystrophy Myotonic Dystrophy Multiple Sclerosis	Yes No	
Do you have pain? Acute Chronic Details:	Yes No	
Mental Health		
Do you have Mental Health problems? Anxiety Depression Bi-polar	Yes No	
Dementia Alzheimer's Disease Short Term Memory Loss	Yes No	
Intellectual Disability? Details:	Yes No	
Can you sign your own consent? If no, who can?	Yes No	

Answer all questions and tick boxes on all of the 4 pages	Please tick
Cognition / Perception / Psychosocial	
Do we need to book an interpreter? Language:	Yes No
Do you have? Hearing Problem Vision Problems Details: Bring your hearing aids and glasses to hospital on admission	Yes No
Do you have any cultural / spiritual issues that we need to know about?	Yes No
Describe:	
Do you have an Advance Care Directive? (If yes, please bring it to the hospital with you)	Yes No
Endocrine / Organ Function	
Do you have Diabetes? Insulin Tablets Diet-controlled Most recent HbA1c value?	Yes No
Do you require a specific diet? If yes, what type of diet?	Yes No
Kidney Disease? Thyroid Disease? Details:	
Liver disease or jaundice? Details:	
Do you have any existing pressure areas?	Yes No
Paintul areas on your skin? Describe:	Yes No
Connective Tissue / Mobility	
Do you have Arthritis? Osteoarthritis Rheumatoid Which joints?	Yes No
Muscular disease or weakness in the arms or legs? Details:	
Bring your walking aids into hospital when you are admitted	Yes No
Have you had any falls in the last 6 months?	
Details:	Yes No
Falls Risk Assessment required? Nurse only to complete on admission 🔛 Yes 🔛 No	
Haematology / Other	
Do you have Anaemia? Are you Anaemic at the moment? Yes No	Yes No
Excessive bleeding or bruising (or family member with this problem)? Details:	Yes No
Other blood disorders (eg. Thalassemia, Sickle Cell)? Details:	Yes No
Blood transfusion? If yes, when:	Yes No
Blood clots? Deep Vein Thrombosis (DVT) Clot in Lung Other:	Yes No
Do you have any other serious illnesses or hereditary diseases not mentioned above	
(eg. Cancer, Polio, Cerebral Palsy, Dwarfism)? Details:	
Have you had a hospital admission for any health problems (excluding operations) in the last 3 years?	
Details:	
Infectious Diseases	
Do you have? HIV AIDS Hepatitis B Hepatitis C Other Hepatitis:	Yes No
Have you ever been told that you have: MRSA VRE Other:	Yes No
If yes, informed by: Where and when:	1

Medications / Drugs				Piedse	euck	
the second						
Do you take any medications/drugs? List ALL regular and occasic	nal medications and tablets o	r attach an	up to			
date drug list from your pharmacist or GP.				Yes	No	
Bring your medications to hospital on day of admission in chemist pack or original bottles/boxes.						
Herbal medicines? NOTE: All herbal medications and noness oil, Glucosamine, Garlic, Ginger) should be stopped at least	 Anti-inflammatories or ential medications (for exain t one week before surgery) 	mple: Fish	oil, Krill	Yes	No	
Do you use a pill box? If yes, who fills it?						
REGULAR MEDICATIONS (Please copy name and dose from pack	II AR MEDICATIONS (Please conviname and dose from packet) Strength (agringer tellet)					
and a set in the set of the set o	strength (eg. per tablet)	Morning	Midday	Evening	Night	
				-		
Vrite any questions or comments						
Vrite any questions or comments						
Vrite any questions or comments						
Vrite any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the	business day before your s	surgery				
Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home:	business day before your s	surgery Vork:				
Write any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Yes No Centrelin	business day before your s	surgery Vork: Dther:				
Vrite any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Yes No Centrelin All the above information is correct to the best of my know	business day before your s	surgery Vork: Dther:				
Write any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Yes No Centrelin All the above information is correct to the best of my know	business day before your s V Workcover C vledge. Signature: (Person completing f	surgery Vork:)ther: ;orm)				
Vrite any questions or comments vdditional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Yes No Centrelin All the above information is correct to the best of my know Vay Surgery Patients Only you are having day surgery you cannot drive for 24 hours after you ome and stay with you for the first 24 hours following anaesthesi	business day before your s v c Workcover C vledge. Signature: (Person completing f our anaesthetic. You must have a.	Surgery Vork: Dther: orm) e a responsi	ble adult t	o accompa	any you	
Vrite any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Do you require a sick certificate? Yes No Centrelin All the above information is correct to the best of my know Day Surgery Patients Only you are having day surgery you cannot drive for 24 hours after you ome and stay with you for the first 24 hours following anaesthesi esponsible adult name:	business day before your s V Workcover C Vledge. Signature: (Person completing f our anaesthetic. You must have a. Phone number:	surgery Vork: Dther: form) e a responsi	ble adult t	o accompa	any you	
Vrite any questions or comments Additional Information Your contact phone number between 3pm and 7pm of the Mobile: Home: Do you require a sick certificate? Yes No Centrelin All the above information is correct to the best of my know Vay Surgery Patients Only you are having day surgery you cannot drive for 24 hours after you one and stay with you for the first 24 hours following anaesthesi esponsible adult name: certify that:	business day before your s v c Workcover C vledge. Signature: (Person completing f our anaesthetic. You must have a. Phone number: rt me home and will be with r	surgery Vork: Other: orm) e a responsi ne for 24 h	ble adult t	o accompa ving anaes	any you thesia.	

_ast reviewed April 2014 Next review April 2016



AJOPS | ORIGINAL ARTICLE PUBLISHED: 24-09-2018

Measuring breast volume in hypertrophy: laser scanning or water displacement?

Tamara Crittenden BAppSc (Hons),^{1,2} Daisy E Veitch,³ Maciej Henneberg PhD DS,⁴ Karen Burford BAppSc OT (Hons),¹ Phillipa van Essen BHSc MPH,¹ Kaveeta Deut BAppSc OT,¹ Kalavani Zeitouneh BAppSc OT (Hons),¹ Elizabeth Lomax BAppSc OT,¹ Philip Griffin MBBS FRACS (Plas),¹ Nicola R Dean PhD FRACS (Plast)^{1,2}

- 1 Department of Plastic and Reconstructive Surgery Flinders Medical Centre Bedford Park, South Australia AUSTRALIA
- 2 College of Medicine and Public Health Flinders University Bedford Park, South Australia AUSTRALIA
- 3 Sharp Dummies Pty Ltd Belair, South Australia AUSTRALIA
- 4 Adelaide Medical School University of Adelaide Adelaide, South Australia AUSTRALIA

OPEN ACCESS

Correspondence

Name: Nicola R Dean

Address: Department of Plastic and Reconstructive Surgery Level 5, Flinders Medical Centre Flinders Drive Bedford Park, South Australia, 5042 AUSTRALIA Email: nicola.dean@sa.gov.au

Telephone: +61 8 82045213

Citation: Crittenden T, Veitch DE, Henneberg M, Burford K, van Essen P, Deut K, Zeitouneh K, Lomax E, Griffin P, Dean NR. Measuring breast volume in hypertrophy: laser scanning

or water displacement? Aust J Plast Surg. 2018;1(2):33-40.

Accepted for publication: 23 May 2018

Copyright © 2018. Authors retain their copyright in the article. This is an open access article distributed under the Creative Commons Attribution Licence which permits unrestricted use, distribution and reproduction in any medium, provided the original work is properly cited.

Abstract

Background: The accurate determination of intact breast volume facilitates preoperative planning for a range of plastic surgical breast procedures. In women with breast hypertrophy, volumetric assessment assists in planning the amount of tissue to be removed during breast reduction surgery to achieve breast symmetry. Further, in jurisdictions where restrictive surgical guidelines exist, measurement of intact breast volume is essential in order to justify breast reduction surgery. Not all practitioners have access to magnetic resonance imaging (MRI) or three dimensional (3D) laser scanning facilities, so the purpose of this study was to determine whether water displacement of the intact breast is an effective substitute method of measurement in women with breast hypertrophy.

Methods: A prospective cohort study was undertaken to measure breast volume using water displacement and 3D laser scanning in breast hypertrophy patients. The volume of a total of 322 breasts were determined using both measurement techniques; 194 preoperatively and 128 at 12 months following breast reduction surgery. Pearson correlations, linear regression and Bland-Altman analyses were used to compare the methods of breast volume assessment.

Results: The mean breast volume according to 3D laser scan was 1440 millilitres (SD = 588 millilitres) and for water displacement was 1419 millilitres (SD = 811 millilitres). There was a strong linear association between breast volumes as measured using water displacement and 3D laser scanning using a Pearson correlation (r = 0.89, p < 0.001). However, using the Bland-Altman analysis, the two methods were found not to be in agreement, with water displacement values consistently larger than 3D scan values.

Australasian Journal of Plastic Surgery

33

Volume 1 Number 2 2018

Conclusions While a strong linear association was found between breast volume measurements made using the water displacement technique and 3D laser scanning, the two methods did not have a high level of equivalence. Water displacement of the intact breast is not an accurate enough tool to be used as a substitute for 3D laser scanning, and therefore not an optimal method of breast volume measurement in women with breast hypertrophy.

Key words: breast volume, water displacement, 3D laser scan, breast hypertrophy, breast reduction

Introduction

Breast hypertrophy is for some women a cause of considerable physical and psychosocial impairment, and adversely affects quality of life. Physical symptoms may include: chronic back, neck, shoulder and breast pain; skin rashes; headaches; shoulder grooving from the constant pressure of bra straps; a reduced capacity for carrying out daily activities and exercise; and numbness or tingling in hands/fingers.¹ Psychosocially, women with breast hypertrophy often display low self-esteem and poor body image, reduced psychosexual function, anxiety and depression.²

Breast reduction surgery is widely known to be the most effective treatment for breast hypertrophy.³ Surgery provides almost immediate symptomatic relief in most cases, and considerably improves the health-related quality of life and wellbeing in women suffering from the functional symptoms of breast hypertrophy.^{4,5} Despite these proven benefits, an increase in the prevalence of breast hypertrophy, and hence in demand for breast reductions, has led to restrictions being placed on surgery by healthcare funders and third-party providers in many jurisdictions. Arbitrary restrictions on either the minimum amount of breast tissue required to be removed at surgery or a body mass index cut-off point have been introduced in many institutions and countries worldwide.6

Measurement of breast volume plays an important role in breast reduction, reconstruction, developmental asymmetry and augmentation. In women with breast hypertrophy it is important for preoperative planning, in intraoperative decisionmaking regarding the amount of tissue to be taken from each breast to achieve symmetry, and where removal of a minimum amount of breast tissue is required to justify surgery.¹⁷⁻²¹ Accurate estimation of breast volume and predicted tissue resection weight also promotes improved counsel to the patient and provides a valuable guide to training surgeons.

A variety of techniques are described in the literature to meet the need for accurate and objective measurement of breast volumes in the clinical setting including magnetic resonance imaging (MRI), mammography, plaster casting, Grossman-Roudner plastic cups, water displacement, anthropometric measurement and 3D surface imaging.⁷¹⁶

In a situation where a woman is undergoing post-mastectomy breast reconstruction, the gold standard for measuring the breast tissue volume for reconstruction is the Archimedes method of water displacement of the mastectomy specimen.²² When measuring breast tissue volume in the intact state, such as in candidates for breast reduction surgery, MRI is reported as being the most accurate.²³ However, MRI is expensive and not always accessible. Three-dimensional laser scanning has also been demonstrated as a valid method of breast volume measurement, $^{11,12,22}\ but$ some centres lack access to this technology. Water displacement of the intact breast has been used in some centres since 1970, but has not previously been validated against more recent methods.24

The purpose of this study is to assess the validity of measuring breast tissue volume using water displacement of the intact breast and to compare this measurement technique with 3D laser scan measurement.

Methods

Patient selection and measurement

A prospective cohort study was performed at Flinders Medical Centre in Adelaide, Australia. All women aged 18 years and above with breast hypertrophy who were eligible to be placed on a waiting list for bilateral breast reduction surgery between March 2007 and February 2016 were informed of the study. Women who met the

Australasian Journal of Plastic Surgery

Volume 1 Number 2 2018

eligibility criteria were invited to voluntarily participate in the study and were provided with a letter and information sheet further outlining the study. Women who were unable to complete written questionnaires, had impaired cognitive capacity, or refused to participate, were excluded from the study. Approval was obtained for this study from the local Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC approval 118.056).

Bilateral breast volumes were obtained preoperatively and/or at 12 months postoperatively by two different techniques-3D laser scanning and water displacement. Patients proceeded immediately from measurement by water displacement to the 3D scanning within the same appointment to allow direct comparison. Some participants had breast volume measurements taken at both the preoperative and 12 months postoperative times, while other participants were measured once as additional measurement was considered onerous for some patients.

Three-dimensional laser body scanning was performed using a Cyberware WBX scanner (Cyberware, Monterey, USA) and Cyslice software (Headus Pty Ltd, Perth, Australia). The Cyberware WBX scanner has four sets of laser heads and cameras that move vertically to scan the surface anatomy of subjects in the upright position. Prior to scanning, the assessor placed 'landmarks' on subjects using palpation to demarcate selected anatomical areas and the breast perimeter. Breast volume was measured according to the validated protocol described by Veitch et al²⁵ and Yip et al.²² Figure 2 illustrates the application of landmarks on a large-breasted woman to section the breast from the torso for volume analysis.

Water displacement measurements based on Archimedes' principle was performed on each breast to determine the volume, as described by Schultz et al.⁷ A large calibrated container was filled with warm water and the patient asked to individually immerse each breast into the container, using a specially modified surgical bed and ensuring that the superior and lateral edges of the breast were at water level (Figure 1). Individual breast volumes were then determined by the amount of water displaced.

Figure 1: Water displacement technique



Figure 2: Volumetric analysis of the breast with Cyberware WBX 3D scan and CySlice software



A=Photograph showing landmark placement following determination of breast borders by palpation B=Scan image with outline of breast and torso boundary C=CySlice image to section the breast portion of the scan from the torso

Statistical analysis

Data was entered into a spreadsheet and statistical analysis performed using SPSS v23.0 statistical software (IBM Corp, Armonk, New York). Pearson coefficients were calculated to assess the linear association between the two measurement techniques.

The strength of any correlation between the two techniques was then determined with a Bland-Altman analysis.²⁶ This analysis plots the mean breast volume measurements for both techniques against their differences. If the two methods are comparable and in agreement, then the differences should be small and the mean of the differences should be close to zero. The 95% limits of agreement were formed using the mean difference in volumes ± 1.96 standard deviation (SD) of the differences

Australasian Journal of Plastic Surgery

Volume 1 Number 2 2018

in volumes. Linear regression analysis was performed to assess proportional bias between the two measurement techniques.

Results

A total of 322 breasts were measured in the study using both 3D laser scanner and water displacement techniques—194 breasts (97 women) were measured preoperatively and 128 breasts (64 women) were measured postoperatively. The mean breast volume according to 3D laser scan was 1440 millilitres (SD = 588 millilitres) and using water displacement was 1419 millilitres (SD = 811 millilitres).

In comparing the breast volumes obtained from 3D laser scanning and water displacement, a Pearson correlation demonstrated a strong positive linear association between the two methods (r = 0.89, p < 0.001) (Figure 3).

Figure 3: Pearson correlation of breast volume measurement—water displacement mean volume versus 3D laser scan mean volume



There is a linear association between the two methods

Figure 4 shows the results of the Bland-Altman plot showing the differences between the two techniques plotted against the averages of the two techniques. A mean difference of -21.7 millilitres with a standard deviation of 399 millilitres was found between the two techniques. While most values were within the 95 percent limits of agreement (a lower limit of -804 millilitres and an upper limit of 760 millilitres), the majority of the data points are widely spread across this range and do not lie close to zero, which would have been

Australasian Journal of Plastic Surgery

the case if there was good agreement between the measured values.

Linear regression analysis confirmed a degree of proportional bias, meaning that one method gave values that were consistently higher or lower than those of the other method. In this instance, the breast volumes obtained from water displacement tended to be larger than those from the 3D scanner, and the difference between the two methods increased as the mean volume increased. This is evident from the plot where there was a discernable trend for more points to be above the line of mean difference in volume as the mean volume increased (Figure 4).

Figure 4: Bland-Altman plot showing mean breast volume versus difference in mean breast volume as measured by water displacement and 3D scanning techniques



While most values lie within the 95% limits of agreement on the plot (±1.96 x SD), there is a large difference in measurement of breast volume using the two methods

The statistical analyses therefore suggest that despite a strong linear correlation between the two methods of breast volume measurement, the measures have low agreement on actual values. The water displacement values were consistently larger than the 3D scan values. The greatest agreement in breast volume measurement between the two methods (within 10 percent) was seen for volumes between 1000 and 2000 millilitres (Figure 5). The findings of this study are in contrast to previous studies conducted in our unit that demonstrated equivalence between 3D laser scanner and direct water displacement of mastectomy specimens, and also between the 3D laser scanner and MRI.

Volume 1 Number 2 2018

Figure 5: The level of agreement between the two methods



Solid green line indicates volumes within 10% agreement and orange dashed line indicates volumes within 20% agreement. Shaded area highlights the range of breast volumes with the highest level of agreement.

Figure 6 demonstrates the breast volumes measured using water displacement and 3D laser scanning in a single patient both preoperatively and postoperatively at 12 months. The amount of breast tissue removed from each breast was 1444 grams, totalling 2888 grams. The patient's body weight remained relatively stable during this period—127 kilograms prior to surgery and 125.3 kilograms at the postoperative time point.

Figure 6. Breast volumes measured using water displacement and 3D laser scanning in a single patient both preoperatively and postoperatively at 12 months



Measurement tune	The operative			1 ost-operative				
ivieasurement type	Right	Left	Total	Right	Left	Total		
WD (ml)	4260	3900	8160	1920	1920	3840		
3D Scan (ml)	2997	3259	6256	1623	1684	3307		
TRW (g)	1444	1444						

Discussion

A large number of techniques for the objective measurement of breast volume have been described in the literature. However, these techniques often vary in accuracy and reliability and some are limited in their application to large-breasted patients. Many studies specifically on women with breast hypertrophy aim to develop a formula for estimating breast reduction weight in reduction mammoplasty rather than the assessment of breast volumes both before and after surgery.^{27:32} This is most likely driven by restrictions placed on breast reduction surgery by healthcare providers or insurance companies which often require surgeons to predict a minimum weight of resected breast tissue. Although a formula to predict tissue resection weight is clinically useful, there remains a need for the accurate determination of absolute breast volumes both in planning breast surgery and in assessing outcomes following surgery.

The use of 3D scanning in the specialty of plastic and reconstructive surgery has increased in recent times. It provides a valuable and non-invasive assessment of the breast for the surgeon, describing the volume, shape, contour, projection and symmetry. This information can not only be useful in the preoperative phase but in the assessment of postoperative outcomes and symmetry. The accurate and objective measurement of breast volume is particularly important in the complex field of breast reconstruction, such as in assessing outcomes following autologous fat grafting techniques¹¹ and preoperatively for unilateral reduction or augmentation for asymmetry.33,34 Three-dimensional scanning technology has also been employed in other areas of plastic surgery including facial symmetrisation 35,36 and for the preoperative planning of nasal surgery.³⁷

The Cyberware WBX laser scanner and CySlice software used in this study have previously been validated against water displacement in measuring mastectomy specimens with a strong correlation demonstrated.²² The same 3D laser scanner and protocol were also found to be strongly equivalent to non-contrast MRI for the assessment of breast volume.¹¹

The most accurate metric of true breast tissue volume is water displacement of mastectomy specimens, however, this is not applicable other than in post-mastectomy reconstruction scenarios.^{15,22,38} Additionally, volume measurement of the intact breast has proven to be more variable due to the challenge of correctly differentiating breast tissue from the chest wall.

Australasian Journal of Plastic Surgery

37

Volume 1 Number 2 2018

This study aimed to determine whether water displacement is a valid lower-cost method of breast volume measurement in women with breast hypertrophy compared to 3D laser scanning. Our measurement of 322 breasts by both 3D laser scan and water displacement showed a close association between the two techniques but the water displacement method had a clear tendency to estimate a larger volume than the 3D scan. A possible explanation for the difference between techniques is that measurement using water displacement requires the patient to be in a prone position, which may change the form of the breast and add axillary or abdominal fat to the breast measurement. In contrast, patients are in the upright position for measurement with the 3D scanner.

The water displacement technique is also highly dependent on the positioning of the patient. Alternative devices for the water displacement technique have been described to address the disadvantages of the immersion technique used in this study.³⁹ Our study supports previous findings that, although water displacement is easily accessible and a low-cost option, the technique has relatively low accuracy for measurement of the intact breast due to poor reproducibility.^{13,40} Threedimensional scanning and MRI display the highest level of accuracy and reliability and provide a comprehensive assessment of volume and shape for breasts in situ. However, these techniques can be cost-prohibitive. Linear anthropometric measurements are less accurate, however, the costs are negligible, they are fast and they have been shown to provide a reliable estimate of breast volume in the clinical setting.^{9,10,14,41}

Another source of difference is that 3D scanning technology allows for a curved cut plane at the back of the breast where it can digitally follow the curved chest wall, while water displacement has by definition a flat cut plane at the water level. The manual positioning of landmarks prior to 3D scanning using palpation to accurately identify the margins of the breast base, and relying on these points later in the scan image, helps to control for potential location error. Key advantages of this study are its large sample size and that the data was collected prospectively. One limitation of this study is that not all participants had breast volume measurements taken at both the preoperative and 12 months postoperative times. It was decided that additional measurement was too onerous for some patients given the commitments associated with voluntarily participating in the study. As a result, both of these techniques were performed on discrete subsets of patients at each time point.

Future extensions to this work would be to incorporate the results of comprehensive anthropometric body shape assessment that was also conducted pre- and postoperatively as part of this study. Analysis of patient-reported outcome measures data collected during the study would also further enhance our understanding of surgery outcomes and of the overall satisfaction in women with reduction in breast volume through surgery.

Conclusion

This study found a strong linear association between the measurement of breast volumes using water displacement and 3D laser scanning techniques in women with breast hypertrophy but the methods have low agreement on actual values. While water displacement may be more convenient and accessible in clinical practice, 3D scanning remains preferable as it has been proven to be a more accurate and reliable technique for the determination of intact breast volume.

Acknowledgements

The authors would like to acknowledge and thank the following people: members of the AFESA research group; Andrea Smallman, nurse practitioner; David Summerhayes and Lynton Emerson for clinical photography; Chris Leigh for the water displacement apparatus; Phil Dench from Headus Australia for the design and volume calculation software; Kathleen Robinette for assistance in scanner selection; Professor David Watson and Professor Julie Ratcliffe for PhD supervision; the Australian Government Research Training Program Scholarship for providing a PhD scholarship at Flinders University.

Australasian Journal of Plastic Surgery

Volume 1 Number 2 2018

Disclosure

The authors have no financial or commercial conflicts of interest to disclose.

References

- Kerrigan CL, Collins ED, Striplin D, Kim HM, Wilkins E, Cunningham B, Lowery J. The health burden of breast hypertrophy. *Plast Reconstr Surg.* 2001;108(6):1591–99.
- 2 Singh KA, Losken A. Additional benefits of reduction mammaplasty: a systematic review of the literature. *Plast Reconstr Surg.* 2012;129(3):562–70.
- 3 Collins ED, Kerrigan CL, Kim M, Lowery JC, Striplin DT, Cunningham B, Wilkins EG. The effectiveness of surgical and nonsurgical interventions in relieving the symptoms of macromastia. *Plast Reconstr Surg.* 2002;109(5):1556–66.
- 4 Mello AA, Domingos NA, Miyazaki MC. Improvement in quality of life and self-esteem after breast reduction surgery. *Aesthet Plast Surg.* 2010;34(1):59–64.
- ⁵ Blomqvist L, Eriksson A, Brandberg Y. Reduction mammaplasty provides long-term improvement in health status and quality of life. *Plast Reconstr Surg.* 2000;106(5):991–97.
- 6 Krieger LM, Lesavoy MA. Managed care's methods for determining coverage of plastic surgery procedures: the example of reduction mammaplasty. *Plast Reconstr Surg.* 2001;107(5):1234–40.
- 7 Schultz RC, Dolezal RF, Nolan J. Further applications of Archimedes' principle in the correction of asymmetrical breasts. Ann Plast Surg. 1986;16(2):98–101.
- 8 Smith DJ, J., Palin WE, J., Katch VL, Bennett JE. Breast volume and anthropomorphic measurements: normal values. *Plast Reconstr Surg.* 1986;78(3):331–35.
- 9 Qiao Q, Zhou G, Ling Y. Breast volume measurement in young Chinese women and clinical applications. *Aesthet Plast Surg.* 1997;21(5):362–68.
- 10 Longo B, Farcomeni A, Ferri G, Campanale A, Sorotos M, Santanelli F. The BREAST-V: A unifying predictive formula for volume assessment in small, medium, and large breasts. *Plast Reconstr Surg.* 2013;132(1):1e–7e.
- 11 Howes BH, Watson DI, Fosh B, Yip JM, Kleinig P, Dean NR. Magnetic resonance imaging versus 3-dimensional laser scanning for breast volume assessment after breast reconstruction. *Ann Plast Surg.* 2017;78(4):455–59.
- 12 Lee WY, Kim MJ, Lew DH, Song SY, Lee DW. Three-dimensional surface imaging is an effective tool for measuring breast volume: a validation study. *Arch Plast Surg.* 2016;43(5):430–37.
- 13 Bulstrode N, Bellamy E, Shrotria S. Breast volume assessment: comparing five different techniques. *Breast*. 2001;10(2):117–23.
- 14 Sigurdson LJ, Kirkland SA. Breast volume determination in breast hypertrophy: an accurate method using two anthropomorphic measurements. *Plast Reconstr Surg.* 2006;118(2):313–20.
- 15 Losken A, Seify H, Denson DD, Paredes AA, Jr., Carlson GW. Validating three-dimensional imaging of the breast. Ann Plast Surg. 2005;54(5):471–76; discussion 7–8.

- 16 Kovacs L, Eder M, Hollweck R, Zimmermann A, Settles M, Schneider A, Endlich M, Mueller A, Schwenzer-Zimmerer K, Papadopulos NA, Biemer E. Comparison between breast volume measurement using 3D surface imaging and classical techniques. *Breast.* 2007;16(2):137–45.
- 17 Klassen A, Fitzpatrick R, Jenkinson C, Goodacre T. Should breast reduction surgery be rationed? A comparison of the health status of patients before and after treatment: postal questionnaire survey. Brit Med J. 1996;313(7055):454–57.
- 18 Koltz PF, Frey JD, Langstein HN. Insurance coverage and reduction mammaplasty: a systematic review of current health care policies. *Plast Reconstr Surg.* 2013;132(4):692e–3e.
- 19 O'Blenes CAE, Delbridge CL, Miller BJ, Pantelis A, Morris SF. Prospective study of outcomes after reduction mammaplasty: Long-term follow-up. *Plast Reconstr Surg.* 2006;117(2):351–58.
- 20 Seitchik MW. Reduction mammaplasty: criteria for insurance coverage. Plast Reconstr Surg. 1995;95(6):1029–032.
- 21 Nguyen JT, Wheatley MJ, Schnur PL, Nguyen TA, Winn SR. Reduction mammaplasty: a review of managed care medical policy coverage criteria. *Plast Reconstr Surg.* 2008;121(4):1092–100.
- 22 Yip JM, Mouratova N, Jeffery RM, Veitch DE, Woodman RJ, Dean NR. Accurate assessment of breast volume: a study comparing the volumetric gold standard (direct water displacement measurement of mastectomy specimen) with a 3D laser scanning technique. *Ann Plast Surg.* 2012;68(2):135–41.
- 23 Choppin SB, Wheat JS, Gee M, Goyal A. The accuracy of breast volume measurement methods: A systematic review. *Breast* 2016;28:121–29.
- 24 Bouman FG. Volumetric measurement of the human breast and breast tissue before and during mammaplasty. Br J Plast Surg. 1970;23(3):263–64.
- 25 Veitch D, Burford K, Dench P, Dean N, Griffin P. Measurement of breast volume using body scan technology (computer-aided anthropometry). Work. 2012;41 Suppl 1:4038–045.
- 26 Bland JM, Altman DG. Comparing methods of measurement: why plotting difference against standard method is misleading. *Lancet.* 1995;346(8982):1085–087.
- 27 Eder M, Papadopulos NA, Kovacs L. Breast volume determination in breast hypertrophy. *Plast Reconstr Surg.* 2007;120(1):356–57.
- 28 Kocak E, Carruthers KH, McMahan JD. A reliable method for the preoperative estimation of tissue to be removed during reduction mammaplasty. *Plast Reconstr Surg.* 2011;127(3):1059–064.
- 29 Sommer NZ, Zook EG, Verhulst SJ. The prediction of breast reduction weight. *Plast Reconstr Surg.* 2002;109(2):506–11.
- 30 Kececi Y, Sir E. Prediction of resection weight in reduction mammaplasty based on anthropometric measurements. *Breast Care*. 2014;9(1):41–45.
- 31 Descamps MJ, Landau AG, Lazarus D, Hudson DA. A formula determining resection weights for reduction mammaplasty. *Plast Reconstr Surg.* 2008;121(2):397–400.
- 32 JZ, Wendel JJ, Zellner EG, Hagan KF, Shack RB, Corlew DS. Association between preoperative measurements and resection weight in patients undergoing reduction mammaplasty. Ann Plast Surg. 2010;64(5):512–15.

Volume 1 Number 2 2018

Crittenden, Veitch, Henneberg et al: Measuring breast volume in hypertrophy: laser scanning or water displacement?

AJOPS | ORIGINAL ARTICLE

- 33 Yip JM, Watson DI, Tiggemann M, Hsia S, Smallman AE, Dean NR. Determinants of breast reconstruction outcome: how important is volume symmetry? J Plast Reconstr Aes. 2015;68(5):679–85.
- 34 Liu C, Luan J, Mu L, Ji K. The role of three-dimensional scanning technique in evaluation of breast asymmetry in breast augmentation: a 100-case study. *Plast Reconstr Surg.* 2010;126(6):2125–132.
- 35 Kovacs L, Zimmermann A, Brockmann G, Gühring M, Baurecht H, Papadopulos NA, Schwenzer-Zimmerer K, Sader R, Biemer E, Zeilhofer HF. Three-dimensional recording of the human face with a 3D laser scanner. *J Plast Reconstr Aes.* 2006;59(11):1193–202.
- 36 Knoops PGM, Beaumont CAA, Borghi A, Rodriguez-Florez N, Breakey RWF, Rodgers W, Angullia F, Jeelani NU, Schievano S, Dunaway DJ. Comparison of three-dimensional scanner systems for craniomaxillofacial imaging. J Plast Reconstr Aes. 2017;70(4):441–49.
- 37 Lekakis G, Claes P, Hamilton GS 3rd, Hellings PW. Three-dimensional surface imaging and the continuous evolution of preoperative and postoperative assessment in rhinoplasty. *Facial Plast Surg.* 2016;32(1):88–94.
- 38 Kayar R, Civelek S, Cobanoglu M, Gungor O, Catal H, Emiroglu M. Five methods of breast volume measurement: a comparative study of measurements of specimen volume in 30 mastectomy cases. *Breast Cancer Basic Clin Res.* 2011;5:43–52.
- 39 Tezel E, Numanoglu A. Practical do-it-yourself device for accurate volume measurement of breast. *Plast Reconstr Surg.* 2000;105(3):1019–023.
- 40 Henseler H, Khambay BS, Bowman A, Smith J, Paul Siebert J, Oehler S, Ju X, Ayoub A, Ray AK. Investigation into accuracy and reproducibility of a 3D breast imaging system using multiple stereo cameras. J Plast Reconstr Aes. 2011;64(5):577–82.
- 41 Brown N, White J, Milligan A, Risius D, Ayres B, Hedger W, Scurr J. The relationship between breast size and anthropometric characteristics. *Am J Hum Biol.* 2012;24(2):158–64.

AFESA – Breast Reduction Measurement Chart

- Measure in millimetres
- Tape measure to be firm but not tight Tape measure and anthropometer to be horizontal at all times
- Patients to "stand tall" and without shoes •
- Zero end of tape measure to be below the number end of •
- the tape measure so that the graduation lines meet.

Tape Measurements

Surgery = BBR ABDO Timeframe = Pre surgery 3 mo post surgery 6 mo post surgery 12 mo post surgery		
Research Code =		
Date:		
Measurer		

Measurement Measurement chart Description (mm) Bra size Post surg bra size = Own bra size = Mark the sternomanubrial point by feel (or 5-6cm below the jugular notch or run fingers between 1st and 2nd ٠ Unisex chest Arms out and place across the lower section of the scapulars Lower arms to side and measure using the cross over hand technique Eye ball apex of breast and measure circumferentially around breast and body Bust Girth • Repeat measurement lower down until highest measurement is achieved Put land marks on for bust point to bust point on the apex of the breast Place tape measure in the crease where the bra ends on the pts back Under Bust Girth Bring the tape measure together along the bra line on the front Intermammary Fold Ask patient to lift breasts. Measure circumference in intermammary fold. Girth Tie elastic around the person well above the waist Waist girth Ask the patient to put the elastic at their waist then ask them to put their hands on their waist /circumference Index fingers of person should line up with elastic- adjust as necessary and mark centre back. ٠ front, left and right. Measure this circumference around markings Hip girth/ Find widest aspect of hips and mark right and left sides ٠ Maximum hip Measure horizontal circumference Bust width- bust Measure between the land marks. This may or may not be the nipple • point to bust point Nipple to Nipple • Ask patient to point to nipples and measure between this point Using anthropometer measure height from behind. Ask pt to stand up straight, remove shoes. Height (Stature) Ensure head is looking forward and eve level is horizontal with ear hole • Suprasternale ٠ Mark the jugular notch (Jugular notch) to • Line up point of anthropometer to jugular notch and take reading ensuring anth. Is vertical ground Pubic symphysis to Ask patient to feel for pubic bone. They should feel soft, then a hard point. ٠ ground Mark this point with a dot and measure from here to the ground Palpate for shoulder point. Release pressure and remeasure to ensure the soft tissue point is • **Bi-Acromial**

Weight and Volumes

•

Shoulder width

Bi-iliac Hip width

Item	Description	Measurement
Weight (kg)	 Zero machine. Remove shoes and heavy clothing/objects 	(kg)
Water displacement LEFT	Shoulder touching the bucket with arm by their side Ensure all of breast immersed in warm water	(mm)
Water displacement RIGHT	 Measure distance from water level to top of the bucket in mm 	(mm)
Scanner LEFT breast volume	Record segmented volume	(ml)
Scanner RIGHT breast volume	Record segmented volume	(ml)
Scanner Torso volume (ml)		(ml)
Scanner TOTAL BODY vol (ml)		(ml)
Data Extraction	By:(name) Scan # Used Date Extracted:	

horizontal with the bony point. Mark the shoulder point once established

Measure from the front. Palpate for iliac crest with flat palms. You may need to work your way

Slide finger up over hip bone to find ridge of hip. Ensure this point is horizontal by feeling

Using the sliding calliper measure bony points from behind.

out to lateral aspect of hip bone

several times. Mark each hip with a dot. Using calliper measure between each bony point